

Effectiveness of Avatars Representing Teachers in M-learning in Saudi Arabia Higher Education

By

Ali Alowayr

This thesis is submitted in partial fulfilment of the requirments for the degree of Doctor of Philosophy (Ph.D)

School of Computer Sciences

Supervisor: Prof. Rachel McCrindle

England, UK September 2017

Abstract

The aim of this research has been to investigate how m-learning can be used to complement traditional learning environments in Higher Education in Saudi Arabia. Specific attention has been paid to how engagement and performance in learning can be influenced by the type of avatar representation of the teacher on the mobile device, which might be in the form of video, audio, image, cartoon or simple text. This study enhances the field of knowledge related to m-learning via three main contributions which are described and developed as the thesis progresses.

Firstly, the research develops, as an extension to the traditional technology acceptance model (TAM), an educational model, MADE-ME (Multi Avatar Delivery Environment for Mobile Education). This model defines how students can interact with different avatar representations of the teacher to deliver learning content. The model shows the relationships between factors such as engagement, interactive elements, gender, major of study, pedagogical performance, etc.

Secondly the research produces a framework MADE-ME (Multi Avatar Delivery Environment for Mobile Education) web-app that enables a range of avatars to represent the teacher in their purpose of delivering interactive learning content via mobile technologies, and which allows students to be tested on how much they have learnt from the content or lesson.

Thirdly, a comprehensive case study is undertaken with student groups studying on a compulsory English language module as part of their higher education in Saudi Arabia to determine how they engaged with the mobile content and how effective their learning was to evaluate and to validate the MADE-ME model and app. The data was collected by a mixed methods approach and used REGRESSION and UNIANOVA techniques for analysing the quantitative data from questionnaires, and a thematic approach for analysing qualitative data from open-ended questions.

The thesis concludes with recommendations for implementing m-learning in Saudi higher education, limitations of the current study and suggestions for further research.

DECLARATION

I confirm that this is my own work and the use of all material from other sources has been properly and fully acknowledged.

Elements of the research presented in this thesis have been published in or submitted to academic journals and conference proceedings. A list of these first-authored publications and submissions can be found at the beginning of the thesis.

Ali Alowayr

Date: 28/03/2018

ACKNOWLEDGEMENT

First and foremost, all praise and thanks be to Allah for giving me the opportunity and ability to pursue this thesis. To whom belong all things in the heavens and on earth. To Him be praise in the Hereafter.

My deepest thanks are due to my beloved parents, Dad, Saeed and my Mom, who sadly passed away in 2005 for their love, encouragement and support throughout my studies. To my wife, and my little daughters Jumanah and Sulaf, I am highly thankful to all of you for your patience and your great sacrifices for me throughout this journey.

I would like to express my sincere gratitude to my supervisor, Professor Rachel McCrindle for her guidance and support during this Ph.D. Special thanks to her and appreciation for the fruitful advice and encouragement.

I would like to acknowledge all of the study participants for their motivated participations and their willingness to express their perceptions and views following the proposed model.

Finally and most importantly, I am extremely thankful my to brothers, sisters, close relatives and all of my friends and colleagues for their support and encouragement to fulfil my Ph.D study.

DEDICATION

To my beloved parents, father Saeed, and my mother Nafra, who sadly passed away in the summer of 2005. Gone but never forgotten.

my lovely wife, Amal

my jewel daughters, Jumanah and Sulaf

my brothers and sisters

Ali Alowayr Publication List

- 1. Alowayr, A., & McCrindle, R. (2017). A case Study to Investigate Students' Readiness, to Use M-learning in Saudi Arabia. *Computing Conference 2017, IEEE*. London, United kingdom.
- 2. Alowayr, A., & McCrindle, R. (2017).Constructing a Multi Avatar Delivery Environment Model for Mobile Education. Accepted in,*International Journal of Computer Science & Information Technology (IJCSIT)*,but not published yet.
- Alowayr, A., & McCrindle, R. (2016). User Interfaces for Mobile Learning in Higher Education in Saudi Arabia. In *International Conference on Information Society (i-Socity 2016)* (pp. 178–179). Dublin, Ireland: IEEE.
- Alowayr, A., & McCrindle, R. (2016). Investigation into Students Preferred Interface on Mobile Devices and their Pedagogic Performance and their Pedagogic Performance. In *E-Learn: World Conference on E-Learning* (pp. 1108–1114). Washington, DC, United States: Association for the Advancement of Computing in Education (AACE).
- 5. Alowayr, A., & McCrindle, R. (2014). Optimisation of e-Learning for Mobile Delivery Platforms. *Conference of Post Graduate at Reading University*.
- Alowayr, A., & Badii, A. (2014). Review of Monitoring Tools for E-Learning Platforms. International Journal of Computer Science and Information Technology, 6(3), 79–86. http://doi.org/10.5121/ijcsit.2014.6306

Table of Content

Abstract

1 Introduction

1.1	Preface of the Area	1
1.2	Research Objectives	3
1.3	Research Questions	5
	Research Contribution or the Novelty of the Research	
1.5	Thesis Organisation	7
1.6	Chapter Summary	9

2 Education System in Saudi Arabia

2.1	Introduction	10
	Education System and Context	
2.3	Culture and Gender in Education in Saudi Arabia	11
2.4	Driving Forces for Technology in Education	12
2.4.1	Current State of Technology in Saudi Arabia Higher Education	14
2.4.2	2 Mobile Learning in Saudi Arabia	15
2.4.3	Importance of English for Saudi Students	17
2.5	Motivation for this Research	18
2.6	Chapter Summary	19

3 Mobile Learning

3.1	Introduction	
3.2	Definition of Mobile Learning	
3.3	Benefits and Opportunities of Mobile Learning	
3.4	Challenges and Barriers of m-Learning	
3.5	Technology Acceptance Model (TAM)	
	Intrinsic Motivation/Engagement	
	Performance Expectancy	
3.5.3	Moderating Effect of Gender	
3.5.4	Perceived Enjoyment	
	Behaviour Intention to Use Technology and Effectiveness	
3.5.6	Framework for Mobile Learning Design Requirement	
3.6	Motivation behind Constructing the M-learning Research Model	
3.6	Chapter Summary	

4 User Experience, Avatar and Learning Theories

4.1	Introduction	. 39
	User Experience(UX)	
	User Experience in Education	
4.2.2	User Experience Design Principles	.41
4.3	Avatars and Mobile Learning Interfaces	. 43
4.3.1	Definition of Avatars	
4.3.2	Avatars in Pedagogy Services	. 45
4.3.3	Avatars and Mobile Learning	. 47
4.3.4	Examples of Virtual Pedagogical Avatars	. 49

4.4	Theories of Learning	. 50
	The Theory of Motivation/Engagement to Learn	
	Learning Styles and Pedagogic Effectiveness	
4.5	Chapter Summary	. 56

5 Research Methodology and Experimental design

5.1 Introduction	
5.2 Research Philosophy and Approach (Layer 1)	59
5.3 Research Methodology Choice (Layer 2)	
5.3.1 Quantitative Methods	
5.3.2 Qualitative Methods	
5.3.3 Mixed Methods	
5.4 Strategies and Tools (Layer 3)	64
5.4.1 Questionnaires	64
5.4.2 Focus Groups	65
5.4.3 Co-creation	65
5.4.4 Experiments	
5.4.5 Grounded Theory	
5.5 Research Design	67
5.5.1 Ethical Approval Procedure	
5.5.1.1 Confidentiality and data protections	
5.6 Pilot Study	69
5.6.1 Subjects Targeted	70
5.6.2 Learning Objectives	72
5.7 Population and Time Horizon (Layer 4)	72
5.7.1 Participants/Sample	
5.8 Experimental - 1 Procedure and Implementation (Layer 5)	74
5.8.1 Procedural Steps	74
5.8.2 Implementation	75
5.9 Data Collection	76
5.9.1 Questionnaires	77
5.9.2 Test phase	
5.10 The Implementation of a Co-creation Workshop (Focus Group)	
5.11 Experimental - 2 Procedure and Implementation (Layer 5)	
5.12 Data Analysis Method	
5.12.1 Quantitative Analysis	
5.12.2 Qualitative Analysis	
5.13 Chapter Summary	

6 The Conceptual Research Model

6.1	Introduction	91
6.2	How the Model is Constructed	91
6.3	The Conceptual Research Model and Hypotheses	94
6.3.1	Variables	96
6.4	The Second Experiment Conceptual Research Model:	
6.5	Chapter Summary	

7 System Implementation (MADE-ME Web-App)

7.1	Introduction	101
	Multi Avatar Delivery Environment for Mobile Education (MADE-ME) Web-	
А	pplication	101
7.3	Development	102
7.3.1	Software	102
7.3.2	2 Access to the Internet	103
7.3.3	3 Location of the MADE-ME web-app Server	103
7.4	User Interface Design	104
7.4.1	Graphic and Information Design.	104
7.5	Use of MADE-ME in Experiment – 1 (Case Study)	105
7.6	Use of MADE-ME in Experiment – 2 (Case Study)	110
7.7	Chapter Summary	117

8 Results of the Study

8.1	Introduction	118
8.2	Recap of Experiments	
8.3	Analysis Overview	119
8.4	Results of Experiment (1) – Questionnaire	
8.4.1	Demographic Results	
8.5	Results of Research Hypotheses – Experiment (1)	
8.6	Results of Experiment (2) – Questionnaire	
8.6.1	Demographic Results	
8.6.2	Participants"Post Questionnaire	
8.6.3	Results of Research Hypotheses - Experiment (2)	
8.7	Qualitative Results	
8.7.1	Responses from an Open-Ended Question of Experiment - 1	
8.7.2	Responses from an Open-Ended Question of Experiment - 2	
8.8	Chapter Summary	

9 Discussion of the Results

9.1 Introduction	
9.2 Discussion of the Research Hypotheses:	
9.2.1 Discussion of Results Related to Experiment 1 (H1) & Experiment 2 (H2)	
9.2.3 Discussion of Results Related to the Experiment 1 (H3) hypothesis	191
9.2.4 Discussion of Results Related to Experiment 1 (H4) & Experiment 2 (H3)	192
9.2.5 Discussion of Results Related to Experiment 1 (H5)	194
9.2.6 Discussion of Results Related to Experiment 1 (H6) & Experiment 2 (H7)	195
9.2.7 Discussion of Results Related to Experiment 1 (H7) & Experiment 2 (H6)	198
9.2.8 Discussion of Results Related to the Experiment 1 (H8) hypothesis	199
9.2.9 Discussion of Results Related to Experiment 1 (H9) & Experiment 2 (H9)	
9.2.10 Discussion results related to Experiment 1 (H10) & Experiment 2 (H8)	
9.2.11 Discussion of Results Related to Experiment 1 (H11) & Experiment 2 (H1	1)205
9.2.12 Discussion of Results Related to the (H4) Experiment 2:	
9.2.13 Discussion of Results Related to the (H5) Experiment 2:	
9.2.14 Discussion of Results Related to the (H10) Experiment 2:	
9.3 Discussion of the Open-Ended Questions	
9.3.1 Discussion of Experiment 1 Open-Ended Comments	

9.4 C	pter Summary	
-------	--------------	--

10 Chapter: Conclusion and Future Works

10.1 Introduction	
10.2 Revisiting the Research Questions	
10.2.1 The Research Objective and Question 1:	
10.2.2 The Research Objective and Question 2:	
10.2.3 The Research Objective and Question 3:	
10.2.4 The Research Objective and Question 4:	
10.2.5 The Research Objective and Question 5:	
10.3 Contributions of the Thesis	
10.3.1 MADE-ME Model	
10.3.2 MADE-ME Web-App	
10.3.3 Case Study	
10.4 Implication of the Research to Saudi Arabian Education	
10.5 The Study's Limitations	
10.6 Future Research	
10.7 Chapter Summary	

11 References

12 Appendices

Appendix A: Research Ethics approval.

Appendix B: Supervisor's letter to the Saudi Arabian Cultural Bureau in London for the data collection.

Appendix C: The Saudi Arabian Cultural Bureau in London approval letter.

Appendix D: The Al-Baha University approval letter to conduct the experiment.

Appendix E: The pre-post questionnaire of the Experiment - 1.

Appendix F: The pre-post questionnaire of the Experiment - 2.

Appendix G: The copy of the exam/test.

Appendix H: A release letter from the main supervisor at Al-Baha University for the completion.

LIST OF FIGURES

FIGURE 1 - 1 AN OVERVIEW OF THE CHAPTER ORGANISATION	9
FIGURE 2 - 1 COUNTRIES WITH THE HIGHEST SMARTPHONES PENETRATION (OUR MOBILE PLANET BY GOOGLE (FOX, 2013)).	14
Figure 3 - 1 Issues relating to access to resources barrier (Jones, 2004)	26
Figure 3 - 2 Relationships between confidence barrier and other barriers (Jones, 2004)	27
Figure 3 - 3 Technology Acceptance Model (Davis, 1985)	30
Figure 3 - 4 Conceptual model for technology acceptance (Davis, 1985)	30
Figure 3 - 5 Proposed Research Model (Maldonado et al, 2010)	31
Figure 3 - 6 Research Framework (Jairak et al., 2009)	32
Figure 3 - 7 Research model of m-learning Determinates (Lowenthal, 2010)	32
Figure 3 - 8 Research model of mobile entertainment adoption (Leong et al., 2013)	33
Figure 3 - 9 Research Model (Aleneziet al. 2010)	34
Figure 3 - 10 A conceptual model of users' satisfaction, behavioural intention, and effectiveness	
toward e-learning (Liaw, 2008).	34
Figure 3 - 11 Consideration for developing effective e-learning (Liaw, 2008).	35
Figure 3 - 12 A framework for M-learning design requirements (Parson et al, 2006)	36
FIGURE 4 - 1 MHCI AS SUPPORT FOR M-LEARNING (BOTHA, 2010)	41
Figure 4 - 2 Avatars in South Korean Cyber Campus (Deeds, 2013)	46
FIGURE 4 - 3 AVATAR MANIFESTATION IN VARIOUS VIRTUAL WORLD SCENARIOS (BARROW, 2009)	47
FIGURE 4 - 4 AVATAR SELECTION AND CUSTOMISATION FOR STUDENTS IN A CLASS (BARROW, 2009)	49
FIGURE 4 - 5 SUMMARY OF MULTIMEDIA LEARNING PRINCIPLES (SEG RESEARCH, 2008)	52
FIGURE 4 - 6 INFORMATION PROCESSING MODEL BASED ON (MAYER, 2005)	55
FIGURE 5 - 1 THE RESEARCH ONION MODEL OF RESEARCH (SAUNDERS & TOSEY, 2012)	58
FIGURE 5 - 2 THE STEPS OF THE SCIENTIFIC METHOD (BRADFORD, 2017)	60
FIGURE 5 - 3 PROCEDURE OF IMPLEMENTING EXPERIMENT (1)	61
FIGURE 5 - 4 CO-CREATION WORKSHOP SESSION MANAGED BY THE RESEARCHER	82
FIGURE 5 - 5 SAMPLE OF PARTICIPANTS PROTOTYPING	83
FIGURE 5 - 6 PROCEDURE OF IMPLEMENTING EXPERIMENT (2)	85
FIGURE 5 - 7 RESEARCH FRAMEWORK FOR CONDUCTING RESEARCH STUDIES	90
FIGURE 6 - 1 STAGES OF DERIVING MADE-ME MODEL	92
FIGURE 6 - 2 CONCEPTUAL RESEARCH MODEL FOR MULTI AVATAR DELIVERY ENVIRONMENT FOR MOBILE EDUCATION	
(MADE-ME) IN SAUDI ARABIA	95

FIGURE 6 - 3 CONCEPTUAL RESEARCH MODEL (MADE-ME) INCLUDING INTERACTION ELEMENTS	98
FIGURE 7 - 1 MADE-ME WEB-APP SERVER LOCATION	104
FIGURE 7 - 2 VISUAL DESIGN AND TECHNIQUE OF THE INTERFACE	105
FIGURE 7 - 3 SCREENSHOT OF CARTOON MOBILE INTERFACE 1	106
FIGURE 7 - 4 SCREENSHOT OF CARTOON MOBILE INTERFACE 2	106
FIGURE 8 - 1 BAR CHART FOR PARTICIPANTS AGE GROUPS	
FIGURE 8 - 2 PARTICIPANTS' MOBILE TYPES OWNERSHIP	
FIGURE 8 - 3 AVERAGE PERIOD OF HAVING SMART MOBILE DEVICE	123
FIGURE 8 - 4 PERCENTAGE OF MOBILE DEVICE DAILY USAGE	124
FIGURE 8 - 5 FEATURES USED BY MALE PARTICIPANTS VIA MOBILE DEVICE CHART (N=78)	125
FIGURE 8 - 6 FEATURES USED BY FEMALE PARTICIPANTS VIA MOBILE DEVICE CHART (N=78)	126
FIGURE 8 - 7 ACTIVITIES RELATED TO LEARNING - MALE PARTICIPANTS CHART (N=78)	127
FIGURE 8 - 8 ACTIVITIES RELATED TO LEARNING - FEMALE PARTICIPANTS CHART (N=78)	128
FIGURE 8 - 9 UNDERSTANDING WHAT DOES M-LEARNING MEAN	129
FIGURE 8 - 10 EXPERIENCE OF USING AVATARS	129
FIGURE 8 - 11 PERCENTAGE OF PARTICIPANTS' GENDER	155
FIGURE 8 - 12 PERCENTAGE OF PARTICIPANTS' FACULTY	155
FIGURE 10 - 1 THE MODEL OF MULTI AVATAR DELIVERY ENVIRONMENT FOR MOBILE EDUCATION (MADE-ME)	219
FIGURE 10 - 2 THE WEB-APP FRAMEWORK OF MULTI AVATAR DELIVERY ENVIRONMENT FOR MOBILE EDUCATION (MA	ADE-ME)

JRE 10 - 2 THE WEB-APP FRAMEWORK OF MULTI AVAIAR DELIVERY ENVIRONMENT FOR MOBILE EDUCATION (MADE-ME)

LIST OF TABLES

TABLE 1 - 1 DEFINITION OF KEY TERMS	3
TABLE 5 - 1 DEMONSTRATION OF RESEARCH PROCEDURE	67
TABLE 5 - 2 VALUES OF CRONBACH'S ALPHA FOR EXPERIMENT 1	70
TABLE 5 - 3 DEMOGRAPHICS PROFILE OF RESPONDENTS	73
TABLE 5 - 4 PROCESS OF DATA COLLECTION WITH GROUP A TABLE 5 - 5 PROCESS OF DATA COLLECTION WITH GROUP B	76
TABLE 6 - 1 DEMONSTRATION OF THE FACTOR SOURCE	93
TABLE 8 - 1 LOGISTIC REGRESSION SYMBOLS	120
TABLE 8 - 2 DEMOGRAPHIC RESULT.	121
TABLE 8 - 3 FEATURES USED BY MALE PARTICIPANTS VIA MOBILE DEVICE	125
TABLE 8 - 4 FEATURES USED BY FEMALE PARTICIPANTS VIA MOBILE DEVICE	126
TABLE 8 - 5 ACTIVITIES RELATED TO LEARNING - MALE PARTICIPANTS	127
TABLE 8 - 6 ACTIVITIES RELATED TO LEARNING - FEMALE PARTICIPANTS	128
TABLE 8 - 7 PERCENTAGES OF TRADITIONAL LEARNING EXPERIENCE	130
TABLE 8 - 8 PERCENTAGES OF MALE FACE-TO-FACE AS A SCREEN CAST TEACHING EXPERIENCE	130
TABLE 8 - 9 PERCENTAGES OF E-LEARNING EXPERIENCE	130
TABLE 8 - 10 PERCENTAGES OF M-LEARNING EXPERIENCE	130
TABLE 8 - 11 PERCENTAGES OF FEMALE FACE-TO-FACE AS A SCREEN CAST TEACHING	130
TABLE 8 - 12 CORRELATIONS BETWEEN QUESTIONS FROM THE QUESTIONNAIRES AND THE SPECIFIC FACTOR'S HYPOTHESIS	131
TABLE 8 - 13 LOGISTIC REGRESSION OF PREFERENCE MODALITY OF LEARNING ON PERFORMANCE EXPECTANCY	133
TABLE 8 - 14 MEAN FOR PERFORMANCE EXPECTANCY STATEMENTS	134
TABLE 8 - 15 UNIANOVA OF LEARNING MODALITY VARIABLES ON ENGLISH LANGUAGE PERFORMANCE EXPECTANCY	134
TABLE 8 - 16 PERFORMANCE EXPECTANCY WHEN USING M-LEARNING WITH MALE STATIC	135
TABLE 8 - 17 MEAN OF GENDER FOR PERFORMANCE EXPECTANCY WHEN USING	135
TABLE 8 - 18 PERFORMANCE EXPECTANCY WHEN USING M-LEARNING WITH ANIMATED AVATAR (VIDEO)	136
TABLE 8 - 19 MEAN OF GENDER FOR PERFORMANCE EXPECTANCY WHEN USING M-LEARNING	136
TABLE 8 - 20 PREFERENCE FOR MULTIMEDIA ON ENGAGEMENT	137
TABLE 8 - 21 GENDER MODERATOR FOR HYPOTHESIS 2	137
TABLE 8 - 22 MEAN OF GENDER MODERATING HYPOTHESIS 2	137
TABLE 8 - 23 PREFERENCE OF FEMALE AVATAR ON ENGAGEMENT (ONLY FOR FEMALE)	138
TABLE 8 - 24 ENJOYMENT BASED ON ENGAGEMENT	139
TABLE 8 - 25 ENGAGEMENT ON PERFORMANCE EXPECTANCY	140
TABLE 8 - 26 GENDER MODERATOR FOR HYPOTHESIS 4	140
TABLE 8 - 27 MEAN OF GENDER MODERATING HYPOTHESIS 4	140

TABLE 8 - 28 MEAN OF MAJOR MODERATING HYPOTHESIS 4	. 140
TABLE 8 - 29 ENGAGEMENT ON PERFORMANCE EXPECTANCY IF THE COURSE IS ENGLISH	.141
TABLE 8 - 30 MEAN OF GENDER FOR THE ENGAGEMENT ON PERFORMANCE EXPECTANCY IF THE COURSE IS ENGLISH	. 141
TABLE 8 - 31 MEAN OF MAJOR FOR THE ENGAGEMENT ON PERFORMANCE	. 141
TABLE 8 - 32 UNIANOVA FOR CONVENIENCE ON PERFORMANCE EXPECTANCY	. 142
Table 8 - 33 Mean of gender moderating hypothesis 5	. 142
TABLE 8 - 34 PERFORMANCE EXPECTANCY ON BEHAVIOURAL INTENTION TO USE M-LEARNING	. 143
TABLE 8 - 35 ENGAGEMENT ON BEHAVIOURAL INTENTION TO USE M-LEARNING	.144
Table 8 - 36 Mean of gender moderating hypothesis 7	.144
TABLE 8 - 37 ENJOYMENT ON BEHAVIOURAL INTENTION TO USE M-LEARNING	. 145
TABLE 8 - 38 ENGAGEMENT ON EFFECTIVENESS (ACTUAL TEST PERFORMANCE)	. 147
Table 8 - 39 Mean of gender moderating hypothesis 9	. 148
TABLE 8 - 40 ENGAGEMENT ON EFFECTIVENESS OF FEMALE AVATAR (ACTUAL TEST PERFORMANCE, ONLY FOR FEMALE)	. 148
TABLE 8 - 41 MEAN OF MAJOR FOR ENGAGEMENT ON EFFECTIVENESS OF FEMALE	. 149
TABLE 8 - 42 PREFERENCE OF MODE OF LEARNING ON EFFECTIVENESS (ACTUAL TEST PERFORMANCE)	. 150
TABLE 8 - 43 EFFECTIVENESS (ACTUAL TEST PERFORMANCE) ON BEHAVIOURAL INTENTION TO USE M-LEARNING	. 152
TABLE 8 - 44 Gender moderator for hypothesis 11	. 153
TABLE 8 - 45 MEAN OF GENDER FOR EFFECTIVENESS (ACTUAL TEST PERFORMANCE) ON BEHAVIOURAL INTENTION TO USE M	-
LEARNING	. 153
TABLE 8 - 46 THE CRONBACH'S ALPHA FOR THE SECOND EXPERIMENT	. 157
TABLE 8 - 47 PREFERENCE FOR MULTIMEDIA ON ENGAGEMENT	. 157
TABLE 8 - 48 GENDER MODERATOR FOR HYPOTHESIS 1	. 158
TABLE 8 - 49 MAJOR MODERATOR FOR HYPOTHESIS 1	. 158
TABLE 8 - 50 MEAN OF MAJOR MODERATING HYPOTHESIS 1	. 159
TABLE 8 - 51 PREFERENCE FOR MULTIMEDIA ON PERFORMANCE EXPECTANCY	. 159
TABLE 8 - 52 GENDER MODERATOR FOR HYPOTHESIS 2	. 160
TABLE 8 - 53 MAJOR MODERATOR FOR HYPOTHESIS 2	.160
TABLE 8 - 54 MEAN OF MAJOR MODERATING FOR HYPOTHESIS 2	. 160
TABLE 8 - 55 ENGAGEMENT ON PERFORMANCE EXPECTANCY	. 161
TABLE 8 - 56 GENDER MODERATOR FOR HYPOTHESIS 3	. 161
TABLE 8 - 57 MAJOR MODERATOR FOR HYPOTHESIS 3	. 162
TABLE 8 - 58 INTERACTIVITY ELEMENTS ON ENGAGEMENT	. 162
TABLE 8 - 59 GENDER MODERATOR FOR HYPOTHESIS 4	. 163
TABLE 8 - 60 MAJOR MODERATOR FOR HYPOTHESIS 4	. 163
TABLE 8 - 61 INTERACTIVITY ELEMENTS ON PERFORMANCE EXPECTANCY	.164
TABLE 8 - 62 GENDER MODERATOR FOR HYPOTHESIS 5	.164
TABLE 8 - 63 MAJOR MODERATOR FOR HYPOTHESIS 5	. 165
TABLE 8 - 64 ENGAGEMENT ON BEHAVIOURAL INTENTION TO USE M-LEARNING	. 165

TABLE 8 - 65 GENDER MODERATOR FOR HYPOTHESIS 6	166
TABLE 8 - 66 MAJOR MODERATOR FOR HYPOTHESIS 6	166
TABLE 8 - 67 PERFORMANCE EXPECTANCY ON INTENTION TO USE M-LEARNING	167
TABLE 8 - 68 GENDER MODERATOR FOR HYPOTHESIS 7	167
TABLE 8 - 69 MAJOR MODERATOR FOR HYPOTHESIS 7	168
TABLE 8 - 70 PREFERENCE FOR MULTIMEDIA ON EFFECTIVENESS (ACTUAL TEST PERFORMANCE)	169
TABLE 8 - 71 GENDER MODERATOR FOR HYPOTHESIS 8	169
TABLE 8 - 72 MEAN OF GENDER MODERATING HYPOTHESIS 8	170
TABLE 8 - 73 MAJOR MODERATOR FOR HYPOTHESIS 8	170
TABLE 8 - 74 MEAN OF MAJOR MODERATING HYPOTHESIS 8	170
TABLE 8 - 75 ENGAGEMENT ON EFFECTIVENESS (ACTUAL TEST PERFORMANCE)	171
TABLE 8 - 76 GENDER MODERATOR FOR HYPOTHESIS 9	171
TABLE 8 - 77 MEAN OF GENDER MODERATING HYPOTHESIS 9	172
TABLE 8 - 78 MAJOR MODERATOR FOR HYPOTHESIS 8	172
TABLE 8 - 79 MEAN OF MAJOR MODERATING HYPOTHESIS 9	172
TABLE 8 - 80 INTERACTIVITY ELEMENTS ON EFFECTIVENESS (ACTUAL TEST PERFORMANCE)	173
TABLE 8 - 81 GENDER MODERATOR FOR HYPOTHESIS 10	173
TABLE 8 - 82 MEAN OF GENDER MODERATING HYPOTHESIS 10	174
TABLE 8 - 83 MAJOR MODERATOR FOR HYPOTHESIS 10	174
TABLE 8 - 84 THE EFFECTIVENESS (ACTUAL TEST PERFORMANCE) ON	175
TABLE 8 - 85 GENDER MODERATOR FOR HYPOTHESIS 11	175
TABLE 8 - 86 MAJOR MODERATOR FOR HYPOTHESIS 11	176
TABLE 8 - 87 BENEFITS AND BARRIERS OF IMPLEMENTING M-LEARNING	

ABBREVIATIONS AND ACRONYMS

- AT: attitude toward
- CAGR: Compound Annual Growth Rate
- CSCL: Computer-Supported Collaborative Learning
- D-learning: Distance Learning
- E-learning: Electronic learning
- HCI: Human computer interactions
- ICT: Information and Communication Technology
- ICT: Information Communication Technology
- LMS: Learning Management System
- MADE-ME: Multi Avatar Delivery Environment for Mobile Education
- MALL: Mobile Assistance Language Learning
- M-learning: Mobile learning
- MOOCs: Massive Open Online Courses
- NCELDE: The National Centre for E-Learning and Distance Learning
- PC: Personal Computer
- PEOU: perceived ease of use
- PU: perceived usefulness
- SA: Saudi Arabia
- SDL: Saudi Digital Library
- SMS: Text Message Service
- STC: Saudi Telecom Company
- TAM: Technology Acceptance Model
- USP: Unique Selling Point
- UTAUT: Unified Theory of Acceptance and Use of Technology
- VLEs: Virtual Learning Environments
- VLEs: Virtual Learning Environments
- WAP: Wireless Application Protocol

1 Introduction

1.1 Preface of the Area

Information and Communication Technology (ICT) has grown tremendously over the past two decades, and this has had a great impact on the education system in general. No longer are classrooms restricted to the teacher-student environment; instead, there is increasing use of technology, to supplement traditional teaching methods. Initially, the use of technology was restricted to the classroom or laboratory environment and, whilst this offered a richer set of teaching resources, the physical limitations of desktop computers meant that students could not access learning materials in a place or location outside of their classrooms or schools (Mupfiga et al., 2017).

Growth in the use of programs for PCs, more affordable laptops, and improvements in communications technology facilitated the development of e-learning, leading to an increase in learner interactions, remote instructors and the delivery of knowledge and content to learners at a time and place convenient to them (Bidin & Ziden, 2013). E-learning courses also provided the opportunity for a social learning environment to be developed, portrayed by participation and interactivity for both learners and teachers (Brindley & Walti, 2009). In the 2000s, virtual learning environments were introduced to facilitate student involvement in the learning process and to provide them with access to a range of educational resources (Guy, 2009), as well as MOOCs (Massive Open Online Courses) as a means of delivering short courses to large cohorts of online learners (Michael, 2016).

Recently there has been a huge growth in the smartphone and tablet market across the world. Forrester research shows that this rate of growth is increasing steeply and by 2019, it is expected to reach 3.5 billion subscribers (Meena, 2014). The percentage of companies that planned to provide m-learning support for their staff rose from 38.5% in 2007 to 51% in 2011 (Quinn, 2011), indicating great potential for mobile applications to have significant penetration in the education marketplace (Meena, 2014). İşman et al. (2015) reported that the revenues generated in the Middle East in 2012 for m-learning products were \$88.3 million, with estimated revenues for 2017 likely to more than double to \$205.4 million.

As a result of this growth, teaching organisations in developed and some developing countries, including Saudi Arabia, are now adopting new teaching methods, informal learning approaches and emerging technologies, such as smartphones and tablets, to support the delivery of learning skills, materials, collaboration, knowledge sharing and lifelong learning (Hawkins, 2016; Bidin & Ziden, 2013). It can be said that m-learning has taken the advantages of e-learning even further by allowing easy access to study resources whilst "on-the-move" as part of blended, informal or distance learning (Ng & Lindgren, 2013; Omale et al. 2010; Nassuora, 2013).

Saudi Arabia is one of the world's fastest growing economies with growth in the education sector and use of ICT being key indicators of this economic development (Alhabeeb & Rowley, 2017; Elyas & Picard, 2010; Almarwani, 2011; Alothmanet al. 2017). To benefit from this growth, the higher education system in Saudi Arabia has been observed to have made significant progress in the implementation and adoption of technologically advanced systems and student learning approaches in response to global shifts towards modern educational methods (Al-Fahad, 2009). Indeed, most Saudi universities have set up e-learning and distance learning "deanships" to integrate the elements of e-learning across the entire educational pathway, from transforming traditional learning methods, to making lessons accessible to all students (Algahtani, 2011), and latterly to a focus on m-learning.

Recent developments in mobile technologies have highlighted their potential for use in formal education. For example, a study by Al-Fahad (2009) on mobile learning technologies and the perception of these technologies, suggests that mobile learning could pave the way to better student retention and understanding, thus becoming a very useful tool that aids education. Nassuora found that despite mobile learning being in the early stages of development in Saudi Arabia, the acceptance rates of m-learning technologies were quite high amongst both students and teachers alike. However, he also stated that full-scale studies of every aspect about m-learning will be needed as m-learning in Saudi Arabia progresses beyond its initial phase (Nassuora, 2013).

While much research has been conducted from various perspectives in mobile learning around the world, as yet, there is a lack of focussed research on the use of avatars for mobile learning globally and particularly in Saudi Arabia, despite studies showing that this might be an important consideration for student engagement and achievement. For example, Ng & Lindgren examined the effects of avatar customisation and narrative on engagement and learning in video games, and have suggested that future studies could look at learners" interactions as a way of achieving excellent teaching and learning outcomes (Ng & Lindgren, 2013), and a study by Falloon showed that the use of avatars could provide for "a powerful, motivating, and educationally valuable learning opportunity" (Falloon, 2010).

The primary aim of this research is to investigate the increasing interest in m-learning from an educational point of view in Saudi Arabia and to ascertain how m-learning can be used as a tool to complement and/or substitute traditional learning environments. Within this general consideration of m-learning, specific attention is paid to the ways in which engagement and performance in learning can be influenced by the type of avatar representation of the teacher on the mobile device, which might be in the form of video, audio, image, cartoon or simple text. In other words, this study investigates how different ways of delivering learning content to students can influence learning outcomes especially when they are away from the traditional classroom. An extension to the traditional technology acceptance model (TAM) (Davis, 1985), is made to describe the interactions between avatar types, student preferences and pedagogic performance for m-learning in a Saudi context.

1.2 Research Objectives

For the purpose of this thesis, key terms are defined as following in the Table 1 - 1.

Term	Definition	Purpose
M-learning	The learning that occurs across multiple contexts, through social and content interactions, using a personal electronic device (Crompton et al. 2016).	Domain of study
Avatar	A visual representation of human characters on the mobile interface in an educational environment (Haake, 2006).	Focus of study
Model	A simplified representation of a real system with any hypotheses required to describe the conceptual system, often statistically. A model concentrates on predicting how factors rely on or influence other factors based on formal	Conceptual solution to problem

Table 1 - 1 Definition of key terms

	relationships that allow the conversion of numerical data into useful information (Adam & Pomerol, 2009).	
Framework	A physical or real representation of the model as layers of an application system that can be implemented with the intension to support the building of something that expands the structure into something useful (Rouse, 2015).	Physical representation of the model
Mobile Web-app	A small specialised program accessed or downloaded onto mobile devices such as iPhones/smart phones, tablets or iPads (Northern Ireland Social Care Council, 2014).	Implementation of the model on digital of the content
Experiment	Division of participants into two or more groups to test the effect of a specific treatment. The group that receives the training or series of runs is called the experimental group (Mertens, 2014).	Case study

To accomplish the main aim of the research, the following objectives apply:

- To identify the main benefits, opportunities and challenges of m-learning when adopted in Saudi Arabia from a students" perspective, and to investigate students" readiness and willingness to use an m-learning approach in their studies within the context of a specific module within their degree courses.
- 2. To investigate the most preferred/engaging avatar representation of a teacher (audio, video, image, cartoon, text) for delivery of learning content via mobile technology.
- 3. To develop an educational model that links delivery of learning content via mobile technologies with pedagogical performance by:
 - a) Setting the variables and factors that align with the research context.

b) Testing a set of hypotheses to determine whether engagement with specific avatar types has a significant impact on pedagogical performance.

4. To design and create the framework for an online web-app that can deliver m-learning content to a mobile device via different avatar representations of the teacher (audio,

video, image, cartoon, text) and which can be used to test the pedagogical effectiveness of each approach by:

a) Constructing different m-learning avatar interfaces.

b) Delivering the content via mobile web-app.

c) Testing students" pedagogic performance of avatar interface type through the mobile we-app.

d) Providing students with the opportunity to co-create and re-design their best mlearning interface framework based on their opinions, preferences and performance.

5. To evaluate the proposed m-learning model and web-app through:

a) Testing the effectiveness of engagement and pedagogical performance through a first stage of experimental design via questionnaires and exam scores.

b) Assessing the second round of experiments based on participants" perceptions through a second stage of experimental design via questionnaires and exam scores.

1.3 Research Questions

Linked to these objectives the central questions of the research are:

- What are the benefits, opportunities and challenges that m-learning can bring to the student population within specific higher education institutions in Saudi? (Objective 1)
- 2) What is the preferred/most engaging way of representing the teacher through an avatar (audio, video, image, cartoon, text) on a mobile device? (Objective 2)
- 3) Is there a significant relationship between students" preference for engagement with particular avatar types and their pedagogic performance; and can such potential relationships be represented on the research model? (Objective 3)
- 4) How can different avatar representations of the teacher be used to develop and deliver learning content via a mobile web-app in order to engage students and improve their pedagogic effectiveness? (Objective 4)

5) What conclusions can be drawn by investigating the links identified through the evaluation of engagement and pedagogical performance in m-learning when the teacher is represented by different avatar types using a cohort of students studying on a degree course at a university in Saudi Arabia? (Objective 5)

1.4 Research Contribution or the Novelty of the Research

This study enhances the field of knowledge related to m-learning via three main contributions which will be described and developed as the thesis progresses, in summary:

- 1- This research develops an educational model, MADE-ME (Multi Avatar Delivery Environment for Mobile Education). This model defines how students can interact with different avatar representations of the teacher to deliver learning content. The model shows the relationships between factors such as engagement, interactive elements, gender, major of study, pedagogical performance, etc.
- 2- This research produces a framework MADE-ME (Multi Avatar Delivery Environment for Mobile Education) web-app that enables a range of avatars to represent the teacher in their purpose of delivering interactive learning content via mobile technologies, and which allows students to be tested on how much they have learnt from the content or lesson.
- 3- A comprehensive case study and investigation of perceptions from student groups studying a particular module in higher education in Saudi Arabia in order to determine how they engaged with the mobile content and how effective their learning was in order to evaluate and validate the MADE-ME model and app. The data was collected by the mixed methods approach: using REGRESSION and UNIANOVA techniques for analysing the quantitative data from questionnaires, and the thematic approach used for analysing qualitative data from the open-ended questions.

1.5 Thesis Organisation

The research takes a scientific approach in order to devise the model and framework that are the key contributions. The thesis has been structured with the following chapters:

Chapter 2 discusses the Saudi Arabian education system, including the impact that culture and gender have on teaching delivery. The importance of introducing technology into higher education in Saudi Arabia and its current use in learning is discussed, as is the importance placed on the teaching of English to Saudi students.

Chapter 3, further introduces the concept of m-learning, and reviews literature related to the benefits and opportunities afforded by m-learning as well as some of the current challenges and barriers associated with it. The chapter also introduces the Technology Acceptance Mode (TAM) and some of its derivatives as a starting point for the development of a model to describe acceptance of m-learning in Saudi Arabia and the motivation for investigating this.

Chapter 4 covers topics related to user experience and how good design can impact on the use of technology for education. The chapter then discusses what avatars are, their origins and how they can be used as part of the learning process, some of the benefits of avatars and examples of avatars in a learning context are also given. It also introduces the integration of Human Computer Interactions (HCI) components and multimedia into the interface design and how these affect/influences by the learning theories and styles.

Chapter 5 reiterates the research aim and questions, and describes the research methodology and experimental design of the studies undertaken during the course of the research. An explanation of the research methodology used is provided, as well as the research approach, ethical considerations and data collection techniques which include pre and post questionnaires/surveys for the first and second studies and co-creation workshop with students in order to incorporate other collaborative and interactive elements.

Chapter 6 describes how the different m-learning technology acceptance models introduced in Chapter 3 can be combined and extended to develop a new pedagogical

model constructed specifically for this research to align m-learning in a Saudi context with human mobile interaction. This model included factors that might affect and optimise the m-learning for English content delivery. It also develops a number of hypotheses to be tested regarding characters in the model such as engagement, enjoyment, convenience, gender, interactive multimedia, performance expectancy, effectiveness and behavioural intention to use m-learning.

Chapter 7 describes the development and implementation of the m-learning webapplication (,,web-app") framework using avatars which were developed to represent the teacher and deployed by using specific software for the selected institution to understand how well students can engage with the m-learning environment using avatars. Moreover, this web-app was also designed to enable the testing of how much a student has learnt from the content or lesson. In addition, this web-app is used to validate the relationships between factors based on the research model.

Chapter 8 presents the findings of the data collected. Qualitative and quantitative analyses are conducted to provide a better perspective of the problem and the current state. The m-learning application using avatars is also evaluated based on a pedagogical perspective. Quantitative data are analysed by using SPSS version 21. The open-ended questions and workshop comments are analysed by adopting the thematic analysis procedure.

Chapter 9 discusses and interprets the findings obtained from the data analysis and also presents the links between the results and the research hypotheses. In addition, consistencies and differences with other studies" findings are covered in this chapter. The research questions are revisited and the theoretical contribution for implementing this approach to learning is discussed.

Chapter 10 highlights the main contribution of this research. The chapter covers a summary of the research model and its implication to the field. It also summarises the developed web-app framework and draws together the recommendations for implementing m-learning in Saudi higher education that been found from the research findings. Lastly, it high lights the limitations and challenges the researcher encountered, and suggestions for further research areas are then presented.

1.6 Chapter Summary

This introductory chapter has presented a brief background to the area of study outlining the research problem, the current situation of m-learning in Saudi Arabia, the aim and its objectives together with the research questions that the study will address. Figure 1 - 1 provides an overview of the chapters" organisation.



The organisation of this thesis is shown in the following diagram.

Figure 1 - 1 An overview of the chapters organisation

The next chapter provides the background and research context for education in Saudi Arabia.

2 Education System in Saudi Arabia

2.1 Introduction

There are various aspects that need to be considered while implementing advanced technologies to assist m-learning in higher education institutions. These aspects not only include the technological concerns but also relate to educational theories, acceptance of the technology, and cultural aspects. Care should be taken to understand the proposed scenario from these perspectives in order to establish how technology can be used to enhance the learning experience. This chapter sets the context of the research by describing the Saudi education system and important factors associated with it that might influence the parameters and focus of the research.

2.2 Education System and Context

The educational system in Saudi Arabia includes a variety of paths to cover the different needs of students. Compulsory education comprises six years of primary school, three years of secondary school, and is followed by three years of high school (Ghaith, 2013).

With regards to education after high school, students can progress to higher educational programs through either studying at public or private universities. In Saudi Arabia, the number of universities has grown over the past decade in response to the increasing population and an era of oil boom (Abubakar et al. 2016). There are currently 52 universities and colleges in Saudi Arabia, including 24 public universities, 8 private universities and 20 private colleges (Alfarani, 2016; MOE, 2015). The public universities are coordinated and funded by the Ministry of Higher Education (Abubakar et al. 2016). The government of Saudi Arabia has allocated a huge budget to this Ministry, recently investing around \$57.9 billion in education, which represents 25 percent of the country's appropriations (Ministry of Finance, 2015). These universities provide a variety of disciplines including, Medicine, Engineering, Computer Science, Sciences, Business Administration and Arts. All curriculums of these mentioned disciplines must be taught in the English language.

The government of Saudi Arabia is looking to improve the quality of teaching and learning in its universities, not only by improving traditional teaching methods, but also by improving learning through provision of various technologies ,increasing flexibility of learning, promoting a better learning atmosphere, improving cognition, enhancing the learning experience, and encouraging students to be more interactive and to actively participate in class through the adoption of new learning technologies including those to support online learning (Al-Fahad, 2009). The advantages of adopting technologies such as mobile learning in higher education for students have been well reported in previous studies, for example, (Al-emran et al. 2016; Aldhaban, 2016) and will be considered in the context of this thesis in Chapter 3 alongside potential challenges. It should be mentioned here, however, that although one of the barriers to implementing m-learning in developing countries is often the high cost of owning a device which is reliable and has multimedia functions to enable learning, Saudi Arabia has one of the world's richest economies and students get paid a monthly allowance while they study in higher education and hence can readily afford such devices (Alfarani, 2016).

2.3 Culture and Gender in Education in Saudi Arabia

Saudi Arabia has a gender-segregated education system which reflects the traditional and cultural expectations of its society. Studies show that the demand for good education in Saudi Arabia is high, and that use of technology is also generally high driven in part by this gender based segregation (Alharthi et al., 2017; Alothman et al., 2017). Further and higher education institutions are using technology to provide teleconferencing and video conferencing to promote the mobility of women. These technologies play a key role in the education sector of Saudi Arabia as they promote and facilitate the acceleration of access to good education without a gender bias (Baki, 2004). However, some researchers argue that although there are these steps taken by the education sector to bridge the gender gap, there needs to be a significant change in the adoption of technology in order for access to education to reach all students and in order to promote an environment where there is significant participation amongst students (Al-Alwani, 2005; Baki, 2004). Additionally, some academic staff members feel that the use of technology does not provide enough incentives (Naveedet al., 2017), leading to some researchers identifying that there is a lack of enthusiasm to implement learning technologies by educational institutions and a lack of resources and trained staff to oversee the implementation in order to harness the benefits of these technological innovations (Al-Alwani, 2005; Naveedet al., 2017).

Gender segregation and the cultural expectations of society in upholding its beliefs and traditions have a profound impact on the education system in Saudi Arabia. Saudi Arabia has a distinct cultural and intellectual stance when compared to Western countries. Various traditions, beliefs and behaviours affect the implementation of technology in the educational sector, and these are quite different to Western culture (Alfarani, 2016), for example, the demographic variables (such as gender) influenced by the cultural traditions and norms of the country. The perceived usefulness of technology and the role technology plays in the education sector are also different when compared with Western counterparts (Hsu, 2013). While implementing technologies for education, there is a significant need for developers to understand these cultural perceptions and to ensure that the design and development of such technologies suit the needs and wants of the learners (Baker et al., 2007). Many researchers support this theory of planned behaviour, for example, Alenezi et al., (2010) and Seliaman & Al-Turki (2012) argue that it is imperative that cultural and social norms are taken into account while developing technology in order to improve its perceived usefulness. Technologies can be successfully implemented, if and only if, they are perceived to be useful by the endusers. For this implementation to be achieved, it becomes mandatory that the end-users can relate well to the technology and that it is visually appealing. The visual appeal of technologies can be achieved when users are presented with an environment which they are used to and which does not offend their beliefs (Al-Alwani, 2005). Therefore, whilst the use of available technologies in an educational field could help to overcome cultural issues and increase pedagogical performance, this can only occur if attention is paid to the users requirements.

2.4 Driving Forces for Technology in Education

In the modern era, the use of technology cannot be understated. Technology plays a key role in teaching and learning. Implementation of good technologies helps overcome various learning barriers and also provides a means to increase student interaction and participation (Brown, 2000). Many researchers support this theory and also argue that the use of technology in classrooms for the purposes of learning needs to be functional and reflect the ,real world^{**} as opposed to just traditional learning (Ghaith, 2013).

Education in Saudi Arabia is one of its fastest growing sectors and is also a key indicator of the economic development of the region (Alothmanet al. 2017). Given the growth rate of this sector, there is tremendous spending potential and universities and other higher education institutions are competing to provide better quality education that is accessible to both female and male students (Alharthi etal, 2017). From a Saudi Arabian context, technology plays an even more important role in the education sector as universities are finding it increasingly difficult to be able to employ skilled professionals to mentor students. Hence, universities are keen on implementing technologies such as smartphones to support the talented lecturers they employ and to help maximise the impact they can have on the quality of teaching (Alhabeeb & Rowley, 2017). With Saudi Arabia being the fourth largest user of the internet and other related technologies in the Arab world, this environment helps to leverage the use of technologies in an educational context (Al-Ghaith et al. 2010; Alothman et al., 2017).

Thus, technologies that promote education and provide opportunities for students to learn at their own pace and also use their creativity have now become very common in the country. Virtual learning environments that help collaboration between students and teachers through technology, such as smart phones, are also becoming a necessity, as opposed to a luxury for higher education institutions, and despite reservations by some teachers, students and other stakeholders universities are subsequently keen on looking at implementing technologies that would help students maximise their potential and help achieve a vibrant learning environment (Al-Fahad, 2009; Almalkiet al. 2013).

One key reason behind the motivation to study m-learning in Saudi Arabia is the rate of penetration of mobile technology in the country and correspondingly the growing potential to tap into the technology and leverage it for use in the academic field. Smartphone penetration is set to rise to 84% in Saudi Arabia by the end of 2016 among the very highest anywhere in the world according to Fox(2013), a figure backed up by historic data and from the Saudi Arabia Consumer Electronics Report 2010, where mobile phone sales accounted for around 22% in the year 2009 with expected the sales growth at the Compound Annual Growth Rate (CAGR) of 7% by 2014 (Report Linker, 2010 as cited in Chanchary & Islam, 2011), and from Our Mobile Planet statistics which show that roughly three in four people in Saudi Arabia own a smartphone, giving the

country one of the highest smartphone penetrations in the world (Fox, 2013) as shown Figure 2 - 1.



Figure 2 - 1 Countries with the highest smartphones penetration (Our Mobile Planet by Google (Fox, 2013)).

2.4.1 Current State of Technology in Saudi Arabia Higher Education

Studies have shown that the Saudi Arabian government and its educational institutions have made investments in terms of finance, time and resources to enhance the education system in the country (Alenezi et al., 2010). The government provides students with technological resources and computer tablets in order to motivate them to pursue education. Most students are provided with tablets, PCs and laptops including access to the internet in order to help them with their education (Alfarani, 2016). This engagement of the government and higher education institutions in Saudi Arabia to harness technological resources and promote a good education system has led to a significant increase in the number of students attending schools and taking up higher education (Alenezi et al., 2010). However, despite Saudi Arabia being categorised as one of the wealthiest countries from an economic point of view, its education environment is still considered to be that of a developing region (Alothman et al., 2017). Correspondingly, the question that arises is why, despite the enormous budget and resources that the Saudi Arabian government invests on developing education in the country, is the standard of education still low?

Many universities in Saudi Arabia constantly use technologies to aid day-to-day teaching. For example, video conferencing is used to facilitate male professors and lecturers teaching in female universities (Ismail et al. 2016). However, the lack of infrastructure in Saudi higher education is a major factor delaying further development, and may even decrease the quality of education in the country. The existence of generic smart technologies are gaining popularity in Saudi Arabian higher education institutions, but the usage of these technologies is small compared to higher education institutions worldwide (Al-zahrani, 2011). Also, Alenezi claims that whilst universities are willing to invest in tech-savvy tools and programs that would be appreciated by both the teachers and students alike, flexibility and improvement of skills are two main areas that universities must consider when implementing such technologies (Alenezi et al., 2010).

2.4.2 Mobile Learning in Saudi Arabia

Knowledge of the growth of e-learning within Saudi Arabia is important for understanding the background to m-learning. In educationally developing countries, such as Saudi Arabia, the evolution of e-learning in higher education has grown steadily (Jairak, Praneetpolgrang, & Mekhabunchakij, 2009). In the 1980s, e-learning started to appear in Saudi universities as an alternative learning approach to traditional face-to-face learning (Chanchary & Islam, 2011). Due to the rapid expansion of technology, personal computers became the most important tools of learning (Al-Fahad, 2009; Alshumaim & Alhassan, 2010), and in many cases, developments in technology have allowed students to study and communicate with teachers and peers through asynchronous tools in terms of distance learning or 'd-learning' (Al-Fahad, 2009).

Growth in technology has also given rise to a number of open universities in Saudi Arabia where e-learning platforms are common, and these open universities make use of e-learning technologies and platforms in order to support their distance education learners (Ismail et al. 2016; Alothman et al., 2017). The increase in education using dlearning can be described as a transitional step from PC learning to learning through mobile devices. Mobile learning was revealed in the 1990s with the successful development of Bluetooth and WAP (Wireless Application Protocol) which enabled students to access course materials 'on the go' (Nassuora, 2013), and which has subsequently developed to provide assisted teaching resources to support the students, helping them to learn at their own pace, revisit material that they are not very comfortable with, and provide them with the flexibility to learn from any location. The concept of mobile learning is therefore not something new in the Saudi Arabian context. There are several applications for mobile devices which have been used at schools, working environments and in everyday life (Huang et al. 2010), and the Saudi government has funded a number of m-learning projects at many of the local universities, including Saudi Electronic University, King Imam Muhammad University, and King Abdul-Aziz University (Badwelan et al. 2016). Moreover, it has invested in a number of related infrastructure sub-projects, such as:

- The National Centre for E-Learning and Distance Learning (NCELDE).
- The National System for the Management of E-learning (JUSUR).
- The Learning Management System (LMS).
- The Saudi Digital Library (SDL).
- The Medicine Program at Qassim University (Almarwani, 2011; Badwelan et al. 2016).

These projects and systems show the rigorous infrastructure for online learning in the country which should support and encourage further integration of m-learning to take place.

In recent years, a number of studies have focused on learners" perceptions towards mlearning, such as users" acceptance of the technology and the impact of applying different m-learning approaches on students over traditional face-to-face learning (Abual-aish & Love, 2013; Alfarani, 2016). Given the high acceptance rates for mobile phone-related technologies in Saudi Arabia researchers believe that the acceptance rate will also be high for the implementation of such technologies in an educational context (Al-Fahad, 2009; Nassuora, 2013). However, an investigation into the acceptance of mlearning in Saudi Arabia by Seliaman and Al-Turki (2012), based on the extensive Technology Acceptance Model (TAM), concluded that, although there is a possibility of technology acceptance and higher perception of usefulness across Saudi Arabian staff and students, there are limitations when the technology does not offer innovation or use the state-of-the-art tools to develop a mobile learning platform (Seliaman & Al-Turki, 2012). Seliaman and Al-Turki also believe that there is a gap in using mobile learning technologies alongside regular education in Saudi Arabia.

2.4.3 Importance of English for Saudi Students

English is one of the most important languages in the world, and subsequently the teaching of English as an International language is becoming an increasingly important module in the universities of many developing countries, including those within Saudi Arabia. English is frequently the communication medium in fields such as education and business. Research from around the world shows that cross-border business communication is most often conducted in English (Geng & Chunling, 2017; Luo & Shenkar, 2017). Many opportunities in international regions and markets are created when learners have a working knowledge of English grammar. Its importance in the global market place therefore cannot be understated, and speaking and reading English opens up opportunities of job employment. It is therefore not surprising that English language is the dominant language of business and commerce within Saudi Arabia as well as the main language of instruction for university courses within the country (Almarwani, 2011).

Consequently, in Saudi Arabia, the English module is a core course to help prepare students for their instruction in English, and it must be completed and passed with a minimum mark of 60 percent by all undergraduate students in their first year of study in all universities (Sedgwick, 2001).

However, when trying to adopt a new technology in education, particularly in Saudi Arabia, there are some factors, such as cultural norms, which can affect the flow of uptake (Almarwani, 2011). The country is following gender-segregation of campuses whereby male instructors are not allowed to teach female students face-to-face. As a result of the gender policy, there is also a lack of female instructors but not of male instructors. Male instructors can therefore teach female students, through only via a closed circuit television system (Algahtani, 2011). In addition, male students cannot meet with female students or directly exchange their knowledge and experience.

Therefore, it is possible that m-learning can be used to more effectively support the current situation, which would help all students in general and female students

specifically, regardless of the segregation, so that both genders can access the same learning materials and achieve more positive outcomes.

2.5 Motivation for this Research

From this review it can be seen that Saudi Arabia is well placed both technologically and economically to implement m-learning within its universities, with the potential for enhancing the quality of its teaching as well as improving access to learning for both genders. The approach of teaching courses online has also been shown to be costeffective enabling more students to be taught with fewer tutors than the number required for traditional face-to-face teaching (Wisneski et al. 2017).

One area where m-learning might have considerable potential is in the teaching of English language. A number of previous language studies focus on the use of mobile devices in different educational settings, including: listening skills, learning new vocabulary via SMS messaging, pronunciation, students" perceptions and students" usage (Eppard et al. 2016), however, there are few empirical studies about the use of mobile devices for teaching language grammar. The importance of this as an area within the context of this research is backed up by the performance and reviews of students from Al-Baha University. At Al-Baha University, students study 90% of their course modules and exams in English. Despite the extensive hours required for the English language module, there is a notable lack of student understanding about the English language in general, and in particular about grammar structures. Although importance is given to this module for university students, their levels of English language knowledge remains very low as noted while conducting this study (the average score of all students in the placement exam was only 28%. In addition, due to the extensive hours required for teaching this course, students frequently complain about having to attend a high number of lectures and they have indicated their desire to learn at their own time and place. Hence the concept of the adoption of m-learning as a way of delivering learning content needs to be investigated in order to tackle this issue.

Investment in developed and developing technologies and, Information and Communication Technologies (ICTs) to serve education in general and the provision of English learning in particular, may in turn improve the level of academic performance and enhanced the field of business.

2.6 Chapter Summary

This chapter has reviewed the educational system in Saudi Arabia, including the cultural aspects and how new technologies feed into this unique culture. This chapter helps to understand the cultural and the educational landscape in the country which needs to be kept in mind while designing and implementing new technology. For technological implementation to be successful, ensuring that the cultural values of the end users are respected plays a major role in determining how well the users are willing to engage with the technology. This chapter also described the introduction of technology into the Saudi education system and has identified the teaching of English language as a potential area that could benefit from the adoption of m-learning.

In the next chapter, the benefits and challenges associated with mobile learning and its introduction in the education systems are discussed along with models that assess acceptance of technology.

3 Mobile Learning

3.1 Introduction

Chapter 2 discussed m-learning within the context of the Saudi Arabian education. This chapter considers the concept of m-learning further and in particular discusses the benefits and opportunities afforded by m-learning as well as some of the challenges and barriers to its adoption. The chapter then looks at how acceptance of m-learning can be measured through the use of the Technology Acceptance Model (TAM) (Davis, 1985), and its derivatives in preparation for enhancing the model further within the context of this research.

3.2 Definition of Mobile Learning

Researchers define mobile learning (m-learning) as the process of learning in various contexts and social interactions by making use of personal electronic mobile devices such as iPhones and Android devices connected to the internet, thereby enabling students to access learning materials (Baek & Touati, 2017; Crompton et al. 2016).Mlearning extends the advantages of e-learning by allowing easy access to study resources whilst "on-the-move" as part of blended, informal or distance learning through the use of mobile technology (Alebaikan, 2010; Park, 2011). M-learning can make learning more accessible, flexible, personalised and attractive as learning can be carried out anytime and anywhere, enabling students to study at their own convenient time and place (Lu, 2013; Helen, 2013; Singh & Reed, 2001). In relation to the current research context, m-learning refers to the use of mobile technologies as a way of delivering learning content to the learner in order to enhance their learning experience through having a representation of the teacher on their mobile device. While e-learning and mlearning can be used as a substitute for traditional face-to-face learning, many researchers believe that both e-learning and m-learning are more effective as an assistance to, or extension of, traditional teaching methods rather than as new independent and isolated tools (Alhassan, 2016). The increasing use of mobile and wireless technologies, and the potential for such technologies to make learning accessible to a wider range of individuals, thus creates great opportunities and exciting new avenues for blended learning.
The concept of blended learning is becoming increasingly popular, not only in the academic world but also in the corporate world. Blended learning can be defined as the combination of online learning and classroom learning on the same platform (Alfarani, 2016). Other definitions focus on the use of a variety of instructional media together with instructional modalities in order to effectively deliver a course module (Manwaring et al., 2017; Vaughan et al. 2014). Mobile devices have become more influential due to the presence of all types of gadgetry that offer the potential to bring different learning methods together (Al-Fahad, 2009). Researchers found evidence of m-learning being a highly significant approach to use for blended learning with learners able to access learning materials at anytime and from anywhere (Alhassan, 2016; Lu, 2013). Given that blended learning modules use technology and various instruction methodologies, it is becoming increasingly used in higher education environments and is gaining a lot of momentum in distance learning to deliver courses effectively (Thomson, 2002; Ng & Lindgren, 2013; Omale et al. 2009; Wiecha et al. 2010). Huang et al.'s study for example, indicated that by providing undergraduate students with facilities, content instruction and information which is available outside of classrooms, distance learning is becoming more acceptable among educational instructors (Huang et al. 2010).

3.3 Benefits and Opportunities of Mobile Learning

M-learning technologies offer a relatively new paradigm of learning that takes into account the experiences of the learner and which enable the learner to approach the lesson at their own pace. Research conducted by Naismith et al. in 2004, discussed the technologies that would help establish a good m-learning environment. Indeed, there are various advantages of an m-learning approach that no other methods of learning can compare with according to Perrin et al. (2006). Recently, Osakwe et al. conducted a comprehensive review on the features and usefulness of mobile technology and how it can be integrated into the educational field (Osakwe et al., 2017). Several studies have suggested that this type of learning environment is best suited to informal learning, including distance learning, part-time learning and open universities, as opposed to formal, physical learning environments (Park, 2011; Seifert, 2014). Perrin et al. (2006), suggest a number of factors which predict the success of m-learning such as: convenience, engagement, interaction, increases motivation, collaboration and

compatibility, which overlap with other identified benefits and advantages of involving mobile technology in learning including:

Portability: due to the size of mobile devices, one of the key advantages of mobile technologies in a learning environment is that the learner can carry them to different locations, making it easy for students to learn out of the classroom (Botha et al. 2010; Romrell et al. 2014). Evans (2008) also assumed that the reason behind students preference for podcasts over more traditional forms of learning was their portability. Additionally, according to Alharthi et al., (2017), the portability of m-learning can "automate" the process of teaching and help in managing the increasing number of students enrolling for higher education in Saudi Arabia.

Convenience: the convenience of learning through a mobile device comes from the flexibility it gives and from overcoming the restriction of needing a specific place and fixed set of time for learning. Several studies have revealed the convenience of using mlearning technologies which enable people to be in contact while outside the reach of conventional communication and learning spaces. For example, m-learning is argued by Guy (2009) to be a method whereby learners from across the globe can facilitate sharing, collaboration and access to data and study materials online via smartphones and other such devices, irrespective of physical barriers such as location and time zone. Mobile technologies have the ability to provide learning materials 24 hours a day and 7 days a week via online connectivity (İşman et al., 2015). However, although overall opinions are positive, Corbeil & Valdes-corbeil (2007) and Zhang and Aikman (2007) claim that the effectiveness of the technology can only be achieved, if and only if, the adopting education institutions have a thoroughly thought out plan to use the technology in addition to conventional teaching.

Repeatability: m-learning allows students to pause and replay desired parts of the information when necessary (Farsi, 2016; Stoicescu & Stanescu, 2015). This advantage of repeatability is unique to online learning over traditional face-to-face learning.

Collaboration and interaction: mobile technology has significantly improved in facilitating easier uploads and downloads of pedagogical materials and enabling collaboration between students worldwide, practises previously restricted in e-learning

environments. Singh et al. (2001) described the use of m-learning applications in combination with traditional resources in the classroom environment as the key factor for increasing learners" participation and collaboration. Furthermore, research has found that using m-learning technologies in a classroom environment leads to increased understanding of concepts, facilitates smoother communication, fosters a collaborative environment and helps continuing education outside the classroom environment (Huang et al., 2010). Learners can also communicate and interact more readily with their instructors and their colleagues, thus solving the obstacles of hiding behind large monitors. Al-Fahad (2009) found that mobile learning will bring new opportunities of learning, and other studies show that learning models in informal environments not only enable educators and educational institutions to share information with relative ease, but also to provide assessment and feedback for the courses and the ability to provide flexible collaborative means of learning (Asiimwe & Grönlund, 2017; Huang et al., 2010; İşman et al., 2015).

Engagement: the evolution of e-learning and m-learning technologies, their advancement to distance learning and informal learning approaches, and the subsequent adaptations of these technologies by higher education institutions are promoting motivating and stimulating study environments (Alhassan, 2016; Huang et al., 2010; Jeng et al., 2010). Algahtani (2011) reported that, the use of online learning with different multimedia elements would provide an enjoyable, interactive and motivational learning environment for students. Recent research claims that, the use of mobile devices in education stimulate motivation, strengthen engagements and deliver content (Asiimwe & Grönlund, 2017, p104). The overall belief is that both e-learning and m-learning gives the instructor a powerful medium through which to control the content, deliver a lecture in a more effective manner and engage students better in the learning process.

Effectiveness: several attempts have been made to provide effective methods to assist learners in the real world and to increase students" achievements through development of m-learning activities (Lai et al., 2016). For instance, the results of research by Hung et al, (2014) showed that, there was a positive learning attitude of the students towards the use of mobile learning when a video-based approach used. The incorporation of mobile technology and pervasive learning can enhance the effectiveness and

accessibility of learning activities. Recently, Kukulska-hulme revealed significant effectiveness though using mobile devices alongside traditional learning techniques in the teaching and learning of the English language (Kukulska-hulme, 2015). In addition, Leong et al. (2013) concluded from an experiment investigating the factors and variables that affect students intentions to use m-learning that perceived usefulness, perceived ease of use and perceived self-efficacy were positively correlated with students" intention to use m-learning.

Freedom: mobile technology meets a variety of students" needs and learning styles (Badwelan et al., 2016). This type of learning provides students with the freedom to choose how they learn or study. Furthermore, studies have shown that learners can also benefit from visualising their learning path and consequently reflecting on their learning experience (Guy, 2009). It is the potential for mobile learning to bridge pedagogically designed learning contexts, facilitate learner-generated contexts, and deliver content (both personal and collaborative), while providing personalisation and ubiquitous social connectedness, that sets it apart from more traditional learning environments (Cochrane, 2010).

In summary, mobile technologies represent a "coming of age" for m-learning with their incorporation of features such as in-built video players for displaying presentations, the portability of the devices, acceptable screen size, easy connectivity, large memory and battery capacity, the ability to support multimedia content and the ability to scan information (Eppard et al. 2016). Furthermore, as mobile technology improves, it becomes synergistic in nature, as the development and growing use of mobile technology emerges on the back of each former improvement to facilitate and enhance learners" collaboration and interaction by means of accessing, discussing and sharing related data via social networks within and across their educational environments (Jeng et al. 2010). There is also extensive research underway on how wireless technologies and hand-held devices could significantly lead to an educational breakthrough by creating a Computer-Supported Collaborative Learning (CSCL) environment that fosters "on-the-move" learning capability available to a wide range of learners globally, potentially surpassing any language or geographical barriers (Alfarani, 2016). Thus, as higher education students continue to depend on technological innovation in and out of

the classroom, it is critical that instructors discover approaches to address their needs (Annetta & Holmes, 2006).

3.4 Challenges and Barriers of m-Learning

Although the previous section focused on the positive advantages and benefits of mobile learning, there also exist challenges and barriers which might delay the adoption of mlearning. A number of researchers believe that inadequate planning processes for implementing e-learning and m-learning environments lead to its failure (Asiimwe & Grönlund, 2017; Bingimlas, 2009; Alhassan, 2016). There are also certain practical issues or barriers whilst implementing advanced technologies for education. For example, although avatars and Virtual Learning Environments (VLEs) provide a great platform for blended learning, when implementing mobile learning platforms, they pose certain restrictions. These restrictions are predominantly due to the limitation posed by mobile hardware and technologies, as despite a tremendous growth in the personal and handheld computing domain, the 3D technology and types of avatars sometimes used, require sophisticated design and high memory space and processing capacity to be able to display the contents effectively (Sohn et al., 2005). Additionally, with the integration of technology in the curriculum or as a part of assisted teaching, there is a paradigm shift in the teaching methodology. This, therefore, puts a lot of pressure on the teacher to be able to use the technology with ease. Furthermore, the design of m-learning interfaces and their usability is another potential barrier for both students and staff to their use. Based on the literature, other challenges and barriers that may significantly impact on the implementation of m-learning include:

Lack of the faculty members' technical expertise: one of the main obstacles of adopting m-learning is the poor experience of the lecturer with regards to designing lessons for use on a mobile device from both technological and pedagogic perspectives. Therefore, the lack of proper training to use the technology can be the major restriction to using the technology (Seifert, 2014; Corbeil & Valdes-corbeil, 2007; Zhang & Aikman, 2007). According to (Al-Azawei et al. 2016), staff resistance to any new experience is a challenge which hinders the online learning uptake. For example, Alabdulaziz & Higgins, (2016) found that the lack of training for teachers and the lack of technical

support on how to use technology effectively were major obstacles for adoption of the technology in teaching to overcome the mathematics difficulties in Saudi Arabia.

Lack of resources (hardware & software): according to Alhassan (2016), the poor reliability and usability of some mobile devices due to their small screen size and limited memory capacity may negatively affect their use of m-learning. This restriction however is increasingly becoming less of an issue as device screen sizes are getting steadily larger and more processing capacity is being added to the newer smartphones and tablets. Hyland (2010) discusses these issues from an access and resourcing point of view and argues that lack of adequate infrastructures to plan and implement the technology, and provision of good software prevents teachers from leveraging the technology to receive maximum benefit and at times provides a negative experience to the students.



Figure 3 - 1 Issues relating to access to resources barrier (Jones, 2004)

Jones has classified the key barriers related to implementing technologies as shown in Figure 3 - 1. For example, poor quality or lack of hardware are reflected in a lack of access to the learning content (Jones, 2004). Having the right resources and the appropriate software to make use of the technology can make the difference between success and failure of the technology implementation (Ghavifekr et al. 2011; Hong & Songan, 2011). This is backed up by recent research which shows how poor infrastructure and limited resources, for example unreliable internet connections, affects access learning materials (Asiimwe and Grönlund, 2017).

Financial abilities: this includes the costs of having reliable smartphones and network connectivity billing. Further to these limitations, the lack of internet connectivity especially for some students who may not be able to afford subscriptions to faster internet tariffs is an issue (Asiimwe and Grönlund, 2017; Bingimlas, 2009; Chanchary and Islam, 2011; Perrin et al., 2006). This is however an issue at individual student level rather than in institutions or universities in Saudi Arabia.

Lack of confidence: Jones (2004) also identified lack of confidence in the use of the technologies by the teacher as potentially having a detrimental effect on the pedagogy, as shown in Figure 3 - 2, a factor agreed by Choy et al.(2009); and Seifert (2014). This barrier against the use of technology in learning and teaching may be due in part to the advanced age of some teachers coupled with a lack of knowledge regarding its benefits (Alabdulaziz & Higgins, 2016).



Figure 3 - 2 Relationships between confidence barrier and other barriers (Jones, 2004)

Less motivation: Ghavifekr et al. (2011) proposed that one of the key factors that can have an adverse effect on the uptake on m-learning is the teachers or students motivation. Al-Azawei et al. (2016), found that 47% of staff and 75% of students" were unwilling or demotivated to use e-learning because of other mentioned challenges. Therefore, significant effort should be placed on extending their academic understanding of the potential impacts of m-learning on different learning aspects such as learning engagement and pedagogical performance.

Poor management procedures: other challenges in using mobile learning environments are more concerned with the education systems and the management approaches that are

used to introduce mobile learning platforms such that they integrate effectively with the existing learning platforms (Naveed et al., 2017). Educators and academic institutions may lack formal procedures and management capabilities to roll out complex systems. For successful implementation, universities and other higher education institutions should therefore have in place management infrastructure along with good ICT resources to support teachers in coping with changes that ensue from the implementation of new technologies and also provide a smooth transition. For teachers and students to be able to appreciate the technology fully (Al-Azawei et al. 2016), they should also effectively implement policies and understand the implications of implementing mobile learning (Kukulska-Hulme et al., 2009).

Poor interactions with peers: Alhassan (2016), found one of the challenges that significantly affects the implementation of m-learning is the lack of interaction between the user and the designed interface. Corbeil & Valdes-corbeil (2007) added that a feeling of isolation or being ,out of the loop'' might occur for those not interested in the technology as opposed to those who embraced it.

Sections β .3 and β .4 have identified a number of benefits and opportunities, and challenges and barriers that can have an influence on the acceptance of the technology. Various researchers, for instance, Annetta & Holmes (2006), Deuchar & Carolyn (2003) and Falloon (2010) have also shown interest in the field of m-learning and have similarly identified both opportunities and challenges. It can be concluded that, acceptance of m-learning is a complex field with instructor, confidence, infrastructure, motivation, financial and institution management all being important dimensions that influence the use of m-learning systems. Another key challenges is the interaction of the students with the avatar online learning interface used to deliver the pedagogical content which may draw their attention and engagement to the material. The challenges that involve identifying areas of development and implementation of mobile learning interfaces in an educational scenario are described in the forthcoming chapters.

3.5 Technology Acceptance Model (TAM)

One of the aims of this research is to investigate students" readiness and willingness to use an m-learning approach in their studies and alongside this to determine their preferred avatar interface representation of the teacher and the interface which assists them most effectively with their learning. It is thus important to be able to model the factors that affect their acceptance of the technology and their performance when using it. Acceptance of technology depends on the perceived usefulness of the proposed technology and the attitude of the end-users towards the technology. The concept of "technology acceptance" began in the late 1970s as a result of many information technology projects failing on the basis of insufficient knowledge by the people using the system (Tabak & Nguyen, 2013). Thus, it became important to predict how well the users would embrace the technology. The Technology Acceptance Model (TAM) was therefore developed to understand if there would be acceptance or rejection of a particular technology (Davis, 1985). Davis predicted that user motivation was dependent on the stimulus obtained from the features of the system being developed and its capabilities. The main three variables of the TAM are , perceived usefulness' (PU), , perceived ease of use' (PEOU) and , attitude toward' (AT) or intention to use the new technology. Furthermore, another critical success framework to be considered for implementing new technology is the Unified Theory of Acceptance and Use of Technology (UTAUT) by Venkatesh et al. (2003)determines which also the factors that influence the students" intention to use a technology. The theory of TAM and UTAUT provide theoretical bases and attempt to empirically compare factors from different technology acceptance models. The remainder of this chapter reviews the literature associated with the TAM and other learning models that have been derived from it with particular emphasis on m-learning acceptance factors.

3.5.1 Intrinsic Motivation/Engagement

In the original Technology Acceptance Model (TAM), the motivation of the users is discussed from three different perspectives, namely, (1) the perceived usefulness of the system that is to be deployed, (2) the perceived ease of use of the system proposed and (3) the users" intention towards using the proposed system (Davis, 1985) as depicted in Figure 3 - 3. The perceived usefulness/performance expectancy of the system is defined

as the level or degree to which the user believes that the system deployed would enhance their current performance while carrying out certain specific tasks.



Figure 3 - 3 Technology Acceptance Model (Davis, 1985).

Subsequently, the attitude and behavioural intention towards using the system is influenced by the perceived ease of use and perceived usefulness. The perceived ease of use also has a direct influence on perceived usefulness.

The perceived ease of use of the system is defined as the degree or level to which the user believes that the system that is to be implemented would require less physical and/or mental efforts in order to perform regular and routine tasks (Davis, 1985). A concept enhanced by Davis as shown in Figure 3 - 4 to incorporate the view that a system's functionalities act as the stimulus to the user who then becomes motivated to use the system with the perception of usability seen in the actual response of using or rejecting the system that has been deployed.



Figure 3 - 4 Conceptual model for technology acceptance (Davis, 1985)

A study "e-learning motivation and educational portal acceptance in developing countries" by Maldonado et al. (2010) empirically modified the UTAUT model by adding an "e-learning motivation" factor and determination of the significance influence of this factor on the use of e-learning (see Figure 3 - 5). The study was conducted in South American with a survey as the main instrument used to collect data from 150

students from 47 different schools. The findings revealed that the motivation variables had a positive influence on behavioural intention.



Figure 3 - 5 Proposed Research Model (Maldonado et al, 2010)

Motivation of students was influenced by their performance, and caused an increase in their engagement, which in turn, affected their behavioural intention to use the system (Maldonado et al. 2010). In addition, Maldonado et al., (2010, p70) claimed that "people will carry out an act only when the desired outcome is to be attained, or they will perform an action that is of value to them".

3.5.2 **Performance Expectancy**

Within the context of this research the factor of performance expectancy is equal to perceived usefulness. The definition of perceived usefulness can be adopted from Daviset al.(1989) as "the degree to which a person believes that using a particular system would enhance his or her job performance". According to Osakwe et al. (2017, p21), performance expectancy and its relationship to mobile technologies is such that "performance expectancy advocates mobile learning will be found useful because it enables individuals to have quick access to information, any time and any place, and on their preferred device". Learners" acceptance of m-learning is a major factor when planning to design a successful m-learning application.

A study conducted by Jairak et al. (2009) used UTAUT based upon TAM and involved 390 higher education students learning through m-learning. The results revealed that performance expectancy, effort expectancy, and social factors have statistically



significant positive relationships with the behavioural intention to use that technology, but the "facilitating conditions" factor did not have a direct relationship Figure 3 - 6.

Figure 3 - 6 Research Framework (Jairak et al., 2009)

In 2010, Lowenthal investigated the UTAUT model factors in a study with 113 higher education students. The results here also highlighted that the effort expectancy and performance expectancy have statistically significant influences on behavioral intention to use an m-learning system. In addition, gender and age as moderators indicated that no statistically significant differences were found that play a role in the acceptance of m-learning. Figure 3 - 7 illustrates the research model of m-learning determinates.



Figure 3 - 7 Research model of m-learning Determinates (Lowenthal, 2010)

3.5.3 Moderating Effect of Gender

A study conducted by Leong et al. (2013) found that the perceived enjoyment factor had a significant impact on mobile entertainment acceptance in Malaysia. Leong et al. (2013) also used gender as a moderator and investigated gender statistically with respect to all other variables to understand how it influences the adoption of m-entertainment between users and found that there were no significant differences between gender as illustrated in Figure 3 - 8.



Figure 3 - 8 Research model of mobile entertainment adoption (Leong et al., 2013)

From a number of studies it can be noted that male students consider the perceived usefulness/performance expectancy more than female students (Koohang, 1989; Ong and Lai, 2006; Shashaani and Khalili, 2001; Leong et al. 2013). However, studies by Gefen and Straub (1997), and Al-emran et al. (2016), found differences suggesting that females perceived higher degrees of usefulness or performance expectancy than males when using computer technology. Taleb and Sohrabi (2012), found that female students have used mobile phones in education more than male students and claime that girls are more skilled in using multimedia, whilst boys are more skilled in advanced communication via using mobile phones.

3.5.4 Perceived Enjoyment

Another study was conducted by Liu (2008), who extended the UTAUT model by adding perceived enjoyment, mobility, attainment value, self-efficacy, and self-management factors in order to investigate learners behavioural intention to use m-learning. Perceived enjoyment is defined by Fosso (2017, p3) as "the extent to which an activity is perceived to be enjoyable in its own right, apart from any performance consequences that may be anticipated". In order to promote learner motivation and engagement, it is important to make learning activities highly enjoyable. There is thus a direct relationship between the enjoyment/happiness and the intrinsic motivation to use a system (Leeet al., 2005; Baek & Touati, 2017). High levels of perceived enjoyment

while undergoing the learning process is a high indicator that students will use that type of technology and will accept it (Aleneziet al. 2010) (see Figure 3 - 9).



Figure 3 - 9 Research Model (Aleneziet al. 2010)

3.5.5 Behaviour Intention to Use Technology and Effectiveness

Some of the studies described above present a link between the acceptance and engagement factors toward the intention to use e-learning, and some show a correlation between the behavioural intention to use that technology in learning and its influence on effectiveness/pedagogical performance (Liaw, 2008). Furthermore, the adopted conceptual model proposed by Liaw includes a number of external variables such as e-learning multimedia instruction as one of the major predictors of e-learning performance and motivation among students see Figure 3 - 10.



Figure 3 - 10 A conceptual model of users' satisfaction, behavioural intention, and effectiveness toward e-learning (Liaw, 2008).

In addition, multimedia instruction variables play a significance role on perceived usefulness/performance expectancy and perceived satisfaction.

Thus, when looking at the model of facilitating effective e-learning (see Figure 3 - 11), it can be noted that there are three fundamental e-learning components which need to be considered when designing effective e-learning: learners" self-efficacy, multimedia formats, and interaction environments (Liaw, 2008). Noteworthy, it is a likely assumption that factors which affect e-learning will be similarly affecting the m-learning.



Figure 3 - 11 Consideration for developing effective e-learning (Liaw, 2008).

3.5.6 Framework for Mobile Learning Design Requirement

Mobile device use has expanded over recent years and many researchers, such as Brown et al.(2006) and Massey et al. (2006), have argued about whether these devices can enhance learning experiences. Brown et al. mapped qualitative data to the conceptual mobile-learning framework proposed by Parsonet al. (2006) that provides systematic support for mobile learning experience design. In addition, this model showed the relationship between the framework factors and m-learning design requirements, and suggested how m-learning applications can be designed with an understanding of these factors and requirements (see Figure 3 - 12).



Figure 3 - 12 A framework for M-learning design requirements (Parson et al, 2006)

Parsonet al., (2006) showed in the above framework that there are four m-learning design requirements: generic mobile design issues, m-learning context, learning experience and learning objectives; and that these factors can engage the learner, facilitating self-motivation and self-regulation which in turn can improve students' learning situations (Brown et al., 2006). Important factors of this framework in relation to the PhD objectives include mobile interface design, media types, and activities to be undertaken by learners when considering outcomes and feedback towards improving engagement, performance and achievement from using m-learning. This framework provides a clear view to be taken in account when designing m-learning contexts which are different to traditional learning in class.

3.6 Motivation behind Constructing the M-learning Research Model

With the development of smartphone devices and the associated functions and facilities they provide, the main focus of this research is on their use within an education environment. In spite of the rapid development of m-learning applications in recent years, studies on the extent to which using different modalities on the mobile device affect attention, motivation and learning performance has seldom been studied (Chih-Ming & Chung-Hsin, 2015). Modality or the mode of content delivery can refer to the use of the most appropriate multimedia, such as video, audio, graphs, etc., for the potential teaching interface. Furthermore, m-learning is a relatively new concept which can help foster human interaction to create interest in learning, thereby improving delivery of courses and increasing the overall understanding of subject matter as it enables students to study at their own pace. Improvements in mobile technologies and

the availability of reliable mobile carriers and wireless networking has provided a great opportunity for both students and educators to improve autonomy of learning (Osakwe et al., 2017; Massey et al. 2006).

Despite a number of studies in the published literature showing the significant benefits of m-learning, it is important to gain an understanding of why some students do not engage with m-learning. According to an investigation by Jairak et al. (2009), the factors that influence the adoption of m-learning are the major principles which should be considered when planning to invest in m-learning. Models investigating individuals" acceptance and intention to use new technologies have been examined by Gefen and Straub (1997), Venkatesh & Morris (2000), Badwelan et al. (2016), and Al-hunaiyyan et al. (2017). Another researcher found that although more than 50% of the learners in his study had no experience with m-learning, they had high levels of acceptance and readiness to engage with m-learning and they had a good perception of how to integrate their learning via mobile devices (Nassuora, 2013). Abu-al-aish & Love (2013) claim that there are a number of issues that may prevent the adoption of m-learning, such as learners not being ready to use m-learning as it is still a new concept to them and hence may require a lot of effort to be implemented.

Abu-al-aish & Love (2013) conducted a study with 174 participants from Brunel University based on the unified theory of acceptance and use of technology (UTAUT) and found that performance expectancy and other factors significantly affected the behavioural intention to use m-learning. Consequently, students" performance expectancy with m-learning affects their performance outcome and hence adoption of an m-learning model is an urgent requirement that needs to be understood.

Saudi researchers in particular are interested in the concept of blended learning and how the implementation of m-learning technologies can be used to assist the traditional learning environment (Alebaikan, 2010; Alfarani, 2016). Previous research has shown that the majority of mobile devices can be integrated into the field of education (Alhassan, 2016; Nassuora, 2013). A number of studies have been focused on employing m-learning in Saudi Arabia, especially with regard to the prospects and challenges that universities and students face whilst implementing the technology (Bingimlas, 2009; Chanchary & Islam, 2011). Although there is a rich understanding of the m-learning area, there are several gaps linked with the effectiveness of this method of learning which require more investigation (Wisneski et al., 2017). One such gap, cited by Laureano-cruceset al. (2016), is the need to study the impact of interfaces and ways of delivering learning materials over the mobile device. Also, from a review of the literature, few studies have examined the influence of higher education students" preferences for learning based on specific multimedia instruction, and students" engagement with the learning process, and their behavioural intention to use m-learning, hence it appears that little work has been done on how students engage with m-learning delivery mechanisms and how to assess what they have learned through mobile devices. Of particular interest is the potential of creating an environment for real-time interaction between learners and the content through mobile devices and to investigate whether there is a connection between students" engagement with/preference for a particular multimedia type and their pedagogic performance.

Therefore, one of the contributions of this research is to develop a model for m-learning that encompasses a range of different multimedia types for use in Saudi Arabia's higher education system. The model enables implementation of good design and user experience practices and incorporates the main elements of m-learning design requirements, such as the design principles, mobile context, Saudi Arabian culture, student learning experiences, how to achieve the best learning outcomes, and how to show whether these factors are linked and influence each other.

3.6 Chapter Summary

This chapter has considered further the concept of m-learning, and in particular it has discussed its key benefits and features, and how they are correlated to education. It also covered the barriers and challenges which might affect the implementation of mobile technology in education. Further, it has discussed how acceptance of a new technology and its influencing features can be modelled.

As a consequence, one of the interesting areas to research further is how mobile interfaces between physical and online learning can be investigated in order to support learning in Saudi Arabia. Another important aspect associated with the development of technology and its acceptance is the user experience and the interaction between learners and the representation of the teacher. This will be discussed in the next chapter.

4 User Experience, Avatar and Learning Theories

4.1 Introduction

Chapter 3 shows that m-learning has become a reality which makes learning available anywhere and anytime, providing opportunities for enriching the learning experience and enhancing learning motivation and preference. One way to ensure that m-learning is effective and that it creates a positive user experience is to improve the design of the user interface by including good use of interactive elements, multimedia components and clear avatar representations of the teacher to engage the learners. This chapter considers the relationships between user experience design, avatar based user interfaces and learning theories and styles.

4.2 User Experience (UX)

A mobile learning approach can be used to deliver learning materials to students, but the materials must be designed properly to compensate for factors such as the small screen display. According to Adham et al., (2016) and Hodhod, (2010), engaging learners and enhancing their motivation to learn and achieve the desired learning outcomes can occur through supplying a graphical interface that shows the representative of the teacher on the online learning environment. Within the context of this research, the m-learning interface must be adjusted to prevent information overload and it should coordinate the interaction between the user and the learning material staking into account any technology limitations. "User experience (UX)" as a concept has evolved gradually over the last two decades, with increasing focus being given to both the design and the user journey, while developing software and technology solutions. There has been a paradigm shift in the way software development takes place, which has moved from focusing primarily on functionality to making the design and approach more user-centric, with the aim of concentrating on the needs of the user rather than on just the functionality of the system (Vermeerenet al. 2010).

The effectiveness of software or a system is enhanced when the user has a good experience. There are various key factors that affect the perceived usefulness and user experience of the system. This perception is affected by the satisfaction of the user from using the software. For the end users to be satisfied, the environment provided by the

system should be stimulating, enhancing and provide an opportunity for learners to engage, interact and involve themselves with the learning materials (Kim et al., 2013). In addition, features of interaction and immersion are important elements to motivate and engage students to learn through virtual learning environments and influence their academic performance (Nguyen et al., 2017; Huang et al., 2010).

In many cases, user experience acts as a distinguishing factor between the success and failure of a software system. When a system concentrates only on the product development cycle to deliver maximum functionality without keeping in mind the robustness and usability of the design, it leads to the failure of the software product. In many cases, user experience makes a product stand out and becomes the product"s Unique Selling Point (USP) which in turn makes users want to try out the software as well as creating a sense of perceived usefulness (Nguyen et al., 2017; Väänänen-Vainio-Mattila, 2008).

4.2.1 User Experience in Education

There are various tools and technologies that are used to promote student interaction and involvement in higher education. Huang et al. (2010, p1179) showed that "interaction is a crucial factor to affect learning performance". Moreover, various distance learning programs use state-of-the-art technologies in order to increase student satisfaction and learning outcomes (Stoddart et al., 2016). Assisted teaching techniques rely on the fact that the whole experience of the student is pleasant and this psychologically gives the student the motivation to interact and engage in the learning process (Machado and Tao, 2007). As proposed by Botha (2010), end-users or "learners" often indicate frustration with the technology as being a major barrier to the use of and participation in technology enhanced learning systems. Consequently, components that might impact on the discourse between the domain requirements and technology affordances, are explored in the field of Mobile Human Computer Interaction (MHCI). Botha (2010) has illustrated the relationship between MHCI and mlearning as shown in Figure 4 - 1.



Figure 4 - 1 MHCI as support for m-learning (Botha, 2010)

Botha (2010) highlights the link between student satisfaction in distance education and assisted learning, and the use of user-centric design principles that can increase the users" interaction with the system and allow them to communicate effectively despite geographic and other barriers. When good user-centric design approaches are followed and industry best practices are used while developing the system, care is taken to ensure that the end product is easy-to-use, user friendly and the entire end user journey is elevated to a higher level (Allen & Bourhis, 2002; İşman et al., 2015).

It can be seen that when user experience design is highly developed, students find the process of education interesting and retention of information is made easier (Huang et al., 2010). Studies with distance education environments that use technologies to assist with learning and which provide a conducive learning environment show that students do not find distance education a barrier anymore, as they feel that the technology facilitates their interaction and makes them feel a part of the system (Machado & Tao, 2007). A number of previous studies have primarily concentrated on how learning outcomes can be improved and how strategies can be implemented in order to blend traditional learning approaches with mobile learning technologies (Vaughan et al., 2014).

4.2.2 User Experience Design Principles

There is a plethora of literature that describes various views on what good user experience design principles are, for example, (Arnold et al.2016; García-Peñalvo and Durán-Escudero, 2017). Most researchers, however, believe that user experience design should not just be related to the engineering process but should also involve good strategies from both a management and organisation perspective. While designing a software solution, the management of the learning institution should actively ensure that

the proposed designs are in line with industry standards and best practices. The cultural norms of the organisation environment also play a key role in determining the design attitude and approach that is taken during the design process both of which can have a great impact on the end design (Alenazi, 2015).

Researchers argue that the best way to create a good user experience is by keeping the design simple and elegant (Botha et al., 2010; Nielsen, 1995). This ensures that the complexities of the design are eliminated and that the user journey is smooth and enjoyable. In addition to simplicity, the design should also take into account Human Computer Interaction (HCI) factors and the design principles that affect the usability of the system and the perception of its usability in the eyes of the end user (Shneiderman & Plaisant, 2004; Nielsen, 1995). When addressing the effectiveness of receiving learning materials in various media formats (e.g. text, graphics, video) and in the right proportions, designers must also keep in mind the goal of the system and the target users of the system (Bates, 2015). Care also should be taken to ensure that any cultural aspects of the target users of the system are considered and that these cultural restrictions and requirements are met. Poor understanding of these concepts will likely lead to the failure of the project and non-acceptance of the system by the end user (Gulliksenet al. 2003).

Games are highly motivating in nature and contain constructive aspects, such as fantasy, control and curiosity that engage the learner. Some researchers believe that game-based theories and approaches for system design can be used, even in cases of non-game based contexts and solutions. Game theory encompasses using various elements of games, such as avatars, characters, video, sound, text and pictures, incorporating them in a traditional application in order to enhance the experience and increase the likeability of the application (Mazlan, 2012). Game theory strongly focuses on user interaction and user inputs to improve the involvement of the user (Ng and Lindgren, 2013). It may also create a virtual world whereby the user can connect to a specific game character, and thus increase the users" interest in the environment and in the system. Implementing game theory concepts includes enriched designs for heuristics and pattern combinations, providing opportunities for interaction and finally making the user comfortable with the virtual reality system (Deterdinget al. 2011).

4.3 Avatars and Mobile Learning Interfaces

Teaching organisations are now adopting new learning methods, informal learning approaches, emerging technologies such as Virtual Environments (VEs), and social applications of m-learning that support collaboration, knowledge sharing and lifelong learning. In some developing countries, mobile phones provide the main communication network available to the public in general and, as such, represent an indispensable channel for expanding accessibility to learning opportunities. This section introduces the concept of avatars and highlights the effect they can have on online learning environments. Virtual Environments (VEs) are defined as artificial environments generated by computers using various graphical technologies to create a perceived sense of artificial reality amongst the users. They may also include artificial, computergenerated humans placing themselves in a virtual two or three dimensional space in a visual manner, as opposed to users having a physical presence (Haake, 2006; Deuchar & Carolyn, 2003). There can be the ability to sense the presence of other people and one's co-presence among them, in a "face-to-face" classroom setting, however, this ability is generally limited to the dates and times during which a particular class is scheduled.

Educators and researchers now believe that learning can happen anywhere given the technological advancements and the paradigm shift in the learning environment (Lu, 2013; Singh & Reed, 2001). The technological landscape has undergone a complete revamp and increasing numbers of advanced technologies are being developed at a previously unimaginable pace. This leads to educators wanting to harness the power of technology in order to provide better pedagogical services to students. The use of state-of-the-art technology has now become more of a requirement in today"s competitive world as higher education institutions and universities becoming increasingly and more interested in acting as a technology enabler for students (Johnson et al. 2015; Thomson, 2002).

The use of human representative technologies, for example by displaying instructor characters as avatars, is becoming increasingly popular in learning, as researchers find that there is a strong positive correlation between the use of avatars and the cognition of the student (Fox et al., 2015; Omale et al., 2009), due to the provision of a positive

social experience and promotion of the social presence of the learner (Mazlan and Burd, 2011). Although the use of avatars or the technology itself contributes to the cognitive presence of the learner, it also increases the involvement of the participants in the learning process which in turn has a positive impact on the cognition of the learner (Omale et al., 2009). The results also show that using avatars in an e-learning environment greatly improves the quality of the learning environment and enhances the teaching and delivery of the subject matter. Avatar-based learning environments and social games thus provide important resources for learning and can help foster an active, effective learning process (Vasalou et al., 2017).

Virtual worlds and avatars provide a means to improve the quality of education, and especially distance education (Lu, 2013), through the provision of avatars that support and enhance the online learning environment. The characteristics of such environments, as described by Lu (2013), include:

- Virtual representation whereby the avatar character is used to portray the teacher or the mentor of the real-world.
- User-generated content, where learners and teachers alike can generate content and actively participate in the discussions. This social attribute fosters and nurtures participation and interaction amongst the participants.
- User interaction where users are able to contribute or interact with the system, pause and learn at their own pace, unlike with traditional methods of teaching.
- Autonomy which gives the learner the power to learn without 'real world' interruptions and provides the flexibility for students to study at their convenient time and place.
- Rich Media Environment (RMA) which includes the use of media such as sound, video, animation, etc. to enhance the learning experience.
- Easy access for authorised users, which enables learners to view and re-view content any number of times to get a better understanding of the topic, unlike with traditional learning environments.

4.3.1 **Definition of Avatars**

Avatars have been defined by Haake (2006) and Chae et al. (2016) as a visual representation of human characters in an educational environment. In another study, Peterson (2006) expressed avatars as an online appearance of signs in a virtual world that are intended to improve communication and interaction in a virtual environment. Deuchar & Carolyn (2003, p255) defined avatars as a channels that allow the user to take an online and a visible persona in the virtual environment and provides them with "the opportunity to engage in surreal and imaginary experiences that transcend the actual world in which they live". Avatars can be defined as pictorial representations whereby human participants are allowed interaction thereby providing a face-to-face e-learning environment between students and the instructors (Biocca & Harms, 2002).

4.3.2 Avatars in Pedagogy Services

Advanced developments in technology and the digital revolution has meant that virtual reality and avatars can now be used across different industries and for different purposes to support collaborative working environments and to support corresponding behavioural shifts and thought patterns (Yoon & Vargas, 2014). The concept of avatars and virtual reality in the context of increased interaction, accessibility and better cognition has been studied by various researchers (Antonacci et al. 2008; Brekel, 2007; Greenwald et al., 2017). Avatars, according to these researchers, have the potential to help overcome the difficulty of face-to-face interactions, crossing the limitations of geographic and language barriers. In doing so, avatars enable the free-flow of communication amongst participants in the group and enhance their social interaction by bringing learners together using a common platform and providing them with the flexibility to stay connected yet anonymous (Adham at al, 2016).

Avatars in a pedagogical perspective can be thought of as characters or figures that can be created and customised as per the needs of the learner. Avatars often give full control of the character and features of the character to the end user. For example, end users are given the option to change its colour, physical features, and appearance details, depending on personal preferences, the simplicity that it provides and the interest that it spurs amongst learners and facilitations within an interactive working environment (Al-Alwani, 2005). The use of avatars in pedagogical services has been presented across world conferences and round tables. Cyber campuses have been used by universities across the globe. Deeds (2013) shows the prototype of avatars used by a South Korean University's cyber campus Figure 4 - 2.



Figure 4 - 2 Avatars in South Korean Cyber Campus (Deeds, 2013)

The research by Deeds (2013) also shows that avatars in many cases help to bridge language barriers in pedagogical services and can be implemented in environments where the teachers and students do not speak a common language.

Barrow (2009) explained that avatars can be used in a wide variety of contexts in education and can provide an engaging platform for learners to participate in engaging learning, and as a means to increase the creative streak in students and pique their interest in the learning space. Figure 4 - 3, shows various online manifestations of avatars (Barrow, 2009).



Figure 4 - 3 Avatar manifestation in various virtual world scenarios (Barrow, 2009).

Most previous studies, for example (Annetta & Holmes, 2006; Falloon, 2010; Ng & Lindgren, 2013), have investigated using avatars in virtual space for informal communication and diversion purposes, however, there is still a lack in the use of avatars in a more formal educational context to optimise the learning and engage students within the framework of learning models and styles.

4.3.3 Avatars and Mobile Learning

This thesis focuses on the uses of avatars in m-learning technologies. The m-learning concept using avatars is gaining a lot of popularity and research conducted over the past two decades has studied its implications, advantages and disadvantages (Kukulska-Hulme et al. 2009; Omale et al. 2009; Adham et al., 2016). Significant developments in game technology and the rapid acceptance of mobile technology has led researchers to consider integrating these technologies in order to effectively leverage the advantage afforded by their combined potential in order to provide an enhanced learning environment (Sharples, 2002).

There are researchers who concentrate on understanding the m-learning environment using avatars in computer games, and specifically in online games e.g. "Massively

Multiplayer Online Role-Playing Games" (MMORPGs) (McManus, 2002; Mazlan 2012). Avatars play an important role in presenting an e-learning lesson by drawing learners' intention and making learning more interesting, personal, and attractive (Chae et al., 2016). Previous studies have primarily concentrated on m-learning environments and the use of avatars from a psychological perspective, considering how such environments can be used for study and research purposes and the implications of using such technologies (Yeeet al. 2007). Social impacts and the impacts of using such technologies are also discussed by researchers (Mitchellet al. 2007), Mitchellet al., for example, primarily focus on the social needs, structures and implications of using such technologies from the point of view of society. M-learning using avatars is still in a relatively early development stage. Research has shown that learners" perceptions towards avatars affects the ways they interact with the system and increase the participation intention in an online learning environment (Blascovich, 2002; Chae et al., 2016).

Avatars, gaming technologies and virtual worlds have always been successful in capturing the attention of young people. The use of avatars in electronic/mobile learning has been taken up by researchers after the success of avatars in computer games (McManus, 2002; Mazlan, 2012). There is a lot of research around adapting specific game techniques and avatars such as CLEV-R (Hodhod, 2010; Jones, 2016; Ng & Lindgren, 2013) and the 3D environment on various platforms, including smartphones, using an interactive game technique approach. The development of this platform is in its initial stages but researchers believe that this project when developed could help students by keeping them engaged and sustaining their interest (Jones, 2016). An avatar can be an important device for presenting m-learning lessons. Adding a face to a module can add interest and motivation for learners. The avatar can also present material in more conversational tones' while the appearance of avatars and the environment, along with their interactions, can affect people's sense of presence in the metaverse as virtual reality learning environments (Annetta & Holmes, 2006; Mazlan, 2012; Greenwald et al., 2017).

When implementing avatars in mobile learning environments, there are various considerations that must be taken into account. These include not only the cost and technological considerations but also the social, psychological and cultural aspects.

These perceptions and ideas need to be catered for when developing avatars in order for the end users to wholly be involved with the avatar learning environment and actively participate in the learning process. Only when there is active participation of the end-user is the usefulness of the technology wholly realised (Omale et al., 2009; Adham et al., 2016).

Studies show that avatars need not just represent the instructor in an educational setting. For children, interest in the subject can be piqued when they are allowed to choose a ,,cool" image and an avatar that they feel represents themselves. This kind of learning environment enables teacher-student communication, and effective tracking of work progress of the student in real-time is made possible. Such communication and activity-based learning can also be used in lower grades to help children concentrate and participate in the class. The activities that the child is involved in, and behavioural aspects that the child shows, can also be recorded and monitored for review by parents as well as teachers as shown in Figure 4 - 4 (Barrow, 2009).



Figure 4 - 4 Avatar selection and customisation for students in a class (Barrow, 2009)

4.3.4 Examples of Virtual Pedagogical Avatars

The fundamental purpose behind including virtual pedagogical avatars in educational applications is their potential to generate engagement with and motivation for learning (Haake, 2006). For example, Falloon (2010, p120) found that using avatars in an educational environment can allow for "a powerful, motivating, and educationally valuable learning opportunity". According to Haake (2006), there are an increasing number of pedagogical avatars that had been used for educational systems, for example,

as instructors, virtual teachers, presenters or guides with the avatar taking the form of a cartoon, a series of still photographs, a 3D character, or a video of an actor (Chae et al., 2016) (see Figure 4 - 5 to Figure 4 - 9).



Figure 4 - 8 Avatars were used to communicate specific messages to target audiences (Falloon, 2010).

Figure 4 - 9 Cartoon avatar

4.4 Theories of Learning

This section aims to link interface design and multimedia content learning theories. The section discusses how the motivation to learn, universal design for learning (UDL) and different, learning styles can be combined in order to create effective pedagogic performance.

4.4.1 The Theory of Motivation/Engagement to Learn

The motivation to learn has been defined in previous studies in a variety of ways, including by MacIntyre et al. (2001) who defined motivation as "an attribute of the individual describing the psychological qualities underlying behaviour with respect to a particular task", and by Huang et al. (2010, p1179) who defined it as an "internal state or condition that activates, guides, and maintains or directs behaviour". Alongside these definitions are suggestions as to why technology can increase motivation to learn. For

example, Guo and Goh, (2015) found that students" engagement and learning motivation is increased when they interact with learning materials through the technology interface; Alabdulaziz & Higgins (2017) reported an increase in student motivation relating to the use of interactive whiteboards in a very interactive student-directed way; and Mazlan (2012) stated that interactivity, feedback and multimedia elements (audio, graphic, video, animation and text) are the significant factors of learners" motivation to learn. Mazlan (2012) also reports that the use of avatars in an online environment might contribute to the motivation of learning. This is due to the elements of the avatar and its ability to represent human characteristics which make the learning more realistic.

Other strategies which help learners to be more engaged and increase their motivation to learn are provision to the students of immediate self-assessment and tests frequent feedback about their learning progress as well as the use of personalised and focused multimedia content to enhance student learning as shown in Figure 4 - 5 (SEG Research, 2008). Research by Han & Finkelstein (2013) strongly supports the idea of the feedback intervention in online learning environments which in turn affect students" learning performance.



Figure 4 - 5 Summary of Multimedia Learning Principles (SEG research, 2008).

These principles also support two important theories which are the Universal Design for Learning (UDL), and Universal Design of Instruction (UDI). Both UDL and UDI focus on how online learning materials can be designed to enable all learners to participate and make the learning outcomes achievable regardless of their abilities, language skills and learning styles (Bühler & Fisseler, 2007). UDL and UDI support the design of instruction and learning such that it is "usable by all students... without the need for adaption or specialized design" (Burgstahler, 2015, p2). Connell et al. (1997), list the principles of universal design for instruction and learning in the context of online learning as following:

1- Equitable use: the design is useful and marketable to people with diverse abilities.

2- *Flexibility in use*: the design accommodates a wide range of individual preferences and abilities.

3- *Simple and intuitive*: use of the design is easy to understand, regardless of the user's experience, knowledge, language skills, or current concentration level.

4- *Tolerance for error*: the design minimizes hazards and the adverse consequences of accidental or unintended actions. Automated tests within an e-learning course should not

only give feedback whether an answer is wrong or right, but also why an answer is wrong (Burgstahler, 2015).

Interestingly, such principles have overlaps with the 10 heuristics of Nielsen (1995) and the 8 golden rules of Shneiderman (2004) for interface design, with selected heuristics as follows:

- *Visibility of system status:* the system should guide the user by providing feedback within good time.
- *Match between system and the real world:* the system should be designed with the learners" language.
- User control and freedom: support undo and redo.
- Consistency and standards: follow platform conventions
- Flexibility and efficiency of use: allow users to tailor frequent actions
- *Recognition rather than recall*: instructions for use of the system should be visible or easily retrievable whenever appropriate.

Thus, principles of UDL & UDI can be applied to the design of e/m-learning and educational technology. Because these principles aim to improve access to learning based on learning styles, digital media can play a significant role in achieving access through the presentation of learning information via a range of media types such as text, images, audio and video as presented in the next section.

4.4.2 Learning Styles and Pedagogic Effectiveness

Learning styles in the context of this thesis may be defined as "the manner in which individuals perceive and process information in learning situations" Brown (2000). There are various learning styles that affect the way students" learn, which may be effective for some learners but not for others (Hawk & Shah, 2007). Most researchers categorise learning styles based on the medium used for teaching, learning and the possibility for improvement in learner performance. A number of researchers, such as Farsi (2016) and Lowenthal (2010) believe that learning styles influence and promote learning based on the personality of the individual and that choosing the right learning style becomes essential in determining how well a student is able to grasp the subject matter. It is important to understand that each student is different and has different

learning preferences and different speeds of learning (Farsi, 2016; Lowenthal, 2010). Within the context of this thesis the VARK model, is adopted which combines four different modalities of learning styles (Visual, Aural, Reading/Writing, and Kinaesthetic). This model claims that learners should have one preference or learning style which takes precedence over other styles (Gilakjani, 2011), and consequently some students may prefer visual styles while others prefer auditory or kinaesthetic styles of learning. Further advantages arise when pulling learners" attention into their own learning styles in the form of "higher interest and motivation in the learning process, increased student responsibility for their own learning, and greater classroom community. These are affective changes, and the changes have resulted in more effective learning". Farsi (2016) defines these styles as follow:

- *Visual preference:* learners are most comfortable with pictures (graphs, images, illustrations, animation/video or pictorial representations) while learning and retaining information.
- *Aural preference:* learners learn by listening to lectures and reading or discussions.
- *Read/write:* learners learn from printed text and words.
- *Kinaesthetic:* learners prefer to learn by doing and they favour interaction with the physical world.

Gilakjani (2011) conducted a study to analyse the learning styles of Iranian university students who were learning English as a Foreign Language (EFL). They completed a questionnaire to determine if they preferred visual, auditory or kinaesthetic learning styles, with results showing that they preferred a visual learning style. The reason behind some learners preferring visual styles is these some learners retain and memorise the information of the subject that they have seen visually better than learning via aurally presented items (Kassaian, 2007). Thus, the use of mobile technologies in collaboration with multimedia in education can lead to individualization. Bonk et al. (2006) confirm that online learning will soon support a greater range of learning styles and individual differences in learning, for instance, blended environments will enable the learner to call up and manipulate pictures, charts, graphs, animations, simulations, and video-clips. Guo & Goh (2015) suggest that providing a visual agent within the

online learning environment could enrich learning experiences and help to attract learners" attention. However, these findings are contrary to a study conducted by Sweller & Van Merriënboer (2005) who found that using multiple styles in teaching can raise the amount of information that needs to be mentally processed. They explain that the delivery of too much information can be an ineffective way to learn and can interfere with the ability of brain to process information and cause unsuccessful retention and integration of information in memory.

Another model is the information processing model based on Mayer (2005) principles, illustrated in Figure 4 - 6.



Figure 4 - 6 Information Processing Model Based on (Mayer, 2005)

Mayer's (2005) principles prove that by using multimedia in collaboration with learning content and by considering the balance between the animation of presentation and its narration it is more likely to be effective. It also more effective when the learner can interact with presentations which reflect on increasing student enjoyment of the experience and improves student pedagogical performance when tested.

The effectiveness or pedagogical performance can be defined as the achievement of goals (Algahtani, 2011). Within the context of this thesis pedagogical effectiveness is defined as "determining whether the interactive learning system accomplishes its objectives within the immediate or short-term context of it is implementation" Reeves & Hedberg (2003, p61). This means that the empirical attempts to improve learners" outcomes by enhancing the learning environment, which aim to measure the impact of m-learning technologies on the achievements of the learners can be identified as an effectiveness. According to Zhang et al. (2006) and Liaw (2008), there are a number of

factors such as the interactive learning activities and the use of multimedia formats during the design phase, which if considered in the context of online learning, they facilitate the effectiveness of the learning environment and increase the potential of students" motivation for the learning. Liaw (2008), also found that multimedia instruction was the biggest predictor which influenced the e-learning effectiveness. This researcher assumed that the higher interactivity can lead to higher engagement of the learner which can give better learning outcome. The use of different settings while designing online learning material has become a promising alternative to the traditional face-to-face learning (Zhanget al. 2006), due to the implication of providing different preferred learning styles which may suit the individual learners" needs and impact on their performance.

4.5 Chapter Summary

This chapter has focused on the mobile interface and the interaction between the learner and learning content. It has identified the user experience (UX) and interface design principles which need to be considered when planning to use mobile technology in education. The chapter has also defined the concept of avatar, has illustrated its benefits and discussed how they could be designed in a pedagogical environment. It has shown how the UX is interlinked with learning theories, and has highlighted some of the key theories that have been used in this thesis such as engagement, learning styles and pedagogical effectiveness, which may play a part in increasing students'' engagement and which might enhance the learning process throughout representations of teachers as avatars on the m-learning interface.

Moving forward from the literature reviews of chapters, 2, 3, and 4, it is important to understand further the needs and requirements of students and how they prefer their courses to be delivered, especially in distance learning through a conceptual research model. It is important for instructors to incorporate learning styles, multimedia and interactive components into their course materials design which in turn supports the learners positively in their learning outcomes. In particular, this research aims to investigate whether or not having a range of avatars can supply opportunities for learners to be engaged in and to interact through interfaces designed with representational characters of their tutors as avatars.
These chapter highlight a number of areas which should be considered in this Ph.D. Firstly, the research requires an extension to the TAM model and its derivatives to take into account additional factors associated with m-learning in this context. It also requires development of m-learning web-app including different avatars that can be used for delivering learning content onto the learners and collecting their perceptions and views associated with their use. The proposed m-learning platform will provide the key multimedia attributes including audio, video and visual characters, as well as incorporating the learning styles of the VARK model to enable learners to visualising, listen, reading and interact with learning content. As a part of the study, the pedagogical performance or effectiveness of each approach will be measured and evaluated by through examining students on the lessons content that has been delivered to the students.

To date there has been little discussion in the literature with regards to how avatars can enhance learning. This research aims to experimentally investigate the factors that influence a proposed acceptance model for m-learning and its variable relationships with by using an English module for Saudi higher education students delivered via the proposed web-app.

To assess the improvement of the participants and their ability to engage and progress well in the module through m-learning, the research has focused on the effectiveness and retention of information of the learner when different avatar types are used to represent the teacher on the mobile device.

By pulling all the elements mentioned in the literature reviews together, they should back up the anticipated outcomes of the proposed research model and the web-app delivery platform. In the next chapter, the research methodology that will be used to investigate the aims and objectives of the thesis will be discussed.

5 Research Methodology and Experimental design

5.1 Introduction

This chapter describes the research philosophy and methodology adopted to effectively answer the proposed research questions. In particular, this chapter discusses the overall research processes and the methods used to accomplish the objectives of this study. The research strategies, techniques and procedures used for data collection instruments and data analysis methods are described. In addition, a detailed description is provided of the experiments and implementation procedures which were used to enhance the research data validity and reliability. The methodological approach adopted follows the "research onion" model proposed by Saunders & Tosey, (2012), as shown in Figure 5 - 1.



Figure 5 - 1 The Research Onion Model of Research (Saunders & Tosey, 2012)

A suitable methodology should guide research towards fulfilling the targeted aims. As shown in Figure 5 - 1, each layer of the "research onion" describes a more detailed and concrete stage of the research process, providing an effective progression through which the research process for a particular study can be designed. The usefulness of this model lies in its adaptability for almost any type of research methodology, its applicability in a variety of contexts and the way in which it sets out the stages through which the researcher must pass when formulating an effective methodology (Saunders & Tosey, 2012). First, the research philosophy requires definition. This creates the starting point for the appropriate selection of the research approach, as indicated in the second layer of

the model. In the third layer, the adopted research strategy defines in more detail the strategies such as experiments, questionnaires, co-creation and grounded theory that are appropriate for the research. The fourth layer identifies the target sample of the study, while the fifth layer represents the stage at which the data collection process is defined in detail and the way in which the selected tools/strategies will be used for data collection. Taken together these layers create a series of stages by which the different methods of data collection can be understood and applied in an ordered and appropriate sequence. This chapter describes this layered approach within the context of the aims and objectives of this research.

5.2 Research Philosophy and Approach (Layer 1)

A research philosophy influences the way in which the research is conducted, the strategies used, the research instruments and different methodologies selected; in short it ensures that the methodologies selected are appropriate to investigate the stated research objectives.

Several major research philosophies have been identified in the Western tradition of science including realism, interpretivism, pragmatism and positivism, the latter sometimes being called "scientific research" (Galliers & Land, 1987).

Positivism or scientific research, offers the perspective that research is concerned with gaining knowledge in a world which is objective, using scientific modes of enquiry, such as experiments and questionnaires where quantitative data is the norm (Flowers, 2009). Related to this approach reality is fixed and exists externally to social actors, is directly measurable and knowable with just one truth (Rubin & Rubin, 2012). The key aim of positivism can be stated as "to collect measurable, empirical evidence in an experiment related to a hypothesis, the results aiming to support or contradict a theory" (Bradford, 2017). The nature of the methods within positivism are structured and analysis of the resultant data is generally achieved through the use of rigorous mathematical and/or statistical techniques. Conversely, interpretivism assumes that people create and associate their own subjective and inter-subjective meanings as they interact with the world around them. In another words, research is based on the idea that it is not objective; it is always a subjective process, using qualitative techniques such as interviews or open-ended questions (Korning & Hebo, 2014). This method may also

accept that there is a reality, however, it cannot be measured directly (Rubin & Rubin, 2012), and allows for an understanding of the participants interpretations of the world (Saunders & Tosey, 2012).

For the purpose of this study, a pragmatist approach has been adopted which allows for the combination of positivism and interpretivism research philosophies. Positivism is used to define cause and effect relationships within a suitable approach as it starts with research questions, constructs hypotheses and designs, executes, and evaluates experiments, which can then be analysed using statistical methods (Bradford, 2017) as shown in Figure 5-2.



Figure 5 - 2 The steps of the scientific method (Bradford, 2017)

Positivism will therefore enable collection of all relevant information related to this study, using the main tools of data collection, such as experimental designs and questionnaires. Interpretivism is adopted to understand and interpret how relevant participants assess the current reality of the m-learning web application through their perceptions, open-ended comments and co-design activities as shown in Figure 5 - 3.



Figure 5 - 3 Procedure of implementing experiment (1)

5.3 Research Methodology Choice (Layer 2)

The purpose of research is to discover answers to questions through the application of scientific procedures. This section discusses the methods to be used for this study. There are two basic approaches to research, the quantitative approach and the qualitative approach, as well as means to incorporate and combine elements of both into the mixed methods approach often associated with the pragmatist research philosophy (Creswell, 2013; Edmonds & Kennedy, 2017).

5.3.1 Quantitative Methods

Quantitative studies rely on experimental design linked to research objectives. According to Mazlan (2012, p55), experimental design is used to "test hypotheses regarding causation, for example, that a particular instructional strategy leads to better student performance". Quantitative methods are usually applied to measure the collection of data with the purpose of verifying hypotheses or theories (Johnson and Larry, 2008) or to measure variables in a quantifiable way (Mertens, 2014). The use of quantitative methods helps to test theories, ensuring they are valid and reliable by assessing their effectiveness in any given circumstance or scenario by using structured and scientific approaches (Creswell, 2013).

Quantitative methods are frequently used when questionnaires are conducted that enable the capture of responses of the participants in a clear and concise manner (Edmonds & Kennedy, 2017). One or more hypotheses can also be tested using statistical methods if the data collection techniques limits the answer choices given to the participants.

In this study, the purpose of the questionnaires was to assess how engaged and motivated the students were to use an m-learning approach and to measure how effective each media type was at enabling students to retain the learning content delivered. As the nature of the research seeks to assess the Human-Mobile Interaction (HMI) between the participant and the mobile device interface, whilst also evaluating the level of interaction taking place during visual, textual and audio learning styles, the researcher is in complete control of the environment and no flexibility is given to deviate.

5.3.2 Qualitative Methods

Qualitative methods capture words, pictures, and artefacts (Mertens, 2014). They help to gain a deeper insight into the problem statement and have been used in this research to analyse the open-ended questions and comments, which are included as part of the post-questionnaire and which allow analysis of participants" overall feedback to identify whether or not there were differences of opinion. These type of questions enabled the researcher to ascertain the participants" perceptions of the benefits and challenges toward implementing m-learning in Saudi higher education. Furthermore, this research has also used focus groups and co-creation workshops with male students to obtain qualitative data, while female students were interviewed in focus groups through the use of video conferencing due to the aforementioned cultural context in Saudi Arabia. Qualitative methods have various advantages including helping a researcher to obtain detailed views of the perceptions of a group of participants which can otherwise be omitted from quantitative methods (Alebaikan, 2010). In this study, the focus groups provided an environment for the researcher to gather in-depth data as questions were flexibly modified by the researcher based on the responses of participants.

5.3.3 Mixed Methods

Mixed methods involve combining the collection of both quantitative and qualitative data in a research study (Mertens, 2014). It has been defined an approach to professional research that incorporates the collection and analysis of qualitative (textual) *and* quantitative (numerical) data (Creswell, 2009; and Ghaith, 2013). For example, a researcher may collect data by quantitative methods through experimental procedures, and then follow this up with structured interviews with a selection of individuals who participated in the experiment to support and explain their views on the experimental outcomes. Alternatively, the researcher may start by collecting data through qualitative approaches and follow this up with quantitative data in an exploratory research approach (Edmonds & Kennedy, 2017).

Early thoughts about the value of the mixed method approach resided in the idea that all methods had bias and weaknesses, and that the collection of both quantitative and qualitative data mitigated the weaknesses of each form of data (Creswell, 2009)

This research uses both quantitative and qualitative methods for the purposes of collecting data in order to gain a better understanding of the research and also explore the problem statement set out in the opening chapter, in a very detailed and in-depth manner. This approach often makes use of one or more quantitative data collection techniques (for example, structured questionnaires) with associated statistical analysis procedures as well as using one or more of the qualitative data collection techniques, (for example, structured interviews, observations and semi-structured focus group) (Creswell, 2013). The purpose of using a mixed method approach in this instance is to co-create ideal avatars which incorporate interactive elements to be used with associated analysis techniques. Setting up of co-creation workshops with students, offered significant potential for participants to assist with further developing their most preferred avatars and adding features they would like to see. The combinations of data collected from the quantitative and qualitative methods are to measure two main objectives:

First: Effectiveness and retention of information that participants have learned and how well they perform pedagogically.

Second: Preferences of avatar type usability and how engaged and motivated the students are to use them.

5.4 Strategies and Tools (Layer 3)

As stated in the above section, there are a number of strategies and tools that can be used for qualitative and quantitative data collection with a mixed methods research approach. These tools include questionnaires, focus groups, co-creation workshops, experimental design and grounded theory.

5.4.1 **Questionnaires**

Baker (1999) defines the questionnaire as "a method of collecting data in which specifically defined groups of individuals are asked to answer a number of questions". Questionnaires often make use of checklists and rating scales. A rating scale is more useful when perception needs to be evaluated on a continuum. There may be different stages to conducting questionnaire surveys, such as using pre-questionnaires and postquestionnaires to participants of taking part in experiments to grape their views before and after an activity. The primary goal of a pre-questionnaire is to facilitate the researcher with the specific information necessary for the study and discourage the participants from diverging from the question (Cheng & Warren, 2006). According to Alenazi (2015), questionnaires area suitable tool for collecting data from large audiences with the purpose of generalising the findings. Assuming questions have been appropriately written, questionnaires are the most appropriate instrument in term of avoiding researcher bias and influence on the participants" responses at the data collection stage, is due to the absence of direct interaction between the researcher and participants (Algahtani, 2011). In addition, the purpose of asking participants to both a pre- and post-questionnaire is to compare levels of achievements and understand whether participants have been affected by the experiments or not. The aim of the questionnaire in this study is to collect data that will allow the investigator to measure differences in participants" perceptions of m-learning and their motivation to learn with it when presented with different avatar representations of the teacher.

5.4.2 Focus Groups

Focus groups can play an important role in enhancing a questionnaire's validity and providing useful information to understand the processes behind observed results and assess changes in people's perceptions of their well-being or views (Norris, 2016). Furthermore, a focus group can be used to improve the quality of survey-based quantitative evaluations by helping generate evaluation hypothesis; strengthening the design of survey questionnaires and expanding or clarifying quantitative evaluation findings (Cochranet al. 2016). Breakwell et al. (2006, p276), define this type of method as "discussion-based interviews that produces a particular type of qualitative data generated via group interaction".

5.4.3 **Co-creation**

Co-creation is a development process whereby design professionals empower, encourage, and guide users to develop solutions by themselves or to help efficiently transfer an innovative solution from users into an institution domain (Joyce, 2017; Vázquez-Casielles et al. 2017). Co-creation encourages the blurring of the role between user and designer, focusing on the process by which the design objective is created. The idea of co-creation in this research is to actively involve users in the design and development of a future online learning, m-learning approach in a creative collaboration

between instructors and students (Pilleret al. 2012). The term co-create/co-design has become popular in mobile phone development, where the two perspectives of hardware and software design are brought into a co-design process. According to Alebaikan (2010), co-creation through focus group workshops create the opportunity for different ideas to be discussed directly and specifically. The reason behind adopting this method is to turn the passive learning materials which might contain huge amount of information into interactive learning materials based on the learners" ideas (Seifert, 2014).

5.4.4 Experiments

The definition of an experiment can be adopted from Montgomery(2017, p2) as "a test or series of runs in which purposeful changes are made to the input variables of a process or system so that we may observe and identify the reasons for changes that may be observed in the output response". Experimentation is the deliberate act of changing one or more process variable or factor in order to observe the effect and the changes have on one or more response variables (Bates, 2015). According to Mertens (2014, p4), the definition of experimental groups is that "the researcher can divide the participants into two or more groups to test the effect of a specific treatment. . . the group that receives the training is called the experimental group". When designing an experiment, there are some elements need to be considered such as defining the population of the study, variables, and the experimental design. When planned to determine which input variables are response to the important input variables is required with subsequent use this model for process or system improvement (Montgomery, 2017).

5.4.5 Grounded Theory

Grounded Theory (GT) is a systematic methodology in the social sciences involving the construction of theory through the analysis of data and is an inductive method used to predict and explain behaviour to build theory (Glaser, 2017). This starts with data being collected from a pilot study then, based on the results of the data, the researcher will rebuild and amend the planning of this experimental materials. GT provides a way to focus on groups of participants, a methodology and a set of methods and analytical tools to investigate and report on what is happening within a specific context. This enables

both quantitative and qualitative data to be collected and analysed using, allowing the researcher to use a variety of tests and materials such as questionnaires, focus groups that may be further developed as the study progresses (Crittenden, 2006).

For this Ph.D study the researcher decided to use a mixed method approach which includes use of questionnaires, experimental designs, co-creations and grounded theory, whereby the stages are going to be built up to validate the proposed hypotheses through implementing this research.

5.5 Research Design

As stated, experimental investigations were performed to explore the common and particular factors for optimal mobile learning using avatars interfaces. The study was designed to measure how engaged and motivated students were when using mobile learning devices with different types of avatars to represent the teacher. Furthermore, the effectiveness of this approach on information retention by students using m-learning was investigated and measured.

Table 5 - 1 demonstrates the activities undertaken deriving the research:

Phases	Activity		
	Design and development phase		
1	Develop the (MADE-ME) research model		
2	Design a pre-questionnaire.		
3	Develop the mobile learning web-application (MADE-ME)		
4	Design and develop a range of avatars for the course contents.		
5	Design the post-questionnaire.		
6	Submit the Ethical of the research.		
Experiments implementations phase			
1	Conduct the pilot study		
2	Implement Experiment 1		
3	Distribute of pre-questionnaire on the main sample of the study		
4	Conduct the learning through the MADE-ME web-app		
5	Develop the m-test through web-app		
6	Distribute of post-questionnaire and collect them when completed		
7	Set up the co-creation workshop		
8	Develop prefered avatar interface types further to incorporate interactive elements		
9	Conduct experiment 2 with the co-created avatar interface.		
10	Collect questionnaires and the m-test result.		

 Table 5 - 1 Demonstration of research procedure

The design and development phases took into consideration the framework that has been proposed as a part of the research while considering the user engagement, design of the interface, mobile platform and the cultural and ethical constraints within Saudi Arabia and its education system.

5.5.1 Ethical Approval Procedure

The ethical dimension identifies those issues that need to be addressed when developing and implementing this research. An approval of the procedure was obtained from the Ethics Committee of the System Engineering School, University of Reading. The ethics application is attached as (Appendix A). The consent of participants is necessary to ensure that all the ethical constraints of the research are met. Issues such as cultural diversity and gender segregation while collecting primary data for the research are important factors in the Saudi context. It was therefore imperative that the ethical issues took account of how primary data was collected and whether there were any issues that the researcher needed to take into consideration while collecting such data and if these would have any implications on the research. Permission to set up the co-creation workshop of students group within the university of Al-Baha University was obtained.

Harris et al.(2008) discuss the importance of how research ethics and list a number of its key principles, for example:

- How to encourage people to participate and the approach that will be taken.
- Displaying the objectives of the research to participants
- Underlining the confidentiality of the data collected from the participants.

The main issues required for this research context which needed consideration were confidentiality, data protection and participant consent.

5.5.1.1 Confidentiality and data protections

There is an ethical and legal requirement to protect the identity of the people who take part in the research process, both in the surveys and in focus groups. Depending on the type of data collected, this could also be a legal requirement. According to Giordanoet al. (2007) when the researcher collects personally identifiable information (PII) from the sample of chosen participants, it is imperative that the researcher takes adequate measures to protect participant information and assure participants the data will not be used except for the purpose of this study only, but that the data can be used for publications connected to the research, such as the thesis and any journal papers that stem from this thesis. Also, it is necessary for the participants to get a thorough understanding about how their details will be stored confidentially and be given the option to not disclose any personally identifiable information. Each participant was assigned a sequential number so that no result or comment can be attributed directly to any named individual but would instead be linked to that number directly. In addition, within this experiment, arrangements for any confidential material generated by the research will be stored securely within the researcher's store and, where appropriate, subsequently be disposed of securely. The researcher, in order to maintain the research ethics, took absolute care not to collect any personal details from the participants if it would not be used in the analysis or help answer the research question (Mertens, 2015).

5.5.1.2 Consent of participants

Researchers collect primary data from a chosen sample or samples. Students who participate in the research process either by means of completing a survey form or attending the focus group must not be subjected to any unethical means or coercion by the researcher (Johnson & Larry, 2008). This research, in line with standard practices, clearly provide the participants with an option to withdraw from the workshop or survey or even decline to participate without having to explain the reasons (Giordano et al., 2007). It is essential for the researcher to therefore obtain explicit permission from the participants to participate in the research by clearly explaining to them the reason for the research and how the data obtained would be used (Harris et al., 2008). In this study, students participated voluntarily in the study, signed a consent form and were informed that they could leave the study at any point they chose.

5.6 Pilot Study

Pilot testing was conducted in order to confirm that all the instruments used in the study were clearly and appropriately presented working clearly. As claimed by Mazlan (2012, p67), pilot testing is "to try the experiment on a few participants first to see whether it makes sense to them, to uncover any serious flaws or problems that might have been

overlooked at the design stage and to generally "fine-tune" the procedure". Pilot studies are useful to ensure the clarity and appropriateness of questions in the questionnaire.

A pilot study with seven Saudi students at the University of Reading was undertaken to ensure readability and clarity of the questions prior to administering the questionnaire to the targeted sample of undergraduate students in Saudi Arabia before and after they undertook a learning activity delivered on their mobile devices via each different avatar types. The level of English for all participants in this pilot was the same level as that of the participants in the research, which is elementary. In this study, particular emphasis was placed on the pre/post questionnaires in order to examine understanding of the meaning of the statements.

The researcher measured the reliability of questionnaires" items by using Cronbach's coefficient Alpha as the common measurement(Harris et al. 2008). The reason behind using this type of test is because it is the most commonly used test to determine the reliability of data when having multiple Likert questions in a questionnaire. The range of the reliability should be between 0 to 1, with coefficients equal to 0.70 or above usually considered adequately reliable (Harris et al. 2008). Cronbach's Alpha was calculated in this research by using SPSS software version 21, See Table 5 - 2.

No	Sections	Number of	Cronbach's
		Questions	Alpha
1	Preference for m-learning mode of delivery	6	0.70
2	Effectiveness/performance expectation	7	0.88
3	Engagements of m-learning	3	0.79
4	Behavioural intention to use	5	0.82
5	Convenience of m-learning	4	0.76
6	Enjoyments of m-learning	3	0.70

Table 5 - 2 Values of Cronbach's Alpha for Experiment 1

From Table 5 - 2, it can be seen that the reliability coefficients are acceptable values of reliability for the research procedures to continue.

5.6.1 Subjects Targeted

The study focused on the English language module, a compulsory course which is mandatory for the foundation year in Al-Baha University. The reason for choosing this module is the extensive hours required for teaching this course. Students complain about the high number of sessions for this course, which are 20 hours face-to-face time plus another 14 hours of six different modules each week with attendance being compulsory. In addition, despite the extensive number of hours for this course, there is a notable lack of student success in terms of English knowledge in general and grammar structures in particular. Furthermore, all Science students are required to study all their courses for the full undergraduate degree modules in English, partially for Arts students. This English model is thus an important element within the Saudi education system and is one where m-learning if successful can have a significant impact on both individual students and education in Saudi as a whole.

English language is the dominant learning and teaching language and it has become almost a necessity for students to learn English if they are wishing to enter a global workforce in Saudi Arabia. While many studies around the world have investigated the use of language learning for improving language skills, most research into language learning strategies involving Arab English Foreign Language learners - particularly Saudi Arabian learners - remain in the early stages of development compared to other nationalities and ethnic groups (Aljuaid, 2010). According to the Education First English Proficiency Index (Alhaisoni, 2012), the English language proficiency level in Saudi Arabia is rated as low.

Research has highlighted some of the main reasons behind the low English language proficiency of Arabic students, such as poor teaching methods and lack of motivation and engagement (Al-khairy, 2013; Al-Tamimi & Shuib, 2009). Therefore, the main purpose of this research is to investigate the benefits, opportunities and challenges that mobile learning brings to the higher education process in Saudi Arabia by creating an environment of real-time interactions amongst learners and their instructors through their mobile devices with application to the English language module.

The researcher contacted an English lecturer in Al-Baha University seeking his support to prepare for him six equivalent English grammar topics together with their exam questions. The lecturer collaborated with another lecturers at the department and they checked the outlines of the English language module which contain 11 main units covering a range of listed topics, one of them being the tenses grammar chapter. This chapter has several units. It should be noted that the contents of all lessons prepared for use in this study were identical in their level of difficulty. These topics included two sub-topics of ,*s*imple present", two sub-topics of ,*s*imple past", ,*p*resent progressive" and ,*f*uture tense". The lecturers also provided the researcher with a copy of the course objectives. To ensure that there is no bias in the outcome of the experiment, the learning requirements, pathways, modes as well as the course goals and objectives remained the same across all the topics, the researcher's supervisor checked them and confirmed the validity of these learning materials.

5.6.2 Learning Objectives

Within the context of the experimental lesson, the students should be familiar with how to use the appropriate context of any grammar and be able to demonstrate understanding the following:

- 1. Pronouns and how they can be used.
- 2. The structure of the basic positive sentence modal.
- 3. The structure of the basic negative sentence modal.
- 4. The question formation of tenses taught.
- 5. The answering formation of any tense's question with the right form.

The selected module was also favoured because it added value to the students" prior knowledge of English language.

5.7 Population and Time Horizon (Layer 4)

According to (Mertens, 2014) the definition of population is "the group to whom you want to apply your results" (P.4). The targeted population of this study was students in Al-Baha University, a public university in Saudi Arabia. Participants were recruited across two different colleges for the foundational year: Science and Arts Colleges. Therefore, Arts and Science faculties were considered the main specialization of those students who participated in this study because they are the main specializations in all Saudi higher education universities. In addition, there are several majors/subject that fall under these two faculties. The Science School contains Chemistry, Physics, Biology and Mathematics disciplines and the Arts school contains English, Arabic language and Islamic disciplines. The research was conducted with both male and female participants, as previously explained. The justification for conducting this study in Saudi Arabia was

because it focuses on whether this approach is most effective and preferred way of learning studying within a Higher Education program in Saudi Arabia.

5.7.1 **Participants/Sample**

The "sample" is the group within a population who will be studied by the researcher (Mertens, 2014). For the context of the research, the researcher selected to use a non-probability sample at a specific (cross-sectional) rather than longitudinal point in time, which can be defined as stated "a sample that deliberately avoids representing the wider population; it seeks only to represent a particular group, a particular named section of the wider population such as a class of students. . . two or three groups of students"(Cohen et al., 2007, p110). Hence, participants were recruited across two different Colleges for the foundational year. Recruitment took place via the Student Affair's office, with four classrooms being selected for the study and the sample being representative of the gender segregation for education in Saudi Arabia for social and cultural factors. A total of 160 questionnaires were distributed and 156 were returned (97.5%). The ages of the students ranged from 18 to 22 years old with a median age of 19 and with 91% of them aged less than 20 years old. The male:female gender split was 50:50. Students participated voluntarily in the study. Table 5 - 3, provides a breakdown of the demographic for respondents for Experiments - 1.

Respondent's Profile	Classification	Frequency	%
Gender	Male	78	50
Gender	Female	78	50
4 32	18-20	145	92.9
Age	21-22	11	7.0
Faculty	Science	83	53.2
Faculty	Arts	73	46.7
	More than 3hr	102	65.3
Average daily hours usage of	2-3 hrs	37	23.7
mobile phones	1-2 hrs	8	5.1
	Less than 1 hr	7	4.4
	Yes, a lot	10	6.4
Lknow m Loorning	Yes, a little	68	43.5
I know m-Learning	Not sure	24	15.3
	No	54	34.6
I heard the term of Avatar/	Yes	43	27.5
Representative of the	No	51	32.6
instructor	Not sure	62	39.7

Table 5 - 3 Demographics Profile of Respondents

5.8 Experimental - 1 Procedure and Implementation (Layer 5)

This section describes the procedure of the first experiment, data collection via (questionnaires) and tests the steps being chronicled as follows:

5.8.1 Procedural Steps

- The researcher received the Ethics approval letter from the Ethics Committee of the System Engineering School at the University of Reading, confirming that there to be no refusal on an ethical basis to conducting the research (Appendix A).
- The researcher provided a letter from his research supervisor, to the head of the Saudi Cultural Bureau in London, seeking permission for the researcher to undertake his research at the Al-Baha University in Saudi Arabia, and to communicate and teach the targeted sample for three months for the first semester from 16.10.2015 to 16.01.2016 (Appendix B).
- The researcher received the approval letter from the head of Saudi Cultural Bureau, confirming that the Al-Baha University had no objection to allowing the researcher to conduct his study at the Colleges of Science and Arts(Appendix C).
- Because the researcher needed to run his study on the foundational year at the two colleges, he asked the Vice Chancellor of Graduate Studies and Scientific Research, at the Al-Baha University to allow the research to go ahead (Appendix D). The Vice Chancellor confirmed the research could go ahead and wrote to the two College Deans to allow the researcher to carry out the study and to provide him with all possible facilitations.
- Because the researcher needed to run his study on the students who were already registered on the English course, he held a meeting with the head of English course and discussed with him the objectives of the research and how they might improve and affect the students" level in English. The head of English's response was positive and he stated that he welcomed such research.
- The researcher asked the manager of the English course to provide him with the targeted number of the sample list names and was provided with a list of four groups of male and female students.

- The researcher then had a face-to-face meeting with the male instructors and a meeting via video conferencing with the female instructors and again discussed the aims and objectives of the proposed study, each of whom welcomed the study.
- The dates and time were arranged with participating instructors in order to run the experiment steps with targeted participants.
- At the beginning of the study, the researcher provided an introduction to each group during the first week of the study about the objectives and aims of the current research and how the steps of the experiments would be carried out. The targeted sample of students were encouraged to participate in the experiment of English language learning via a mobile device.
- In carrying out the experiment, the first phase was to identify the source of information both from primary and secondary data. All primary data was collected from the sample identified in step one by using semi-structured pre-questionnaires and the second phase was obtained from the literature review. The researcher circulated the pre-questionnaire manually amongst the participants and collected them within 20 minutes for each group.
- In order to examine whether or not there were initial significant improvements in the students" experience and performance of English language grammar, a paper based pretest was conducted on all participants for 15 minutes prior to each lesson.
- After that, students used their smartphones devices such as iPhone and Android to receive English language course materials through the MADE-ME web application. Participants received mini-lectures via the five different user interfaces as (1) text; (2) static image; (3) cartoon; (4) audio; (5) video. The same lessons were provided to both the male and female students, plus one more which was a female static image avatar. The justification for why each one of the participants interacted with each method of learning was because that allowed them to compare preferences and engagement.
- Male participants were then given one more lesson by the researcher via a traditional face-to-face lecture, while the manager of English modules in the female departments was in charge of giving the same lesson to the female participants.

5.8.2 Implementation

In this experiment, the researcher proposed that the study develop five types of mobile interfaces using avatars to represent the teacher for male students and six types of interface for female foundational year student. The five lessons were as video, audio, static, cartoon and text, further the sixth type of female was (female static image avatar). Participant were taking two lessons per day as two different modes of delivery via mobile smartphones. The researcher randomised the links of lessons to ensure that the students did not accumulate their information and progress well on the later test rather than the earlier. He divided each group of the main four groups into two sub-groups (A and B) in order to randomise the order of lessons between the two groups. For example, one group undertook the first day's video and audio lessons, while the other group undertook cartoon and text lessons and vice versa for the next day/session. See Table 5 - 4; Table 5 - 5.

Table 5 - 4 Process of data collection with group A

Day no.	Activity
1	Pre-questionnaire
2	Pre-test of text - cartoon and its post-test
3	Pre-test of audio - static image and its post-test
4	Pre-test of F2F - video and its post-test
5	Post-questionnaire

Table 5 - 5	Process	of data	collection	with	group B
-------------	---------	---------	------------	------	---------

Day no.	Activity
1	Pre-questionnaire
2	Pre-test of video- F2F and its post-test
3	Pre-test of text - cartoon and its post-test
4	Pre-test of audio - static image and its post-test
5	Post-questionnaire

Students were given five-minute breaks after finishing each lesson and they undertook the post-test immediately. The researcher then collected the post-test from the participants. Five days after the beginning of the experiment, each group had completed the learning via each of the different the modes of delivery, and at this point, the researcher distributed the post-questionnaire to all participants and collecting them within 20 minutes.

5.9 Data Collection

Data collection was undertaken over a three month period from September 2015 to December 2015. All data were collected from the research instruments were as pre/posttests, pre/post-questionnaires and information from the co-creation workshop which was facilitated by the researcher. The researcher was the primary coder of the data, checking scale reliability and was responsible for assessing all code consistencies. Data privacy was protected through the distribution and separation procedures. The personal data collected, such as the demographic information through questionnaires, were coded by numbers in order to not identify individual participants. Moreover, the test results were protected by coding them with the same code number as the questionnaires and, then once converting them into a digital format as Microsoft Office Excel sheet, the raw questionnaire were stored securely offsite. The digital data is protected by password on a personal computer. Data backups were made and stored on another device. The Excel sheet was imported into Statistical Package for Social Science (SPSS) version 21.

5.9.1 Questionnaires

As stated, the questionnaires were completed in two stages: one before the implementation of the experiment (pre-questionnaire) and the other after the experiment was conducted (post-questionnaire). The aim of the pre-questionnaire was to obtain student responses as preliminary information such as:

- Demographic information e.g. gender, age and academic disciplines of study.
- Demographic of mobile device e.g. average use of mobiles device and types of owned phones.
- M-learning experience and gathering inputs on the m-learning platform that would be launched or used.
- The student preferences of multimedia types and identifying avatars.
- The attitudes of users towards using the technology in education.
- The initial perception of engagement for using m-learning.

These information helped to understand the perception of all the participants before using the MADE-ME app to understand their acceptance of the technology. A further pos-questionnaire, based on the outcomes of the initial questionnaire, was circulated to the participants sample to collate information regarding the usability aspects of the MADE-ME app and to determine how likely m-learning with an avatar would be successfully implemented in Saudi Arabian higher education institutions. Students completed the post-questionnaires based on how they would feel if mobile technologies were to be used for learning on a core course in their first year of their undergraduate degree. This post-questionnaire was divided into the following categories:

- Demographic information.
- Effort/Important.

- Engagement/Motivation.
- Enjoyment/Interest.
- Students" expecting their Effectiveness/Usefulness of retaining information taught by the approach.
- The perceived intention to use the technology.

The demographic information section gathered data about participant's age, gender, faculty, year of study and asked also for the technological demographic, such as general daily mobile usage and for how long they had owned them, whether participants had used mobile devices in any education-related activities and to what extent they rated features such as the convenience of mobile learning. Furthermore, in the post questionnaire, participants were also asked to rate how well they thought they had done in the test using their favourite interface and mode of lesson delivery using 5-points, ranging from 5- Great; 4- Good; 3- Average; 2- Poor; 1- Do not know.

Questionnaires were paper based and needed to be filled by hand in order to collect the data only for the target sample. There was a short introduction at the start of the questionnaire to explain the objective of the study and to reassure participants that their data would be held confidentially and that responses would not be attributable to any individual. M-learning terminology was defined within the questionnaire as it was new to their academic environment. The design of the questionnaire went through several stages:

- Both questionnaires (pre and post) were designed with reference to previous studies" questionnaires in the literatures. Suitable phrases were taken from these studies and some were modified and rephrased to suit the requirements of the current research. The reason was to assist in ensuring the validity and reliability of the questions was maintained.
- The investigator then submitted the draft questionnaire to his supervisor, at the University of Reading and also to academic staff in the English department at the University of Al-Baha in Saudi Arabia, who suggested amendments and valuable comments.
- In order to ensure the validity of the pre and post questionnaires used in this study, they were reviewed by three researchers/educationalists: two were experts in computer

science/m-learning to provide feedback on the questions and their relevance to the research and whether they able to measure what they intended investigate. The third expert was a statistician in the Statistical Services Centre at the University of Reading, to evaluate and ensure the selection of the appropriate statistical techniques for the data analysis.

- After refining the questionnaire by making the required amendments, the prequestionnaire consisted of 45 statements and the post questionnaire consisted of 42.
- As all the students who participated in the study were Arabs, who generally have low levels of English language understanding and who may have found difficulties when responding to these questionnaires in English, the researcher translated the questionnaires into Arabic after the supervisor's approval.
- The Arabic version of the questionnaires was then submitted to two separate specialists in linguistics at the Al-Baha University whose first language is Arabic, to ensure the clarity, the correct translation between the languages and to give their opinions on the translation, and also to ensure the correction of the Arabic grammatical phrasing of the statements. The researcher asked two of his colleagues as a lecturers in the English department in Saudi Arabia and native Arabic speakers to translate the Arabic version of the questionnaire to English without looking at the original English version.
- The investigator subsequently made a comparison of the two English versions, where the researcher found no significant differences between the two versions in terms of meaning, although, his supervisor recommended him to do some minor amendments.
- The questionnaires used in this study consisted primarily of closed question checklists and Likert scales and rankings, but there were also several open-ended questions that gave students the opportunity to express their views regarding their perceived potential benefits and negatives toward m-learning.

The measurement scale that has been used in this questionnaire was based on an ordinal measurement ranking of a 1-5 point Likert scale of agreement. As stated by Dawes (2008), the five point Likert scale is the most commonly used scale in any study and its validity and reliability are better and more accurate than for scales with fewer points. The scale to be used in these questionnaires is as follows (Appendix E):

- 1 = Strongly Disagree
- 2 = Disagree
- 3 = Neutral or Undecided

- 4 = Agree
- 5 = Strongly Agree
- Participants undertook the experiments and completed the comprehensive pre and post questionnaires in order to investigate their preferences for and performance with five different user interfaces: (1) text; (2) static image; (3) cartoon; (4) audio; (5) video for males, plus and an extra interface for female students using a female static image.
- The questionnaires" reliability was checked to verify the coefficient (Cronbach's Alpha) was high which more than (0.7) in this case study and was shown to meet the reliability requirements in SPSS, as was shown in Table 5 2 in Chapter 5 (p70).

5.9.2 Test phase

Learning performance or pedagogical effectiveness was evaluated by designing tests based on the learning content. To measure the students" effectiveness using this approach, assessments (grades on/across all modes of delivery) were performed. The test/exam for this study was prepared by a specialist team of academic staff from the English department. Participants in this study conducted the pre- and post-tests. The purpose of the pre-test was to ascertain what aspects of English grammar they knew and were familiar with. Both tests were identical in order to measure and assess the students" progress using the experimental approach. The questions required the students to rewrite the correct structure of each sentence using two different forms. To explain this in more detail, if the students were given the sentence in a positive format, they were required to re-write the sentence in both a negative and question format and vice versa of that tense grammar. The team checked the correctness of the questions with regard to the scientific content. In addition, they ensured the suitability of each question and that they were relative to the objectives of the lessons. Furthermore, they ensured the language of each question was clear and easily understood. The average time each exam required was 15 minutes. According to the statistical analysis, the pilot study confirmed that all lessons" exams questions had the same level of difficulty, which showed consistency between these ways of learning. Analytical results provided the evidence that these tests were strongly reliable in evaluating learning performance.

5.10 The Implementation of a Co-creation Workshop (Focus Group)

Focus groups or co-creation workshops are being increasingly used in the field of education for "gathering data on attitudes, values and opinions" (Cohen, et al. 2007, p.376). Structured focus groups are used to reveal detailed information from the participants as they communicate and interact within the group. In this manner, cocreation can contribute towards validity of the research and insure the real views of participants and their perceptions are provided. According to Liaw (2008), by considering the responses of students who participated in e-learning courses, a greater understanding of the reasons why learners are often not satisfied with the e-learning experience can be gained. It should be noted that learners were passively learning with all the previous content delivery mechanism, however, in this stage, the researcher set up a workshop with a number of invited participants with the aim of working with them to co-create, customise and reinvent the avatars as active or interactive learning interfaces including extending features they liked and discarding what they did not like. As highlighted by Liaw (2008, p868), "learning activities in which learners play active roles will engage and motivate students" learning more effectively than learning activities where learners are passive".

Co-creation is about working together for a strong community and more effective social design. It starts from the idea that designs are successful only when the people being served are involved (Maenpaa, 2012). Participants will then get the chance of choosing the best avatars and design their own mobile learning interfaces. In addition, in this experiment the researcher intended to investigate student preferences for the best way to learn through mobile phone technology. The researcher selected seven participants from each of the two classes (Arts and Science), drawing students from both the male and female classes. The total number of workshop participants was 28 students and they were divided into four groups. The selection was based on random selection from the previous participants in order to find different experiences. The workshop was carried out in a suitable and quiet place at the College of Science in Al-Baha University (See Figure 5 - 4). The workshop the students undertook lasted between 45 minutes to one hour.



Figure 5 - 4 Co-Creation Workshop Session Managed by the Researcher (published with consent of the participants)

The researcher concentrated in this session on the creativity of learners, based on their views and perspectives on the previous experiment and added new elements which they missed and wished to have while learning through mobile phones. The researcher and groups of students went through various activities during the workshop; for example:

- 1. A ,brainstorming" session about the effectiveness of m-learning for the English module.
- 2. Sketching ideas on how to optimize the learning process on paper worksheets.
- 3. They were encouraged by the researcher to design a storyboard and prototype their engaged mobile interface (See Figure 5 5).
- 4. They were asked deeper questions based on the initial results of their previous questionnaire, and they were asked further about the advantages and obstacles of English learning regarding the teaching methods.
- 5. Students tried to improve or merge elements from the different types of m-learning interfaces.



Figure 5 - 5 Sample of participants prototyping

The researcher collected and analysed the qualitative data for the purpose of designing a new prototype and developing the mobile application based on the final result. In summary, students wanted a balance between passive learning and active/interactive learning to engage them more with the content, which could help them to retain the information they have learned. According to the data collected, the web-app needed to have the following functions and features:

- The ability to keep students motivated and engaged by providing exercises in between concepts.
- The ability to provide instant feedback.
- The ability to direct the learner to revise information if their answer was wrong.

All of these characteristics and features are highlighted by Eppard et al. (2016). Additionally, students pointed out their enthusiasm to learn from the animated instructor (video interface avatar). Finally, when the researcher completed the case study and collected the required data, he received a release letter from the main supervisor who monitored the experiment at Al-Baha University which confirmed the success of completion (Appendix H).

5.11 Experimental - 2 Procedure and Implementation (Layer 5)

Based on data from the first experiment, three types of avatars (audio; video; cartoon) were further developed with 28 students in a co-creation workshop in order to incorporate other collaborative and interactive elements to support the learning process. A second experiment was conducted using the avatar interfaces co-created by the students, to compare levels of engagement and pedagogic performance in relation to mobile learning and traditional teaching of English as a foreign language to 103 students in the same higher education institution in Saudi Arabia.

This section discusses the procedure of the second experiment data collections and tests. The experiment went through the following stages:

- This second round of experimentation was based on the results of the initial testing, with the same objectives and sample population and environment in Al-Baha University in Saudi Arabia.
- The design of the lesson was based on using of Articulate Storyline 2 software and PowToon website as explained in details in Chapter 7.
- For the data collection method, questionnaire which contains closed questions and open ended questions designed through the identical stages that been used in first experiment (Appendix F).
- The experiment was conducted between 23/11/2016 to 15/12/2016.
- The researcher contacted the Heads of the English course, in the male college and in the female college, initially by email and subsequently by phone, to select the two groups from each gender in the Science and Arts classes.
- The researcher asked the managers to provide him with the targeted number of sample list names, then with the contact with students" affairs, they provided him with the four groups of male and female lists of names.
- The researcher sent the instrument tools, such as a letter detailing the instructions for the experiment and also electronic pre-test and electronic questionnaire, as links by email to the two managers. This experiment underwent several steps see Figure 5 6.



Figure 5 - 6 Procedure of implementing experiment (2)

- When all the participants understood how the system worked, they started by connecting their mobile devices to the internet and followed the steps in the instructions" letter by accessing the pre-test for 15 minutes via their smart phones.
- After finishing the pre-test and submitting their tests electronically, they were given a web link which took them through three lessons (audio, video and cartoon) lasting approximately 30 minutes in total. The participants used headphones to hear the lessons without disturbing others. This period gave the students extra time to review the learning material. The lesson was divided into three modalities of delivery and each mode had the intervention of two exercises.
- The following day, participants were able to access the second link which directed them to a second test, identical to the pre-test. After submitting electronically, they received a link which took them through the questionnaire for 20 minutes. The tests were multi-choice questions about the lecture content.
- After the post-test, each participant received the final link which required them to fill out a questionnaire to assess her or his perceived motivation/engagement, interaction and expected effectiveness and also to give feedback on the web-app system and their learning experience.

In this experiment, the researcher proposed that the study developed three types of mobile interfaces within interactive elements for each delivery mechanism. Based on the grammar they the students learning in their lessons, they undertook exercises with immediate auto feedback if their answers were correct; they were then able to carry on to the next concept of the lesson. If they answered incorrectly, the system provided feedback informing them of their incorrect answer and transferred the learner back to the point in the lesson linked to this question. Learners were able to go back to that later question and try to answer it through the previous process. With the intention here was to show whether providing learners with feedback and self-assessment could effectively transform the learning process.

5.12 Data Analysis Method

This section presents the type of quantitative and qualitative data collected during this research. In order to achieve the objectives of this research (identifying the participants" motivation/engagement) and to test the research hypotheses, questionnaires were

conducted as the main tool to garner participants" views and perceptions, and the pre/post-tests were the main data used to analyse their pedagogical effectiveness/achievement quantitatively. Additionally, the open-ended questions and co-creation workshops were analysed qualitatively. Accordingly, this section describes the core methods and tests used to analyse these datasets.

5.12.1 Quantitative Analysis

The quantitative data obtained from the post-questionnaire for the first and second experiments, were the primary data source for this study, and were analysed using the statistical analysis software, "SPSS" (Statistical Package for Social Science). Statistical methods helped the researcher to compute specific and numerical values relating to the research questions. The research questions not only considered the perceptions of the participants but also focused on obtaining facts and figures, thus limiting the flexibility of response of the participants. It also led the participants to answer only the point in question, which might sometimes make them resistant (Mertens, 2015). In addition, this type of data enabled the researcher to make comparison between learning methods in term of preferences, engagement and effectiveness. The method also allowed the researcher to run different analyses and tests on the addressed hypotheses, which answered significant portions of the main research questions. Noteworthy, was that these statistical analyses techniques were chosen based on statisticians" advice from the Statistical Services Centre at the University of Reading. Based on the post-questionnaire completions, the data was converted into SPSS software. The following factors were considered the most important for the data analysis:

- 1. Analysis of the preferred mode for content delivery and avatar teacher representation: determined the content delivery modality most favoured by students, which theoretically translate to an increase in learning experience and learning effectiveness.
- 2. Analysis of performance expectancy (usefulness): explored how well students felt they had understood the learning content according to the different learning modalities.

- 3. *Analysis of engagement:* determined which method of learning delivery was most interesting and held students attention while they were learning and which extended their motivation to learn and progress well in their education.
- 4. *Analysis of enjoyment:* related to the learning modality engagement which led to the feeling of an attractive learning environment which might influence achievement in the learning process and outcomes positively.
- 5. *Analysis of convenience:* explored the variables of convenience and differences according to gender.
- 6. *Analysis of behavioural intention to use m-learning:* determined the adoption of mlearning. It was important to analyse the students" intention to use the technology and particularly which mode of delivery was the best according to their engagement and the effectiveness of their learning.
- 7. Analysis of effectiveness (achievement):similar to the engagement factor but exploring which students had pedagogically performed (test score) well with these modes of learning delivery.

While the participants had their own views about how much the mobile web-app improved their engagement, preferences, enjoyment, intention to use and pedagogical performance/effectiveness, this did not necessarily mean that they succeeded in having the right method of learning. Therefore, the qualitative approach of open-ended questions and co-creation workshops, allowed them to explain the quantitative findings further, with specific reference to the questions related to the hypotheses.

5.12.2 Qualitative Analysis

This study adopted a mixed methods approach, using quantitative data to display results and then using qualitative data to support or reject the initial data (Creswell et al. 2011). Further to the quantitative analysis, more insightful analysis was extracted from the use of open-ended questions and through co-creation workshops. Participants were asked question during the workshop about key of interest, as well as getting them involved in the customisation of their ideal avatar-based user interfaces for content delivered.

One of aims of this research was to reveal and explore the benefits, opportunities and challenges that m-learning might bring to the higher education process in Saudi Arabia, in order to understand how to improve m-learning by optimising the learning interfaces.

Some of the participants responded to the open-ended questions. After analysing the data quantitatively, the researcher transcribed all comments from the open-ended questions in preparation for analysis. The themes for this research were explored via five phases of process according to Kabilan(2016):

- 1) Familiarisation with the data: reading all the data to be familiar with.
- Creating primary code: the primary codes were created by phrases or/and keywords directly linked to the ideas and views.
- 3) Searching for themes: grouping all data relevant to potential themes.
- Reviewing themes: checking all created themes and drawing them as a thematic (map) of the analysis.
- 5) *Naming and defining themes:* generating a clear name or definitions for each theme (category) in order to write the final analysis and link them to the research questions and literature reviews.

Comments from the open-ended questions were read several times to identify proper themes and categories. Then, each theme was coded to specifically describe the master theme. As the comments were written in Arabic. The main themes and students" comments were translated into English in order to be used in the current research.

Additionally, outside of the main research questions, after conducting the experiments, a couple of the participants" instructors were also interviewed to provide their perceptions and feedbacks with regard to the students" motivations and progression. Their comments were also analysed and are included in the results in Chapter 8 to add extra insight to the research findings.

Overall, the study methodology progressed through several stages which began with the research paradigms and ended with the data collection techniques. The most important layers are illustrated in Figure 5 - 7.



Figure 5 - 7 Research Framework for Conducting Research Studies

5.13 Chapter Summary

This research has adopted both *positivism* and *interpretivism* in a mixed methods approach, encompassing collection of both qualitative and quantitative data. Research was focussed on learning content via a range of different mobile interface avatars and students" engagements and performance with the learning materials was assessed according to a set of hypotheses developed from a theoretical model, MADE-ME (Multi Avatar Delivery Environment for Mobile Education) which is described in Chapter 6, and a MADE-ME delivery platform described in Chapter 7.

A pilot study with seven Saudi students at the University of Reading was undertaken to ensure the reliability, readability and clarity of the questions prior to administering the pre and post questionnaires to 156 undergraduate students in Saudi Arabia before and after they undertook a learning activity delivered on their mobile devices via each of the five different avatar types. In order to explore further student preferences and performance, four groups of seven students participated in a co-creation workshop enabling them to design their own mobile learning interface. The outcomes of these workshops together with the comprehensive analysis of the student responses to the questionnaire were then used to inform a second round of experiments with 103 participants. Results from study are documented in Chapter 8 and discussed in Chapter 9.

6 The Conceptual Research Model

6.1 Introduction

This chapter describes the construction of the MADE-ME research model, which is one of the key outputs of this research. MADE-ME, or "Multi Avatar Delivery Environment for Mobile Education", shows the relationship between learners" preferences for learning modality, engagement, enjoyment and the effectiveness (pedagogical performance). This model builds on a combination of several m-learning models and learning theories and extends the traditional technology acceptance model (TAM) by adding new variables to fulfil the objectives of this research. The purpose of MADE-ME is to determine the factors that affect higher educational students" intention to adopt m-learning and to mitigate obstacles that may affect the success or failure of this approach in Saudi Arabia. Thus, this research aims to investigate learners" intrinsic engagement factors on the behavioural intentions factor towards the use of m-learning technology, and to discover to what extent these factors influence the effectiveness/pedagogical performance of the student when learning is via mobile technologies. As a result of the proposed model, perceived performance expectancy and perceived engagement both contribute to the learners" behavioural intention to use the m-learning web-app (MADE-ME).

6.2 How the Model is Constructed

The delivery of effective m-learning content depends on various issues such as the technology acceptance model (TAM) and its extended models, theories of learning, human computer interaction (HCI), the local environment, and the learning being undertaken. Moreover, highlighting interactivity the learning element, engagement/preferences, benefits and barriers, and how these together affect the learning achievements/outcomes of students, have contributed to the construction of the MADE-ME web-app. In this research, the model and its hypotheses were developed based on the previous models described in the literatures. A series of case studies were conducted and the data collected and analysed statistically. Based on the results of the first case study which both qualitatively and qualitatively collected data and the cocreation workshop, the model has been adjusted and extended to incorporate additional factors. It also has been verified again through a second round of case study

experiments in order to validate the final version of the mode through testing its hypotheses based on the participants" responses and data. The flow stages of constructing and extending the MADE-ME research model is shown in Figure 6 - 1. The main findings behind constructing that model are summarised here as they have helped to answer the current research questions and objectives.



Figure 6 - 1 Stages of deriving MADE-ME model

Table 6 - 1 explains how the integrated factors of the MADE-ME model have been informed by the literature reviews and by the case studies in this research.
Table 6 - 1 Demonstration	of the	factor	source
---------------------------	--------	--------	--------

Key factor of model	Derived from/influenced by	Result
Intrinsic engagement or motivation	The proposed research model for the study of e-learning motivation and acceptance in developing countries based on extension the UTAUT by (<i>Maldonado et al, 2010</i>).	Inclusion in Model 1 & 2
Performance expectancy	The research framework which used UTAUT based upon TAM by (<i>Jairak et al., 2009; Lowenthal, 2010</i>).	Inclusion in Model 1 & 2
Gender	The research model of mobile entertainment adoption by <i>Leong et al., (2013)</i> ; and by <i>Lowenthal (2010)</i> who investigated the UTAUT	Inclusion in Model 1 & 2
Major/subject	Additional moderator based on the research context	Inclusion in Model 1 & 2
Enjoyment	The Research model of <i>Alenezi etal.</i> , (2010) who empirically investigated the enjoyment and other factors to determine their influencing on students intention to use technology. Also, <i>Liu</i> (2008) extended the UTAUT model by adding perceived enjoyment.	Inclusion in Model 1
Multimedia instruction	A conceptual model of users' satisfaction, behavioural intention, and effectiveness toward e-learning by <i>Liaw</i> (2008).	Inclusion in Model 1 & 2
Convenience	Additional factor based on the research context	Inclusion in Model 1
Interactivity elements	Additional factor based on the first case study and focus group data.	Inclusion in Model 2
Behavioural intention	A conceptual model of users' satisfaction, behavioural intention, and effectiveness toward e-learning by <i>Liaw</i> (2008); Alenezi etal., (2010)&(Lowenthal, 2010)which based on UTAUT.	Inclusion in Model 1 & 2
Effectiveness or pedagogical performance	A conceptual model of users' satisfaction, behavioural intention, and effectiveness toward e-learning by <i>Liaw</i> (2008).	Inclusion in Model 1 & 2

6.3 The Conceptual Research Model and Hypotheses

This research, as mentioned earlier, uses qualitative and quantitative techniques to effectively answer the questions about the technology. In Saudi Arabia, where the higher education system currently depends on the instructor delivering the course directly to the students, the concept of assisted technology is quite new. Making it work alongside regular teaching approaches therefore requires a lot of change, with students needing to believe that the technology can bring potential benefits to the existing system. This knowledge also paves the way for understanding the expectations of users about the technology and for making improvements to make it more user friendly and acceptable. Hence. measuring the perceptions of engagement/preference and effectiveness/pedagogical performance of the users towards the use of new technology requires a deeper understanding of these perceptions and intentions to use the technology. To date little importance has been given to predicting the perceived performance expectations of the technology, thus leaving a gap in the research about how well such technologies can be used alongside regular pedagogical methods.

Among all acceptance models, there is a need to build a conceptual model based on the previous frameworks of mobile learning contexts and which in this case align with the Saudi Arabian educational context. The conceptual model is used to investigate the current research factors and find out how well, if m-learning was deployed, the users would be able to accept the changes and embrace the technology. Saudi Arabia has a unique culture, which in turn is reflected in its education environment. Government and the education ministry requires gender segregation in schools and universities classes. Therefore, there is a need to add the gender variable as moderator on all investigated factors to find out the differences between genders and what impact these differences have on intentions to use m-learning. Moreover, one of the higher education ministry rules is that female students are not allowed to carry their mobile phones into the university campus for religious purposes; therefore, the research suggests that this might affect their perception of convenience. As a consequence, ,,convenience'' was an added factor in the research model in order to investigate its influence on the performance expectancy and to identify any differences between gender.

In addition, within the context of the research objectives, the subject/major was added as another moderator on all investigated factors. Arts and Science faculties were considered to be the main specializations of the students participating in this study because they are the main specializations in all Saudi higher education universities. In addition, there are several subject/majors that fall under these two faculties.

Based on the combination of models from prior studies of mobile learning, the researcher adopted factors that would assist this research, such as: Performance Expectancy, Intrinsic Engagement, Enjoyment, Multimedia Instructions Preference, Convenience, Effectiveness and Behavioural Intention to use m-learning, which together combine to create a new extension model to the TAM which includes these mentioned factors integrated together in the MADE-ME model. The MADE-ME model for Saudi Arabia used in this PhD is showed in Figure 6 - 2.



Figure 6 - 2 Conceptual Research Model for Multi Avatar Delivery Environment for Mobile Education (MADE-ME) In Saudi Arabia

These elements are the main factors/variables of the MADE-ME framework and they help to ensure that the implementation of the technology is successful and proves useful. As discussed in the section on the TAM in Chapter 3, acceptance of technology is based on being able to predict intentions of users based on their beliefs and prior actions. When cultural expectations are not met, the users do not feel comfortable with the environment and will not use the technology. Similarly, learning expectations, styles and outcomes determine the perceived usefulness/performance expectancy of the technology and so, when not met, would not entice the user to use the technology. If the design principles and the mobile context are not well thought out, this too leads to either the failure in the technology being appealing or the perception of motivation is not met, thereby leading to implementation failure.

6.3.1 Variables

According to Farsi (2016), when planning to conduct a scientific experiment, it is necessary to predetermine the dependent and independent variables of the study. Dependent variables are those variables affected by the independent variable and which can be called the outcome. With regard to this study, the various learning approaches for the English language (cartoon, video, audio, text, static image and face-to-face) are the independent variables. The aim is therefore to see how these variables influence the dependent variables, which are grouped into four main categories: engagement, enjoyment, effectiveness and the intention to use this approach of learning.

Based on the above explanation, the thesis proposes the following hypotheses:

- *H1. Preference for multimedia instruction will positively influence students' performance expectancy.*
- *H1_a.Gender will positively moderate the effect of preferred multimedia instruction on the performance expectancy.*
- H1_b.Major/subject of study will positively moderate the effect of preferred multimedia instruction on the performance expectancy.
- H2.Preference for multimedia instruction will positively influence students' intrinsic engagement
- *H2_a.Gender will positively moderate the effect of preferred multimedia instruction on students' intrinsic engagement.*
- H2_b.Major of study will positively moderate the effect of preferred multimedia instruction on students' intrinsic engagement.
- H3. Intrinsic engagement will positively influence students' enjoyment.
- H3_a.Gender will positively moderate the effect of intrinsic engagement on the students' enjoyment.
- H3_b.Major/subject of study will positively moderate the effect of intrinsic engagement on the students' enjoyment.
- *H4.Intrinsic engagement will positively influence students' performance expectancy.*
- *H4_a.Gender will positively moderate the effect of intrinsic engagement on the students' performance expectancy.*
- *H4_b.Major/subject of study will positively moderate the effect of intrinsic engagement on the students' performance expectancy.*
- *H5. The convenience of m-learning will positively influence students' performance expectancy.*
- H5_a.Gender will positively moderate the effect of the convenience of *m*-learning on students' performance expectancy.

- H5_b.Major/subject of study will positively moderate the effect of the convenience of m-learning on students' performance expectancy.
- *H6. The performance expectancy will positively influence students' behavioural intention to use m-learning.*
- *H6_a.Gender will positively moderate the effect of the performance expectancy on the behavioural intention to use m-learning.*
- H6_b.Major/subject of study will positively moderate the effect of the performance expectancy on the behavioural intention to use m-learning.
- *H7. Intrinsic engagement will positively influence students' behavioural intention to use m-learning.*
- *H7_a.Gender will positively moderate the effect of intrinsic engagement on the students' behavioural intention to use m-learning.*
- *H7_b.Major/subject of study will positively moderate the effect of intrinsic engagement on the students' behavioural intention to use m-learning.*
- *H8. Enjoyment will positively influence students' behavioural intention to use m-learning.*
- H8_a.Gender will positively moderate the effect of enjoyment on the students' behavioural intention to use m-learning.
- H8_b.Major/subject of study will positively moderate the effect of enjoyment on the students' behavioural intention to use m-learning.
- H9.Intrinsic engagement will positively influence students' effectiveness.
- H9_a.Gender will positively moderate the effect of intrinsic engagement on the students' effectiveness.
- H9_b.Major/subject of study will positively moderate the effect of intrinsic engagement on the students' effectiveness.
- H10. Preference for multimedia instruction will positively influence students' effectiveness.
- $H10_a$. Gender will positively moderate the effect of preferred multimedia instruction on the effectiveness.
- H10_b.Major/subject of study will positively moderate the effect of preferred multimedia instruction on the effectiveness.
- H11. The effectiveness of m-learning will positively influence students' behavioural intention to use m-learning.
- *H11_a.Gender will positively moderate the effect of effectiveness on the students' behavioural intention to use m-learning.*
- H11_b.Major/subject of study will positively moderate the effect of effectiveness on the students' behavioural intention to use m-learning.

6.4 The Second Experiment Conceptual Research Model:

Based on the results from the first model that have been analysed for this research (see chapter 8 for details) a number of factors were highlighted that needed to be amended, either by adding them to or deleting them from the model. The findings confirmed the necessity of two factors, convenience and enjoyment being considered when adopting new way of learning. These factors should however not be included in the second research model because they were considered not to be treated separately and they were encompassed in the wider context of engagement and behavioural intention factors. In addition, participants identified that the use of multimedia with the intervention of interactive elements may affect their intrinsic engagement and performance expectations. Figure 6 - 3, presents the new version of the research model followed by the amended research hypotheses.



Figure 6 - 3 Conceptual Research Model (MADE-ME) Including Interaction Elements

This second MADE-ME model has been reconstructed and adjusted based on the first case study and the co-creation work shop that was undertaken in order to adjust some of the model hypotheses as well as testing and verifying them again. The second series of these proposed hypotheses is presented as follows:

- *H1. Preference for multimedia instruction will positively influence students' intrinsic engagement.*
- H1_a.Gender will positively moderate the effect of preferred multimedia instruction on the intrinsic engagement.
- H1_b.Major/subject of study will positively moderate the effect of preferred multimedia instruction on the intrinsic engagement.

- H2. Preference for multimedia instruction will positively influence students' performance expectancy.
- *H2_a.Gender will positively moderate the effect of preferred multimedia instruction on the performance expectancy.*
- H2_b.Major/subject of study will positively moderate the effect of preferred multimedia instruction on the performance expectancy.
- H3. Intrinsic engagement will positively influence students' performance expectancy.
- H3_a.Gender will positively moderate the effect of intrinsic engagement on the students' performance expectancy.
- H4_b.Major/subject of study will positively moderate the effect of intrinsic engagement on the students' performance expectancy.
- *H4. The interaction activities in m-learning will positively influence students' intrinsic engagement.*
- H4_a.Gender will positively moderate the effect of interaction activities in m-learning on students' intrinsic engagement.
- H4_b.Major/subject of study will positively moderate the effect of intrinsic engagement on the students' intrinsic engagement.
- *H5. The interaction activities in m-learning will positively influence students' performance expectancy.*
- *H5_a.Gender will positively moderate the effect of interaction activities in m-learning on students' performance expectancy.*
- H5_b.Major/subject of study will positively moderate the effect of interaction activities in m-learning on students' performance expectancy.
- *H6. Intrinsic engagement will positively influence students'* behavioural Intention to use m-learning.
- $H6_a$. Gender will positively moderate the effect of intrinsic engagement on the behavioural Intention to use m-learning.
- H6_b.Major/subject of study will positively moderate the effect of intrinsic engagement on the students' behavioural Intention to use m-learning.
- *H7. The performance expectancy will positively influence students' behavioural intention to use m-learning.*
- *H7_a.Gender will positively moderate the effect of performance expectancy on the behavioural intention to use m-learning.*
- *H7_b.Major/subject of study will positively moderate the effect of performance expectancy on the students' behavioural intention to use m-learning.*

- H8. Preference for multimedia instruction will positively influence students' effectiveness.
- H8_a.Gender will positively moderate the effect of preferred multimedia instruction on the effectiveness.
- H8_b.Major/subject of study will positively moderate the effect of preferred multimedia instruction on the effectiveness.
- H9. Intrinsic engagement will positively influence students' effectiveness.
- H9_a.Gender will positively moderate the effect of intrinsic engagement on the students' effectiveness.
- H9_b.Major/subject of study will positively moderate the effect of intrinsic engagement on the students' effectiveness.
- H10. The interaction activities in m-learning will positively influence students' effectiveness.
- H10_a.Gender will positively moderate the effect of interaction activities in m-learning on students' effectiveness.
- H10_b.Major/subject of study will positively moderate the effect of intrinsic engagement on the students' effectiveness.
- H11. The effectiveness of m-learning will positively influence students' behavioural intention to use m-learning.
- H11_a.Gender will positively moderate the effect of effectiveness on the students' behavioural intention to use m-learning.
- H11_b.Major/subject of study will positively moderate the effect of effectiveness on the students' behavioural intention to use m-learning.

6.5 Chapter Summary

This chapter has shown how the MADE-ME research model has been derived from several previous models, the important factors of which were integrated into this research and were used to investigate the factors or variables set out earlier. Based on the different elements covered in this research (TAM, human computer interaction and the learning theories), this research has constructed two version of the MADE-ME model based on each other to meet the research objectives. The MADE-ME model is one of the research contributions of this thesis to the field of learning and engagement. The development of the MADE-ME web-app which enables the conceptual theory to be applied in practice through the delivery of learning content to students is presented in the next chapter.

7 System Implementation (MADE-ME Web-App)

7.1 Introduction

Chapter 6 described the educational research model which links delivery of learning content via mobile technologies with pedagogical performance. In order to validate and test its hypotheses empirically, the MADE-ME web-application was developed. This chapter presents the process of designing and creating the framework for the MADE-ME online web-app that can deliver m-learning content to a mobile device via different avatar representations of the teacher (audio, video, image, cartoon; text) and which can be used to test the pedagogical effectiveness of the content been delivered based on that model. Further, it will cover the development of the MADE-ME web-app system, including highlighting the environment of the two experiments set up to prove this web-app and get useful feedback on the model, the user interface design, and software used for development.

7.2 Multi Avatar Delivery Environment for Mobile Education (MADE-ME) Web-Application

One of the research objectives of this study is to find the most preferred/engaging avatar representation of a teacher for delivery of learning content via mobile technology in order to enhance learning through a variety of avatar interfaces and multimedia content delivered through a web based application. This research proposes and develops an application, Multi Avatar Delivery Environment for Mobile Education (MADE-ME) as the web-app platform for delivering mobile content to optimise learning outcomes via mobile assisted language learning. The development of the web-app framework is based on the feedback and perceptions of the participants. The MADE-ME application is used in this research to deliver English content, supported by some explanation in the Arabic language, for Saudi students. The most significant element of the platform is that it provides learners with the opportunity to learn through exercises and the feedback they answer correctly. Further, the instructor can provide a number of different types of tests, such as multiple choice, true or false, drag and drop and/or open-ended questions.

In order to create this environment, a number of functional and non-functional requirements were identified. With regards to functional requirements the web-app has the ability to:

- Display screen avatars as representations of the teacher.
- Deliver learning content via a mobile phone.
- Enable content to be in varying multimedia formats.
- Allow multi-language delivery.
- Test students on what they have learnt.

Non-functional features include:

- The incorporation of good user interface design principles.
- The concept of usability with regards to how the app works.
- Its reliability with regard to availability and content delivery.
- The ease of use of the interface.
- The ease of learning of the apps" features.

Focusing on the requirements enables evaluation of how easily users can perform their goals on the web-app which influences their engagement/experience with the learning material and improved pedagogical performance for the English course being used as an example in this research context.

The novelty of this app is that it can be used to deliver any course material in any language to any organisation around the world in order to support the delivery of multimedia content to enhance learning and teaching.

7.3 Development

This section describes the software used, access to the internet and the location of the web-app server for the MADE-ME web-app.

7.3.1 Software

In order to develop the web-app for use in smartphones for this research, three software tools have been used in combination for the development phase. Firstly, the online website "PowToon" was used to create an animated cartoon avatar to be used as one of the avatar representations of the teacher for the purpose of drawing the learners"

attention and engagement. Secondly, Camtasia software was used to record the voice over for the slides as presentation of the content. Thirdly, the Articulate Storyline 2 software was used to construct the lessons with the addition of the different type of avatars. Articulate also enabled the mobile web-app to embed interactive elements into its framework such as providing tests and immediately assessing learners" progress by providing feedback in both textual and verbal forms. Finally, the Articulate Storyline software enabled final version of the MADE-ME app to be published as HTML5 which is reliable and mobile friendly with regards to the app being made available and connected with the internet. Noteworthy, the Google form website was used for setting up the questionnaires questions in order to collect the respondents" data electronically.

7.3.2 Access to the Internet

The MADE-ME web-app can be accessed anytime and anywhere via an Internet connection. In this research, most of the male students in the experiment were running the learning app at the university campus. However, in terms of the internet facility, there was no access to the Wi-Fi Internet from the students" classrooms, so the researcher bought a high-speed internet modem from an STC (Saudi Telecom Company). The small modem device was portable and easy to use in any classrooms. In addition, students who had the service of the internet available on their mobile devices were encouraged to use and connect to their own internet networks. However, the female students preferred to learn through the use of the app within their home with their convenience.

7.3.3 Location of the MADE-ME web-app Server

The web-app system was hosted by the researcher's own server located in Europe. In order to publish the lessons of the experiments which can be accessed by any of the participants, the researcher subscribed and created a server for two years with a registered domain. The domain name was Saudielearning.org (see Figure 7 - 1). The mobile web-app enabled participants to access the system and learning materials through their own mobile phones and to achieve their learning requirements from their location in Saudi Arabia.

Hostin an account to	g Account	S
- An		
saudielea	rning.org	

Figure 7 - 1 MADE-ME Web-App Server Location

7.4 User Interface Design

It is important to take on board the principles and practical elements associated with designing good mobile user interfaces. Interactive electronic systems can be successful when they combine a different number of factors such as usability, reliability, functionality and performance. These factors are equal in their level of importance and are linked with each other. In other words, any failure in any of these factors could cause failure of the whole system (Mayhew, 1999). The user interface design depends on adopting a structured process to the design incorporation of good user interface design principles and adherence to guidelines related to the design of web pages. Robbins (2007) and Mazlan (2012) set out different types of tasks which should be included in designing Web pages such as ensuring that the design of interface and easy to use has strong graphic and information design.

7.4.1 Graphic and Information Design.

The definition of "graphic design" is what the users see on the web page interface, such as texts and graphics (Noble & Bestley, 2016), with the information design being how the information is organised on the web page. Mazlan (2012) proposed a number of suggestions and guidelines when creating visual designs. For instance, the amount of information should be controlled and not presented all at one time. In addition, information need to be organised by presenting elements clearly at the centre of the screen while the colours of the text and background should be consistence across all aspects of the interface. It is also important to provide easy access to the system and hence the MADE-ME interface also included a navigation button in the appropriate

position on the screen. Figure 7 - 2, shows the integration of these elements into the design of the MADE-ME web-app.



Figure 7 - 2 Visual design and technique of the interface

7.5 Use of MADE-ME in Experiment – 1 (Case Study)

The purpose of this research was to design a number of different lessons using different types of multimedia avatars to represent the teacher and then to test learners to engage their level of success with each. The intent was to provide m-learning interfaces which allowed participants to access and learn from these lectures. All lecture materials and content were similar in terms of the difficulty level of the concept that is being learnt. Lessons were designed with the teacher having different types of interfaces such as text, audio, static image, video and cartoon and were accessed by all students as a web interface application for ease of use across all mobile platforms (iOS, Android, Microsoft etc.).

Within the context of this thesis and in order to prove this work and validate the proposed model hypotheses, two case studies and a co-creation work shop were conducted.

In the first case study, the participants were 75 male students and 75 female students who undertook the learning process via delivery on their mobile phones of a number of lessons with different multimedia content representative of the English module they undertake on their degree course.

In the first lesson, the interface was designed to have different multimedia such as text, voice over and animated cartoon characters as shown in Figure 7 - 3 and Figure 7 - 4. In addition, the interface had the control buttons for pause, go forward or back to the lesson.



Figure 7 - 3 Screenshot of cartoon mobile interface 1

	مثال على تركيبة الجملة المثبته للفاعل They	
	Subject + verb to be + verb + ing	
\bigcirc	They are working now	
		1

Figure 7 - 4 Screenshot of cartoon mobile interface 2

Further, each of the lessons had their own structure and media inclusion, such as lessons using only text, text with audio and text with video.

The second lesson was a text lesson designed as a series of Camtasia slides using colour conventions for different constructs of English grammar. This lesson gave the participants the opportunity to download it as an e-book via the iBook (or equivalent) application on their mobile device. The main benefit with this approach is that learners only needed to connect to the internet at the beginning of the lesson to access and download it onto their mobile devices, after which they are able to access and continue learning from the lesson without requiring a continuous internet connection. See Figure 7 - 5 & Figure 7 - 6, for examples of these text based learning screens.



Figure 7 - 5 Screenshot of text mobile interface 1

Figure 7 - 6 Screenshot of text mobile interface 2

The third lesson developed as an audio lesson, again using the Camtasia software. The researcher has recorded his voice over each presented piece of text (see Figure 7 - 7. Thereby combining text and audio as the learning medium.



Figure 7 - 7 Screenshot of audio mobile interface

The fourth lesson included the instructors" static image. This lesson combined three features (text, voice over and instructor's image). To maintain the student motivation, the images different from slide to slide based on the presentation context, for example, left and right landside see (Figure 7 - 8 & Figure 7 - 9).



Figure 7 - 8 Screenshot of static image mobile interface 1

المذمول به فعل مساعد ملضى Subject + Verb to be + Object	تركيبة الجمله المثبته
He	المجموعه الاولى
It + was	
-որվիսը.	
-uhlippine	

Figure 7 - 9 Screenshot of static image mobile interface 2

The fifth lesson was given as a video interface, which included both text and an animated video of a human character with a normal voice (Figure 7 - 10 to Figure 7 - 11). This representation of the learning process was designed to make the students feel like they were in a traditional face-to-face teaching setting, but with the added value of having navigation control buttons as in any normal video (Figure 7 - 12).



Figure 7 - 10 Screenshot of video mobile interface 1



Figure 7 - 11 Screenshot of video mobile interface 2

1.00	•০০০ য় ক Done 5:53	2:41 pm		✓ 98% → + -11:16	
	الفاعل Subject + آ	العاده المتكرره فعل اساء erb + Object/	لجمل المثبته	أمثله على ال	
	He		ئۇلى	مثال على المجموعة ال	
\mathbf{O}		erb+s + habit	العاده		
	He drinks coffe	e every day. يوم	و يشرب القهوة كل	هو	
	· · · · ·		••	×ĸ	

Figure 7 - 12 Control navigation of video mobile interface

As has been noted, in terms of the unique culture in Saudi Arabia, female students can be taught by both genders, however, male instructors can only teach them through the use of technology rather than via face-to-face settings. On the other hand, female instructors are *not* allowed to teach male students even through the use of technologies. In this research, students of both genders undertook all lessons one to five, but the female students took one extra lesson through the mobile web-app using a female instructor's voice (see Figure 7 - 13).



Figure 7 - 13 Screenshot of female mobile interface

All participants also had comparative lesson delivered via traditional class teaching. Once participants had undertaken each lesson they took post-test to assess their performance and measure the effectiveness of learning with regard to the content of the lesson. Finally, post-questionnaire was undertaken (quantitatively and qualitatively) by the participants to measure the extent of their engagement with the different ways of delivering course materials.

The purpose of Experiment - 1 was to get participants to use the mobile web-app in order to explore which approach/es they preferred and to takes their top preferences further into co-creation and development sessions in order to influence positively their pedagogical performance. Indeed, both case study 1, and co-creation workshop data were used as preliminary analysis to support the design and development of the second experiment even though the full analysis, results and discussion of the information for both of them are presented in detail in Chapters 8 and 9.

7.6 Use of MADE-ME in Experiment – 2 (Case Study)

After analysing the data of the first experiment, it was found that participants most preferred to learn from video, then the cartoon interface and lastly the audio interface. Conversely however, in relation to the participants retention and understanding of the information delivered and their performance in the test it was the audio highest interface that was found to be the most effective, then the cartoon interface, followed by the video at the lowest level of success as shown in (Figure 7 - 14). The text interface was at low level regarding both the preference and performance.

Weight	Audio	Cartoon	Video
Highest			
Moderate			
Lowest	0		
	Preference	A Performa	nce

Figure 7 - 14 Diagram of the compensation between preference and performance

Based on the first analysis of the participants comments and those stemming from the co-design/creation workshop which allowed the participants to express and design their own mobile interface, the theory of Universal Design for Learning (UDL) was adopted,

which is an educational framework that can accommodate individual differences in learning styles and increase the flexibility and effectiveness of the learning environment. This also enables those students who have particular challenges to be given special assistance as it enables the implementation of specific multimedia types to meet and support all students" diverse learning needs. For this purpose, the three most popular types of learning (audio, video, cartoon) were taken forward and redesigned as a second experiment where by students again undertook lessons material where all was similar in terms of the difficulty of level concepts (see Figure 7 – 15 to Figure 7 - 17. The participants for Experiment - 2 comprised 103 students from Al-Baha University who received instructions about the experiment and mechanisms of delivery of course material via the MADE-ME application interfaces, allowing them to access and interact with content of the English module.



Figure 7 – 15 Screenshot of audio mobile interface



Figure 7 - 16 Screenshot of video mobile interface



Figure 7 - 17 Screenshot of cartoon mobile interface

The approach for Experiment - 2 delivering content with interactivity exercise using the mobile MADE-ME web-app can be summarised in the following flow chart (Figure 7 - 18).



Figure 7 – 18 The flow-chart for MADE-ME web-app process

One of the main comments that was strongly highlighted by the participants was for the app to have a facility for assessing their progress during their learning. Responding to this, a self-assessment tool was developed and delivered through the MADE-ME mobile web-app system and was used for the English course, as shown in Figure 7 - 19.



Figure 7 - 19 Screenshot of frequent exercise activities

This tool gave participants the opportunity to receive immediate feedback on the exercises they undertook following each concept to be learnt (Figure 7 - 20). In addition, another very useful and important tool, was added to MADE-ME for the learners, providing them with the opportunity to look back into any specific point in the lesson to learn how to get the answer right when the feedback message showed the answer they gave was incorrect (Figure 7 - 21). The Articulate Storyline 2 software was used to construct the lessons that integrated these tools.

	Choose the co	rrect answer: الاجابه الصديحه	اختار /ي ا). (write)
0	● did○ did	correct. اجابتک ماطنه،انفر طی زر الاماب القاحد، ککی تشرف طی الام الام الدی الفاحد فریب الی الفاحد WHILLEH	نین لا نا
	مرده ليده الماضي اليسيط	writed •••	

Figure 7 - 20 Screenshot of instance feedback interface for incorrect answer



Figure 7 - 21 Directed to this part which explain how to answer the exercise

To ensure learners had understood the concept they could not move forward from the lesson until they had chosen the right answer. When they answered correctly, they received a direct feedback message to show their success and they were allowed to move to the next part of the lesson (see Figure 7 - 22).

Choose th	ابه الصحيحه: ne right answer:	إختار/ي الاجا	
She	Correct صحيح	هي تار	
🔿 is pla	That's right! You selected the correct response. المسنت الجابئك مسميمه،القر على رز "استمر" لتنتقل للشريمه التأليه	- 3	
🖲 is pla	Continue		
🔿 playin			
🔘 are pl	aying		
		-up hope	

Figure 7 - 22 Screenshot of the correct answer feedback

In order to assess student performance the next step in the process was the inclusion of an online mobile exam which was designed as a Google form. The design of the app took each student through all the mandatory lessons and materials until the final interface, which had the link to the test. In other words, the researcher tried to avoid any student bias which would affect their result of the performance; hence, students could not access the exam link until they had passed all lessons successfully and completed all the exercises to reach the final interface. Participants were able to access the exam via the given link and they were then required to fill-in their demographic information such as name, gender and subject discipline. There were three sections of the exam in terms of the English language content (1) present simple, (2) past simple, (3) present continuous, based on the three delivery mechanisms (1) audio, (2) video, and (3) cartoon as shown in to Figure 7 - 23 to Figure 7 - 26.



Figure 7 - 23 Screenshot of posttest part (1)

Figure 7 - 24 Screenshot of posttest part (2)

• —	• —
الماشتى اليمرط Past simple	المعتدارع المستمر Present continuous
ستكون الخمس استكه القائمة إختبار (جابة واحده من الأربع خبارات في زمن الناضي البسيط	ستقرن الخمس اسكنه لقائمه إختيار إجباء واحدد من الأربيع خيارات في زمن الحضارع المستمر
1. You English language yesterday. (Speak). Negative *	1. You English language right now. (Speak). Negative *
O did not speaked	O do not speaking
O did not speak	are not speaking
O did not speaks	not speaking
O are not speak	 are speaking not
2. She Tennis last month. (Play). Positive *	2. Shetennis at the moment. (Play). Positive *
O did paly	is playing
🔿 playd	playing
is played	O plays
played	O does playing
3. They the school at 2.00pm yesterday. (Leave). Positive *	3. They for their exams just now. (Study). Positive *
O leave	 studying
O did leave	do studying
O did leaved	are study
O leaved	 are studying
4. Itat 8.00 o'clock in the morning. (Start). Negative *	4. It to move right now. (Start). Negative *
O did not start	does not starting
is not started	 not starting
O did not started	is not starting
O does not start	
5. I the chicken burger last night. (Like). Negative *	5.1 the chicken burger at the moment. (eat). Negative * and on ot eating Summer
\odot	

Figure 7 - 25 Screenshot of posttest part (3)

Figure 7 - 26 Screenshot of posttest part (4)

An online questionnaire tool was also developed to collect the participants" perceptions and views toward their engagement with each style of learning. Immediately after they had submitted their exam, the link to the questionnaire was shown. There was a short introduction at the beginning of the online questionnaire, which presented the objectives of the research and provided participants with some guidance of how to complete the questionnaire correctly (see Figure 7 – 28 to Figure 7 - 29), and the copy of the questionnaire as Appendix F. The respondents" data was collected and stored securely online for later analysis.

• —
Questionnaire experiment 2
ا مرکزه های شمان قدر می کنیم را اینه قدر رویدن بعض می تخریم قدر ماد این و است قب می ایند افتوادش می مارد می شواید نام ایران شده این می نابد شه انتریش می ایند این انتریش این می ایران قدر ایران می است قد آن از انت پندری تمید هد زیشتر ، مقد ایرا مرکز می 15 دقیه رستون ومیدانتش که می شریه راعه اهدام می اعترایی و آسمین. • Required
] - اكلب اسماك الكلائي بالعربي؟ * ۲- اكلب اسماك الكلائي بالعربي؟
2- ما هر جندانه؟ *
عر () س ()
3- ما هو تخمیسانه مینکیند؟ * عثر ⊖ غاب ⊖
مع در ابتك السابقة أو الحالية هل سنق لك أن أخذت درس علمي (محاضره) من خلال جهاز الجوال قبل هذ الدر اسه *
м () г
ريما
التدريس السابعة (المسوتين، القيديري والكرتوني)، من مسئلة إخلاق كري الغرار. الذي يعتر عن وجهة تطرك بدعة أذا أنسبل التعلم عن طريق تشانية الحوال إذا كانت التائمة تحتوي على الأثن
الله العمل على عليه المولي عليه الحوان الله كانته العليه الحواري علي الادي 4- صرت المعلم مع طرور النص المتروح أي (صرتم) * ارتش يتما ه

Figure 7 - 27 Screenshot of online questionnaire (1)



Figure 7 - 28 Screenshot of online questionnaire (2)



Figure 7 - 29 Screenshot of online questionnaire (3)

Initial analysis of the exam results revealed positive effectiveness of the MADE-ME app on students" learning English language concepts. The questionnaire also showed the participants" enthusiastic attitude toward using this app in their future learning. Detailed analysis and results will be presented in the next chapter.

7.7 Chapter Summary

This chapter began by describing the implementation of the MADE-ME web-app including an outline of the tools used for the development and hosting of the web-app. The chapter then described two experiments that were conducted with the students to assess their preference for different avatar representations of the teacher when learning content is being presented, and the means where by students" performance with each learning styles could be tested, and their feedback on their experience of using the web-app gathered. In summary, the MADE-ME app can teach, engage, assess and guide the learner towards a positive and effective approach to online learning of English language concepts. The result of the experiments will be presented in more detail in the next chapter which is based on the analysis of implemented experiment data.

8 Results of the Study

8.1 Introduction

The aim of this study is to investigate the ways in which engagement and performance in learning can be influenced by the type of avatar representation of the teacher on the mobile device which might be in the form of video, audio, image, cartoon or text. In particular, the purpose of the research is to investigate how m-learning can be used to assist the teaching of compulsory English modules within Higher Education courses in Saudi Arabia. Based on the research objectives and the methodology used, this research conducted two experiments and one co-creation workshop. The later experiment built upon the results of the previous experiment, the users" perceptions and views that they expressed in the open-ended questions of experiment one and a co-creation workshop. This chapter starts with presentation of the data obtained from the questionnaires described in Chapter 5, followed by an analysis of the proposed hypotheses which were presented in Chapter 6, in order to support or reject them. The MADE-ME model factors are analysed in terms of the relationships existing between them, their influence on students" intentions to use m-learning, and which avatar types are the most effective for the learning process. Accordingly, the full analysis of these hypotheses and the findings provided in this chapter address the main research questions of the study.

8.2 Recap of Experiments

In order to achieve the key aim of the research, a number of related objectives with associated research questions were identified and a number of experiments/case studies undertaken to resolve them. As a result of the research, the MADE-ME model has been constructed as well as the MADE-ME web-app designed and developed to deliver learning material. To evaluate the model through the use of the mobile web-app, students from Al-Baha University, Saudi Arabia were recruited to be involved in the study which comprised two case studies and a co-design workshop for data collection reasons. For the first case study, 156 participants learnt 5 lessons (audio, video, carton, image, and text) via their mobile phones, plus one further lesson which was delivered as traditional face-to-face teaching. Material was selected from their English module with the purpose offering the avatar type that the students found most engaging and most effective with regards to learning outcomes. Subset of these participants subsequently

took part in the co-creation workshop and redesigned their preferred m-learning avatar interface. The model and the mobile app were amended and redeveloped based on the students" feedback, and a second experiment was undertaken with a further 103 students from Al-Baha University, again in order to determine if the existing factors in the model could be used to improve the learning process. Detailed analysis of this data is presented in the coming sections.

8.3 Analysis Overview

This section addresses the techniques that have been used to analyse the data, as well as describing the specific approaches undertaken for both the quantitative and qualitative responses.

Statistical data analyses were performed on the data collected from the tests and participant questionnaires as quantitative analysis. In addition, the participants" perceptions and discussions of usability at the focus group workshop have been considered and analysed using the qualitative method of thematic analysis.

For the quantitative data, significant differences between Likert ratings are assessed. All analyses of the collected quantitative data have been carried out using the Statistical Package for the Social Science (SPSS), the results of the analysis being used later in the thesis to provide recommendations based on the statistical result to inform and perhaps even change the process of learning in the Saudi Higher Education. Based on these *potential* results, the researcher might find the best preference for learners is to learn through traditional classes (face to face). Alternatively, learning via mobile devices using favoured avatars might be shown to be the preferred option either outright or as blended learning whereby m-learning can assist the traditional teaching and learning process. Furthermore, the results of the analysis will confirm the extent to which the use of avatars affects students" learning and retention of information.

In order to assess the research hypotheses the following tests were used:

• The UNIANOVA: this type of analysis is used to analyse the interaction and relationship of one or more independent variables on a single dependent variable and to indicate the differences that are found (Liu et al. 2007). That is, the UNIANOVA procedure provides regression analysis and analysis of variance for one dependent variable by one or more factors and/or variables (SPSS 21, Help command). Within

the context of the current research, it is used to analyse the influences of a particular preferred method of learning delivery, motivation and engagement to learn via that method, as well as other factors related to the effectiveness and intention to use that type of m-learning. If the P significance value is less than or equal 0.05 (P \leq 0.05), the factor has a significant effect on the outcome variable; however, if the P significance value is greater than 0.05 (P \geq 0.05), the factor has no statistically significant relationship with the outcome factor (Frost, 2015).

- **Pairwise Comparisons**: this test was used for this research as a sub-test under the UNIANOVA test to compare the interaction between students" gender, faculty and other dimensions of factors such as motivation, performance expectancy, effectiveness and intention to use according to the learning delivery mechanisms.
- LOGISTIC REGRESSION: this type of analysis is the appropriate regression test to be used when the predicted outcome variable (dependent) is dichotomous (binary) on a set of independent variables. In this research, it analyses the differences and relationships between the different avatar types of learning delivery. Logistic regression results are represented by acronyms as shown in Table 8 1. The P-value is the significance value of the tested factor on the dependant factor. It is identical to the UNIANOVA technique whereby if the P-value is less than 0.05, this indicates the significance of the relationship and that it has statistical influence; however, if the P-value greater than 0.05, this indicates that the influence of that relationship does not have statistically significant differences between the two variables.

5	
Symbols	Meaning
Р	The significance value, if $P < 0.05$ is significant; and $P > 0.05$ is not significance.
Exp (B)	The expectation ratio/odds value.

 Table 8 - 1 Logistic Regression symbols

• Cronbach Alpha: this test is a common measurement of research instrument reliability when using SPSS (Harris et al. 2008). As mentioned in section 5.6 to show reliability, the minimum value should generally be that of 0.7. In regard to this research, the Cronbach Alpha test showed all factors that have been used were highly

accepted, sufficiently reliable and that questionnaires were carefully designed to collect data regarding students" attitude toward m-learning (see Table 5 - 2).

These statistical techniques were chosen based on consultation with Statistical Advisors in the Statistical Services Centre at the University of Reading, from the beginning of designing the questionnaires through to the analysis stages. The advisors confirmed at each stage that these were common techniques to adopt in research such as this and that they were extremely suitable tools to use for this research to achieve its aim and objectives.

8.4 Results of Experiment (1) – Questionnaire

8.4.1 **Demographic Results**

This section of the results presents the analysis of the first experiment demographic information. This includes gender, age, faculty of study/major, year of study, average period of having a mobile device, average daily hours spent using mobile devices, the type of mobile device(s) and activities and services related to mobile devices. In this section, the researcher measured statistically the participants'' frequency and percentages.

Gender Group

The participants in this experiment were split 50:50 according to gender, with 78 students from each sex. Table 8 - 2, illustrates the demographic data.

Respondent's Profile	Classification	Frequency	%
Gender	Male	78	50
Gender	Female	78	50
Ago	18-20	145	92.9
Age	21-22	11	7.0
Faculty	Science	83	53.2
	Arts	73	46.7
Average daily hours usage of mobile phones	More than 3hrs	102	65.3
	2-3 hrs	37	23.7
	1-2 hrs	8	5.1
	Less than 1 hr	7	4.4
I know m-Learning	Yes, a lot	10	6.4
	Yes, a little	68	43.5
	Not sure	24	15.3
	No	54	34.6
I heard the term of Avatar/	Yes	43	27.5
Representative of the	No	51	32.6
instructor	Not sure	62	39.7

Table 8 - 2 Demographic result

Age Groups

From Figure 8 - 1, 93% of the participants were aged between 18-20 years old and the remaining participants aged between 21-22 years old were only 7%, which means that which means that the sample population was representative of the targeted population for the study. Data shows the 18-22 years old group having the highest usage of upcoming mobile technologies.



Figure 8 - 1 Bar Chart for Participants Age Groups

Faculty Groups

The participants' responses about their intended study disciplines after their foundation year are illustrated in Table 8 - 2 that 53% of the participants are currently classified in Science departments and 47% of the participants in Arts department.

Participants' mobile devices ownership

Students were asked about their ownership of mobile devices to determine if their devices possessed the smart features required to support m-learning. Figure 8 - 2 shows that the vast majority of the participants own at least one mobile phone and a few have two devices. 98% of students have or have owned a smartphone or tablet, while only 1.2% of the participants have basic phones. Accordingly, most of the higher education students in Saudi Arabia own smartphones, which they use for both learning and non-learning purposes (Alhassan, 2016). First, iPhones and then Samsung Galaxies were the

most popular brands of device. Ownership of these high-quality devices, which readily connect to the internet and wi-fi increases the likelihood of them being easily utilised in educational and learning environments.



Figure 8 - 2 Participants' Mobile Types Ownership

Year of Study

All the participating students are fresh year students (foundational year) in one of the Saudi public universities.

Period of having Mobile Smart Devices

To ascertain their familiarity with the technology, participants were asked about how long they had been using their mobile devices, which in the majority of cases was between 1 and 3 years and frequently more (see Figure 8 - 3). This suggests that students are familiar with mobile technology and as such are well placed to use these technologies for educational purposes if they so wished and content is available.



Figure 8 - 3 Average period of Having Smart Mobile Device

Hours Spent on the Mobile Phone (per day)

Participants were also asked for how long on average per day they used their mobile devices with the majority of responses (65%) being in excess of 3 hours per day and a

further 24% of participants using them for 2-3 hours per day (see Figure 8 - 4). Only 4% use them for less than an hour a day.



Figure 8 - 4 Percentage of Mobile device daily usage

General Activities and Services Used by the Mobile Device

Looking in more detail at the current use of mobile devices by the participants Table 8 - 3 - Figure 8 - 5, male and Table 8 - 4 - Figure 8 - 6, female), it can be seen that a variety of activities are undertaken related to communication, information gathering and media interaction. Male participants are seen to use their phones for communication activities more than female participants, with 65% of male students, for example, using their mobile devices often or regularly for phone calls, compared to only 32% of female students. The numbers using Facebook for fun was surprisingly small, but "WhatsApp" emerged with both males and females as the most popular way for chatting and exchanging media. Browsing the web for fun and watching videos for fun are other highly popular activities with both genders. The number of participants browsing the web for education were low compare to browsing the web for fun.

Activities via mobile device	Not at all %	Seldom %	Some %	Regularly %	Often %
Phone calls	1.3	9.0	24.4	41.0	24.4
Text message	11.5	32.1	12.8	26.9	16.7
Using Twitter	19.2	7.7	23.1	38.5	11.5
Using Facebook for fun	59.0	20.5	6.4	7.7	6.4
Using WhatsApp	3.8	7.7	5.1	50.0	29.5
Calendar	35.9	25.6	14.1	10.3	14.1
Reading articles books	20.5	23.1	23.1	15.4	16.7
Watching videos for fun	2.6	6.4	23.1	37.2	30.8
Playing games	14.1	10.3	21.8	37.2	16.7
Map facility	30.8	28.2	19.2	11.5	10.3
Taking, sending photos	7.7	12.8	21.8	42.3	15.4
Taking uploading videos	24.4	16.7	30.8	11.5	16.7
Browsing web for fun	2.6	9.0	15.4	47.4	21.8
Browsing web for education	48.7	17.9	11.5	9.0	11.5

 Table 8 - 3 Features used by male participants via mobile device



Figure 8 - 5 *Features used by male participants via mobile device chart* (*n*=78)

Activities via mobile device	Not at all %	Seldom %	Some %	Regularly %	Often %
Phone calls	3.8	30.8	33.3	11.5	20.5
Text message	14.1	47.4	12.8	6.4	19.2
Using Twitter	23.1	12.8	24.4	14.1	25.6
Using Facebook for fun	80.8	3.8	3.8	5.1	6.4
Using WhatsApp	3.8	5.1	7.7	50.0	33.3
Calendar	32.1	23.1	17.9	12.8	14.1
Reading articles books	17.9	28.2	21.8	12.8	19.2
Watching videos for fun	2.6	6.4	15.4	37.2	37.2
Playing games	11.5	21.8	29.5	15.4	21.8
Map facility	42.3	21.8	15.4	9.0	9.0
Taking, sending photos	2.6	5.1	11.5	39.7	41.0
Taking uploading videos	16.7	20.5	15.4	26.9	17.9
Browsing web for fun	1.3	9.0	10.3	44.9	32.1
Browsing web for education	38.5	35.9	16.7	5.1	3.8

 Table 8 - 4 Features used by female participants via mobile device



Figure 8 - 6 *Features used by female participants via mobile device chart* (n=78)

Activities Related to Learning

To explore the potential educational uses of mobile devices in more detail, participants were polled about what education-related activities they had undertaken or would consider undertaking via their mobile devices, with results presented in Table 8 - 5 - Figure 8 - 7 for male participants and Table 8 - 6 - Figure 8 - 8 for female participants. Again, a range of activities has been experienced with the most popular being "checking course timetables" (males and females), "viewing educational videos" (especially females) and "submitting coursework" (males), although viewing lectures as screencasts

appears to be a less popular activity. Whilst female participants frequently listed that they would not undertake educational activities via mobile devices, one popular option amongst the females (60.5%) was their willing to register for their courses via mobile devices. This might be due to faster access to the web via mobile devices than traditional computers, or it could possibly be attributed to the cultural norms and religious aspects of Saudi Arabia, whereby females are not allowed to go by themselves to internet services (e.g. Internet café) outside of the home.

Learning Activities	Would not do %	Might do %	Would do %	Have done %
Register For Courses	17.1	11.8	47.4	23.7
Check Course timetable	6.4	11.5	15.4	66.7
Check Course Syllabus	20.5	35.9	25.6	17.9
Listen to lectures as screencasts	32.4	25.7	17.6	24.3
View Educational Videos	27.3	39.0	14.3	19.5
Submit Course work	21.1	23.7	18.4	36.8
Access library account	51.3	25.6	15.4	7.7
Access library database	39.0	35.1	20.8	5.2
Use Social media for education	60.3	24.4	7.7	7.7
pay study fees	44.2	31.2	14.3	10.4
Take online tests	19.5	36.4	15.6	28.6

Table 8 - 5 Activities related to learning - Male participants



Figure 8 - 7 Activities related to learning - male participants chart (n=78)

Learning Activities	Would not do %	Might do %	Would do %	Have done %
Register For Courses	10.5	17.1	60.5	11.8
Check Course timetable	7.8	18.2	20.8	53.2
Check Course Syllabus	30.7	26.7	20.0	22.7
Listen to lectures as screencasts	44.6	27.0	13.5	14.9
View Educational Videos	25.6	32.1	9.0	33.3
Submit Course work	24.7	33.8	15.6	26.0
Access library account	66.2	19.5	9.1	5.2
Access library database	54.5	24.7	16.9	3.9
Use Social media for education	59.7	16.9	6.5	16.9
Pay study fees	39.5	36.8	6.6	17.1
Take online tests	35.1	21.6	12.2	31.1

Table 8 - 6 Activities related to learning - female participants



Figure 8 - 8 *Activities related to learning - female participants chart* (*n*=78)

Prior familiarity of m-learning functionality

80% of the male participants understood partially or totally what m-learning involved, whereas this figure was 75.4% for female participants. This means that approximately a quarter of the participants did not understand the meaning of m-learning prior to this study. This data is illustrated in Figure 8 - 9.
• Do you understand what m-learning involves?



Figure 8 - 9 Understanding what does m-learning mean

Prior experience of using avatars

43% of participants reported that they had some prior knowledge of avatars, whilst 36% reported that they did not know about or had not seen avatars prior to this study, and a 20% of both genders reported that they were unsure. These results are illustrated in Figure 8 - 10.



Figure 8 - 10 Experience of using avatars

It appears from Figure 8 - 10 that with female students were slightly more experienced in using avatars than male students. As with, 47.4% of female students having some experience of using avatars compared to only 34.6% of male students.

Participants' experience of learning types

Data relating to the experience of student learning modes (face-to-face; e-learning; mlearning; male or female on screen cast at the class), whether in their previous or current education, is presented in the following tables.

teaching)						
Regularly						
82.1%						
89.7%						
85.9%						

Table 8 - 7 Percentages of traditional learning experience

Table 8 - 8 Percentages of male face-to-face as a screen cast teaching experience

	How much have you experienced with face-to-face screen cast with male instructor at the class							
	Not at all Seldom Some Regularly							
Female	26.3%	14.5%	48.7%	10.5%				
Male	51.3%	20.5%	23.1%	5.1%				
	39.0%	17.5%	35.7%	7.8%				

Table 8 - 9 Percentages of e-learning experience

	How much have you experienced with online learning (e-learning) via PCs or laptops							
	Not at all Seldom Some Regularly							
Female	49.4%	27.3%	19.5%	3.9%				
Male	56.4%	19.2%	20.5%	3.8%				
	52.9%	23.2%	20.0%	3.9%				

Table 8 - 10 Percentages of m-learning experience

	How much have you experienced with mobile learning (m-learning) via phones or tablets							
	Not at all	Not at all Seldom Some Regularly						
Female	60.5%	18.4%	13.2%	7.9%				
Male	74.4%	11.5%	7.7%	6.4%				
	67.5%	14.9%	10.4%	7.1%				

Table 8 - 11 Percentages of female face-to-face as a screen cast teaching experience for female only

	How much have you experienced with face-to-face screen casts with female						
	instructor at the class						
	Not at all	Seldom	Some	Regularly			
Female	35.9%	20.5%	30.8%	12.8%			

As illustrated in the above tables, the most common way of learning that the students have experienced is through the traditional face-to-face method, with learning occurring from listening to a male instructor presenting the lesson through a screen cast also being popular. The reason percentage of female students who have experienced screen casting is due at least in part to the country"s educational system including issues of culture and gender segregation. Students had least experience with m-learning with 67.5% of the participants having not practically experienced that method of learning. Even though the results in Table 8 - 10 show that they are highly informed and familiar with the term of m-learning (76%). These findings support the research objectives to investigate and explore the factors that engage and enhance learners to learn via m-learning effectively.

8.5 Results of Research Hypotheses – Experiment (1)

At the centre of the research is the MADE-ME model which includes a set of hypotheses that have been developed to answer the research questions. Table 8 - 12 illustrates the correlations between the questions from the questionnaires and each specific factor's hypothesis.

Statement	H _n	Hypothesis					
Performance expectancy							
Do you think accessing course material on your mobile device with an avatar helped you to learn the material more efficiency than the same content presented in traditional formats (face-to-face)		H1a. Gender will positively moderate the effect of preferred multimedia instruction on the performance expectancy.					
I think that the m-learning would be a more effective way to learn English language than learning face-to- face in classes							
I think m-learning would strengthen my participation when learning English language	H1 _{a,b}						
I think having an m-learning interface with an avatar would positively assist me to learn English		H1b. Major of study will positively					
I think having an avatar approach would increase my understanding of information that I have learnt		moderate the effect of preferred multimedia instruction on the performance expectancy.					
I think having an avatar approach would increase my retention of information that I have learnt							
Intrinsic Engagement							
Did the content delivered to you by a mobile device engage you		H2a. Gender will positively moderate the effect of preferred multimedia instruction					
If you had more content delivered on your mobile device would you be more motivated to learn English than via traditional teaching (face-to-face)	H2 _{a,b}	on students" intrinsic engagement.					

Table 8 - 12 Correlations between questions from the questionnaires and the specific factor's hypothesis

Behavioural intention to use m-learning	ç	
I think the portability of mobile device plays a strong factors in enabling me to learn anywhere and anytime		m-learning on students" performance expectancy.
I think the repeatable and pause features of lessons on the mobile device would be more convenient to me than a traditional class.	H5 _{a,b}	students" performance expectancy. H5b. Major of study will positively moderate the effect of the convenience of
I think the m-learning would be more convenient to me than learning face-to-face in class		H5a. Gender will positively moderate the effect of the convenience of m-learning on
Convenience of m-learning		
I would like to learn with a static female avatar via a mobile device with text + voice		
I would prefer to learn with a cartoon photo avatar via mobile devices with text + speech		H4b. Major of study will positively moderate the effect of intrinsic engagement on the students" performance expectancy.
would prefer to learn with a talking head male avatar via mobile devices with text + video		H4h Major of study will positively
I would prefer to learn with a static male avatar via mobile device with text + voice	H4 _{a,b}	
I would prefer to learn through mobile devices including text + voice via mobile device (but no image)		
I would prefer having an m-learning interface without an avatar		H4a. Gender will positively moderate the effect of intrinsic engagement on the students" performance expectancy.
I would prefer to learn in the normal traditional face-to-face class		HA. Conden illinoidinel and added
Multimedia preferences (mode of delive	ery)	
I think having an interface with an avatar will be fun to learn from		moderate the effect of intrinsic engagement on the students" enjoyment.
I think having access to materials on mobile devices would be a fun interaction between content-learner	H3 _{a,b}	effect of intrinsic engagement on the students" enjoyment. H3b. Major of study will positively
Did you enjoy having content delivered to you by a mobile device		H3a. Gender will positively moderate the
screen cast of male instructor at the class Learning Enjoyment		
I find a female face to face instructor more engaging to me than learning via		instruction on students" intrinsic engagement.
I think having access to the English materials on my mobile devices would enhance my motivation to learn English language		H2b. Major of study will positively moderate the effect of preferred multimedia

I plan to continue using my mobile device for receiving educational content (if available)		H6a. Gender will positively moderate the effect of the performance expectancy on the behavioural intention to use m-learning.
I believe extending the concept of the avatar to other courses would be useful.		
Would you like to undertake future courses with integrated mobile learning for English language	H6 _{a,b}	H6b. Major of study will positively
Would you like to undertake future courses with integrated mobile learning for other courses		moderate the effect of the performance expectancy on the behavioural intention to use m-learning.
Would you recommend m-learning with an avatar interface to other students		

The results of the hypotheses based on the answers expressed in the questionnaires are as follow:

H1. Preferences and Performance Expectancy

The first hypothesis states that:

Preference for multimedia instruction through mobile device (or traditional F2F) will positively influence students' performance expectancy.

In order to determine the relationship of these two factors, the Logistic regression test was used. This type of test shows statistically if any of these types of learning has a significant influence on the performance expectancy of the students. The results of this test are illustrated in the following table (see Table 8 - 13).

Table 8 - 13 Logistic Regression of preference modality of learning on performance expectancy

Preference for learning method	В	Sig.	Exp(B)	95% C.I.	for EXP(B)
Yes or No				Lower	Upper
I like the traditional F2F	948	.170	.388	.100	1.502
I like the m-learning with text interface	.382	.630	1.464	.310	6.910
I like the m-learning with audio interface	.188	.794	1.206	.295	4.928
I like the m-learning with M/Image interface	.074	.917	1.077	.269	4.317
I like the m-learning with video interface	2.197	.007	8.994	1.829	44.219
I like the m-learning with cartoon interface	.031	.968	1.031	.225	4.730
I like the m-learning with F/Image interface	.391	.603	1.479	.338	6.466
Age	504	.124	.604	.318	1.148
Faculty	982	.161	.375	.095	1.478
Constant	9.171	.134	9613.821		

Table 8 - 13 indicates the performance expectation of students with regards to their believe that accessing course materials in general on their mobile device with an avatar will help them learn the content better than if the same content was presented in a traditional format (face-to-face). The findings here highlight that there is a statistically significant relationship between the outcome and the preference of students to learn with when the "talking head male avatar" as a video (p = 0.007) on their mobile device.

The mean for each of the five statements about performance expectancy when learning English language materials was calculated, then a UNIANOVA test was conducted to find the relationship between the modality of learning and the students" perceived performance expectancy/usefulness. Table 8 - 14 show these statements and the means of the responses:

Question No	Statement	Mean
1	I think that the m-learning would be a more effective way to learn English language than learning face-to- face in classes.	3.37
2	I think m-learning would strengthen my participation when learning English language.	3.53
3	I think having an m-learning interface with an avatar would positively assist me to learn English.	3.51
4	I think having an avatar approach would increase my understanding of information that I have learnt.	3.64
5	I think having an avatar approach would increase my retention of information that I have learnt.	3.51

 Table 8 - 14 Mean for performance expectancy statements

To demonstrate the potential of this approach and its suitability for the application, Table 8 - 15 show the results of these factors when applied learning materials specifically to English language.

Table 8 - 15 UNIANOVA of learning modality variables on English language performance expectancy

Source	Type III Sum	Type III Sum Mean		Sig.
	of Squares	Square		
I like the traditional F2F	.238	.238	.751	.390
I like the m-learning with text interface	.006	.006	.020	.887
I like the m-learning with audio interface	.059	.059	.185	.669
I like the m-learning with male Image interface	3.817	3.817	12.034	.001
I like the m-learning with video interface	7.196	7.196	22.689	.000
I like the m-learning with cartoon interface	.009	.009	.029	.865
I like the m-learning with female image	.499	.499	1.574	.215
Faculty	.324	.324	1.021	.317

What is interesting in this data is that, when the statements indicated English language as the main learning material, the results show that two modality had significant results. There is a statistically significant relationship between the outcome and students'' modality preference of learning through a mobile device with a male static avatar (p = 0.001) and with the video interface (p < 0.001). This hypothesis is moderated by two factors (gender; major).

Gender:

 $H1_a$. Gender will positively moderate the effect of preferred multimedia instruction on the performance expectancy.

Major/subject:

H1_b. Major of study will positively moderate the effect of preferred multimedia instruction on the performance expectancy.

With regards to $H1_a$, male students expectations that they would perform better when using mobile learning with a male static avatar was 0.3 points higher than female students, as shown in the following table.

Table 8 - 16 Performance expectancy when using m-learning with male static avatar for English content

(I)	(J)	Mean	Std.	Sig. ^b	95% Confidence	
Gender	Gender	Difference (I-J)	Error		Interval for Difference ^b	
					Lower	Upper
					Bound	Bound
Female	Male	326*	.119	.007	561	091
Male	Female	.326*	.119	.007	.091	.561

Table 8 - 17 Mean of gender for performance expectancy when usingm-learning with male static avatar for English content

Gender	Mean	Std. Error	95% Confidence Interval		
			Lower Bound	Upper Bound	
Female	3.346 ^a	.083	3.181	3.510	
Male	3.672 ^a	.085	3.504	3.839	

In addition, when learning via mobile devices with animated avatars (video), similarly to the previous result, male students also expected to perform better by 0.3 points higher than female students see Table 8 - 18.

Pairwise Comparisons						
(I)	(J)	Mean	Std.	Sig. ^b	95% Co	nfidence
Gender	Gender	Difference	Error		Interval for	
		(I-J)			Difference ^b	
					Lower	Upper
					Bound	Bound
Female	Male	269*	.115	.021	496	042
Male	Female	.269*	.115	.021	.042	.496

Table 8 - 18 Performance expectancy when using m-learning with animated avatar (video)

Table 8 - 19 Mean of gender for performance expectancy when using m-learning with animated avatar (video)

Gender	Mean	Std. Error	95% Confidence Interval		
			Lower Bound	Upper Bound	
Female	3.375 ^a	.082	3.214	3.537	
Male	3.645 ^a	.081	3.484	3.805	

However, for H1b there was no significant difference between the two major groups Arts and Science when learning via mobile devices with static male avatars and also when learning via mobile devices with animated avatars with P-value = 0.31.

Table of summary 1 Hypothese 1 results					
Hypothesis	Status				
H1 with two modality (video, static image)	Supported				
H1a. Gender	Supported				
H1b. Major/subject	Rejected				

H2. Preferences and Intrinsic Engagement

The second hypothesis states that

Preference for multimedia instruction will positively influence students' intrinsic engagement.

This hypothesis was moderated by two factors (gender; major).

Gender:

 $H2_a$. Gender will positively moderate the effect of preferred multimedia instruction on students' intrinsic engagement.

Major/subject

H2_b.Major of study will positively moderate the effect of preferred multimedia instruction on students' intrinsic engagement.

The UNIANOVA used to test the main hypothesis and the result are illustrated as follows in Table 8 - 20:

Source	Type III Sum	Mean	F	Sig.
	of Squares	Square		
I like the traditional F2F	2.701	2.701	3.239	.074
I like the m-learning with text interface	.463	.463	.555	.457
I like the m-learning with audio interface	2.219	2.219	2.661	.105
I like the m-learning with male Image interface	13.415	13.415	16.085	.000
I like the m-learning with video interface	4.669	4.669	5.599	.019
I like the m-learning with cartoon interface	9.251	9.251	11.093	.001

 Table 8 - 20 Preference for multimedia on engagement

Table 8 - 20 shows that there is a clear trend of increasing preference for interfaces which contain an avatar. Results show that there is a statistically significant relationship between engagement and students" preference to learn via m-learning with the static male avatar (p < 0.001); m-learning with the video avatar (p = 0.01); and m-learning with a cartoon avatar (0.001).

With regard to $H2_{a}$, there are statistically significant differences between genders. Male students" engagement was 0.5 points higher than female students" engagement when learning via mobile devices with content delivered by an avatar as shown in Table 8 - 21.

Pairwise Comparisons							
(I)	(J)	Mean	Std. Error	Sig.b	95% Confidence		
Gender	Gender	Difference (I-J)			Interval for Difference		
					Lower	Upper Bound	
					Bound		
Female	Male	534*	.158	.001	847	221	
Male	Female	.534*	.158	.001	.221	.847	
*. The me	*. The mean difference is significant at the .05 level.						

 Table 8 - 21 Gender moderator for hypothesis 2

 Table 8 - 22 Mean of gender moderating hypothesis 2

Dependent Variable: I think having access to the English materials on my mobile devices would enhance my motivation to learn English language						
Gender	Mean	Std. Error	95% Confidence Interval			
			Lower Bound Upper Bound			
Female	3.214a	.110	2.997	3.432		
Male	3.749a	.111	3.528	3.969		

In contrast, the result of the relationship between these factors and the major/subject as a moderator showed no significant differences between the two disciplines of Arts and Science.

Hypothesis	Status
H2 with three modality (video, static image,	Supported
cartoon)	
H2a Gender	Supported
H2b Major	Rejected

Both male and female students participated in each of the different modalities of learning, and additionally females" students conducted further m-learning lesson with a static female avatar. The result indicated that there is a statistically significant relationship between female students engagement and m-learning with a static female avatar (p = 0.001), as illustrated in the Table8 - 23. The findings also showed no statistically significant differences between the Arts and Science on disciplines of study.

 Table 8 - 23 Preference of female avatar on engagement (Only for female)

Tests of Between-Subjects Effects						
Source	Type III Sum	Mean	F	Sig.		
	of Squares	Square		-		
I like the m-learning with female Image interface	15.115	15.115	12.146	.001		
Major/subject	.105	.105	.084	.772		

H3. Engagement and Enjoyment

In this hypothesis, the researcher was investigating the relationship between the engagement factors and how they can affect students" enjoyment when they have access to materials on mobile devices, and in particular whether or not that would be perceived as a fun interaction between the content and the learner. The hypotheses was stated as:

Intrinsic engagement will positively influence students' enjoyment.

The researcher conducted a UNIANOVA test to explore how the enjoyment of using mlearning can be predicted by intrinsic motivation to have positive correlations in the model. As illustrated earlier in the model, gender and major of study are both moderating this hypothesis as following:

Gender:

H3a. Gender will positively moderate the effect of intrinsic engagement on the students' enjoyment.

Major/subject:

H3b. Major of study will positively moderate the effect of intrinsic engagement on the students' enjoyment.

The tests results are shown in the following table:

Tests of Between-Subjects Effects						
Source	Type III Sum of	Mean	F	Sig.		
Squares		Square				
Engagement 4748	131.120	131.120	53.512	.000		
Gender	3.264	3.264	1.332	.250		
Major/subject	.543	.543	.221	.639		

Table 8 - 24 Enjoyment based on engagement

The result revealed that there is a statistically significant relationship between perceived enjoyment and students" beliefs that having access to English materials on their mobile devices would enhance their motivation to learn English language (p < 0.001). Results showed that enjoyment of using m-learning was predicted by intrinsic motivation.

For the H3a and H3b, the result can be shown from Table8 - 24, that there were no significant differences between the groups, neither for gender or faculty/major of students as (p = 0.12) and (p = 0.6) respectively, with the P-values for both are > 0.05.

Table of summary 3 Hypothesis 3 results				
Hypothesis	Status			
H3. Engagement and Enjoyment	Supported			
H3a. Gender	Rejected			
H3b. Major/subject	Rejected			

H4. Engagement and Performance Expectancy

The fourth hypothesis tested whether or not *the engagement factor influences the performance expectancy factor*. The gender and faculty used as moderators of this hypothesis were as follows:

Gender:

H4a. Gender will positively moderate the effect of intrinsic engagement on the students' performance expectancy.

Major/subject

H4b. Major of study will positively moderate the effect of intrinsic engagement on the students' performance expectancy.

At this point, the general statement of engagement was tested by the UNIANOVA test to predict its influence on the performance expectancy or usefulness of learning. With the result presented in Table 8 - 25:

<i>Table 8 - 25 Engagement on performance expectancy</i>

Tests of Between-Subjects Effects				
Source Type III Sum of		Mean	F	Sig.
Squares		Square		
Engagement q4748	28.247	28.247	35.767	.000
Gender	4.141	4.141	5.244	.023
major/subject	.533	.533	.675	.413

 Table 8 - 26 Gender moderator for hypothesis 4

Pairwise Comparisons						
(I) Gender	(I) Gender (J) Mean Std. Sig.b 95% Confidence					fidence
	Gender Difference Error Interval for Difference					Difference
(I-J) Lower Upper						
					Bound	Bound
Female	Male	455*	.161	.005	774	136
Male	Female	.455*	.161	.005	.136	.774
*. The mean difference is significant at the .05 level.						

Table 8 - 27 Mean of gender moderating hypothesis 4

Gender	Mean	Std. Error	95% Confidence Interval	
			Lower Bound	Upper Bound
Female	1.440a	.112	1.219	1.661
Male	1.895a	.116	1.666	2.125

Table 8 - 28 Mean of major moderating hypothesis 4

Faculty	Mean	Std. Error	95% Confidence Interval	
			Lower Bound	Upper Bound
Science	1.727a	.105	1.520	1.935
Art	1.608a	.111	1.389	1.828

As illustrated from Table 8 - 25, there is a statistically significant relationship between the performance expectancy if learning was via mobile devices with an avatar for any course, and the intrinsic engagement (p < 0.001). In addition, there are significant differences between genders. Male students showed statistically that their expectation of performance was 0.5 higher than for female students, because they engaged with that way of learning. However, regarding the faculty, there was no statistical differences between groups of the two disciplines.

On the other hand, when the specific statement related to the engagement of the English language course was tested to predict its influence on the performance expectancy factor, the result presented in Table 8 - 29:

Table 8 - 29 Engagement on performance expectancy if the course is English					
Source	Type III Sum of	Mean Square	F	Sig.	
Squares					
Engagement q57	269.419	269.419	124.291	.000	
Gender	.023	.023	.011	.917	

7 029

3 2 4 3

074

7 029

Major/subject

T-11. 0 20 E · c .1 1. 1

Table 8 - 30 Mean of gender for the engagement on performance expectancy if the course is English

Gender	Mean	Std. Error	95% Confidence Interval	
			Lower Bound	Upper Bound
Female	6.928 ^a	.173	6.585	7.270
Male	6.882 ^a	.183	6.521	7.243

Table 8 - 31 Mean of major for the engagement on performance expectancy if the course is English

Faculty	Mean	Std. Error	95% Confidence Interval	
			Lower Bound	Upper Bound
Science	7.126 ^a	.163	6.803	7.449
Art	6.683 ^a	.188	6.311	7.055

As Table 8 – 29 shows, there is a significant relationship between the students" performance expectancy when they were learning via English materials and their motivation to access the module materials through mobile phones to learn the English content (p < 0.001). However, for H4a and H4b, there were no significant differences between groups as shown in Table 8 - 30 and Table 8 - 31.

Table of summary 4 hypothesis 4 results

Hypothesis	Status
H4. Engagement and Performance Expectancy	Supported
H4a. Gender	Rejected
H4b. Major/subject	Rejected

H5. Convenience and Performance Expectancy

In this stage, the researcher tested the convenience factor of m-learning to find out whether or not that factor effects the performance expectancy. The hypothesis stated and the two moderators (gender, faculty) used are as follows:

H5. The convenience of m-learning will positively influence students' performance expectancy.

Gender:

H5a. Gender will positively moderate the effect of the convenience of m-learning on students' performance expectancy.

Major/subject:

H5b. Major of study will positively moderate the effect of the convenience of *m*-learning on students' performance expectancy.

When the UNIANOVA was used to test these hypotheses, the results were presented as follows in Table 8 - 32:

Table 6 - 52 ettimite infor convenience on performance expectance,					
Source	Type III Sum	Mean Square	F	Sig.	
	of Squares				
(convenient) Meanq53_54_55	26.834	26.834	33.138	.000	
Gender	5.745	5.745	7.094	.009	
Major/subject	.080	.080	.098	.754	

Table 8 - 32 UNIANOVA for convenience on performance expectancy

Table 8 - 33	Mean of gen	der moderati	ng hypothesis 5
Gender	Mean	Std. Error	95% Confidence I

Gender	Mean	Std. Error	95% Confidence Interval	
			Lower Bound	Upper Bound
Female	1.338 ^a	.105	1.131	1.545
Male	2.009 ^a	.106	1.800	2.219

From Table 8 - 32, there is a significant relationship between the performance expectancy factor and the convenience factor (p < 0.001).

With regard to gender as a moderator, there are statistically significant differences between genders. The results show that male students perceived m-learning to be more convenient for them with a 0.7 point higher mean than female students. Therefore, male students expect m-learning will increase their usefulness and performance compared to female students. However, regarding the major factor when moderating the current hypothesis, there were no differences between the two discipline groups.

Table of summary 5 hypothesis 5 results	
Hypothesis	Status
H5. Convenience and Performance Expectancy	Supported
H5a. Gender	Supported
H5b.Major/subject	Rejected

— 11 *a* **51 4** • **5**

H6. Performance Expectancy and Behavioural Intention

Based on the research model, the hypotheses number 6 stated that:

The performance expectancy will positively influence students' behavioural intention to use m-learning.

The two moderators added to be as following:

Gender:

H6a. Gender will positively moderate the effect of the performance expectancy on the behavioural intention to use m-learning.

Major/subject:

H6b. Major of study will positively moderate the effect of the performance expectancy on the behavioural intention to use m-learning.

The UNIANOVA test was conducted to investigate this hypothesis and the results are illustrated in Table 8 - 34:

Source	Dependent Variable	Type III Sum	Mean	F	Sig.
	Intention to use m-	of Squares	Square		
	learning				
(Expectancy) sumq56_58	q7172	289.643	289.643	137.732	.000
	q818283	443.615	443.615	101.350	.000
Gender	q7172	.048	.048	.023	.880
Gender	q818283	4.689	4.689	1.071	.302
Major/subject	q7172	.031	.031	.015	.903
Wajoi/Subject	q818283	.717	.717	.164	.686

Table 8 - 34 Performance expectancy on behavioural intention to use m-learning

* q7172 & q818283 are the intention to use m-learning statements

As the above table reveals, if there was a statistically significant relationship between the intention to use m-learning and students" performance expectations of learning via mobile devices, it would effectively strength their participation to learn English language compared to the traditional face-to-face learning (p < 0.001). However, regarding the H6a and H6b, there were no significant differences whether between the genders and the two faculties.

Table of summary 6 hypothesis 7 results

		Hypothesis			Status
Н6.	Performance	Expectancy	and	Behavioural	Supported
Inten	tion				
Н6а.	Gender				Rejected
H6b.	Major/subject				Rejected

H7. Intrinsic Engagement and Behavioural intention

Similarly, another important factor, "intrinsic engagement", was used as a predictor to influence the behavioural intention to use m-learning. Based on the research model, the

two moderators" factors (genders and faculty) were added into this hypothesis as following:

Gender:

H7a. Gender will positively moderate the effect of intrinsic engagement on the students' behavioural intention to use m-learning.

Major/subject:

H7b. Major of study will positively moderate the effect of intrinsic engagement on the students' behavioural intention to use m-learning.

The result of the hypothesis test is presented in the Table 8 - 35:

Source	Dependent Variable:	Type III Sum	Mean	F	Sig.
	Intention to use m-le.	of Squares	Square		•
Engaged when used m-learning	q7172	141.748	141.748	45.234	.000
Engaged when used in-learning	q818283	306.959	306.959	57.163	.000
Gender	q7172	10.395	10.395	3.317	.071
Gender	q818283	.954	.954	.178	.674
Major/subject	q7172	.158	.158	.050	.823
Major/Subject	q818283	2.629	2.629	.490	.485

Table 8 - 35 Engagement on behavioural intention to use m-learning

Table 8 - 36 Mean of gender moderating hypothesis 7

Dependent Variable	Gender	Mean	Std. Error	95% Confidence Interval						
Intention to use m-learning				Lower Bound	Upper Bound					
a71 72	Female	7.301 ^a	.226	6.853	7.748					
q71-72	Male	6.939 ^a	.241	6.464	7.414					
q81-83	Female	5.244 ^a	.296	4.658	5.830					
481-85	Male	5.148 ^a	.315	4.526	5.771					
a. Covariates appearing in the model are evaluated at the following values: q4748 = 2.8733, Age =										
19.07.										

* q71-72 & q81-82-83 are the intention to use m-learning statements

As shown in Table 8 - 35, there is a statistically significant relationship between the dependent variable (intention to use m-learning) and the students" engagement to learn via mobile devices (p < 0.001). In addition, there is a statistically significant relationship between the dependent variable and the students" beliefs that having access to the English materials on their mobile devices would enhance their motivation to learn English language (p < 0.001). However, there were no significant differences between gender and faculty.

Although the gender is not statistically significant (p = 0.07), the interesting findings from the second table is that the mean of female students (7.3) was higher than the mean

of male students (6.9). In other words, those female students who were intending and planning to use the m-learning in future were slightly more engaged with the approach of m-learning compared to the male students.

Table of summ	ary 7	hypothesis	7	results	

Hypothesis	Status
H7.Intrinsic Engagement and Behavioural intention	Supported
H7a. Gender	Rejected
H7b. Major/subject	Rejected

H8. Enjoyment and Behavioural Intention

Further to these predictor factors and based on the research model, there was another important factor (Enjoyment of learning) as an independent variable which might have an influence on the behavioural intention to use m-learning. The hypothesis was stated as:

Enjoyment will positively influence students' behavioural intention to use m-learning.

The researcher conducted a UNIANOVA test to explore how the students" intention to use m-learning can be predicted by enjoyment to have positive correlations in the model. As illustrated earlier in the model, the gender and major of the study are moderating this hypothesis to be as following:

Gender:

H8a. Gender will positively moderate the effect of enjoyment on the students' behavioural intention to use m-learning.

Major/subject:

H8b. Major of study will positively moderate the effect of enjoyment on the students' behavioural intention to use m-learning.

The tests findings are presented in the following table.

Source	Dependent Variable	Type III Sum	Mean Square	F	Sig.
	Intention to use m-	of Squares			
	learning				
(Enjoyment) q46	q7172	167.927	167.927	57.593	.000
	q818283	245.821	245.821	42.599	.000
Gender	q7172	6.811	6.811	2.336	.129
Gender	q818283	.437	.437	.076	.783
Faculty	q7172	.500	.500	.171	.679
raculty	q818283	4.374	4.374	.758	.385

Table 8 - 37 Enjoyment on behavioural intention to use m-learning

The above table revealed that there is a statistically significant relationship between the intention to use m-learning and the students" enjoyment in which they believed accessing the materials on mobile devices was a fun interaction between content and learner (p < 0.001). Results showed that the intention of using m-learning was significantly predicted by enjoyment. For the two moderators, the results indicate no significant difference between the males and females groups or the Arts and Science students.

Table 1	Hypotheses	8	results	summery

Hypothesis	Status
H8. Enjoyment and Behavioural Intention	Supported
H8a. Gender	Rejected
H8b. Major/subject	Rejected

H9. The Engagement and Pedagogical Performance

As the research objective is aiming to increase and improve the students" pedagogical performance and outcome, the research model assumed in (H9)as follows:

The intrinsic engagement positively influence the students' effectiveness.

As illustrated earlier in the model, the gender and major of the study are moderating this hypothesis to be as following:

Gender:

H9a. Gender will positively moderate the effect of intrinsic engagement on the students' effectiveness.

Major/subject:

H9b. Major of study will positively moderate the effect of intrinsic engagement on the students' effectiveness.

After running the test of these hypotheses, the results were as follows in Table 8 - 38:

Source	Post-test of each method as	Mean	Sig.
		Square	
	m-learning with text	58.277	.062
	m-learning with static male avatar	55.810	.043
Encocomont	m-learning with cartoon avatar	31.497	.025
Engagement	m-learning with audio avatar	22.747	.064
	m-learning with video avatar	66.826	.028
	face-to-face teaching	2.529	.492
	m-learning with text	142.852	.001
	m-learning with static male avatar	27.332	.155
Gender	m-learning with cartoon avatar	39.098	.013
Gender	m-learning with audio avatar	51.468	.006
	m-learning with video avatar	138.794	.002
	face-to-face teaching	4.611	.354
	m-learning with text	14.329	.266
	m-learning with static male avatar	7.553	.454
Faculty	m-learning with cartoon avatar	26.997	.038
racuity	m-learning with audio avatar	25.201	.052
	m-learning with video avatar	1.335	.754
	face-to-face teaching	.388	.788

 Table 8 - 38 Engagement on effectiveness (actual test performance)

The Table 8 - 38 indicates that there are statistically significant relationships between the performance outcome and the students" engagements to learn through m-learning when the interface was: a static male image avatar (p = 0.04), a cartoon avatar (p = 0.02) and a video interfaces avatar (p = 0.02). This means that students performed well in these three ways of learning because they had engaged with them while learning the English lessons. Moreover, regarding the hypothesis when moderated by gender, there were significant differences between genders in their test scores. Female students outperformed male students by 2 points than in m-learning with a textual interface, by 0.9 points higher in m-learning with cartoon interface, by 1.5 points higher in mlearning with audio, and by 2.5 points higher in m-learning with videos, as shown in Table 8 - 39.

Pairwise Comparisons							
Learning method	(I) Gender	(J) Gender	Mean	95% Confidence Interval for			
** Through mobile device			Difference	Differ	ence		
5			(I-J)	Lower Bound	Upper Bound		
Text **	Female	Male	2.069*	.834	3.303		
lext **	Male	Female	-2.069*	-3.303	834		
static male avatar **	Female	Male	1.480	.148	2.813		
	Male	Female	-1.480	-2.813	148		
cartoon avatar **	Female	Male	.987*	.084	1.889		
	Male	Female	987*	-1.889	084		
audio avatar **	Female	Male	1.512*	.580	2.444		
audio avatar ··	Male	Female	-1.512*	-2.444	580		
video avatar **	Female	Male	2.507*	1.165	3.849		
video avatar **	Male	Female	-2.507*	-3.849	-1.165		
face to face teaching	Female	Male	.673	168	1.513		
face-to-face teaching	Male	Female	673	-1.513	.168		

 Table 8 - 39 Mean of gender moderating hypothesis 9

Based on estimated marginal means

*. The mean difference is significant at the .05 level.

The second sub-hypothesis concerns weather the moderator is the major/subject. The result showed that there were statistically significant differences between the two groups, with Science students engaging and outperforming Arts students by 0.9 points in m-learning with a cartoon interface (avatar).

Because the female students undertook an additional lesson through the mobile device as a static female avatar, the test was done for this type of learning separately and the results presented in Table 8 - 40.

performance, only for female)							
Source	Type III Sum of	Mean Square	F	Sig.			
	Squares						
Engagement 57	3.170	3.170	.270	.605			
Major/subject	61.400	61.400	5.232	.025			

Table 8 - 40 Engagement on effectiveness of female avatar (actual test performance, only for female)

The results indicate that there was no statistically significant relationship between the results (test scores) of that type of learning and the engagement (p = 0.6); however, there were significant differences between the two major groups. Science students engaged and performed by 1.7 points higher than Arts students in m-learning with a static female avatar. See the mean differences of this findings in this table.

	Pairwise Comparisons						
(I)	(J)	Mean	Std.	Sig.b	95% Co	nfidence	
Faculty	Faculty	Difference	Error		Interval for Differenc		
		(I-J)			Lower	Upper	
					Bound	Bound	
Science Art 1.662* .727 .025 .214 3.11					3.110		
Arts	Science	-1.662*	.727	.025	-3.110	214	
*. The m	ean differ	ence is signific	cant at th	e .05 leve	el.		

 Table 8 - 41 Mean of major for Engagement on effectiveness of female

 avatar (actual test performance, only for female)

H10. Preference and Pedagogical Performance.

This hypothesis investigated whether or not the preference of face-to-face or any multimedia instruction has an influence on the pedagogical performance (effectiveness). The hypothesis (H10) was stated as follows:

Preference for multimedia instruction will positively influence students' effectiveness.

The following moderators added:

Gender:

 $H10_a$. Gender will positively moderate the effect of preferred multimedia instruction on the effectiveness.

Major/subject:

H10_b. Major of study will positively moderate the effect of preferred multimedia instruction on the effectiveness.

The results of these hypotheses presented in Table 8 - 42, reveal that there is a statistically significant relationship between the students" preference for using mlearning video interface and the result of face-to-face test (p = 0.001). There is also a statistically significant relationship between gender and the outcomes which is the result of using face-to-face learning (p = 0.001). Female students preferred and outperformed male students by 0.7 points in face-to-face learning method compared to the m-learning. The results presented as in the follows Table 8 - 42:

Source Way of learning preferred	Dependent Variable	Mean Square	F	Sig.
· · · ·	m-learning with text	.110	.010	.922
	m-learning with static male avatar	.023	.002	.968
	m-learning with cartoon avatar	1.206	.175	.676
Face-to-face method	m-learning with audio avatar	2.529	.362	.549
	m-learning with video avatar	1.890	.144	.705
	face-to-face teaching	5.867	1.242	.267
	m-learning with text	.105	.009	.924
	m-learning with static male avatar	16.308	1.152	.285
no looming suith tout	m-learning with cartoon avatar	1.314	.191	.663
m-learning with text	m-learning with audio avatar	9.766	1.396	.240
	m-learning with video avatar	6.991	.533	.467
	face-to-face teaching	.164	.035	.853
	m-learning with text	4.390	.386	.536
	m-learning with static male avatar	41.237	2.912	.090
n loomino mith on line and a	m-learning with cartoon avatar	.037	.005	.942
m-learning with audio avatar	m-learning with audio avatar	1.536	.220	.640
	m-learning with video avatar	17.686	1.349	.248
	face-to-face teaching	.028	.006	.939
	m-learning with text	10.614	.933	.336
m-learning with static male avatar	m-learning with static male avatar	11.503	.812	.369
	m-learning with cartoon avatar	2.091	.304	.582
	m-learning with audio avatar	11.872	1.697	.195
	m-learning with video avatar	.214	.016	.899
	face-to-face teaching	14.341	3.036	.084
	m-learning with text	1.147	.101	.751
	m-learning with static male avatar	45.176	3.190	.076
	m-learning with cartoon avatar	4.829	.702	.404
m-learning with video avatar	m-learning with audio avatar	1.592	.228	.634
	m-learning with video avatar	4.620	.352	.554
	face-to-face teaching	58.327	12.349	.001
	m-learning with text	6.171	.542	.463
	m-learning with static male avatar	1.057	.075	.785
1	m-learning with cartoon avatar	.470	.068	.794
m-learning with cartoon avatar	m-learning with audio avatar	11.825	1.691	.196
	m-learning with video avatar	27.997	2.135	.146
	face-to-face teaching	4.356	.922	.339
	m-learning with text	.596	.052	.819
	m-learning with static male avatar	5.782	.408	.524
Conten	m-learning with cartoon avatar	2.109	.307	.581
Gender	m-learning with audio avatar	2.287	.327	.568
	m-learning with video avatar	20.055	1.530	.218
	face-to-face teaching	54.069	11.447	.001
	m-learning with text	33.076	2.908	.091
	m-learning with static male avatar	26.710	1.886	.172
	m-learning with cartoon avatar	35.482	5.158	.025
Faculty/ Major	m-learning with audio avatar	14.694	2.101	.150
	-		1.029	.312
	m-learning with video avatar	13.487	1.029	.312

Table 8 - 42 Preference of mode of learning on effectiveness (actual test performance)

In addition, from Table 8 - 42, there are statistically significant differences between the groups of major toward the outcomes when the learning was by using m-learning as a cartoon interface (p = 0.02). Science students preferred and outperformed Art students by 1.1 points in m-learning with cartoon avatars compared to the other ways of learning.

Table of summary 8 hypothesis 10 results

Independent (preference)	Dependant (outperformed)	Status
M-learning with video avatar	Face-to-face leaning	Supported
Female gender	Face-to-face leaning	Supported
Science faculty	M-learning with Cartoon avatar	Supported

H11. Pedagogical Performance and Behavioural Intention

The last hypothesis from that research model states that:

The effectiveness of m-learning will positively influence students' behavioural intention to use m-learning.

The moderators have been added to this hypothesis. The sub-hypotheses will be as the following:

Gender:

H11a. Gender will positively moderate the effect of effectiveness on the students' behavioural intention to use m-learning.

Major/subject:

H11b. Major of study will positively moderate the effect of effectiveness on the students' behavioural intention to use m-learning.

The results of these hypotheses are presented in Table 8 - 43.

Source (post-test for)	Dependent	Type III Sum	Mean	F	Sig.
	Variable	of Squares	Square		
m-learning with text	q7172	11.904	11.904	2.816	.096
in-learning with text	q818283	.491	.491	.065	.800
m-learning with static male avatar	q7172	.643	.643	.152	.697
III-learning with static male avata	q818283	3.093	3.093	.407	.525
m-learning with cartoon avatar	q7172	.987	.987	.233	.630
in-learning with cartoon avatai	q818283	29.726	29.726	3.911	.050
m-learning with audio avatar	q7172	1.152	1.152	.273	.602
	q818283	1.200	1.200	.158	.692
m learning with video aveter	q7172	11.201	11.201	2.650	.106
m-learning with video avatar	q818283	1.488	1.488	.196	.659
face-to-face teaching	q7172	1.467	1.467	.347	.557
lace-to-lace teaching	q818283	4.259	4.259	.560	.455
m-learning with static female avatar	q7172	3.699	3.699	.8002	.373
In-learning with static lennale avatai	q818283	2.584	2.584	.308	.580
Condon	q7172	.114	.114	.027	.870
Gender	q818283	20.452	20.452	2.691	.103
Maion/archiaat	q7172	.076	.076	.018	.894
Major/subject	q818283	.000	.000	.000	.995

Table 8 - 43 Effectiveness (actual test performance) on behavioural intention to use m-learning

Q7172 & q818283= intention to use factor.

Even though the results of (H10) show that the students" preference to perform well using m-learning with video, and the cartoon avatar was not significant in term of performance, the results did not agree with this. When the actual post-test results were tested as a direct predictor on the behavioural intention to use (H11), the result revealed that the preference of cartoon avatars has an influence on the intention to use m-learning, with (p = 0.05). However, for those students who preferred video avatars, their video test results did not show evidence of the influence of that way of learning on the intention to use, with (p = 0.1).

For the H11_{a&b}, because learning through the cartoon avatar was statistically significant, the gender and faculty have been investigated further. The result showed that there is statistically a significant difference between genders towards their preference for cartoon avatars, which influenced their intention to use m-leaning (p = 0.001). Table 8 - 44 shows that male students intended to use m-learning with cartoon avatars by 1.6 points higher than female students. However, in the second sub-hypothesis, which is about the major/subject, the results show that no significant difference between groups.

Table 8 - 44 Gender moderator for hypothesis 11

Pairwise Comparisons							
(I)	(I) (J) Mean Std. Sig.b 95% Confidence						
Gender	Gender	Difference	Error		Interval for Difference		
		(I-J)			Lower	Upper	
					Bound	Bound	
Female	Male	-1.550*	.442	.001	-2.423	677	
Male	Female	1.550*	.442	.001	.677	2.423	
*. The me	ean differer	nce is signific	cant at th	e .05 level.			

Table 8 - 45 Mean of gender for effectiveness (actual test performance) on behavioural intention to use m-learning

Gender	Mean	Std. Error	95% Confidence Interval	
			Lower Uppe	
			Bound	Bound
Female	4.446a	.311	3.830	5.061
Male	5.995a	.313	5.376	6.615

Table of summary 9 hypothesis 11 results

Independent (effectiveness)	Dependant (outperformed)	Status
M-learning with cartoon avatar	Intention to use m-learning	Supported
Male gender (for cartoon)	Intention to use m-learning	Supported
Major/subject	Intention to use m-learning	Rejected

Up to this stage, the analysis and results of the first case study ended even though the discussion and interpretations of these data will be presented in details in Chapter 9. The following table 10, provides a summary of the research hypotheses results.

H _n	Hypothesis Statement	1	ain H. Sig. value	n G	Sig. noder. ænder		Sig. 10der. najor		
H ₁	Preference for multimedia instruction will positively influence students" performance expectancy.	I V	0.001 <0.001		0.007	-	N/S		
\mathbf{H}_2	Preference for multimedia instruction will positively influence students "intrinsic engagement	I <0.001 V 0.01 C 0.001			0.001		N/S		
H ₃	Intrinsic engagement will positively influence students" enjoyment.	<	0.001		N/S		N/S		
H ₄	Intrinsic engagement will positively influence students" performance expectancy.	<	0.001		N/S		N/S		
H_5	The convenience of m-learning will positively influence students "performance expectancy.	<0.001		0.009		0.009 N		N/S	
H ₆	The performance expectancy will positively influence students "behavioural intention to use m-learning.	<0.001		N/S		N/S			N/S
H ₇	Intrinsic engagement will positively influence students" behavioural intention to use m-learning.	< 0.001		N/S			N/S		
H ₈	Enjoyment will positively influence students" behavioural intention to use m-learning.	<	0.001		N/S		N/S		
H9	Intrinsic engagement will positively influence students" effectiveness.	I C V	0.04 0.02 0.02	T C A V	0.001 0.01 0.006 0.002	С	0.03		
H ₁₀	Preference for multimedia instruction will positively influence students" effectiveness.	v	0.001	F	0.001	С	0.02		
H ₁₁	The effectiveness of m-learning will positively influence students" behavioural intention to use m- learning. = Moderator (gender, subject); V= video; A= audio; C= ca	С	0.05	С	0.001		N/S		

Moder. = Moderator (gender, subject); V= video; A= audio; C= cartoon; I= image; T= text; F= face-to-face.

N/S= Not Significant

The results and analysis of the second case study will be illustrated in section 8.6.

8.6 Results of Experiment (2) – Questionnaire

8.6.1 **Demographic Results**

This section presents the analysis of the second experiment on demographic information. This includes gender, age, faculty of study/major, average period of having mobile devices, average hours spent on mobile daily usage and the type of mobile

device and some activities related to mobile devices. For this part of the research, the researcher measured statistically the participants" frequency and percentages.

Gender

The participants in these experiments were 38:62 according to the gender, with 39 male students and 64 female students. Figure 8 - 11 illustrates the gender data.



Figure 8 - 11 percentage of participants' gender

Faculty Groups

Participants responses to the choices of which faculty they were willing to continue their studies is illustrated in Figure 8 - 12. 53% of the participants currently classified in the Science departments and 47% of the participants were from the Arts department foundational year. It should be mentioned the percentages included both genders.



Figure 8 - 12 percentage of participants' faculty

8.6.2 Participants' Post Questionnaire

Students have been asked the following questions:

In your previous and/or current education, have you experienced m-learning:

Options	Frequency	Percentage
No	55	53.4
Maybe	21	20.4
Yes	27	26.2
Total	103	100.0

In which order did you find the lessons most engaging:

	Audio	Video	Cartoon
Female	37.5%	48.4%	14.1%
Male	30.8%	61.5%	7.7%
Total	35.0%	53.4%	11.7%



In which order do you think has helped you learnt most effectively:

	Audio	Video	Cartoon
Female	40.6%	43.8%	15.6%
Male	43.6%	53.8%	2.6%
Total	41.7%	47.6%	10.7%



Cronbach's Alpha was calculated in this experiment again by using SPSS software version 21 (see Table 8 - 46).

No	Sections	Number of Questions	Cronbach's Alpha
If de	livery was via mobile device as an audio mode	Questions	
1	Engagement	2	0.70
2	performance expectation	3	0.77
If de	livery was via mobile device as an video mode		
3	Engagements	2	0.79
4	performance expectation	5	0.86
If de	livery was via mobile device as an cartoon mode		
5	Engagement	2	0.85
6	performance expectation	3	0.89
Beha	vioural intention		
7	Intention to use m-learning	3	0.88

Table 8 - 46, indicates that the reliability coefficients are acceptable for conducting the research procedures.

8.6.3 **Results of Research Hypotheses - Experiment (2)**

H1. Preferences and Engagement

The first hypothesis states as follows:

Preference for multimedia instruction will positively influence students' intrinsic engagement.

In order to determine the relationship of these two factors, the UNIANOVA test was used. This type of test shows statistically whether or not these types of learning have a significant influence on the interface engagement. The result of testing this hypothesis is illustrated in Table 8 - 46.

Table 8 - 47 Preference for multimedia on engagement							
I like having an m-learning	Type III Sum of	Mean Square	F	Sig.			
interface with:	Squares						
Voice (audio) media instruction	.224	.224	.372	.543			
Video media instruction	2.326	2.326	3.868	.051			
Cartoon media instruction	14.900	14.900	18.994	.000			

Table 8 - 47 Preference for multimedia on engagement

The Table 8 - 47 shows that there were statistically significant relationships between the engagement factor of learning when the mode of delivery as a cartoon and video media instructions with (p < 0.001; p = 0.05) respectively.

Gender:

In order to know whether or not gender shows differences between those two factors (preference for the multimedia type) and (engagement), the hypothesis states the:

Gender will positively moderate the effect of preferred multimedia instruction on the intrinsic engagement.

The result of analysing this hypothesis is illustrated in Table 8-48:

<i>Table</i> 6 - 46	Table 8 - 48 Genaer moderator for hypothesis 1									
Mode of	(I)	(J)	Mean	Std.	Sig. ^a	95% Confidence				
delivery	Gender	Gender	Difference	Error		Interval for Difference				
			(I-J)			Lower	Upper			
						Bound	Bound			
Audio	Male	Female	.045	.158	.775	269	.359			
Video	Male	Female	066	.158	.676	379	.247			
Cartoon	Male	Female	237	.183	.198	600	.126			

Table 8 - 48 Gender moderator for hypothesis 1

The findings from Table 8 - 48 shows that there were no significant difference between genders in the three types (audio; video; cartoon) of delivery (p = 0.7; 0.6; 0.1).

When the current hypothesis moderated with the major, the statement of the hypothesis would be as follows:

Major/subject:

*H1*_b. *Major of study will positively moderate the effect of preferred multimedia instruction on the intrinsic engagement.*

The result of analysing this hypothesis is illustrated in the following Table 8 - 49:

10000 17	Table 6 Ty major moderator for hypothesis 1									
Mode of	(I)	(J)	Mean	Std. Error	Sig. ^a	95% Confidence				
delivery	Faculty	Faculty	Difference			Interval for Difference				
			(I-J)			Lower	Upper			
						Bound	Bound			
Audio	Science	Arts	.300	.154	.054	006	.606			
Video	Science	Arts	.198	.156	.207	111	.506			
Cartoon	Science	Arts	.534*	.178	.003	.181	.887			

Table 8 - 49 Major moderator for hypothesis 1

From Table 8 - 49, the findings indicate that there is only a statistically significant difference between the engagement of students when the hypothesis is moderated by the faculty if learning was by cartoon multimedia (p = 0.003). The science students were higher than Arts student by 0.5 points when the learning was undertaken through a mobile device with a cartoon avatar interface, as shown in the following table:

Table 8 - 50 Mean of major moderating hypothesis 1									
Faculty	Mean	Std. Error	95% Confidence Interval						
			Lower Bound	Upper Bound					
Science	3.806 ^a	.121	3.567	4.046					
Arts	3.272 ^a	.134	3.005	3.539					
a. Covaria	tes appearing	g in the mod	el are evaluated	at the following					
values: I liked having an m-learning interface with Cartoon media									
instruction	= 3.019.								

Table 8 - 50 Mean of major moderating hypothesis 1

H2. Preferences and Performance Expectancy

The second hypothesis states that:

Preference for multimedia instruction will positively influence students' performance expectancy.

The result of analysing this hypothesis is illustrated in the following Table 8 - 51:

Table 8 - 51 Preference for multimedia on performance expectancy							
I like having an m-learning	Type III Sum of	Mean	F	Sig.			
interface with:	Squares	Square					
Voice (audio) media instruction	1.438	1.438	2.413	.124			
Video media instruction	.860	.860	1.425	.235			
Cartoon media instruction	10.459	10.459	14.815	.000			

Table 8 - 51 Preference for multimedia on performance expectancy

The result from Table 8 – 51 indicates that there was a statistical significant relationship between students" performance expectancy and their preferred multimedia when learning was via mobile device only with the cartoon avatar (p < 0.001). However, for the other modes of delivery (audio and video), no there were statistically significant differences between the types of learning and the performance expectancy.

This hypothesis is moderated by two factors (gender and faculty). The test of this hypothesis was conducted to investigate the relationship and differences between participants.

Gender:

 $H2_a$. Gender will positively moderate the effect of preferred multimedia instruction on the performance expectancy.

The result of analysing this hypothesis is illustrated in the following Table 8 - 52:

Table 8 - 52 Gender moderator for hypothesis 2

Mode of	(I)	(J)	Mean	Std. Error	Sig. ^a	95% Confidence	
delivery	Gender	Gender	Difference			Interval for Difference	
			(I-J)			Lower	Upper
						Bound	Bound
Audio	Male	Female	.008	.157	.961	305	.320
Video	Male	Female	.141	.158	.375	454	.173
Cartoon	Male	Female	.301	.173	.086	645	.043

Table 8 - 52 shows that there were no significant differences between participants" gender for the three types of learning (audio; video; cartoon).

Major/subject:

 $H2_{b}$. Major of study will positively moderate the effect of preferred multimedia instruction on the performance expectancy.

The result of analysing this hypothesis is illustrated in the following Table 8 - 53:

<i>Tuble</i> 0 - <i>J</i> .	Tuble 6 - 55 Major moderator for hypothesis 2									
Mode of	(I) Faculty	(J) Faculty	Mean	Std.	Sig. ^a	95% Confidence				
delivery			Difference	Error		Interval for Difference				
			(I-J)			Lower	Upper			
						Bound	Bound			
Audio	Science	Arts	.262	.153	.091	043	.566			
Video	Science	Arts	.094	.156	.548	215	.403			
Cartoon	Science	Arts	.395*	.169	.021	.060	.730			

Table 8 - 53 Major moderator for hypothesis 2

From the above table, the result indicates that there is only a statistically significant difference between the engagement of students when that hypothesis was moderated by the major if learning was by cartoon multimedia (p = 0.02). The Science students were higher than Arts student by 0.4 points when learning through m-learning with a cartoon interface, as shown in the following table:

Table 8 - 54 Mean of major moderating for hypothesis 2									
Faculty	Mean	Std. Error	95% Confide	ence Interval					
			Lower Bound	Upper Bound					
Science	3.855a	.115	3.628	4.083					
Arts	3.461a	.128	3.208	3.714					
a. Covaria	tes appearing	g in the mod	el are evaluated	at the following					
values: I liked having an m-learning interface with Cartoon media									
instruction	= 3.019.								

H3. Engagement and Performance Expectancy

The third hypothesis states that:

will positively influence students' Intrinsic engagement performance expectancy.

The result of analysing this hypothesis is illustrated in the following Table 8 - 55:

Source	Dependent Variable: Performance expectancy if	Type III Sum of Squares	Mean Square	F	Sig.
	learning via	1	1		
	Audio	1.915	1.915	5.627	.020
Engagement	Video	10.389	10.389	28.073	.000
	Cartoon	43.488	43.488	171.908	.000

 Table 8 - 55 Engagement on performance expectancy

From Table 8 - 55, the results shows that there were statistically significant relationships between engagement and performance expectancy when learning via audio, video, and cartoon (p = 0.02; p < 0.001; p < 0.001). However, in terms of that hypothesis being moderated by gender, the hypothesis was stated and is presented as follows:

Gender:

 $H3_a$. Gender will positively moderate the effect of intrinsic engagement on the students' performance expectancy.

The result of analysing this hypothesis is illustrated in the following Table 8 - 56:

Performance	(I)	(J)	Mean	Std.	Sig.	95% Co	onfidence
expectancy if	Gender	Gender	Difference	Error		Inter	val for
learning via			(I-J)			Difference	
						Lower	Upper
						Bound	Bound
audio	Male	Female	.036	.122	.766	205	.278
video	Male	Female	.133	.127	.297	119	.385
cartoon	Male	Female	.134	.105	.204	342	.074

 Table 8 - 56 Gender moderator for hypothesis 3

The findings of the above table show that there were no significant differences between participants in terms of gender in the three ways of learning: p-value = 0.7 for audio; p-value = 0.2 for video; p-value = 0.2 for cartoon. Furthermore, if the current hypothesis is moderated by major, the statement of the hypothesis and its findings can be illustrated as follows:

Major/subject:

H3_b. Major of study will positively moderate the effect of intrinsic engagement on the students' performance expectancy.

The result of analysing this hypothesis is illustrated in the following Table 8 - 57:

Table 8 - 57 Major moderator for hypothesis 3

Performance	(I) Faculty	(J) Faculty	Mean	Std.	Sig. ^a	95% Co	onfidence
expectancy if			Difference	Error		Inter	val for
learning via			(I-J)			Diffe	erence ^a
						Lower	Upper
						Bound	Bound
audio	Science	Arts	016	.121	.898	256	.225
video	Science	Arts	076	.126	.550	327	.175
cartoon	Science	Arts	012	.105	.910	219	.196

The above table shows that there were no any significant differences between participants in term of two different faculties in the three ways of learning.

H4. Interactivity and Engagement

The fourth hypothesis states that:

The interaction activities in m-learning will positively influence students' intrinsic engagement.

This hypothesis has been tested using the UNIANOVA technique. The results appear in the following Table 8 - 58:

Tuble 0 50 Inter	denviry elements on eng				
Source	Dependent Variable:	Type III	Mean	F	Sig.
	Engagement if	Sum of	Square		
	learning via	Squares			
Self-assessment	audio	8.088	8.088	15.590	.000
and feedback	video	9.622	9.622	17.975	.000
interaction	cartoon	12.902	12.902	15.947	.000

Table 8 - 58 Interactivity elements on engagement

The above table shows there were statistically significant relationships between the engagement of students and their interactions, which was by the self-assessment and feedback elements while learning via the three ways of delivery: audio with p < 0.001, video with p < 0.001, and cartoon with p < 0.001.

When the H_{4a} was moderated by the gender, the statement of the hypothesis illustrated as follows:

Gender:

Gender will positively moderate the effect of interaction activities in m-learning on students' intrinsic engagement.

The results of that investigation indicate that there were no significant differences between participants in term of gender (p-value = 0.8; 0.3; 0.1) for the three ways of delivery. The table 8 - 59, provides the evidence of that:

Table 8 - 59 Gender moderator for hypothesis 4

Performance	(I)	(J)	Mean	Std.	Sig.	95% Co	onfidence
Engagement	Gender	Gender	Difference	Error		Inter	val for
if learning via			(I-J)			Diffe	erence
						Lower	Upper
						Bound	Bound
audio	Male	Female	.031	.148	.832	262	.325
video	Male	Female	.138	.150	.360	160	.436
cartoon	Male	Female	276	.185	.138	642	.091

In addition, for the second sub-hypothesis, which under the fourth hypothesis is moderated by the faculty, stated as follows:

Major/subject:

Major of study will positively moderate the effect of interaction activities through m-learning on the students' intrinsic engagement.

The statistical test has been undertaken and the result appears in the following Table 8 - 60:

Dependent	(I)	(J)	Mean	Std.	Sig	95% Con	fidence
Variable:	Faculty	Faculty	Difference	Error	_	Interval for	
Engagement if	-	-	(I-J)			Differ	ence
learning via						Lower	Upper
						Bound	Bound
audio	Science	Arts	.030	.154	.847	275	.334
video	Science	Arts	008	.156	.960	317	.302
cartoon	Science	Arts	.343	.192	.076	037	.724

 Table 8 - 60 Major moderator for hypothesis 4

The result indicates that there were no significant differences between participants in terms of different faculties when that factor used as a moderator between interaction activities through m-learning on the students" intrinsic engagement. The p-value of the audio mode of delivery is 0.8; for video avatar is 0.9 and lastly for the cartoon avatar is 0.1.

H5. Interactivity and Performance Expectancy

The fifth hypothesis states that:

The interaction activities in m-learning will positively influence students' performance expectancy.

The UNIANOVA test has been conducted on this hypothesis and the result illustrated in the following table:

Tuble 6 - 01 Intel	eractivity elements on performance expectancy						
Source	Dependent Variable:	Type III	Mean	F	Sig.		
	Performance expectancy	Sum of	Square				
	if learning via	Squares					
Self-assessment	audio	7.116	7.116	13.336	.000		
and feedback	video	3.682	3.682	6.424	.013		
interaction	cartoon	8.792	8.792	12.341	.001		

The result of analysing this hypothesis is illustrated in the following Table 8 - 61:

The Table 8 - 61 presents the significant evidence of the three ways of delivery if they have interaction activities elements which would positively influence students" performance expectancy. The P-value for the mean square of learning via audio is statistically significant with (p < 0.001); via video is significant with (p = 0.01) and via cartoon is significant with (p = 0.001). When this hypothesis is moderated by gender, the result clearly show no statistically significant differences between participants among the three ways of delivery with p = 0.62 for audio; p = 0.23 for video and p = 0.057 for the cartoon, as shown in Table 8 - 62.

Table 8 - 61 Interactivity elements on performance expectancy

Gender:

 $H5_a$. Gender will positively moderate the effect of interaction activities in m-learning on students' performance expectancy.

The result of analysing this hypothesis is illustrated in the following Table 8 - 62:

Dependent Variable: is the	(I)	(J)	Mean	Std.	Sig.	95% Confidence	
performance expectancy if	Gender	Gender	Difference	Error		Interval for	or Difference
learning via:			(I-J)			Lower	Upper
						Bound	Bound
audio	Male	Female	.074	.150	.623	224	.371
video	Male	Female	.185	.155	.238	124	.493
cartoon	Male	Female	.334	.173	.057	678	.010

Table 8 - 62 Gender moderator for hypothesis 5

The results from Table 8 - 62 show that there were no significant differences between gender in the three types of learning. Further to that moderator, the other moderator, *faculty*, provides no significant differences between participants in term of the two different faculties as shown in the Table 8 - 63.

Major/subject:

Major of study will positively moderate the effect of interaction activities in m-learning on students' performance expectancy.
The result of analysing this hypothesis is illustrated in the following Table 8 - 63:

Tuble 0 05 major moderan	or jor nype	incois 5					
Dependent Variable: is the	(I)	(J)	Mean	Std.	Sig.	95% Confidence	
performance expectancy if	Faculty	Faculty	Difference	Error		Interval for Difference	
learning via:			(I-J)			Lower	Upper
						Bound	Bound
audio	Science	Arts	004	.156	.980	313	.305
video	Science	Arts	039	.161	.811	359	.282
cartoon	Science	Arts	.263	.180	.147	094	.620

Table 8 - 63 Major moderator for hypothesis 5

H6. The Engagement and Behavioural Intention

The sixth hypothesis states that:

Intrinsic engagement will positively influence students' behavioural intention to use m-learning.

In order to test the above hypotheses, the UNIANOVA was used to check statistically the influence of students" engagement of any delivery ways and whether it might affect the behavioural intention to use m-learning or not. The results are shown as follows in Table 8 - 64:

Table 8 - 64 Engagement on behavioural intention to use m-learning									
Source: Engagement if	Type III Sum	Mean	F	Sig.					
learning via	of Squares	Square							
audio	.657	.657	1.799	.183					
video	14.652	14.652	40.093	.000					
cartoon	.834	.834	2.281	.134					

Table 8 - 64 Fraggement on behavioural intention to use m-learning

From Table 8 - 64, the data clearly shows a statistically significant positive influence on the behavioural intention to use m-learning only if the learning was undertaken using a video avatar, since it was the most engaging way with p < 0.001. This hypothesis moderated with the gender factor and the sub-hypothesis stated and presented its result as following:

Gender:

 $H6_a$. Gender will positively moderate the effect of intrinsic engagement on the behavioural Intention to use m-learning.

The result of analysing this hypothesis is illustrated in the following Table 8 - 65:

Table 8 - 65	Gender n	noderator f	or hypothe	esis 6

Dependent Varia	ible: (I)	(J)	Mean	Std.	Sig.	95% Confidence	
Engagement if	Gender	Gender	Difference	Error		Interval for	Difference
learning via			(I-J)			Lower	Upper
						Bound	Bound
audio	Male	Female	.088	.148	.556	382	.207
video	Male	Female	.134	.122	.277	109	.377
cartoon	Male	Female	.016	.160	.921	301	.333

The Table 8 – 65 provides the statistical evidence that there were no significant differences between participants in term of gender. The p-value for the audio is 0.55; for the video is 0.27, and for the cartoon avatar is 0.92. On the other hand, the faculty moderated the same hypothesis. The sub-hypothesis which moderated by the faculty stated and the results are presented as follows:

Major/subject:

 $H6_b$. Major of study will positively moderate the effect of intrinsic engagement on the students' behavioural intention to use m-learning.

The result of analysing this hypothesis is illustrated in the following Table 8 - 66:

Tuble 6 66 Major moderator jor hypothesis 6										
Dependent Variable is	(I)	(J)	Mean	Std.	Sig.	95% Confidence				
Performance expectancy	Faculty	Faculty	Difference	Error		Interval for Difference				
if learning via:			(I-J)			Lower	Upper			
						Bound	Bound			
audio	Science	Arts	.063	.147	.668	228	.355			
video	Science	Arts	.038	.123	.757	207	.283			
cartoon	Science	Arts	.039	.161	.810	280	.358			

Table 8 - 66 Major moderator for hypothesis 6

Table 8 – 66 demonstrates that there were no significant differences between participants in terms of the different majors for the intention to use m-learning, even for the three different ways of delivery audio with p = 0.66; video with p = 0.75; and cartoon with p = 0.81.

H7. Performance Expectancy and Behavioural Intention

Furthermore, another factor which may influence the students" behavioural intention to use was the performance expectancy. The statement of hypothesis seven is presented as follows:

The performance expectancy will positively influence students' behavioural intention to use m-learning.

In order to test the above hypothesis, UNIANOVA was used to check which ways of learning they expected would increase their usefulness, improve their performance and statistically influences students intentions to use m-learning in future Table 8 - 67.

Table 8 - 67 performance expectancy on intention to use m-learning									
Source:	Type III Sum	Mean	F	Sig.					
Performance expectancy	of Squares	Square		_					
if learning via									
audio	5.486	5.486	12.675	.001					
video	3.431	3.431	7.926	.006					
cartoon	1.245	1.245	2.877	.093					

Table 8 – 67 shows that, from a statistical point of view, the result of that hypothesis. Students show their intention to use m-learning specifically when the interface includes audio and video, because they perceived their performance if taught by these ways would be higher with P-value = 0.001 for audio; and with P-value = 0.006 for video. Students show their unwillingness statistically to use m-learning if delivered via a cartoon avatar with p-value = 0.09; hence, they did not expect to perform better while learning by that way.

The sub-hypothesis of the current hypothesis statement and testing analysis presented as follows:

Gender:

Gender will positively moderate the effect of performance expectancy on the behavioural Intention to use m-learning.

The result of analysing this hypothesis is illustrated in the following Table 8 - 68:

Dependent Variable is	(I)	(J)	Mean	Std.	Sig.	95% Confidence	
Performance expectancy	Gender	Gender	Difference	Error		Interval for	Difference
if learning via			(I-J)			Lower	Upper
						Bound	Bound
audio	Male	Female	.118	.141	.407	163	.398
video	Male	Female	.139	.144	.335	146	.424
cartoon	Male	Female	.061	.158	.698	252	.375

Table 8 - 68 Gender moderator for hypothesis 7

From Table8 - 68, the analysis indicates no significant differences between participants in term of gender. In order to know the differences between participants in term of majors, the same test was conducted for the sub-hypothesis, with the following results:

Major/subject:

H7_b. Major of study will positively moderate the effect of performance expectancy on the students' behavioural Intention to use m-learning.

The result of analysing this hypothesis is illustrated in the following table:

Tuble 6 - 69 Major moderator jor hypothesis 7									
Dependent Variable is	(I)	(J)	Mean	Std.	Sig.	95% Confidence			
Performance expectancy	Faculty	Faculty	Difference	Error		Interval for Difference			
if learning via			(I-J)			Lower	Upper		
						Bound	Bound		
audio	Science	Arts	.084	.140	.548	193	.361		
video	Science	Arts	.152	.143	.293	133	.437		
cartoon	Science	Arts	.080	.156	.610	230	.390		

Table 8 - 69 Major moderator for hypothesis 7

Table 8 - 69 shows the result clearly that there were no significant differences between participants in term of the two different majors.

H8. Preferences and Pedagogical Performance

In addition, based on the research model, other two factors illustrated to be tested to find whether or not is there a relationship between the preference for multimedia and the effectiveness (actual test performance). The statement of hypothesis eight is presented as follows:

Preference for multimedia instruction will positively influence students' effectiveness.

In order to test the above hypothesis, the UNIANOVA test was used to check which preferred way of multimedia instruction would positively increase students" effectiveness and improve their test outcomes/performance. The following Table 8 - 70 illustrates the result.

Source Preferring with	m-learning	Dependent	Type III Sum of Squares	Mean Square	F	Sig.
audio avatar		Post Audio	.159	.159	.055	.815
video avatar		Post-Video	.730	.730	.282	.597
cartoon avatar		Post-Cartoon	5.151	5.151	2.054	.155

 Table 8 - 70 Preference for multimedia on effectiveness (actual test performance)

The result of Table 8 - 70 indicates there was no statistically significant evidence to support that hypothesis with p-value = 0.81 for the audio; p-value = 0.59 for the video; and p-value = 0.15 for the cartoon avatar.

The current hypothesis was then moderated with the gender factor:

Gender:

 $H8_{a}$. Gender will positively moderate the effect of preferred multimedia instruction on the effectiveness.

The results were as follows:

Dependent (J) Mean Std. Sig.b 95% Confidence (I) Variable Gender Gender Difference Error Interval for Difference (I-J) Lower Upper Bound Bound -.828* .029 -1.567 -.088 Post-Audio Male Female 372 Post-Video Female -.883* -1.582 Male 352 .014 -.184 Post-Cartoon Male Female -1.056* 346 .003 -1.744 -.369 *. The mean difference is significant at the .05 level.

Table 8 - 71 Gender moderator for hypothesis 8

In terms of using gender as a moderator for the current hypothesis, the students" preference for the three ways showed statistical differences between participants based on their post-tests results. The result of analysing the variable of audio avatar preference on its actual test outcome was statistically significant, with p-value = 0.02. In addition, the analysis of the second variable, *learning via video avatar preference*, on its actual test outcome is showing significant with p-value = 0.01. Finally, the analysis of the third variable, learning via cartoon avatar preference on the actual test outcome is statistically significant, with p-value = 0.003.

Female students scored higher than male students with 0.8 points for the audio test, with 0.9 points for the video test, and for 1 point for the cartoon test:

Dependent Variable	Gender	Mean	Std.	95% Confidence Interval	
			Error	Lower Upper	
				Bound	Bound
Post-Audio	Female	2.050a	.222	1.609	2.492
	Male	1.223a	.298	.630	1.816
Post-Video	Female	2.290a	.210	1.872	2.708
Post-video	Male	1.407a	.282	.846	1.968
Post-Cartoon	Female	3.606a	.207	3.196	4.017
Post-Cartoon	Male	2.550a	.278	1.998	3.102

 Table 8 - 72 Mean of gender moderating hypothesis 8

In order to know the differences between participants in term of majors, the same test was conducted for the sub-hypothesis:

Major/subject:

H8_b. Major of study will positively moderate the effect of preferred multimedia instruction on the effectiveness.

The result of analysing this hypothesis is illustrated in the following Table 8 - 73:

Dependent	(I)	(J)	Mean	Std.	Sig.b	95% Confidence		
1					515.0			
Variable	Faculty	Faculty	Difference	Error		Interval for	Difference	
			(I-J)			Lower	Upper	
						Bound	Bound	
Post-Audio	Science	Arts	.452	.361	.213	265	1.169	
Post-Video	Science	Arts	.442	.341	.198	236	1.121	
Post-Cartoon	Science	Arts	.745*	.336	.029	.078	1.412	
*. The mean di	fference is	significant	at the .05 le	vel.				

Table 8 - 73 Major moderator for hypothesis 8

The Table 8 - 73 indicates there were significant differences between the two majors" participants in their final test result of the cartoon avatar, with p-value = 0.02. The Science students are higher with 0.7 points than Arts students in their final test of the cartoon avatar lesson:

Table 8 - 74 Mean of major moderating hypothesis 8									
Dependent	Faculty	Mean	Std.	95% Confidence					
Variable			Error	Inte	rval				
				Lower	Upper				
				Bound	Bound				
Post-Audio	Science	1.863	.246	1.373	2.352				
rost-Audio	Arts	1.410	.271	.871	1.949				
Post-Video	Science	2.070	.233	1.606	2.533				
Post-video	Arts	1.627	.257	1.117	2.137				
Post-Cartoon	Science	3.450	.229	2.995	3.906				
Post-Cartoon	Arts	2.706	.252	2.204	3.207				

 Table 8 - 74 Mean of major moderating hypothesis 8

H9. The Engagement and Pedagogical Performance

Furthermore, another factor which may influence the students" effectiveness (actual test performance) was the intrinsic engagement. The statement of hypothesis nine is presented as follows:

Intrinsic engagement will positively influence students' effectiveness.

In order to test the above hypothesis, the UNIANOVA test was used to check which way engagement positively influenced students" effectiveness and improved their test outcomes/performance and the results presented in Table 8 - 75:

Habie 0 75 Engageme	Table 6 75 Engagement on effectiveness (detaat lest performance)										
Source: Engagement	Dependent	Type III Sum	Mean	F	Sig.						
if learning via	Variable	of Squares	Square								
audio	Post-Audio	3.730	3.730	1.325	.253						
video	Post-Video	3.276	3.276	1.252	.266						
cartoon	Post-Cartoon	.761	.761	.298	.587						

 Table 8 - 75 Engagement on effectiveness (actual test performance)

It appears from the above table that there was no statistically significant evidence supporting that hypothesis with p-value = 0.25 for the audio; p-value = 0.26 for the video; and p-value = 0.58 for the cartoon avatar. In order to add the gender factor as a moderator for the current hypothesis, the statement and its analysis is presented as follows.

Gender:

 $H9_a$. Gender will positively moderate the effect of intrinsic engagement on the students' effectiveness.

The result of analysing this hypothesis is illustrated in the following Table 8 - 76:

Dependent	(I)	(J)	Mean	Std.	Sig.b	95% Confidence				
Variable	Gender	Gender	Difference	Error		Interval for Difference				
			(I-J)			Lower	Upper			
						Bound	Bound			
Post-Audio	Male	Female	916*	.374	.016	-1.659	173			
Post-Video	Male	Female	955*	.360	.010	-1.671	239			
Post-Cartoon	Male	Female	-1.032*	.356	.005	-1.739	325			
*. The mean di	*. The mean difference is significant at the .05 level.									

Table 8 - 76 Gender moderator for hypothesis 9

As illustrated in Table 8 - 76, there were statistically significant differences between participants in term of gender in the three ways of delivery. The p-value of the audio post-test is 0.01; for the video post-test is 0.01; and for the cartoon post-test is 0.005. The mean of the female students is higher with 0.9 points in the audio score test; with

1.0 point in the video score test; and 1.0 point as well in the cartoon score test than male students:

Tuble 8 - 77 Mean of genuer moderating hypothesis 9									
Dependent	Gender	Mean	Std.	95% Confidence					
Variable			Error	Interval					
				Lower Upper					
				Bound	Bound				
Post-Audio	Female	2.081a	.220	1.644	2.519				
rost-Audio	Male	1.165a	.298	.572	1.758				
Post-Video	Female	2.314a	.212	1.892	2.735				
Post-video	Male	1.358a	.288	.787	1.930				
Post-Cartoon	Female	3.603a	.210	3.186	4.019				
Post-Cartoon	Male	2.571a	.284	2.006	3.135				

 Table 8 - 77 Mean of gender moderating hypothesis 9

The second sub-hypothesis of the current hypothesis was to use *major/faculty* used as a moderator:

Major/subject:

H9_b. Major of study will positively moderate the effect of intrinsic engagement on the students' effectiveness.

The result of analysing this hypothesis is illustrated in the following Table 8 - 78:

Dependent	(I)	(J)	Mean	Std.	Sig.a	95% Confidence		
Variable	Faculty	Faculty	Difference	Error		Interval for	or Difference	
			(I-J)			Lower	Upper	
						Bound	Bound	
Post-Audio	Science	Arts	.444	.367	.230	287	1.174	
Post-Video	Science	Arts	.433	.354	.225	271	1.137	
Post-Cartoon	Science	Arts	.667	.350	.060	029	1.362	

Table 8 - 78 Major moderator for hypothesis 8

The above analysis indicates no statistically significant differences between participants in term of different majors/faculties among the three different ways of the delivery. The p-value of the students in the audio score test is 0.28; for the video score test is 0.22; and for the cartoon score test is 0.06.

1able 8 - 79 Mea	n of major	moaerati	ing nypothe	esis 9	
Dependent Variable	Faculty	Mean	Std.	95% Confidence Interval	
			Error	Lower Bound	Upper Bound
Post-Audio	Science	1.845a	.245	1.359	2.331
	Arts	1.401a	.274	.855	1.947
Post-Video	Science	2.053a	.236	1.584	2.521
rost-video	Arts	1.620a	.265	1.094	2.146
Post-Cartoon	Science	3.420a	.233	2.957	3.883
rost-Cartoon	Arts	2.753a	.261	2.234	3.273

Table 8 - 79 Mean of major moderating hypothesis 9

H10. Interactivity and Pedagogical Performance

Again, another factor which may influence the students" effectiveness (actual test performance) was the interactivity element. The statement of hypothesis nine is presented as follows:

The interaction activities in m-learning will positively influence students' effectiveness.

The tenth hypothesis was proposed to determine that the intervention of the interaction activities would positively influence and increase students" effectiveness/outcome performance. The UNIANOVA test was conducted and the results presented as follows in Table 8 - 80:

Source	Dependent	Type III Sum	Mean	F	Sig.
	Variable	of Squares	Square		
The interaction activities	Post-Audio	17.237	17.237	6.598	.012
(self-assessment and	Post-Video	12.936	12.936	5.351	.023
feedback elements)	Post-Cartoon	9.115	9.115	3.683	.058

 Table 8 - 80 Interactivity elements on effectiveness (actual test performance)

As illustrated in Table 8 – 80, there were statistically significant relationships and influence of the interactive activities (quick exercise, feedback and self-assessment) on the students" pedagogical performance and effectiveness if the learning was undertaken via m-learning with audio avatar with p = 0.01; and via m-learning with video avatar with p = 0.02. However, the result indicates no significant influence of that interactivity elements on students" performance when they learnt via cartoon avatars, with p = 0.058. The gender variable moderated the current hypothesis which stated:

Gender:

 $H10_a$. Gender will positively moderate the effect of interaction activities in m-learning on students' effectiveness.

The result of analysing this hypothesis is illustrated in the following Table 8 - 81:

Dependent	(I)	(J)	Mean	Std.	Sig.	95% Confidence			
Variable	Gender	Gender	Difference	Error		Interval for Difference			
			(I-J)			Lower Upper			
						Bound	Bound		
Post-Audio	Male	Female	769*	.353	.032	-1.471	066		
Post-Video	Male	Female	785*	.340	.023	-1.461	110		
Post-Cartoon	Male	Female	947*	.344	.007	-1.631	263		
*. The mean di	*. The mean difference is significant at the .05 level.								

Table 8 - 81 Gender moderator for hypothesis 10

There were statistically significant differences between participants among the three delivery ways in term of the two different genders. The p-value of the audio post-test score is 0.03; for the video post-test score is 0.02; and for the cartoon post-test score is 0.007.

The female students were higher than male students, with 0.8 points in the audio and video post-test score; and with 0.9 point in the cartoon test score as shown in Table 8 - 82:

Dependent	Gender	Mean	Std.	95% Confidence	
Variable			Error	Interval	
				Lower Upper	
				Bound	Bound
Post-Audio	Female	2.034a	.211	1.615	2.453
rost-Audio	Male	1.265a	.284	.700	1.830
Post-Video	Female	2.256a	.203	1.853	2.659
Post-video	Male	1.471a	.273	.927	2.015
Post-Cartoon	Female	3.583a	.205	3.175	3.991
Post-Cartoon	Male	2.636a	.277	2.086	3.186

Table 8 - 82 Mean of gender moderating hypothesis 10

The second sub-hypothesis of the current hypothesis is when the major/faculty factor used as a moderator. The hypothesis would be stated and presented its analysis as following.

Major/subject:

H10_b. Major of study will positively moderate the effect of intrinsic engagement on the students' effectiveness.

The result of analysing this hypothesis is illustrated in the following Table 8 - 83:

Dependent	(I)	(J)	Mean	Std.	Sig.	95% Confidence		
Variable	Faculty	Faculty	Difference	Error		Interval f	or Difference	
			(I-J)			Lower	Upper	
						Bound	Bound	
Post-Audio	Science	Arts	.093	.361	.797	625	.812	
Post-Video	Science	Arts	.047	.348	.893	644	.738	
Post-Cartoon	Science	Arts	.475	.352	.180	224	1.174	

 Table 8 - 83 Major moderator for hypothesis 10
 Image: 10

As the above table indicates no any statistically significant differences between participants in relation to the major/faculty moderator.

H11. The Pedagogical Performance and Behavioural Intention

The last hypothesis in the proposed model aimed to find out statistically any one of the delivery ways based on its test score has the influence on students" intention to use that way on the m-learning. The hypothesis statement and analysis is presented thus:

The effectiveness of m-learning will positively influence students' behavioural intention to use m-learning.

The result of analysing this hypothesis is illustrated in the following Table 8 - 84:

intention to use m-learning										
Source	Type III Sum	Mean	F	Sig.						
	of Squares	Square								
Post-Audio	2.792	2.792	3.920	.051						
Post-Video	2.999	2.999	4.230	.043						
Post-Cartoon	1.021	1.021	1.361	.247						

Table 8 - 84 The effectiveness (actual test performance) onintention to use m-learning

From Table 8 - 84, the result clearly indicates there is a statistically significant influence between the video test score and the intention to use that way via mobile devices in future.

With regard to the sub-hypotheses of the current hypothesis, which moderated the *gender* and *faculty/major*, the following statements and results are presented:

Gender:

 $H11_a$. Gender will positively moderate the effect of effectiveness on the students' behavioural intention to use m-learning.

Major/subject:

H11_b. Major of study will positively moderate the effect of effectiveness on the students' behavioural intention to use m-learning.

The result of analysing this hypotheses is illustrated in the following Table 8 - 85:

Table 8 - 85 Genaer moderator for hypothesis 11										
Dependent	(I)	(J)	Mean	Std.	Sig.	95% Confidence				
Variable	Gender	Gender	Difference	Error		Interval for Difference				
			(I-J)			Lower Upper				
						Bound	Bound			
Post-Audio	Male	Female	.079	.192	.681	302	.460			
Post-Video	Male	Female	035	.170	.838	372	.303			
Post-Cartoon	Male	Female	096	.198	.629	491	.298			
*. The mean diffe	*. The mean difference is significant at the .05 level.									

 Table 8 - 85 Gender moderator for hypothesis 11

 Table 8 - 86 Major moderator for hypothesis 11

Dependent	(I)	(J)	Mean	Std.	Sig.	95% Confidence	
Variable	Faculty	Faculty	Difference	Error		Interval for	Difference
			(I-J)			Lower Upper	
						Bound	Bound
Post-Audio	Science	Arts	.147	.183	.425	217	.511
Post-Video	Science	Arts	178	.177	.318	530	.174
Post-Cartoon	Science	Arts	.187	.186	.319	183	.557

From Table 8 - 85 and Table 8 - 86, it can be seen that the moderators of *gender* and *major* showed no significant differences between participants among the three delivery ways (audio, video, and cartoon) avatars lessons.

The following table 11, provides a summary of the research hypotheses results for the second case study.

H _n	Hypothesis Statement	Main H. Sig. P_value		Sig. moder. Gender	Sig. moder. major	
H ₁	Preference for multimedia instruction will positively influence students" intrinsic engagement.	V C	0.05	N/S	С	0.003
H ₂	Preference for multimedia instruction will positively influence students" performance expectancy.	C	<0.001	N/S	С	0.02
H ₃	Intrinsic engagement will positively influence students" performance expectancy.	A V C	0.02 <0.001 <0.001	N/S		N/S
H ₄	The interaction activities in m-learning will positively influence students" intrinsic engagement.	A V C	<0.001 <0.001 <0.001	N/S		N/S
H ₅	The interaction activities in m-learning will positively influence students" performance expectancy.	A V C	<0.001 0.01 0.001	N/S	N/S	
H ₆	Intrinsic engagement will positively influence students"behavioural Intention to use m- learning.	v	<0.001	N/S		N/S
H ₇	The performance expectancy will positively influence students" behavioural intention to use m-learning.	A V	0.001	N/S	N/S	
H ₈	Preference for multimedia instruction will positively influence students" effectiveness.	N/S		A 0.02 V 0.01 C 0.00 3	С	0.02
H9	Intrinsic engagement will positively influence students" effectiveness.	N/S		A 0.01 V 0.01 C 0.00 5	N/S	
H ₁₀	The interaction activities in m-learning will positively influence students" effectiveness.	A V	0.01	A 0.03 V 0.02 C 0.00 7	-	N/S
H ₁₁	The effectiveness of m-learning will positively influence students" behavioural intention to use m-learning.	v	0.04	N/S		N/S

Moder. = Moderator (gender, subject); V= video; A= audio; C= cartoon; I= image; T= text; F= face-to-face; N/S=Not Significant.

Up to this stage, the quantitative analysis and results of the second case study ended even though the discussion and interpretations of these data will be presented in details in chapter 9.

The results of qualitative data will be described in details in the coming section.

8.7 Qualitative Results

8.7.1 Responses from an Open-Ended Question of Experiment - 1

To learn more about the factors that influence the MADE-ME web application effectiveness and to enrich the quantitative data, the questionnaire concluded with openended questions to collect respondents" opinions. The first two questions asked the students about what they saw as the potential benefits to them from m-learning, and what the challenges and barriers they thought they face. In fact, 90 questionnaires out of the completed 156 were returned with comments on the benefits and advantages and 83 questionnaires were returned with comments regarding the challenges of m-learning. The participants" comments were various, so they were coded and analysed by the following themes:

Themes	Number of students responded		
Advantages and benefits of adopting m-learning	(n=90)	%	
Pause and repeatable features of the lesson	47	52	
Easy and fast to search and get the information	20	22	
Freedom to decide when and where each lesson will be learned (convenience)	33	37	
Fast to get, digest and understand the content	18	20	
Engage and motivate leaner to learn	6	7	
Provide concentrations and focus on the content	5	6	
Challenges and barriers of adopting m-learning	(n=83)	%	
Low internet connection	15	18	
Lack of communications with other peers and instructors to ask about things not understood	12	13	
Screen size may not display proper content	8	10	
Escape from learning to other entertainments or social media	13	15	
Lack of attention, interaction and motivation of the content	20	24	
Limited batteries capacity	4	5	
High internet subscription fees	3	4	

Table 8 - 87 Benefits and barriers of implementing m-learning

Benefits and Advantages

From Table 8 - 87, some students identified particular benefits and advantages of using m-learning. More than half of the respondents commented that the features of "pause" and "repeat the lesson" was one of the significant advantages of using mobile device for learning. 33 of the respondents felt there was a freedom to learn when it suited them. Others agreed that the use of m-learning is "Easy and fast to search and get the

information". About 18 students clearly stated: "The use of mobile device is fast to get, digest and understand the content". Lastly, some of students mentioned that learning through the mobile device may engage and enhance their motivation to learn and that this approach to learning would increase their concentration and ability to focus on the learning materials.

Challenges and Barriers

On the other hand, the respondents on the other question, regarding the challenges of using m-learning, provided invaluable insights for this research. About 20 of the respondents expressed that lack of attention, interaction and motivation with/for the learning materials was the highest barrier of using mobile phones in learning. Another fifteen students noted the low level of internet connections was an obstacle for them. Moreover, the lack of communications with other peers and instructors to enable clarification was commented by 12 students. Nevertheless, some respondents (13 students) highlighted the problem of being distracted from learning by other entertainments or by social media is a common issue. 12 respondents mentioned the issues regarding the poor display on the mobile screen, the limited battery capacity and lastly, high internet subscriptions limited the use of mobiles in learning.

Liking and Disliking

The participants were asked a series of open-ended questions to reflect their perceptions and analyse their views toward their usage of m-learning with avatars. The questions included asking them what did they like most about m-learning using avatars; what did they dislike; what did they feel that was engaging about m-learning using avatars; how could the avatars they have used be improved; and whether they had other comments they wished to make about their experiences of receiving course content using an avatar.

The open-ended comments in relation to what students liked about m-learning with avatars were varied. Some participants identified particular points such as "The use of m-learning with avatar provided me the concentrations and it's attracted my attention into the presented lesson". Many learners showed their attraction about the learning by writing it was: "Attractive". Others agreed that "It makes the learning easier and quicker to understand the content". One student clearly stated "The use of animated avatar is a new engaging way of learning and it's interesting to watch. I hoped that, it could apply

on the rest of my course materials". Thus, many of the participants advocated the method of using avatar through video and they described it as the best way to increase their motivation to learn.

On the other hand, some students stated that they disliked using the text interface only for learning. Another also wrote, "It's not engaging to learn from the text interface". The majority of the students commented on this type of question with the statement: "Nothing we can mention". A few of students complained about the long timeline of the lessons.

Feeling of Motivation

The above comments related to the learners" likes and dislikes toward using m-learning with avatars, but they also described their feeling about how their engaging about m-learning. Many of the participants praised their feeling of engaging with m-learning with avatars, with enormous number of positive characteristics, such as that approach provided "Understanding and comfort", or "Motivated and attracted"; "I got benefits from this app which I did not expected"; "I felt happiness"; "It's excellent and attracted the attention"; "I enjoyed a lot"; "It helped to enhance the learning empower the understanding"; "It is very nice and interesting"; "The method of learning through the current app helped me a lot to understand English grammar". Other statements as "The features of pausing and repeating the lesson at anytime and anywhere made the m-learning is engaging, comfortable, interesting, convenience, and effective" showed the effect of m-learning to engage students to be independent learners.

8.7.2 Responses from an Open-Ended Question of Experiment - 2

The second experiment questionnaires design was based on the earlier experiments" results to investigate further perceptions and to understand in more detail participants" views toward the use of m-learning with avatars through open-ended questions. Participants were asked direct questions such as: "The earlier findings showed more male students than female students preferred m-learning to traditional face-to-face lecturing. Why do you think this might be the case?". "Both male and female students particularly liked learning with a video interface. Why do you think this might be the case?". "Audio was the least preferred but was found to enable the students to perform best. Why do you think this might be the case?". "Female students preferred female

rather than male audio. Why do you think this might be the case?" and "Do you think it would be the same for a video presentation (for female students only)". "Science students liked using m-learning more than Arts students, however, Arts students intended to continue using m-learning after the experiment more than Science students. Why do you think this might be the case?". Lastly, "Is there anything else would like to add about m-learning interfaces or m-learning in comparison to face-to-face teaching in general".

• Open-ended Comments in Relation to the Above Questions

The earlier case study results showed:

Male versus Female:

More male students than female students preferred m-learning to traditional face-to-face lecturing. Why do you think this might be the case?

With regard to the first question which asked about reasons behind these findings which showed male students preferred m-learning to traditional face-to-face learning compared to female students, participants commented different views. The most commonly stated comment by the female students was: "Because we are liking to learn in real learning environment or similar to that via online learning, which we can see it's difficult for us to have, due to the culture aspects where the female instructor could not uncover her face while presenting the lesson in an open access source". Other female students supported that case by claiming that "Female students prefer to learn via the learning method that provides full connection to the lecturer in order to collaborate and ask questions with any difficulties during the lesson". One anticipated comment by female participant was that "Not all female students are preferring face-to-face lectures over m-learning, hence I'm one of those who prefers m-learning a lot". Other learners expressed some issues that they thought are the reasons behind that case; for instance: "Maybe some female students could not afford to own electronic devices (smartphones) or if they have them, maybe they find them difficult to connect to the internet".

Student preferences video:

Both male and female students particularly liked learning with a video interface. Why do you think this might be the case?

Furthermore, this question had been asked in order to investigate the reasons behind why almost all of the participants liked to learn through the video avatar compared to the other modes of delivery. The main comments were written as a response to questions, such as this response: "The Preference for the video avatar was that it includes mixture of multimedia like (real human character, sound, text and animated instructor) which makes the lesson attracting and interesting... and that in turn lasts in the mind for a while". This previous comments can combine two objectives of the current research together for the video avatar type of delivery which are the engagement and effectiveness. Another student justified their preference by saying that: "The video avatar is the closest type of learning to the traditional face-to-face lesson and so is similar to the real environment . . . the body language of the teacher is important for English language learning". The main objective of the animated avatar is to grab learners" attention, a point supported by participant comments such as "The video avatar is grabs our attention and helps us to focus which make it easy to retention . . . also, the nicest and most effective thing in that teaching method is the use of exercises and activities with the immediate feedback, which is the reason we feel comfortable during learning because we can learn and check our knowledge, but not in front of people". Hence, the majority of students enjoyed the video avatar and activities used in that mode of delivery through the MADE-ME web-app. The reality of interaction with the content affects the quality of m-learning.

Performance better by audio but not preferred

Audio was the least preferred but was found to enable the students to perform best. Why do you think this might be the case?

The responses on this question were very few and the vast majority of comments can be summarised by this response: maybe because the concentrations were on the learning materials only with the sound and not distracted by other things.

Female student preferred female avatar

Female students preferred female rather than male audio. Why do you think this might be the case?

Regarding this question, some students agreed that females find it easier to learn from the same gender instructor due to the culture norms and traditions in Saudi Arabia. Many participants wrote that, because female teachers are close to female students, they can understand our needs and the best way to engage and teach us. However, other female students" comments can be summarised by this student: "We do not mind and we have no preference of gender".

Science versus Arts

Science students liked using m-learning more than Arts students, however, Arts students intended to continue using m-learning after the experiment more than Science students. Why do you think this might be the case?

Regarding the findings that showed Science students preferred using m-learning more than Arts students, more Arts students intended to continue using m-learning after the experiment than Science students. One participant commented:

"Maybe that kind of learning through that application helps to memorise information that been presented which can meet the Arts students styles for their future learning, however, this application may not be suitable for Science students for their future modules which based on brainstorming and learning by practice in labs".

Additional comments

Is there anything else you would like to add about m-learning interfaces or m-learning in comparison to face-to-face teaching in general?

In this final open-ended question, participants expressed their views of their overall perceptions on the study. Students showed their acceptance and happiness toward their usage of m-learning with avatar. The following examples illustrate the students" responses: "This type of learning is interesting"; "This study is excellent and we hope to use the proposed application identically in our future studies"; "The study is fruitful and interesting... I really understand the whole content"; "The learning through m-learning is motivated and enhanced the learning"; "We got a lot of benefits from this study and hoped to continue use for our learning", and lastly, some students wished to use this mobile application for other courses to be distance or blended learning.

8.8 Chapter Summary

This chapter provided the results that were extracted (quantitatively and qualitatively) from the experiments (1 & 2) that were conducted with the learners" participants from Al-Baha University in Saudi Arabia. The chapter outlined students" opinions and perceptions toward the most preferred way of learning and the most effective in terms of pedagogical performance. It covered the analysis techniques used to analyse the data.

The results revealed that the students (both male and female) who owned smartphones devices, used them for many activities in both social and educational aspects, and they were highly motivated to use them further if study opportunities arise. It also investigated the research hypotheses based on the proposed model and revealed the relationships between the factors. Key findings of logistic regression and UNIANOVA analyses revealed that there was a strong support for the use of m-learning as part of a blended learning process. It also revealed that whilst students has have a preferred avatar type through which to receive static content, they did not always correspond with their most effective learning outcomes until interactive interface elements were added to the interface mode of delivery. The results indicated that, m-learning within animated avatar (video) was regarded as a better method of learning in terms of preference and effectiveness, and it's only the mode which had significant influence on students intention to use in future with p = 0.04.

Interestingly findings, in terms of the moderators, the preference of science students to learn via the cartoon avatar had statistically significant influence on performance expectancy and on actual pedagogical performance (effectiveness) with p = 0.02, and on intrinsic engagement with p = 0.003. On the other hand, in terms of the gender as a moderator, female students were higher than male students according to engagement, pedagogical performance, and intention to use m-learning.

Lastly, the chapter concluded by presenting the qualitative data and views that been grouped from the open-ended questions which presented the benefits and challenges of implementing m-learning in higher education in Saudi Arabia.

In summary, this research focuses specifically on the mode of learning content delivery and engagements via the interfaces, therefore, findings and results for using MADE-ME web applications in education were showing positive attitude towards adopting that approach.

A more detailed interpretation and discussion of these results will be given in the next Chapter 9.

9 Discussion of the Results

9.1 Introduction

This chapter aims to interpret the results of the first and second experiments described in Chapter 8. It provides a discussion of the quantitative findings and supports them with the qualitative perceptions of the participants. The overall objective of this chapter is to answer the research questions and determine whether or not providing different teaching delivery modalities via mobile devices can influence learning outcomes and optimise the learning process for students on the English language module at Al-Baha University in Saudi Arabia. Regarding the main research question of this study, it is important for the study to determine whether there is "*a connection between student engagement and/preference for particular avatar types and their pedagogic performance*", and whether an m-learning app with an avatar interface can be developed for use in higher education in Saudi Arabia with specific application to the teaching of English language?

The chapter discusses the findings from experiments 1 and 2 with a particular focus on the hypotheses developed as part of the MADE-ME whether or not the use of interactive and engaged avatar elements through the m-learning interface affects the students" intention to use that type of learning.

9.2 Discussion of the Research Hypotheses:

9.2.1 Discussion of Results Related to Experiment 1 (H1) & Experiment 2 (H2)

The first hypothesis in Experiment 1 and the second hypothesis in Experiment 2 stated that, "*The preference for one of the learning modalities will positively influence the performance expectancy/usefulness of students*". This hypothesis tested whether or not the adoption of avatars in mobile technology was helpful for learning the content compared to the same content presented in a traditional face-to-face format. The results were in favour of m-learning. The findings provide evidence that there is a statistically significant relationship between the performance expectancy/usefulness and preference of students to learn with a mobile interface, which included a "talking head" male avatar as a video. Adham et al. (2016) confirmed the need for an avatar tool which can be

effective and supportive to Saudi learners during their online learning. These results further support the idea of preferences for learning through online technologies rather than via traditional learning formats. Artino (2010, p275) found that:

"Students who preferred to take future courses in an online format (as opposed to face-to-face) also reported greater confidence in their ability to learn online (self-efficacy) and greater satisfaction with their recent online learning experience".

This finding is consistent with that of Chen & Wu (2015) who investigated the effects of different video lecture types on sustained attention, emotion, cognitive load, and learning performance and found that "the voice-over presentation type generates the highest sustained attention and enhanced students" learning performance". "Mazlan (Doctoral thesis, 2012), reported that participants expressed their readiness and desire to use avatars in their future learning materials".

Interestingly, there were also differences in responses related to the preferred mode of delivery and its potential impact on performance expectations, particularly when the materials of learning are for an English lesson. The findings indicate that there was a statistically significant relationship between the outcome and students" modality preference for learning through mobiles with static male avatars, and mobiles with animated video avatar interfaces. This is in line with the statement made by Peterson (2006, p79), who stated the use of "avatars enhanced the subjects" sense of telepresence and that the learners made use of their communicative features during the interaction".

In addition, when the current hypothesis was investigated in the second experiment, the result showed that there was a statistically significant relationship between students" performance expectancy and the preferred multimedia, if learning was through mobile devices with the cartoon avatars. The lack of a standardised outcome measure makes it difficult to interpret these results with confidence. The common denominator between the results of the first and second study is the presence of the avatar in the three approaches of learning (video, static male, cartoon) on the mobile interface, which may have enhanced the learners" motivations.

When this hypothesis was moderated by gender, the results showed that there were statistically significant differences between the two gender groups in favour of male students. Male students expected to perform better when they used mobile learning with static male avatars and animated video avatars by 0.3 points higher than females. However, in the second study, the findings indicate no statistically significant differences between genders in the three types of learning. This result may be explained by the fact that female students are starting to accept the approach of mobile learning combined with avatars when the interactive elements are integrated into the lessons.

A possible explanation for this might be that the positive responses toward having human avatar characteristics on mobile interfaces while learning online from the male point of view is based on the feeling of the instructor being "present". One of the participants claimed that "the person interacts with the animated avatar because of the feeling of being in a real class environment". A number of participants commented that, because of the cultural aspect and social norm, the female instructor could not present her face on the online learning interface, which of course is beneficial if English language learners can focus on body language, and that it was therefore easier for the male instructor.

When the current hypothesis was moderated in Experiment - 1 with the major/subject of study, the results show that there were no significant differences between the two groups. These results are different from the second study results, where the findings show that Science students gained higher rates than Arts students when the learning through mobile devices with cartoon avatar interfaces. These results support the findings of Hong et al., (2003) who found statistical differences between students in term of major/subject which was in favour of science students compared to those in the humanities. The reason for this may because the cartoon characters increased the imaginary world of Science students and consequently their performance expectations.

Key Findings:

The preference for a cartoon mode of delivery was the only one to have a significant relationship with the student perceptions with regard to their increasing performance with p = 0.02. Science student showed significantly higher rates than the Arts students in terms of that mode.

9.2.2 Discussion of Results Related to Experiment 1 (H2) & Experiment 2 (H1)

The second hypothesis in Experiment 1 and the first hypothesis in Experiment 2 stated that "*preference for multimedia instruction will positively influence students' intrinsic engagement*". Testing of this hypothesis indicates that there is a statistically significant relationship between engagement and students" preference to learn via m-learning with static male avatars, m-learning with video avatars, and m-learning with cartoon avatars. Interestingly, the correlation between the two factors of this hypothesis is related to the presence of the three avatars on the mobile interface. This supports the findings of Mazlan (2012), where the use of an avatar in comparison with a text avatar and non-avatar in the online learning environment was the factor that engaged students to learn. It is anticipated that this approach of adopting avatars on the m-learning interface will engage learners to be motivated and more active in the learning process. This finding is also consistent with findings by González et al. (2013) who stated the role of the avatar in learning as

"The digital representation of the individuals within the virtual world. It has an ability to perform actions and to simulate humanto-human interactions to increase engagement and hence learning" as reported in (Adham et al. 2016, p87).

The result of this hypothesis can be supported by the findings of the second experiment which showed that there was a statistically significant relationship between the engagement factor of learning if the mode of delivery was a video with cartoon media instructions. These results were similar to the findings of the first experiment, in that when the lesson was delivered as a video with cartoon avatar interfaces via the mobile device, both approaches significantly increased students" engagement. Again, interestingly, the correlation between the two factors of this hypothesis is related to the presence of avatars on the mobile interface giving the sense of a human presence.

When this hypothesis was moderated by gender, the result indicates that there were statistically significant differences between genders. Male students engaged more than female students when the content was delivered via mobiles with an avatar representation of the teacher. This result reflects Wang & Yeh, (2013, p408), who

reported that "students preferentially chose a pedagogical agent of the same gender". Some participants expressed the belief that male students are always seeking to find the easiest way to learn the information but female students like to learn through the traditional face-to-face teaching in order to discuss and clarify any ambiguous points. One such female student presented her motivations and continued her comments by adding:

"Now, I think after this current study ..., many of the female participants desire this approach of m-learning for their future studies and they will change their minds to use m-learning besides traditional learning".

As a consequence, when the gender is moderated in the current hypothesis in the second experiment, the results showed no significant differences between genders in the three types of learning, which contradicts the first experiment's findings.

Furthermore, when this hypothesis was moderated by the major/subject in the first experiment, the results revealed no significant differences between the Arts and Science groups. The researcher attributed this result to the enthusiasm of both major/subject groups to find out a new way of learning which could help them to improve their learning in this core module. However, there were importance differences between the two groups in terms of different majors/subject in the second experiment. Science students were higher than Arts students in terms of motivation and engagement when learning through m-learning with cartoon avatar interfaces. The reason for this may be because the cartoon characters increased the imaginary world of Science students and consequently their motivation. These results support the findings of Turan (2014), who recommended that the need for visual and audio cartoon characters for education is necessary for increasing student achievement in math lessons. In addition, in terms of the effectiveness of engagement elements on science students, Alabdulaziz & Higgins, also found that the motivation of students for learning mathematics increased with the use of technology (Alabdulaziz & Higgins, 2017).

There were extra modalities of learning which were tested separately because they were particularly for female students only. That last mode of delivery was hence through mlearning with static female avatars. The results indicate a statistically significant relationship between female student engagement and m-learning with static female avatars. This result tallies with Kim and Baylor (2007) who stated "The majority of female students chose female avatars as their learning partners and were likely to choose a female ... as "most like themselves".

Key Findings:

The preference for mode of delivery and students" perceptions has the largest impact on engagement. In terms of the moderators, the major/subject of Science students shows significantly higher values than the Arts students with p = 0.003.

9.2.3 Discussion of Results Related to the Experiment 1 (H3) hypothesis

This hypothesis stated that "intrinsic engagement will positively influence students' *enjoyment*". The results from the third hypothesis show that there is a statistically significant relationship between perceived enjoyment and students" beliefs that having access to English materials on their mobile devices have enhanced their motivation to learn English language. In fact, the results showed that learners who were more intrinsically motivated to learn via m-learning experienced higher levels of enjoyment in learning than those who were less motivated. These results are in agreement with researchers who suggested that engagement with the learning method, would increase the level of enjoyment (Baek & Touati, 2017; Hodhod, 2010), and this in turn would lead to acceptance and intention to reuse the same learning approach. Leong et al. (2013) confirmed the association between perceived enjoyment, which was an important factor, and enhancement of users" intentions towards adopting mobile entertainment. Furthermore, İşman et al. (2015) claimed that, when using the perceived enjoyment as an external variable integrated into TAM, the findings of the study revealed a positive influence on the perceived usefulness, which ultimately increased students" attitudes toward m-learning. Concerning the possible link between enjoyment and intrinsic engagement, the findings suggest a direct relationship between these two variables.

These factors may explain the relatively strong correlation between the intrinsic engagement of the learning approach and that directly reflection on the enjoyment of using that way of learning. This hypothesis was one of the main objectives of the current study and it is supported statistically.

The current hypothesis was tested again to find out if there were differences between groups of gender, as well as between groups of majors/subject. The analysis results indicate no significant differences between groups for either gender and/or major/subject. The findings are in line with Leong et al. (2013), who concluded there were no significant gender differences in the adoption of mobile entertainment. Leong et al. (2013) suggested justification of this was the equal opportunities which is given to both genders from the government's policy to gain expertise and knowledge.

These results, though, differ from Wang & Yeh (2013, p408) who claimed that "students who learned with a male pedagogical agent showed more interest in the learning materials than students with a female pedagogical agent".

Key Findings:

The level of enjoyment experienced in a mobile learning is influenced by the learner's level of intrinsic engagement.

9.2.4 Discussion of Results Related to Experiment 1 (H4) & Experiment 2 (H3)

The fourth hypothesis in Experiment 1 and third hypothesis in Experiment 2 was stated as "the intrinsic engagement will positively influence students' performance expectancy". This hypothesis has been tested for two purposes. Firstly, it used the mean of responses for a number of statements which regard to learning being for a general course across all modules and whether or not the perceived engagement of learning content influenced the performance expectancy/usefulness. The second usage of the hypothesis was to test the result for when the course material was for teaching English language. The results of both tests revealed that there was a statistically significant relationship between the performance expectancy if learning was via a mobile device with an avatar for any course and the intrinsic engagement, as well as if the course was for teaching language English. Accordingly, the motivation or the engagement of elearning was construed to affect the performance expectancy, which has a direct influence on intention to use the technology (Maldonado et al. 2010). Lee et al. (2005) reported the performance expectancy of online learning is an outcome of the intrinsic motivation. The study concluded that performance expectancy is the key driver of usage behaviour and intentions. To support the above findings, the same hypothesis was tested again in the second experiment and the results revealed the same results. Findings illustrated there were statistically significant relationships between the engagement and performance expectancy regardless of the learning delivery method used (audio, video, cartoon).

With regard to *gender* as a moderator, the result of the first test of the hypothesis showed that there were significant differences between genders; however, the second test revealed no significant differences between males and females when the students were engaged in learning English language. The male students engaged more than the female students, which reflected positively on their performance expectation. It is difficult to explain this result, but it might be related to the preference of female students to learn via traditional face-to-face methods, as opposed to the male students preferred to learn via mobile phones. The findings when the learning materials are for English language teaching were in line with Maldonado et al. (2010) who used the gender factor as a moderator in their model and found no significant differences between males and females. Both genders can be equally motivated toward using e-learning. The results from the same hypothesis in the second experiment revealed the same findings, which showed no significant differences between groups of genders.

Further investigation was conducted into the current hypothesis with the major/subject as the moderator. The findings suggest no significant differences between major/subject groups. Arts and Science can be equally engaged and motivated toward the use of mlearning which was consistent with the findings of Leong et al. (2013). The possible explanation for this result may be that both Arts and Science have the same level of English teaching in higher education, and both are seeking to enrich their English information as a core module in higher education, the new learning approach encouraged their motivation to interact with its environment. It is also worth noting that the major/subject moderation in the second experiment did not show significant differences between groups.

Key Findings:

Engaged students perceived that they were learning and improving.

9.2.5 Discussion of Results Related to Experiment 1 (H5)

The hypothesis stated "the convenience of m-learning will positively influence students" performance expectancy". The results from the fifth hypothesis show that there was a statistically significant positive relationship between the performance expectancy factor and the convenience factor. Perrin et al. (2006) explained the differences between traditional face-to-face teaching and online teaching, then concluded with findings in favour of online learning in terms of convenience and time. They mentioned that students can access the information for learning at their own convenient time and that is a key advantage to m-learning. Furthermore, the authors claimed that "Learners are not restricted to a specific physical environment, a particular delivery channel, or a fixed set of times for undertaking training and education" (p32). Their results corroborated the ideas of Singh & Reed (2001), who suggested that the convenience of technologies in blended learning improved the effectiveness and learning experiences. According to Al-Fahad (2009), mobile learning in pedagogic aspects is seen as the learning process that is considered outside of the conventional classroom setting and with the help of learning devices such as smartphones and tablets etc., specifically in educational activities, individuals tend to keep their learning activities in continuous form. Considering the mobile learning system, the learner obtains educational and informational opportunities quickly, since it minimises the physical presence or distance (Huang et al. 2010).

However, the findings of the current study do not support research by Artino (2010), who found the learners perceived the course contents were useful and interesting when the learning was in a face-to-face format. In addition, Moor & Kearsley (2005, p275) likewise has stated: "While students often appreciate the convenience of online learning, if given the choice, many would rather complete courses in a traditional, classroom-based format". Artino (2010), assumed that the findings are not entirely clear and need further investigations. However, Alfarani (2016) considered the convenience factor to be the key predictor of m-learning adoption in Saudi Arabia universities.

With regard to the interventions of gender and major/subject, the results revealed there was statistical evidence showing the difference between the two gender groups. Male students perceived that the use of mobile phones in education was more convenient than female students and that would impact positively on their usefulness. The possible explanation for this may be related to Saudi female students not being allowed to carry mobile devices onto the campus. Males, by contrast, who responded to the use of mobile technologies as not being important were minimal at only 5.7%, since there were able to carry their devices even in classes at the universities. The researcher investigated this point more by conducting direct open-ended question for the female students. The majority of responses from the participants stated something similar to this student who said, that:

"Girls' behaviours in learning is to ask the instructor questions directly and collaborate with others when there is an ambiguity... Sometimes you need an immediate answer for your question to construct the upcoming knowledge and that would be difficult when learning via m-learning".

On the other hand, when the moderator was the major/subject, the results revealed no significant differences between the Arts and Science. Again, the research investigated modules in a core course for all disciplines" students, therefore, with the connectivity to the internet and the small weight of mobile devices, all students felt that m-learning would allow them to be used at their convenience, regardless the place and time. An instructor of those participants proposed a reason for this factor:

"M-learning seems to be convenient to our world days, as you know, most of the students spend a lot of time in their phones". Furthermore, the instructor stated "It's a great idea. Simply, because most of them spend their whole time on their cell phones".

9.2.6 Discussion of Results Related to Experiment 1 (H6) & Experiment 2 (H7)

The sixth hypothesis in Experiment 1 and the seventh hypothesis in Experiment 2 was *"The performance expectancy will positively influence students' behavioural intention to use m-learning"*. The results from this hypothesis show that there was a statistically

significant relationship between the intention to use m-learning and students" performance expectations if they were learning via mobile devices with them believing, it would effectively strengthen their participation in learning of the English language, compared to traditional face-to-face learning. That is, those with high performance expectancy toward using m-learning had a higher intention to use m-learning than those with lower performance expectancy. This result is in line with Alfarani's (2016, p180) findings which were from teachers rather than students" perceptions. The study concluded that:

"The faculty members responded with high performance expectancies (i.e. those who believe that using m-learning in their teaching will be beneficial to them) have a tendency to accept m-learning more than faculty members with lower performance expectancies and willing to adopt m-learning in the present and in the future".

In the main, the current results supports many other researches who found a positive relationship between performance expectancy and the user's behavioural intention to use m-learning from the students" point of views (Al-Gahtani et al. 2007; Iqbal & Bhatti, 2015; Kim et al. 2013; Nassuora, 2013; Wang et al. 2009; Osakwe et al., 2017). However, Jairak et al. (2009) did not find a significant influence between these factors. In researching the factors that determine learners" acceptance of mobile technology, Jairak et al. found that the perceived performance expectancy was one of the important factors which impact on the intention to use m-learning.

It is worth noting that in the second experiment, the hypothesis test aimed to find out the specific way(s) of mobile interface delivery that the students intended to use caused by their perceptions to perform better. The findings show that there were statistically significant relationships between the intention to use m-learning and the performance expectancy when the interface included audio and video modes of delivery. These results were consistent with Wang et al., (2009) who found the performance expectancy was shown to be the strongest predictor of behavioural intention to use m-learning.

When the research model hypothesised the current hypothesis with gender and major/subject as moderators, the result of these analyses revealed that no significant differences between groups in term of gender or major/subject. The results are

consistent with Al-Gahtani et al.'s (2007) findings that "Performance expectancy had a positive effect on intention, but found no interacting effect with performance expectancy and either gender or age on intention to use the technology". One of the female participants of the current study believed that she derived benefits from the MADE-ME app, then commented in the open-ended questions that:

"Now, I think after this current study ... many of the female participants desire this approach of m-learning for their future studies and they will change their minds to use m-learning besides traditional learning".

Moreover, from a number of studies, for example, Alfarani (2016); Leong et al. (2013); Wang et al., 2009; Osakwe et al., 2017) it has been found that there are no statistical significant differences between the above factors when moderated by gender. The possible justification for this might be that male and female students in Saudi Arabia have become more experienced and skilled in using advanced m-learning methods; therefore, they become more equal in terms of their perception toward the new learning approach, irrespective of gender. So, when the current hypothesis was moderated with the external factors of *gender* and *major/subject* in the second hypothesis the results were consistent with the earlier results which indicate there were no significant differences between groups in terms of gender or major/subject among the three ways of learning (video, audio, cartoon).

These results are in contrast to earlier findings from Ong and Lai (2006); Shashaani and Khalili (2001); Koohang (1989); Venkatesh & Morris (2000), and (Leong et al., 2013), who found male students' expectations regarding the benefits and usefulness factors were greater than female students, for example, Venkatesh & Morris, (2000) found that "Males were greatly affected by attitude towards the adoption of a new technology".

In term of the major/subject, the results are in agreement with Algahtani's (2011) findings which showed no significant differences between sciences and arts students which he proposed may be due to popularity of the new technologies and high rate of usage among all majors/subject of studies.

9.2.7 Discussion of Results Related to Experiment 1 (H7) & Experiment 2 (H6)

The seventh hypothesis in Experiment 1 and the sixth hypothesis in Experiment 2 stated "intrinsic engagement will positively influence students' behavioural intention to use m*learning*". The results from this hypothesis show that there is a statistically significant relationship between the intention to use m-learning and the students" belief that having access to the English materials on their mobile devices would enhance their motivation to learn English language. The study confirms that the students" intrinsic engagement is associated with their behavioural intention to use m-learning. Consistent with the literature, this research found that participants who reported using mobile phones for learning English language engaged with them and this reflected on their intention to use the approach in their future learning (Maldonado et al., 2010). Maldonado et al. (2010, p76) concluded their study by stating that "student e-learning motivation plays a significant role in e-educational portal use and adoption in developing countries". These findings are in line with those of previous studies (Lee et al., 2005; Leong, 2013) who found that motivation factor impacts on students" attitude and intention toward using the technology. Furthermore, Kim et al. (2013) found a strong positive relationship between the users" engagement and the intention to continue using their smartphones.

There is notable evidence based on analysis of the second experiment for the same hypothesis describing the impact of engagement factor on the intention to use the technology factor. More specifically, the researcher was aiming to determine which one of these approaches has a statistically significant relationship. The previous experiment showed a significant relationship between the intention to use m-learning and the students" engagement to learn via mobile devices in general. The results from the second experiment for the same hypothesis showed that there was only one of the three ways of learning (video, audio, cartoon) that the students" engaged with and intended to use in future learning. The results showed a statistically significant positive relationship between the students on their mobile device. It is clear from the results that, after adopting the interactive elements and receiving immediate feedbacks provided by the app, that learners appreciated the video learning approach and were not so keen to use the other approaches. It may be that the reason behind this difference is because that mode of

learning delivery requires body movement. One of the research objectives was to find the most engaging and effective method of learning; therefore, by obtaining the results, the researcher anticipated these findings prior to conducting the study, after which were supported statistically. The results are in line with Baek & Touati (2017) who showed the intention to use mobile games is significantly predicted by intrinsic motivation. According to Boutsika (2014, p125) "The pedagogical strategies should encourage student participation in interaction". Osunkoya & Chern (2013) confirmed that realistic movements by the avatar are now becoming the natural interaction for the HCI. This mode of delivery, with its interaction elements, is creating a strong bridge between humans and graphical interfaces.

When moderated with gender and major/subject, the results indicate no statistically significant differences between the two groups in terms of *gender* or *major*. These results were supported by the results of the second experiment which presented the exact same results. These findings are consistent with Maldonado et al. (2010, p78), who reported that:

"We did not find gender as a moderator in our model. These findings suggest that, in Peru, male and female students can be equally motivated toward use of e-learning portals and similar polices can be used to motivate both genders toward e-learning".

These results further demonstrate the importance of m-learning motivation in higher educational use among learners and confirm their intention to use that approach of learning.

Key Findings:

Students who have more positive attitudes towards using mobile learning are more intrinsically motivated to learn. Students who have positive intentions to use m-learning with animated avatar (video) were more motivated.

9.2.8 Discussion of Results Related to the Experiment 1 (H8) hypothesis

This hypothesis stated "enjoyment will positively influence students' behavioural intention to use m-learning". Perceived enjoyment is quite related to perceived intrinsic

motivations. Therefore, some researchers included the enjoyment factor in the research model as an important structure, for example Baek & Touati (2017). The results from this hypothesis show that there was a statistically significant relationship between the intention to use m-learning and the students" enjoyment, in which they believed accessing into the materials on mobile devices was a fun interaction between content and learner. Liu (2008) and Baek & Touati (2017), adopted a framework for mobile learning for future research and integrated the perceived enjoyment as a critical structure. This finding broadly supports the work of other studies in this area linking the enjoyment variables with the students" intention to use the e-learning. Alenezi et al. (2010) and Leong et al.(2013) found that "perceived enjoyment has a significant relationship on the students' intention of online learning and directly influences their intention". In consequence, this research assumed the enjoyment factor is a very important factor in creating positive intentions to use e-learning in the higher education environment. Furthermore, a study conducted to investigate the students" intention of use for web-based learning systems concluded that the results demonstrate that the enjoyment of students toward using that system has a significant impact on students" intention to use the system (Saadé et al. 2008). The results are also consistent with previous findings which revealed a direct link between enjoyment and the intention towards using the technology (Lee, 2009; Wu & Liu, 2007). To sum up, there is a positive relationship between learners" enjoyment of the learning approach with the adoption of/the intention to use that way of learning. In fact, results showed that learners who were keener to use m-learning in future learning experienced higher levels of enjoyment in their learning than those students who were less enthusiastic to use mlearning. Concerning the possible link between enjoyment and intention to use that technology, the current findings suggest a direct relationship between these two variables.

When the two moderators of *gender* and *major/subject* are used with the current hypothesis, the results revealed that there were no significant differences between the groups. This finding is consistent with that of Leong et al. (2013) who conducted a study on behavioural intention to use mobile entertainment, who found there were no significant moderating effects of gender when they asked "Does gender really matter?".
The level of enjoyment can predict the behavioural intention to use m-learning.

9.2.9 Discussion of Results Related to Experiment 1 (H9) & Experiment 2 (H9)

The ninth hypothesis in both Experiment 1 and Experiment 2 stated "intrinsic engagement will positively influence students' effectiveness". The result for this hypothesis show that there was a significant relationship between performance outcomes and the students" engagement to learn through m-learning when the interface was a static male, cartoon, and video avatar. This result may be explained by the fact that the pictorial representation of the human character on the mobile interfaces draw learners" attention, which help them to memorise what they have learnt and which leads to more effective learning. The results support Rebolledo-Mendez et al. (2008), who conducted a study of using avatars in Computer-Aided Instruction and found that intrinsic motivation/engagement helped students to achieve greater degrees of learning. Another study on avatars and how they may be used effectively in e-learning also confirmed these findings, stating "...avatars seem to have a beneficial effect on learner motivation and concentration during learning" (Wang et al. 2005). Students were motivated and had increased learning potential when the avatar was used in the classroom (Mazlan, 2012). Other studies agreed that, if the learner was fully motivated and engaged with the learning environment, the outcome was likely to be better performance and achievement in learning (Baek et al. 2015; Logan et al. 2011).

In contrast to earlier findings, however, no evidence of the relationships between the engagement factor and the real performance effectiveness was detected in the second experiment. The result of analysing this hypothesis showed no statistically significant relationship between these two factors. These results concur with Baek & Touati, (2017), who found no direct link between motivation and game achievement. However, this finding is in contrast with previous studies (Unrau & Schlackman, 2006; Jurisevic et al. 2008; Baek et al. 2015), which found a direct link between motivation and game achievement. The findings of the current research suggest that a learner's motivation level did not determine how well they achieved in the learning. One possible explanation is that there may be other considerations, such as the lack of

competitiveness that impacted learners" motivation to perform in the exam. The researcher suggests future investigations on the competitiveness elements in education and whether it influences learners" motivations and pedagogical performance.

When the current hypothesis was moderated by *gender* the results show that female students significantly engaged with and then performed higher than male students when the learning was as a text, cartoon, audio, and video avatar on the mobile interfaces. Ghaith (2013) also showed a significant impact for female students when they used the blended learning methods, which led them to better academic achievement compared to traditional face-to-face learning alone. To confirm those findings, the results from the second hypothesis revealed the same evidence of showing the differences between groups in term of gender. The mean of the female students was higher than the mean of male students in the three modes of deliveries (audio; video; cartoon) in terms of performance in exam results.

On the other hand, when this hypothesis was moderated with students" major/subject of study, the findings showed that Science students engaged and outperformed their Arts student counterparts in m-learning with a cartoon interface (avatar). These results are consistent with those found by Hong et al., (2003). The possible suggestion of these findings may due to the familiarities and experiences of science students with the use of technologies such as in labs more than arts students. However, the findings of the second study highlighted no significant differences between groups in term of *major/subject*.

9.2.10 Discussion results related to Experiment 1 (H10) & Experiment 2 (H8)

The tenth hypothesis in Experiment 1 and the eighth hypothesis in Experiment 2 stated "*preference for multimedia instruction will positively influence students' effectiveness*". The result of this hypothesis showed that there is a statistically significant relationship between the students "preference for using m-learning video interface only and the result of the face-to-face test. These factors may explain the relatively strong correlation between the face-to-face learning and the mobile learning as a video lesson. Wu (2015, p119) reported that:

"the video of a lecturer may give learners a sense of interacting with an actual person while watching a video lecture. In other words, the video of a lecturer may foster a sense of social presence".

Some researchers Church (2004); Valenzeno (2003) suggest that the lecturer video might improve learners learning in other ways. Another study by Baylor & Ryu (2003) can confirm the findings of this hypothesis which investigated the effectiveness of using avatars in online learning for building communities and presence in educational environments. Therefore, the results suggested that the avatar provided "a sense of presence that is the catalyst for community and learning". The animated avatar (video) provides learners with the feeling of an instructor-like figure and that students perceived a strong positive effect (Baylor & Ryu, 2003). Other studies have suggested a strong preference (Allert, 2004; Thomas et al., 2002). Multimedia instruction is an important predictor enhancing e-learning effectiveness (Liaw, 2008) and therefore, the quality of multimedia enhances students" positive attitudes toward e-learning.

However, this outcome is contrary to the second study findings which demonstrates no evidence for learning style preference to predict learner achievement. The results of analysing this hypothesis showed no statistically significant evidence to support the hypothesis. Regardless of the learners" mode of delivery preference, the mean scores were very similar in the three ways of learning (video, audio, cartoon). This outcomes are consistent with Al-azaweiet al. (2016) who found and suggested that preference for learning styles had no correlation with students" academic performance. These results suggest that even though learners prefer a particular mode of delivery, they do not necessarily improve their knowledge and remember the information at the exam. These findings are in line with prior literature see, for example, (Gomes & Mendes, 2010; Prajapati et al. 2011). The level of pedagogical performance or achievement is not significantly predicted by learners" mode of delivery preferences (learning styles) (Baek & Touati, 2017). To sum up, whilst students have a preferred avatar type through which to receive static content, this preference did not always correspond with their most effective learning outcomes until interactive interface elements were added to the interface mode of delivery.

When the current hypothesis was moderated with *gender*, the result shows that female students preferred and outperformed male students in face-to-face learning methods compared to the m-learning. Artino (2010) reported that, although many students were satisfied with the convenience of online learning, many of them would rather continue their courses in traditional face-to-face in classrooms. However, the findings of the second experiment showed that there were also significant differences between groups in terms of gender. Female students scored higher than male students in the three modes of delivery (audio; video; cartoon) in term of performances and exam results.

When moderated with *major/subject* the results show that Science students preferred and outperformed Arts students in m-learning with cartoon avatars compared to the other ways of learning. The second experiment results support these findings and revealed identical results. Interestingly, the second hypothesis in the second experiment was to investigate whether or not the preference for multimedia has an influence on performance expectancy which was moderated by the major/subject, with the result revealing that the cartoon mode of delivery was only the one found that has a significant relationship according to the major/subject. Science students showed significantly higher scores than the Arts students with p = 0.02. Here, in the current hypothesis which aimed to investigate whether or not the preference for multimedia has an influence on the actual pedagogical performance according to the major, the results indicated that the Science students scored higher than Arts students in their final test of the cartoon avatar lesson with p = 0.02. In contrast, the study by Al-Fahad (2009) indicated that there were no significant differences between learners in terms of their major of study for Arts and Medicine students when they were offered mobile learning to improve their retention.

Key Findings:

Students" preferences for a learning mode of delivery is not linked to pedagogical performance (effectiveness). In terms of the moderators, the preference for cartoon avatar mode of delivery had statistically significant differences between majors according to the influence on performance expectancy. Science students showed significantly higher scores than the Arts students with p = 0.02. Interestingly, when testing the same hypothesis on the actual performance, the results are identical with p = 0.02.

9.2.11 Discussion of Results Related to Experiment 1 (H11) & Experiment 2 (H11)

The eleventh hypothesis in both Experiment 1 and Experiment 2 stated "*the effectiveness of m-learning will positively influence students' behavioural intention to use m-learning*". The result of this hypothesis shows that there is a statistically significant relationship between the intentions to use m-learning with a cartoon avatar. The possible explanation for this might be that learners constructed their knowledge in this mode of learning from the previous modes which, in turn, helped them to enrich their information and after which they performed better in the exam. This study confirms that the effectiveness of m-learning is associated with the intention to use that approach of learning. These results reflect those of Liaw (2008, p873) who also found that "there was a significantly high correlation (r = 0.70) between learners" behavioural intention to participate in e-learning and e-learning effectiveness". The use of mobile technologies in combination with multimedia and wireless infrastructures in educational environments increases the effectiveness of these technologies as learning tools which in turn improves mobile learning adoptions in future (Alfarani, 2016).

However, the same hypothesis used again in the second experiment and the result indicates that there is a statistically significant influence between the video test score and the intention to use that way via mobile devices in future, a result anticipated by the researcher and then ultimately found. The researcher's assumptions were based on the co-creation workshop findings and the participants' perceptions. The video mode of delivery was found to be the most engaging approach but was not the most effective; therefore, in the second study, integration of these interactive elements included self-assessments and immediate feedback, with the conclusions showing that their approach can be the most engaging and effective way of learning via mobile devices. This results support the findings of Zhang et al.(2006, p15) who found that "students in the elearning environment who are provided with interactive video achieved significantly better learning performance and a higher level of learner satisfaction than those in other settings". Interestingly, there were no significant relationships between the students'' results and intention to use m-learning with cartoon avatars and the test result of that way of learning.

When moderated with *gender*, the results showed that male students intended to use mlearning with cartoon avatars more than female students in the first experiment. However, the second experiment results indicated that no were significant differences between groups in term of gender among the three ways of learning. When moderated with *major/subject*, the result show that no significant difference between groups which is the same results of the second experiment.

Key Findings:

Students intend to use m-learning with a video avatar because they found it to be the best performed way of learning.

9.2.12 Discussion of Results Related to the (H4) Experiment 2:

Hypothesis four in Experiment 2 stated that "the interaction activities in m-learning will positively influence students' intrinsic engagement". The findings showed that there were statistically significant relationships between the engagement of students with the interactivity elements, which was assessed by the self-assessment and feedback elements while learning was via the three ways of delivery: audio, video and cartoon. It has been proven that feedback in education is a powerful tool which enhances learning (Economides, 2006). In accordance with the present results, a previous study by Perrin et al. (2006) demonstrated that immediate feedback supports learners immediately after an activity. In order to deliver and control feedback to a very large number of students allowing them to learn at the same time despite their differences, the learning should be given through technology. According to Alabdulaziz & Higgins (2017, p586) "Individualized learning through a computer can allow a student to observe the speed at which they achieve their targets, providing feedback on current performance, and maybe motivating students to continue with their tasks". For instance, it might bring reassurance to the learner after the exam starts, may reduce the learners" susceptibility to panic when he/she answers incorrectly and may also encourage and congratulate the learner on his/her effort. While feedback was not provided in the previous experiment, participating students suggested it would be a beneficial feature in the mobile app as learners wanted to see how well they are doing at the same time. Eppard et al. (2016) reported that, according to the data collected for their research which was about making suggestions for choosing Apps for a foundational level English program in the United

Arab Emirates, participants expressed their requirements to have some features in apps, such as instant feedback. Based on this finding, the researcher used that function in the current study, with the MADE-ME App incorporating a process of error correction and presenting instant feedback to guide the learner to do more learning reviews on the information belonging to a question that has been answered incorrectly, which in turn increased learners" knowledge, completing a loop of learning that could drive the achievement of learning objectives. As reported in Bahrin's (2011) thesis, one of the engaging elements in game-based learning is outcomes and feedback. Algahtani (2011) claimed that the use of multimedia and interactive modes of learning contributed in the evaluation of the effectiveness of e-learning in Saudi universities, and by providing instant feedback that significantly reflected immediate benefits, including high motivation and more enjoyment of learning. The results of this hypothesis is consistent with human computer interaction research which proved the use of technology has the ability to improve student interaction (Lee at al, 2012).

When the current hypothesis was moderated with *gender* and *major/subject*, the results indicated that there were no significant differences between groups in term of gender or major among the three ways of learning. The aspect of feedback interactions observed by the researcher was an interesting function that been used in the web app and was appreciated by the majority of the students regardless their genders or majors.

Key Findings:

Students" perceptions of the importance of learning interactivities had a strong influence on engagement.

9.2.13 Discussion of Results Related to the (H5) Experiment 2:

Hypothesis five in Experiment 2 stated that "the interaction activities in m-learning will positively influence students' performance expectancy". Interestingly, the results of this hypothesis showed that there were positive significant correlations between the feedback activities and the students" performance expectancy, regardless of the mode of delivery. According to Economides(2006) and Robert (2012), feedback activities may increase the learner's belief on a test's usefulness and that may relax their concerns of test fairness. Derouin et al., (2005) suggested that when feedback is given frequently

and immediately following a response from the learner in e-learning, it is helpful to learners and makes them knowledgeable.

When the current hypothesis was moderated with *gender* and *major/subject*, the results indicated there were no significant differences between groups in term/subject of gender nor major among the three ways of learning of video, audio and cartoon.

9.2.14 Discussion of Results Related to the (H10) Experiment 2:

Hypothesis ten in Experiment 2 stated that "the interaction activities in m-learning will positively influence students' effectiveness". The results of this hypothesis showed that there were statistically significant relationships and influences of the interactive activities (quick exercise, feedback and self-assessment) on the students" pedagogical performance and effectiveness if the learning was via m-learning with audio and video avatar. This significant effectiveness result stemming from the audio mode of delivery was found in the first experiment. In contrast, the video mode of delivery was not so successful. As a result of this the MADE-ME model was extended in order to incorporate the interaction features and the findings showed that the video mode also significantly improved student performance. This implied that students who had more interactions with the content were able to perform better in m-learning settings. In the current research, the web-app is found to be a flexible learning tool which allows interactions with human characters, including movement and voice and that something agreed with by Farsi (2016). According to Tang & Byrne (2007), the level of interaction increased learners" participation in the learning activities, which in turn influenced positively the learning outcomes. The approach to learning used in this study is an assistive learning approach and the results are in line with previous studies, such as those reported in Ghaith (2013, p169), which suggested that blended learning shows better efficiency and effectiveness of training in the U.S.; the results of the study illustrated that 77% of U.S institutions currently use blended learning. Sonak et al. (2002) conducted a study on how feedback affects learner performance through the internet, and the results here showed a positive significant relationship between feedback via the internet and students" academic achievement. Further to that, Bates (2015) concluded a case study by saying that feedback supports learners to learn and apply the self-learning strategies which helps them to succeed in online learning.

When moderated by *gender*, the results show that female students score higher than male students in the audio, video and cartoon post-test score. A possible reason for this may be related to female students feeling as if they were in a real class which can present the info, assessing and testing their knowledge about the subject matter, and then providing them with their progress via the auto feedbacks. However, no significant differences were found when moderated with *major/subject*.

9.3 Discussion of the Open-Ended Questions

Open-ended questions were used at the end of the questionnaires which can be characterised as qualitative data. The relationship between qualitative and quantitative methods is complementary rather than exclusive enriching the data by exploring issues further and supplementing the quantitative work as part of the validation process. The discussion of these are as follows:

9.3.1 Discussion of Experiment 1 Open-Ended Comments

Advantages/benefits of Mobile Devices for Student Learning

Looking at Table 8 - 87, many students (n=47) pointed out that the "pause and repeatable" features of the lessons on the mobile device were the most useful advantages of that approach of learning. Students found the ability to "stop" and "pause" lessons and to fast-forward or rewind materials in order to repeat and/or focus on any part of the content, was the feature which distinguished m-learning over traditional face-to-face learning. This is consistent with earlier studies (Kukulska-hulme et al., 2015; Evans, 2008; Farsi, 2016). According to the statistical analysis of this study, significantly more students believed that revising and learning from m-learning materials made them more receptive and effective than learning via traditional face-to-face. M-learning has an additional advantage, which is the ability to access information quickly. A factor identified by a number of students (n=20) who perceived the use of m-learning provided the ease and speed with which information could be found. They felt that mobile devices allowed them to reach course materials quickly, a finding confirming the investigations of Algahtani (2011) and Gikas & Grant (2013). This advantage was also mentioned by one of the instructors, who she was asked: "What benefits do you expect that this kind of education through mobile device will bring to the students?". She stated:

"... It gives them the access to review whatever they need in seconds"

Table 8 - 87, also shows that students (n=33) believed that the convenience of learning from mobile devices was a useful aspect of m-learning. Other, m-learning allows students to access learning materials at a time that is convenient and from a location at which were not required to meet there by reducing the need to physically move from place to place for lectures This theme of m-learning advantages has been discussed elsewhere (Al-emran et al., 2016; Liaw, 2008). Because of the connectivity to the internet and the small weight of mobile devices, students felt that it would allow them to be used at their convenience, regardless of place and/or time. An instructor of these participants described this feature as:

"M-learning seems to be convenient to our world today, as you know, most of the students spend a lot of time in their phones"

Further:

"... It's a great brilliant idea. Simply, because most of them spend their whole time on their cell phones"

And again:

"They are interested. There are some reasons like they have their phones almost all day"

In addition, Table 8 - 87 presents other advantages of adopting m-learning in Saudi higher education, such as it being easy to understand the content, and the increased engagement and motivation, which they believe would help them to concentrate on the learning materials (n=18; 6; 5, respectively). The provision of increased concentration and enhanced learning through m-learning are all perceptions associated with the positive intentions, attitudes and usefulness components of the TAM, leading the researcher to assume that they have changed their way of learning from the routine face-to-face learning to incorporating technology in their learning which enhance their motivation of learning. The results from this research supports findings of Chen & Wu (2015) who concluded that the usage of multimedia technology in education positively affects learning processes and encourages learners to concentrate. Some learners commented as following:

"... Especially to break the daily routine and to attract the attention of more students"

In addition,

"I think the use of technology in education helps us to understand with faster, stronger and more helping keeping information in the mind of the student . . . and create an atmosphere of fun that helps to reduce the boredom during the lecture at the classroom"

Challenges of Mobile Devices for Student Learning

Although the participants considered m-learning was helpful, a number of challenges and barriers from learning via mobile devices were also identified. Table 8 - 87, shows some of these challenges. Some students (n=15) identified that the low speed of internet connection is one of the issues which prevented the acceptance of using m-learning in Saudi Arabia. This study realised that low internet connectivity, whether at home or in the university, was related to the land's geographic factors of having many large mountains, which disturbs the implementation of m-learning, as highlighted by the prior studies (Mirza & Al-Abdulkareem, 2011; Tarus et al., 2015). M-learning has an additional barrier, as identified by a number of students (n=12) who stated that students" objections m-learning were based on miscommunications a lack of interaction with other peers and instructors to ask about concepts not understood. When students studied via the distance learning format, it minimised the level of discussion, as well as the level of contact, among students. Liaw (2008) also found that this is a concern when implementing m-learning. The researcher therefore determine this point as an objective to improve the autonomy of the learning and to convert the learners experience from being passive to interactive. One of the interviewee's instructors supported the authors" assumption and expressed that:

"M-learning may stimulate students' self-learning mode and change the way of learning and receiving English language by practicing the language or answer some electronic tests .. All that because the English learning required learning with practice and understanding until the student reaches the top of the level of knowledge and information" Furthermore, looking at Table 8 - 87, a small number of students (n=8) claimed that learning via using mobile device may be disturbed by the small size of the screen. Whilst this was certainly a problem when using old fashion types of devices such as Nokia and Panda; it is seen based on the statistical analysis of this study that, all of the sample of this population owned smartphones such as iPhones and Galaxy, which have greater success of screen size and user acceptance. In addition, there were a very small number (n=4) who claimed the limited batteries of the mobile device may be an issue in implementing m-learning, however, this could be solved with the new device versions and by portable chargers. Moreover, "escape" from learning to other entertainments or social media was pointed out by (n=13) students something that can be tackled by distance learning tracking which encourages learners to complete the lesson and remain focused. Table 8 - 87, also shows a lack of awareness, attention and motivation in mlearning as one of the barriers that might make so students avoid the use of m-learning; this could be solved by having professional and well-designed interfaces of learning. Providing instructors with training courses and technical support on how to use technology in education environment would tackle this issue and reduce the number of staff who are against the adoption of technology in learning (Alabdulaziz & Higgins, 2016). Other barriers that have been highlighted by participants, include the difficulty of tracking and monitoring students" feedback; students postponing completion of the lesson; and reluctance to receive the lesson because it disturbed other functions of their mobile device. A further barrier from the student's point of view is high internet subscription fees although only (n=3) highlighted this case. From this small number of students, researchers do not assume that this is a big issue, especially as all Saudi higher education students are joining universities in their country free of fees and they are also receiving a monthly allowance from the government to cover all studying requirements (Alfarani, 2016).

9.4 Chapter Summary

This chapter has discussed and interpreted the results from the previous chapter quantitatively and qualitatively, seeking to answer the research questions regarding the way in which m-learning can be used effectively for learning the English language. The first part of the chapter discussed the evaluation of the results that based on statistical tests to find out the relationships and correlations between factors and evidence from the literature was also given to support these findings. The summary of the main findings confirm the preference for mode of delivery and students" perceptions has the largest impact on engagement. In addition, it also showed that the level of enjoyment experienced in a mobile learning is influenced by the learner's level of intrinsic engagement. Also, the study identified that students" preferences for the learning mode of delivery and their engagement for any learning styles were not linked to pedagogical performance (effectiveness), consistent with previous studies which reported similar findings (Al-azawei et al. 2016; Prajapati et al. 2011). The chapter provided a discussion of the quantitative findings and supported them with the qualitative perceptions of the participants in the second part of the chapter. The potential advantages and challenges of implementing m-learning among Saudi learners from their perceptions have been discussed. This study can confirm that the use of learning interactivities while designing the m-learning lessons were the strongest factor which influence m-learning effectiveness.

The summary of the main contributions of the work are presented in the next chapter.

10 Chapter: Conclusion and Future Works

10.1 Introduction

This chapter summaries the research undertaken, highlights its novelty and the implications for the field. The aim of the research was to investigate how different ways of delivering learning content to students can influence learning outcomes and optimize the learning process especially when they are away from the traditional classroom. More specifically, this study focused on investigating how m-learning can be used to assist the teaching of compulsory English modules within Higher Education courses in Saudi Arabia.

The chapter begins by revisiting each of the objectives of the research and the corresponding research questions, highlighting how each was addressed during the research. The chapter then summaries the major contributions to the field stemming from the research and proposes a set of recommendations linked to the implementation of m-learning in Saudi Arabia. Finally the study identified the limitations of the current study and identified future work directions.

10.2 Revisiting the Research Questions

To evaluate if this thesis was successful in achieving its objectives, this section gives a review and insight into the main questions it set out to address. The overall aim of this research was to investigate the increasing interest in m-learning from an educational point of view in Saudi Arabia and to ascertain how m-learning can be used as a tool to complement and/or substitute traditional learning environments, with specific attention being paid to the ways in which engagement and performance in learning can be influenced by the type of avatar representation of the teacher on the mobile device. Associated with this aim a number of objectives was identified and a number of research questions defined.

10.2.1 The Research Objective and Question 1:

 To identify the main benefits, opportunities and challenges of m-learning when adopted in Saudi Arabia from a students" perspective, and to investigate students" readiness and willingness to use an m-learning approach in their studies within the context of a specific module within their degree courses. The research question which associated with this objective was as follows:

• What are the benefits, opportunities and challenges that m-learning can bring to the student population within specific higher education institutions in Saudi?

This question was answered by reviewing the literature as a means of collecting information to seed the development and support of the research model. Within this question, the potential benefits and barriers of m-learning from the literature, were identified. From open-ended questions in the questionnaire and from the discussions occurring during the co-creating workshop which expanded participants comments and explored their views and perceptions further.

10.2.2 The Research Objective and Question 2:

 To investigate the most preferred/engaging avatar representation of a teacher (audio, video, image, cartoon, text) for delivery of learning content via mobile technology.

The research question which associated with this objective was as follows:

• What is the preferred/most engaging way of representing the teacher through an avatar (audio, video, image, cartoon, text) on a mobile device?

This question was answered by conducting the first case study which explored the different types of that could be used to represent the teacher when learning content is delivered via mobile phone, with feedback being collected from the pre and post questionnaires for Experiment - 1.

10.2.3 The Research Objective and Question 3:

- To develop an educational model that links delivery of learning content via mobile technologies with pedagogical performance by:
 - a) Setting the variables and factors that align with the research context.
 - b) Testing a set of hypotheses to determine whether engagement with specific avatar types has a significant impact on pedagogical performance.

The research question which associated with this objective was as follows:

• Is there a significant relationship between students" preference for engagement with particular avatar types and their pedagogic performance; and can such potential relationships be represented on the research model?

This question was answered by reviewing current technology acceptance models and extending them to take into account key factors associated with this study in order to create the conceptual MADE-ME model (Multi Avatar Delivery Environment for Mobile Education) described in Chapter 6, and the definition of a corresponding set of hypotheses by which engagement and pedagogical performance can be assessed.

10.2.4 The Research Objective and Question 4:

- To design and create the framework for an online web-app that can deliver mlearning content to a mobile device via different avatar representations of the teacher (audio; video; image, cartoon; text) and which can be used to test the pedagogical effectiveness of each approach by:
 - a) Constructing different m-learning avatar interfaces.
 - b) Delivering the content via mobile web-app.
 - c) Testing students" pedagogic performance of avatar interface type through the mobile we-app.
 - d) Providing students with the opportunity to co-create and re-design their best m-learning interface framework based on their opinions, preferences and performance.

The research question which associated with this objective was as follows:

• How can different avatar representations of the teacher be used to develop and deliver learning content via a mobile web-app in order to engage students and improve their pedagogic effectiveness?

This question was answered by designing and developing The MADE-ME web app framework as detailed in Chapter 7. Five types of multimedia (audio, video, cartoon, image, text) were initially implemented to deliver English learning material to learners through that mobile app, followed by adjustment of the avatar interface on the basis of a co-creation workshop.

10.2.5 The Research Objective and Question 5:

- To evaluate the proposed m-learning model and web-app through:
 - a) Testing the effectiveness of engagement and pedagogical performance through a first stage of experimental design via questionnaires and exam scores.
 - b) Assessing the second round of experiments based on participants" perceptions through a second stage of experimental design via questionnaires and exam scores.

The research question which associated with this objective was as following:

• What conclusions can be drawn by investigating the links identified through the evaluation of engagement and pedagogical performance in m-learning when the teacher is represented by different avatar types using a cohort of students studying on a degree course at a university in Saudi Arabia?

To address this question, two experiments/case studies were conducted. The m-learning web app framework and the research model were further redeveloped based on participants" feedbacks and perceptions stemming from the co-creation workshop. Once the design stage was finalized, implementation and evaluation of the final model was undertaken in the second experiment, which enabled the researcher to analyse participants" thoughts via quantitative analysis as well as qualitative open-ended comments.

10.3 Contributions of the Thesis

Stemming from this work there been three key contributions to the field and a number of recommendations which made to the Ministry of Higher Education in Saudi Arabia regarding m-learning. This study contributes to the field of m-learning delivery both theoretically and practically by investigating the effectiveness of learning via the use of mobile avatar interfaces in higher education. This thesis presents its key contribution as following.

10.3.1 MADE-ME Model

This study fills a gap in the literature by providing a new conceptual research model. This model extends the current technology acceptance model (TAM) by including a number of important factors related to the Saudi Arabian m-learning context. The model factors were obtained from a combination of educational models in prior developed studies of online learning and it extends them by adding new variables. The proposed model of this research, MADE-ME (Multi Avatar Delivery Environment for Mobile Education), is an important output from this work. The MADE-ME has been used to identify how students can use different ways of learning through mobile devices to receive English language content. The model provides hypotheses and criteria to ensure successful implementation of the technology and to evaluate the engagement and experience of the students using the technology. The Saudi country has a unique culture, in particular, it follows gender segregation in schools and universities classes which in turn is reflected in the education environment. Therefore, there is a need to add the gender variable as a moderator on all investigated factors to find out the differences between genders and what impact these differences have on intentions to use mlearning. Further, this model defined how students can interact with different avatar representations of the teacher to deliver learning content by identifying how the interactive elements affected the effectiveness of m-learning and how interactivity can be improved in m-learning applications. As a consequences, this new model of evaluation could be utilised in further research to evaluate the engagements and effectiveness of m-learning at higher education institutions.

The MADE-ME model of (Multi Avatar Delivery Environment for Mobile Education) which is illustrated in Figure 10 - 1, gives the ability for educators and instructional designers to consider how they might design new learning activities for their mobile learners.



Figure 10 - 1 The model of Multi Avatar Delivery Environment for Mobile Education (MADE-ME)

10.3.2 **MADE-ME Web-App**

Another output from this study is the design and development of a web-app platform for delivering learning content on mobile phones. the MADE-ME (Multi Avatars Delivery Environment for Mobile Education) web-app has been developed to deliver learning content with the optimisation of learning outcomes in mind. The MADE-ME web-app enables the delivery of lessons using a range of avatars to represent the teacher (audio, video, cartoon, image, text) and to deliver interactive learning content to students see Figure 10 - 2. The most significant element of the app and its platform is that it is designed to provide learners with the opportunity to learn through exercises and receive immediate feedback. The app is also designed to test their retention of the content they have been delivered. Further, the instructor can provide different types of tests such as multiple choice, true or false, drag and drops and/or open-ended questions. The mobile app can be accessed anytime and anywhere via an internet connection and it can be used to deliver course material in any language from any point in the world. For the purpose of this study the MADE-ME app was used to deliver English language teaching material, but it could be used to deliver content for any educational course.



Figure 10 - 2 The web-app framework of Multi Avatar Delivery Environment for Mobile Education (MADE-ME)

10.3.3 Case Study

The third key contribution from this thesis is a comprehensive case study investigating the perceptions of a group of students studying an English language module in Al-Baha university in order to determine how they engaged with the mobile content and how effective their learning was when using the MADE-ME app.

In light of the results that were found which were identified while implementing the research and based on a review of the literature, the researcher groups some recommendations in order to achieve the objective of m-learning implementation in Saudi Arabian universities.

10.4 Implication of the Research to Saudi Arabian Education

The study provides positive information for the Saudi Arabian Higher Educational Ministry, and this information may also help them change their conditions and regulations toward using new learning approaches such as a blended learning method combining m-learning with traditional face-to-face teaching. The findings of the study show this approach has been appreciated by students, and the thesis presents the importance of m-learning engagement and effectiveness that could encourage pedagogical policy makers to adopt this approach of blended learning in Saudi universities. In addition, the results of this research provide researchers, educators, m-learning application developers and practitioners with important guidelines for designing successful mobile learning application and how to motivate learners about the

advantages of m-learning higher education study. The findings of this research will not only assist m-learning application developers and practitioners to design better userexperienced interfaces and promote this new approach to potential users, but also provide insights into research on mobile learning acceptance. In light of the thesis findings, the research recommends Saudi universities to consider adopting m-learning through applications such as the MADE-ME web-app to improve the learning process as well as increasing engagement and effectiveness of learning this type of app. This type of application or similar could facilitate and serve the cultural norms in the country such as gender issues. Whilst the MADE-ME app supports easy delivery of online content to mobile devices, for this to be universally acceptable, however, it will still be necessary for Saudi universities to provide training courses for educators on how to use such m-learning web-apps and incorporate multimedia and interactivity elements.

10.5 The Study's Limitations

Despite the many strength of this research, there are a number of ways in which it could be improved linked to scope, culture and the available technical infrastructure. These can be summarised as follows:

- The research has been limited to two faculties" of students (Arts and Science) at one of the Saudi public universities (Al-Baha University). Furthermore, because Al-Baha University is a new university, it may not be typical if compared to older universities in Saudi Arabia and findings cannot be generalised to other (old or new universities) populations.
- Even though three case studies with three groups of participant have been conducted, this is still a relatively small number of participants and further trials are needed to confirm accuracy of findings and generalisations to other populations.
- The researcher tried to conduct the second experiment with the same sample of students who undertook the first experiment but they had moved into the second year where English language is not a core module. Although there were 19 respondents, this was too small a number for valid comparisons to be made for this set.
- Due to cultural norms, the female avatar had to be used as a static image and it was not possible to present a female instructor with an uncovered face on an online teaching platform in Saudi Arabia.

- The rules of the Saudi Higher Education Ministry do not allow female students to bring their mobile phones to the university campus, therefore, the female participants may have found that the m-learning approach to be less convenient to them than the male counterparts.
- Due to unreliable internet connections in the male university campus, the students" perceptions may have been negatively affected against the behavioural intention to use m-learning or against the convenience factor.
- The research model focused on only eight of the subscales to investigate: engagement, enjoyment, performance expectancy, convenience, effectiveness, interactive and the behavioural intention to use m-learning. Literature reviews have been undertaken comprehensively on these factors, however, the findings of the study contribute to knowledge about these subscales alone.
- This experiment did not track or capture students" login activity to the application.

10.6 Future Research

The limitations of this study as well as the promising findings to date create opportunities for future research which are outlined as following:

- 1- Further studies can be undertaken with more to confirm the current findings and also to increase the generalisation of the research model and the web-app. In addition, further factors can be explored and integrated into the MADE-ME model.
- 2- The experiments could be extended to include students from different faculties or subjects of study; students from different cities across Saudi Arabia; students from different urban versus nomadic cultural backgrounds; students at public versus private universities; and those at Middle Eastern universities in general.
- 3- The experiments involving students can examined from their instructors" point so that they can become informed about potential benefit and limitations of implementing mlearning in the educational process. As mentioned earlier, out of the scope of the research, the researcher conducted a short interview with the participants" English instructors and their supervisor who supported this as a point to be considered in future research by commenting that:

"As an English Supervisor, I found that the experiment is really more effective for both students and teachers. In somehow, this experiment gave us a general overview of how to teach students the English course as a second language through a new way of teaching via convenience technology which I obviously realised its impact on their effectiveness and improvement".

- 4- It would be useful to analyse data collected from the 2^{nd,} 3rd and fourth year students at the same university and compare the findings with the main participants of this study (foundational year). Students at these higher levels are studying most of their courses in English and they may realise the benefit of this approach or have different concerns about it as an approach.
- 5- The research addressed some cultural challenges with female students and their teachers. That is due to the cultural restrictions of gender segregation within Saudi Arabia. Female students should have m-learning with female teachers only inside the campus of the university. However, this will require a female researcher to conduct this experiment for female students which may encourage and motivate them to learn from someone of the same gender.
- 6- Further research can be conducted to design and develop educational lessons specifically for mobile devices instead of programmes which are prepared to be delivered via PCs and then optimised to work on mobile devices.
- 7- Hosting the mobile web app on the university's online learning portal such as Moodle would provide the opportunity to capture and track students" login activities, providing rich log files that track student activities within the system and indicate the extent of their engagement with m-learning.
- 8- Conducting similar research with other taught modules being provided in Al-Baha University to explore whether or not there are subject specific differences between the research models factors.

10.7 Chapter Summary

This final chapter has revisited the research objectives and questions and how they have been addressed in this thesis. The three main contributions of the work to the field (model, web-app, and case study) have been summarised and recommendation from, limitations of, and opportunities for further work have been highlighted.

11 References

- Abu-al-aish, A., & Love, S. (2013). Factors Influencing Students "Acceptance of M-Learning: An Investigation in Higher Education. *The International Review of Research in Open and Distance Learning*, 14(5), 1–11. Retrieved from http://www.irrodl.org/index.php/irrodl/article/view/1631
- Abubakar, I. R., Al-Shihri, F. S., & Ahmed, S. M. (2016). Students" assessment of campus sustainability at the University of Dammam, Saudi Arabia. *Sustainability (Switzerland)*, 8(1), 1–14. http://doi.org/10.3390/su8010059
- Adam, F., & Pomerol, J.-C. (2009). Better Executive Information with the Dashboard Approach. In *Encyclopedia of Information Science and Technology* (pp. 335–340). University College Cork, Ireland: IGI Globa.
- Adham, R., Lundqvist, K., & Parslow, P. (2016). The Use of Avatars in Gender Segregated Online Learning within MOOCs in Saudi Arabia. *Global Learn 2016*. Retrieved from https://www.learntechlib.org/p/172713
- Al-Alwani, S. (2005). Brarriers to integrating information technology in Saudi Arabia Science Education.
- Al-azawei, A., Al-bermani, A., & Lundqvist, K. (2016). Evaluating the Effect of Arabic Engineering Students "Learning Styles in Blended Programming Courses Research Objectives, 15, 109–130.
- Al-Azawei, A., Parslow, P., & Lundqvist, K. (2016). Barriers and Opportunities of E-Learning Implementation in Iraq : A Case of Public Universities. *International Review* of Research in Open and Distributed Learning, 17(5).
- Al-emran, M., Elsherif, H. M., & Shaalan, K. (2016). Investigating attitudes towards the use of mobile learning in higher education. *Computers in Human Behavior*, 56, 93–102. http://doi.org/10.1016/j.chb.2015.11.033
- Al-Fahad, F. (2009). Students" Attitudes and Perceptions towards the Effectiveness of Mobile Learning in King Saud University, Saudi. *The Turkish Online Journal of Educational Technology*, 8(2). Retrieved from http://eric.ed.gov/?id=ED505940
- Al-Gahtani, S. S., Hubona, G. S., & Wang, J. (2007). Information technology (IT) in Saudi Arabia: Culture and the acceptance and use of IT. *Information and Management*, 44(8), 681–691. http://doi.org/10.1016/j.im.2007.09.002
- Al-Ghaith, W., Kawano, L., & Sandhu, K. (2010). Factors Influencing the Adoption and Usage of Online Services in Saudi Arabia. *The Electronic Journal on Information Systems in Developing Countries*, 1(40), 1–32. http://doi.org/.
- Al-hunaiyyan, A., Alhajri, R., & Al-sharhan, S. (2017). Investigating the determinants and age and gender differences in the acceptance of mobile learning. *International Journal of Interactive Mobile Technologies*, *11*(4), 4–16.
- Al-khairy, M. H. (2013). English As a Foreign Language Learning Demotivational Factors As Perceived By Saudi Undergraduates. *European Scientific Journal*, 9(32), 365–382.
- Al-Tamimi, A., & Shuib, M. (2009). Motivation and attitudes towards learning English: A study of petroleum engineering undergraduates at Hadhramout University of Sciences and Technology. *GEMA Online Journal of Language Studies*, 9(2), 29–55.
- Al-zahrani, A. M. (2011). Preparing Pre-service Teachers for the Effective Use of Educational Technology in Saudi Arabia Submitted by.
- Alabdulaziz, M., & Higgins, S. (2016). Obstacles to technology use when addressing Saudi primary students" mathematics difficulties. *International Journal of Engineering Research and Technology*, 5(10), 412–417. http://doi.org/10.1063/1.2756072
- Alabdulaziz, M., & Higgins, S. (2017). Understanding Technology Use and Constructivist Strategies when Addressing Saudi Primary Students â€TM Mathematics Difficulties.

International Journal of Innovative Research in Science, Engineering and Technology, *6*(1), 105. http://doi.org/10.15680/IJIRSET.2017.0601105

Aldhaban, F. A. (2016). *PDXScholar Exploratory Study of the Adoption and Use of the Smartphone Technology in Emerging Regions : Case of Saudi Arabia*. Portland State University. Retrieved from

http://pdxscholar.library.pdx.edu/cgi/viewcontent.cgi?article=3656&context=open_acce ss_etds

- Alebaikan, R. a. (2010). Perceptions of Blended Learning in Saudi Universities. *Thesis for the Degree of Doctor of Philosophy, University of Exeter*, (September), 350.
- Alenazi, G. (2015). Institutional Support for Academic Sta to Adopt Virtual Learning Environments (VLEs) in Saudi Arabian Universities. University of Durham.
- Alenezi, A., Karim, A., Malek, A., & Veloo, A. (2010). An Empirical Investigation into the Role of Enjoyment, Computer Anxiety, Computer Self-Efficacy and Internet Experience in Influencing the Students" Intention to Use. *Turkish Online Journal of Educational Technology*, 9(4), 22–34. Retrieved from http://eric.ed.gov/?id=EJ908069
- Alfarani, L. A. K. (2016). Exploring the Influences on Faculty Members "Adoption of Mobile Learning at King Abdulaziz University, Saudi Arabia. *E-Theses for the Degree* of Doctor of Philosophy, University of Leeds, School of Education, (February).
- Algahtani, A. (2011). Evaluating the Effectiveness of the E-learning Experience in Some Universities in Saudi Arabia from Male Students "Perceptions. *Durham University E-Theses for the Degree of Doctor of Philosophy*, 27, 123–129.
- Alhabeeb, A., & Rowley, J. (2017). Critical success factors for eLearning in Saudi Arabian universities. *International Journal of Educational Management*, 31(2), 131–147. http://doi.org/10.1108/IJEM-01-2016-0006
- Alhaisoni, E. (2012). Language Learning Strategy Use of Saudi EFL Students in an Intensive English Learning Context. Asian Social Science, 8(13), 115–127. http://doi.org/10.5539/ass.v8n13p115
- Alharthi, A., Alassafi, M. O., Walters, R. J., & Wills, G. B. (2017). An exploratory study for investigating the critical success factors for cloud migration in the Saudi Arabian higher education context. *Telematics and Informatics*, 34(2), 664–678. http://doi.org/10.1016/j.tele.2016.10.008

Alhassan, R. (2016). Mobile Learning as a Method of Ubiquitous Learning: Students" Attitudes, Readiness, and Possible Barriers to Implementation in Higher Education. *Journal of Education and Learning*, 5(1), 176. http://doi.org/10.5539/jel.v5n1p176

Aljuaid, H. (2010). Language learning strategies: Perceptions of female Saudi EFL learners. Lancaster University Postgraduate Conference in Linguistics & Language Teaching, 2– 24.

Allen, M., & Bourhis, J. (2002). Comparing student satisfaction with distance education to traditional classrooms in higher education: A meta-analysis. *The American Journal of Distance Education*, 16(2). Retrieved from http://www.tandfonline.com/doi/abs/10.1207/S15389286AJDE1602_3

Allert, J. (2004). Learning style and factors contributing to success in an introductory computer science course. *Proceedings IEEE International Conference on Advanced Learning Technologies (ICALT04)*, 33(1), 184–188.

http://doi.org/10.1109/ICALT.2004.1357442

Almalki, G., Finger, G., & Zagami, J. (2013). Introducing SMART Table Technology in Saudi Arabia Education System. ... *Journal of Advanced Computer Science* & ..., 4(2), 46–52. Retrieved from http://www.thesai.org/Downloads/Volume4No2/Paper_7-Introducing_SMART_Table_Technology_in_Saudi_Arabia_Education_System.pdf
Almarwani, M. A. (2011). ML for EFL : Rationale for mobile learning. In *International* Conference ICT for Language Learning, 4th edition.

- Alothman, M., Robertson, J., & Michaelson, G. (2017). Computer usage and attitudes among Saudi Arabian undergraduate students. *Computers and Education*, *110*, 127–142. http://doi.org/10.1016/j.compedu.2017.02.010
- Alshumaim, Y., & Alhassan, R. (2010). Current availability and use of ICT among secondary EFL teachers in Saudi Arabia: Possibilities and reality. *Paper Presented at the Globle Learn Asia Pacific 2010 Conference on Learning and Technology*, 523–532. Retrieved from http://www.editlib.org/pv/34227
- Annetta, L., & Holmes, S. (2006). Creating presence and community in a synchronous virtual learning environment using avatars. *International Journal of Instructional Technology and* Retrieved from http://itdl.org/Journal/Aug_06/Aug_06.pdf#page=31
- Antonacci, D., DiBartolo, S., Edwards, N., Fritch, K., McMullen, B., & Murch-shafer, R. (2008). The power of virtual worlds in education: A second life primer and resource for exploring the potential of virtual worlds to impact teaching and learning. *Angel Learning*, (July), 1–8. Retrieved from
 - http://scholar.google.com/scholar?hl=en&btnG=Search&q=intitle:The+Power+of+Virtu al+Worlds+in+Education+:+A+Second+Life+Primer+and+Resource+for+Exploring+th e+Potential+of+Virtual+Worlds+to+Impact+Teaching+and+Learning#0
- Arnold, B. J., Koehler, M. J., & Greenhalgh, S. P. (2016). Design Principles for Creating and Maintaining Immersive Experiences in Educational Games. SITE Conference 2016.
- Artino, A. R. (2010). Online or face-to-face learning? Exploring the personal factors that predict students" choice of instructional format. *Internet and Higher Education*, 13(4), 272–276. http://doi.org/10.1016/j.iheduc.2010.07.005
- Asiimwe, E. N., & Grönlund, Å. (2017). Practices and challenges in an emerging m-learning environment Mathias Hatakka. *International Journal of Education and Development Using Information and Communication Technology (IJEDICT)*, 13(1), 103–122.
- Badwelan, A., Drew, S., & Bahaddad, A. A. (2016). Towards Acceptance M-Learning Approach in Higher Education in Saudi Arabia. *International Journal of Business and Management*, 11(8), 12–30. http://doi.org/10.5539/ijbm.v11n8p12
- Baek, Y., & Touati, A. (2017). Exploring how individual traits influence enjoyment in a mobile learning game. *Computers in Human Behavior*, 69, 347–357. http://doi.org/10.1016/j.chb.2016.12.053
- Baek, Y., Xu, Y., Han, S., & Cho, J. (2015). Exploring Effects of Intrinsic Motivation and Prior Knowledge on Student Achievements in Game-based Learning. *Smart Computing Review*, 5(5), 368–377. http://doi.org/10.6029/smartcr.2015.05.001
- Bahrin, Z. S. (2011). Mobile game-based learning (mGBL) engineering model. *E-Thesis in University Utara Malaysia*. Retrieved from http://etd.uum.edu.my/2807/
- Baker, E. W., Al-Gahtani, S. S., & Hubona, G. S. (2007). The effects of gender and age on new technology implementation in a developing country: Testing the theory of planned behavior (TPB). *Information Technology & People*, 20(4), 352–375. http://doi.org/10.1108/09593840710839798
- Baki, R. (2004). Gender-Segregated Education in Saudi Arabia: Its Impact on Social Norms and the Saudi Labor Market. *Education Policy Analysis Archives*, 12(28), 1–12. Retrieved from http://eric.ed.gov/?id=EJ852310
- Barab, S., Thomas, M., Dodge, T., Carteaux, R., & Tuzun, H. (2005). Making learning fun: Quest Atlantis, a game without guns. *Educational Technology Research and Development*, 53(1), 86–107. http://doi.org/10.1007/BF02504859
- Barrow, L. (2009). Virtual worlds project proposal, (June).
- Bates, A. W. (2015). *Teaching in a Digital Age Gidelines for Designing Teaching and Learning*. a Creative Commons Attribution-NonCommercial 4.0 International.

Retrieved from http://teachonline.ca/sites/default/files/pdfs/teaching-in-a-digital-age_2016.pdf%5Cn(http://opentextbc.ca/teachinginadigitalage/

- Baylor, A. M. Y. L., & Ryu, J. (2003). the Effects of Image and Animation in. J. Education Computing Research, 28(4), 373–394.
- Bidin, S., & Ziden, A. A. (2013). Adoption and Application of Mobile Learning in the Education Industry. *Procedia - Social and Behavioral Sciences*, 90(InCULT 2012), 720– 729. http://doi.org/10.1016/j.sbspro.2013.07.145
- Bingimlas, K. (2009). Barriers to the successful integration of ICT in teaching and learning environments: A review of the literature. *Eurasia Journal of Mathematics, Science & Technology*, 5(3), 235–245. Retrieved from http://ejmste.com/v5n3/EURASIA v5n3.pdf#page=53
- Biocca, F., & Harms, C. (2002). Defining and measuring social presence: Contribution to the networked minds theory and measure. *Proceedings of PRESENCE*, 1–36. Retrieved from

http://scholar.google.com/scholar?hl=en&btnG=Search&q=intitle:Defining+and+measu ring+social+presence+:+Contribution+to+the+Networked+Minds+Theory+and+Measur e#0

- Blascovich, J. (2002). A theoretical model of social influence for increasing the utility of collaborative virtual environments. *Proceedings of the 4th International Conference on Collaborative Virtual Environments*, 25–30. http://doi.org/10.1145/571878.571883
- Botha, A., Herselman, M., & van Greunen, D. (2010). Mobile user experience in a mlearning environment. Proceedings of the 2010 Annual Research Conference of the South African Institute of Computer Scientists and Information Technologists on -SAICSIT '10, (August 2016), 29–38. http://doi.org/10.1145/1899503.1899507
- Boutsika, E. (2014). Kinect in education: A proposal for children with autism. *Procedia Computer Science*, 27(Dsai 2013), 123–129. http://doi.org/10.1016/j.procs.2014.02.015
- Bradford, A. (2017). What is Science. Retrieved from http://www.livescience.com/20896-science-scientific-method.html
- Brekel, A. van den. (2007). Get your consumer health information from an avatar!: health and medical related activities in a virtual environment. *European Association for Health Information &* Retrieved from

http://www.eahil.eu/conferences/2007Krakow/www.bm.cm-

- uj.krakow.pl/eahil/proceedings/oral/vanBrekel.pdf
- Breakwell, G. M., Fife-Schaw, C., Hammond, S. and Smith, J. (2006) Research Methods in Psychology, 3rd Edition, London: Sage.
- Brindley, J., & Walti, C. (2009). Creating effective collaborative learning groups in an online environment. *The International Review of Research in Open and Distance Learning, University of Oldenburg, Germany*, (2000). Retrieved from http://www.irrodl.org/index.php/irrodl/article/vi.ew/675/1271
- Brown, H. D. (2000). Principles of language teaching and learning. *White Plains, NY: Longman*, (4th ed). Retrieved from http://www.csun.edu/~SB4310/How People Learn.pdf
- Brown, R., Ryu, H., & Parsons, D. (2006). Mobile helper for university students: a design for a mobile learning environment. *Proceedings of the 18th Australia Conference on Computer-Human Interaction: Design: Activities, Artefacts and Environments*, 297– 300. http://doi.org/10.1145/1228175.1228227
- Bonk, C., Kim, K. J. & Zeng, T. (2006). Future Directions of Blended Learning in Higher Education and Workplace Settings. In Bonk, C. & Graham, C. R. (Eds.) Handbook of

Blended Learning: Global Perspectives, Local Designs. (pp. 550- 567). San Francisco: Pfeiffer.

- Bühler, C., & Fisseler, B. (2007). Accessible E-Learning and Educational Technology -Extending Learning Opportunities for People with Disabilities. In *Michael E. Auer.Conference ICL2007, September* (pp. 1–11). Retrieved from https://hal.archivesouvertes.fr/hal-00257138/
- Burgstahler, S. (2015). Universal design of instruction (UDI): Definition, principles, guidelines, and examples. *Disabilities, Opportunities, Internetworking, and Technology* (*Do-It*), 1–4. Retrieved from http://www.washington.edu/doit/universal-design-instruction-udi-definition-principles-guidelines-and-examples
- Chae, S. W., Lee, K. C., & Seo, Y. W. (2016). Exploring the Effect of Avatar Trust on Learners" Perceived Participation Intentions in an e-Learning Environment. *International Journal of Human-Computer Interaction*, 32(5), 373–393. http://doi.org/10.1080/10447318.2016.1150643
- Chanchary, F. H., & Islam, S. (2011). MOBILE LEARNING IN SAUDI ARABIA -PROSPECTS AND CHALLENGES M- CHALLENGES. International Arab Conference on Information Technology, Najran University, Saudi Arabia (ACIT'2011).
- Chen, C., & Wu, C. (2015). Effects of different video lecture types on sustained attention, emotion, cognitive load, and learning performance. *Computers & Education*, 80, 108– 121. http://doi.org/10.1016/j.compedu.2014.08.015
- Cheng, W., & Warren, M. (2006). Having second thoughts: Student perceptions before and after a peer assessment exercise. *Studies in Higher Education*, 22(2), 233–239. http://doi.org/10.1080/03075079712331381064
- Chih-Ming, C., & Chung-Hsin, W. (2015). Effects of Different Video Lecture Types on Sustained Attention, Emotion, Cognitive Load, and Learning Performance. In *Proceedings - 2015 IIAI 4th International Congress on Advanced Applied Informatics, IIAI-AAI 2015* (Vol. 80, pp. 108–121). http://doi.org/10.1109/IIAI-AAI.2015.225
- Choy, S., Berkner, L., Lee, J., & Topper, A. (2009). Academic Competitiveness and SMART Grant Programs: First-Year Lessons Learned. *US Department of Education*. Retrieved from http://eric.ed.gov/?id=ED504171
- Cochran, J. D., Baker, H. M., Benson, D., & Rhea, W. (2016). Business Student Perceptions of Online Learning: Using Focus Groups for Richer Understanding of Student Perspectives. Organization Management Journal, 13(3), 149–166. http://doi.org/10.1080/15416518.2016.1218195
- Cochrane, T. D. (2010). Exploring mobile learning success factors. *ALT-J, Research in Learning Technology*, *18*(2), 37–41. http://doi.org/10.1080/09687769.2010.494718
- Cohen, L., Manion, L., & Morrison, K. (2007). *Research Methods In Education. Sixth Edition, London Routledge*.
- Connell, B. R., Jones, M., Mace, R., Mueller, J., Mullick, A., Ostroff, E., et al. (1997). The principles of universal design. www.ncsu.edu/ncsu/design/cud/ about_ud/udprinciplestext.htm
- Corbeil, J., & Valdes-corbeil, M. E. (2007). Are you ready for mobile learning? *Educause Quarterly*, *30*(2), 51–58. Retrieved from http://www.learning-centric.net/mobile.cc/relatedinfo.pdf
- Creswell, J. (2013). Research design: Qualitative, quantitative, and mixed methods approaches. *SAGE Publications International Educational Professional Publisher*. Retrieved from

http://scholar.google.com/scholar?hl=en&btnG=Search&q=intitle:Research+design+qua litative,+quantitative,+anad+Mixed+Methods+Approaches#0

Crittenden, E. (2006). Grounded Theory : A Research Methodology for e-learning.

Malaysiain Journal of Distance Education, 8(1), 1–13.

- Crompton, H., Burke, D., Gregory, K. H., & Gräbe, C. (2016). The Use of Mobile Learning in Science: A Systematic Review. *Journal of Science Education and Technology*, 25(2), 149–160. http://doi.org/10.1007/s10956-015-9597-x
- Davis, F. (1985). A Technology Acceptance Model for Empirically Testing New End-User Information Systems. *Theory and results, Massachusetts Institute of Technology, Cambridge, MA*, (December 1985), 291.
- Davis, F., Bagozzi, R., & Warshaw, P. (1989). User Acceptance of Computer Technology: A Comparison of two Theoretical Models. *Mangagement Science*, *35*(8), 982–1002. http://doi.org/10.1287/mnsc.35.8.982
- Dawes, J. (2008). Do data characteristics change according to the number of scale points used? An experiment using 5-point, 7-point and 10-point scales. *International Journal of Market Research*, *50*(1), 61–77. http://doi.org/Article
- Deeds, D. W. (2013). Avatars Don "t Steal Souls Virual Worlds education Roundtable, (June).
- Derouin, R. E., Fritzsche, B. A., & Salas, E. (2005). E-Learning in Organizations. Journal of Management, 31(6), 920–940. http://doi.org/10.1177/0149206305279815
- Deterding, S., Sicart, M., Nacke, L., O'Hara, K., & Dixon, D. (2011). Gamification. using game-design elements in non-gaming contexts. *Proceedings of the 2011 Annual Conference Extended Abstracts on Human Factors in Computing Systems - CHI EA '11*, 2425. http://doi.org/10.1145/1979742.1979575
- Deuchar, S., & Carolyn, N. (2003). The Impact of Avatars and 3D Virtual World Creation on Learning. Proceedings of the 16th Annual NACCQ, Palmerston North New Zealand July, 2003 (Eds) Mann, S. and Williamson, A. Www.naccq.ac.nz, 255–258. Retrieved from http://www.citrenz.ac.nz/conferences/2003/papers/255.pdf
- Economides, A. (2006). Emotional feedback in CAT (Computer Adaptive Testing). International Journal of Instructional Technology & Distance Learning, 3(2), 11–16. Retrieved from

http://www.researchgate.net/publication/27381331_Emotional_feedback_in_CAT_(Computer_Adaptive_Testing)

- Edmonds, W. A., & Kennedy, T. D. (2017). An applied guide to research designs: Quantitative, qualitative, and mixed methods. Retrieved from https://books.google.co.uk/books?hl=en&lr=&id=x32zDAAAQBAJ&oi=fnd&pg=PP1 &dq=quantitative+approach+and+the+qualitative+approach++mixed+methods&ots=sy ZdDmKmfY&sig=3CvqNGOHPuqI-w-ae9ZVBHx9EvI#v=onepage&q=quantitative approach and the qualitative approach mi
- Elyas, T., & Picard, M. (2010). Saudi Arabian educational history: Impacts on English language teaching. *Education, Business and Society: Contemporary Middle Eastern Issues*, *3*, 136–145. http://doi.org/10.1108/17537981011047961
- Eppard, J., Nasser, O., & Reddy, P. (2016). The next generation of technology: Mobile apps in the english language classroom. *International Journal of Emerging Technologies in Learning*, 11(4), 21–27. http://doi.org/10.3991/ijet.v11i04.5293
- Evans, C. (2008). The effectiveness of m-learning in the form of podcast revision lectures in higher education. *Computers and Education*, *50*(2), 491–498. http://doi.org/10.1016/j.compedu.2007.09.016
- Falloon, G. (2010). Using avatars and virtual environments in learning: What do they have to offer? *British Journal of Educational Technology*, *41*(1), 108–122. http://doi.org/10.1111/j.1467-8535.2009.00991.x
- Farooq, U., Schafer, W., Rosson, M. B., & Carroll, J. M. (2002). M-education: Bridging the gap of mobile and desktop computing. *Proceedings IEEE International Workshop on*

Wireless and Mobile Technologies in Education, WMTE 2002, 91–94. http://doi.org/10.1109/WMTE.2002.1039226

Farsi, M. (2016). interactive Islamic Prayer. Durrham E-Theses, 463.

- Fleming, J. (1998). Web navigation: designing the user experience. In *Chapter 5: Interface* and Interaction design (p. 288).
- Flowers, P. (2009). Research Philosophies Importance and Relevance. *Leading Learning and Change*, 1(3), 1–5. Retrieved from http://www.networkedcranfield.com/cell/Assignent Submissions/research philosophy - issue 1 - final.pdf
- Fosso Wamba, S., Bhattacharya, M., Trinchera, L., & Ngai, E. W. T. (2017). Role of intrinsic and extrinsic factors in user social media acceptance within workspace: Assessing unobserved heterogeneity. *International Journal of Information Management*, *37*(2), 1–13. http://doi.org/10.1016/j.ijinfomgt.2016.11.004
- Fox, J., Ahn, S. J. G., Janssen, J. H., Yeykelis, L., Segovia, K. Y., & Bailenson, J. N. (2015). Avatars versus agents: A meta-analysis quantifying the effect of agency on social influence. *Human-Computer Interaction*, 30(5), 401–432. http://doi.org/10.1080/07370024.2014.921494
- Fox, Z. (2013). *The 15 Countries With the Highest Smartphone Pentration. Our Mobile Planet*. Retrieved from http://mashable.com/2013/08/27/global-smartphone-penetration/#m5OriXWnsaqI
- Galliers, R. D., & Land, F. F. (1987). Choosing appropriate information systems research methodologies. *Communications of the ACM*, *30*(11), 901–902. http://doi.org/10.1145/32206.315753
- García-Peñalvo, F. J., & Durán-Escudero, J. (2017). Interaction Design Principles in WYRED Platform. In P. Zaphiris & A. Ioannou (Eds.), *International conference on Learning and Collaboration Technologies*. (pp. 371–381). Cham: Springer International Publishing. http://doi.org/10.1007/978-3-319-58515-4_29
- Gefen, D., & Straub, D. W. (1997). Gender differences in the perception and use of e-mail: An extension to the Technology Acceptance Model. MIS Quarterly, 21(4).
- Geng, & Chunling. (2017). On the Teaching Innovation of Business English Teaching: A study on Multimodal Communicative Competence of Ethnic Universities. *Theory and Practice in Language Studies*, 7(4), 322.

http://doi.org/http://dx.doi.org/10.1108/17506200710779521

- Ghaith, O. (2013). The Impact of Blended Learning on Female Student-Teachers in Kuwait. *A Thesis in School of Sport and Education Brunel University*, 215.
- Ghavifekr, S., Hussin, S., & Ghani, M. F. A. (2011). The Process of Malaysian Smart School Policy Cycle : A Qualitative Analysis. *Journal of Research and Reflection in Education*, 5(2), 83–104.
- Gikas, J., & Grant, M. M. (2013). Mobile computing devices in higher education: Student perspectives on learning with cellphones, smartphones & social media. *Internet and Higher Education*, *19*, 18–26. http://doi.org/10.1016/j.iheduc.2013.06.002
- Giordano, J., O'Reilly, M., Taylor, H., & Dogra, N. (2007). Confidentiality and autonomy: the challenge(s) of offering research participants a choice of disclosing their identity. *Qualitative Health Research*, 17(2), 264–275. http://doi.org/10.1177/1049732306297884

Glaser, B. (2017). Discovery of grounded theory: Strategies for qualitative research. Retrieved from

https://books.google.co.uk/books?hl=en&lr=&id=GTMrDwAAQBAJ&oi=fnd&pg=PP1 &dq=definition+of+grounded+theory+learning+&ots=JsUaMAqxd0&sig=E642CtpDg_ R5jM8dd2UkNVbabYo#v=onepage&q=definition of grounded theory learning&f=true

- Gomes, A. J., & Mendes, A. J. (2010). A study on student performance in first year CS courses. Proceedings of the Fifteenth Annual Conference on Innovation and Technology in Computer Science Education - ITiCSE '10, 113. http://doi.org/10.1145/1822090.1822123
- González, m. A., santos, b. S. N., vargas, a. R., martín-gutiérrez, j. & orihuela, A. R. 2013. Virtual Worlds. Opportunities and Challenges in the 21 st Century. *Procedia Computer Science*, 25, 330-337.
- Greenwald, S. W., Kunert, A., Weimar, B., Beck, S., Weimar, B., Fröhlich, B., & Weimar, B. (2017). Technology and Applications for Collaborative Learning in Virtual Reality. In 12th International Conference on Computer Supported Collaborative Learning CSCL 2017 (pp. 719–726). Pennsylvania.
- Gulliksen, J. Goransson, B. Boivie, I. Blomkvist, S. (2003). Key principles for user-centred systems design Gulliksen Jan.
- Guo, Y. R., & Goh, D. H.-L. (2015). Affect in Embodied Pedagogical Agents. Journal of Educational Computing Research, 53(1), 124–149. http://doi.org/10.1177/0735633115588774
- Guy, R. (2009). *The evolution of mobile teaching and learning*. information science press. Retrieved from http://books.google.com/books?hl=en&lr=&id=Cz5SCEaHNAMC&oi=fnd&pg=PR9& da=Tha+Evolution+of+Mabila+Teaching+and+Learning&ots=U4_6PVh7Xv&scig=6lk
 - dq=The+Evolution+of+Mobile+Teaching+and+Learning&ots=U4_6RYh7Xx&sig=6lk LGRxih-_wSlU1qgeFbK7_-mQ
- Haake, M. (2006). Virtual Pedagogical Agents-Beyond the Constraints of the Computational Approach. *Lucs.lu.se*. Retrieved from http://www.lucs.lu.se/wp-content/uploads/2013/09/haake licentiate thesis 2006.pdf
- Han, J. H., & Finkelstein, A. (2013). Understanding the effects of professors" pedagogical development with Clicker Assessment and Feedback technologies and the impact on students" engagement and learning in higher education. *Computers and Education*, 65, 64–76. http://doi.org/10.1016/j.compedu.2013.02.002
- Harris, S., Sherritt, L., Holder, D., Kulig, J., Shrier, L., & Knight, J. (2008). Reliability and Validity of the Brief Multidimensional measure of religiousnessSpirituality Among adolescents. *International Electronic Journal of Elementary Education*, 47(4), 438–457. http://doi.org/10.1007/s
- Hawk, T., & Shah, A. (2007). Using learning style instruments to enhance student learning. *Decision Sciences Journal of Innovative Education*, *5*(1), 1–19. Retrieved from http://onlinelibrary.wiley.com/doi/10.1111/j.1540-4609.2007.00125.x/full
- Hawkins, D. (2016). Designing Mobile Learning Environments To Support Teacher-Led Field Trips Within Informal LearningEnvironments. The University of Texas at San Antonio. Retrieved from https://search.proquest.com/docview/1793669157?pqorigsite=gscholar
- Hodhod, R. (2010). Interactive Narrative for Adaptive Educational Games : Architecture and an Application to Character Education. *Ph.D. Thesis in Computer Science Department*, (February).
- Hong, K., Ridzuan, A., & Kuek, M. (2003). Students" attitudes toward the use of the Internet for learning: A study at a university in Malaysia. *Journal of Educational Technology & Society*, 6(2), 29–39. http://doi.org/10.1177/1473095211400151
- Hong, K., & Songan, P. (2011). ICT in the changing landscape of higher education in Southeast Asia. *Australasian Journal of Educational ...*, 27(8), 1276–1290. Retrieved from http://www.ncbi.nlm.nih.gov/pmc/articles/PMC3045104/
- Hoppe, H., Joiner, R., Milrad, M., & Sharples, M. (2003). Guest editorial: Wireless and Mobile Technologies in Education. *Journal of Computer Assisted Learning*, 19(3), 255–

259. http://doi.org/10.1046/j.0266-4909.2003.00027.x

- Hsu, L. (2013). English as a foreign language learners" perception of mobile assisted language learning: a cross-national study. *Computer Assisted Language Learning*, 26(3), 197–213. http://doi.org/10.1080/09588221.2011.649485
- Huang, H.-M., Rauch, U., & Liaw, S.-S. (2010). Investigating learners" attitudes toward virtual reality learning environments: Based on a constructivist approach. *Computers & Education*, 55(3), 1171–1182. http://doi.org/10.1016/j.compedu.2010.05.014
- Huang, Y., Hwang, W., & Chang, K. (2010). Guest Editorial-Innovations in Designing Mobile Learning Applications. *Journal of Educational Technology & Socitey*, 13(3). Retrieved from http://www.thai-library.org/Resource/0000005765.pdf#page=6
- Hung, I. C., Yang, X. J., Fang, W. C., Hwang, G. J., & Chen, N. S. (2014). A context-aware video prompt approach to improving students^{**} in-field reflection levels. *Computers and Education*, 70, 80–91. http://doi.org/10.1016/j.compedu.2013.08.007
- Hyland, N. E. (2010). Social Justice in Early Childhood Classrooms What the Research Tells Us, (January).
- Iqbal, S., & Bhatti, Z. A. (2015). Investigation of University Student Readiness towards Mlearning using Technology Acceptance Model, *16*(4), 83–103.
- Ismail, R., Lai, Y. M., Ayub, Z. A., Ahmad, A. R., & Wan, C. Da. (2016). Kingdom of Saudi Arabia. In Y. M. Lai, A. R. Ahmad, & C. Da Wan (Eds.), *Higher Education in the Middle East and North Africa: Exploring Regional and Country Specific Potentials* (pp. 127–146). Singapore: Springer Singapore. http://doi.org/10.1007/978-981-10-1056-9_7
- İşman, A., Willis, J., Fahme, D., & Eric Zhi, F. L. (2015). Computer-Assisted Pronunciation Learning in a Collaborative Context: A Case Study in Taiwan. *TOJET: The Turkish Online Journal of Educational Technology – October 2015*, 14(4).
- Jairak, K., Praneetpolgrang, P., & Mekhabunchakij, K. (2009). An acceptance of mobile learning for higher education students in Thailand. *The Sixth International Conference* on eLearning for Knowledge-Based Society, Thailand, (December), 36.1-36.8. Retrieved from http://ijcim.th.org/SpecialEditions/v17nSP3/36_Full_Kallaya Jairak_Online.pdf
- Jeng, Y.-L. ., Wu, T.-T. ., Huang, Y.-M. ., Tan, Q. ., & Yang, S. J. H. . (2010). The add-on impact of mobile applications in learning strategies: A review study. *Educational Technology and Society*, 13(3), 3–11. Retrieved from http://www.scopus.com/inward/record.url?eid=2-s2.0-
 - 77958083212&partnerID=40&md5=4eaf58630947b4b0f36df11920f3fac4
- Johnson, B., & Larry, C. (2008). Educational Research: Quantitative, Qualitative, and Mixed Approaches. *SAGE Publications, Inc*, (3), 1–12.
- Johnson, L., Adams Becker, S., & Estrada, V., and Freeman, A. (2015). *NMC Horizon Report: 2015 Higher Education Edition*. Austin, Texas: The new media Consortium. Retrieved from http://cdn.nmc.org/media/2015-nmc-horizon-report-HE-EN.pdf
- Jones, I. (2016). Do the locomotion, Obstinate Avatars, Dehiscent Performances, and the Rise of the Comedic Video Game. The velvet light TRAP (Vol. 77).
- Joyce, A. (2017). Co-creation and Design Thinking to Envision More Sustainable Business Models: A Foresight Design Approach for Organizational Sustainability of SME Manufacturers. In J. Bellemare, S. Carrier, K. Nielsen, & F. T. Piller (Eds.), Managing Complexity: Proceedings of the 8th World Conference on Mass Customization, Personalization, and Co-Creation (MCPC 2015), Montreal, Canada, October 20th-22th, 2015 (pp. 173–193). Cham: Springer International Publishing. http://doi.org/10.1007/978-3-319-29058-4 14
- Jurisevic, M., Gla? zar, S. A., Pu? cko, C. R., & Devetak, I. (2008). Intrinsic motivation of pre-service primary school teachers for learning chemistry in relation to their

academic achievement. International Journal of Science Education, 30(1), 87e107. http://dx.doi.org/10.1080/09500690701455879.

- Kabilan, M. K. (2016). Using Facebook as an e-portfolio in enhancing pre-service teachers" professional development. *Australasian Journal of Educational Technology*, *32*(1), 19–31. Retrieved from https://www.scopus.com/inward/record.uri?eid=2-s2.0-84961833603&partnerID=40&md5=b2e78aeb06710df9e1a648ed09df1fd2
- Kassaian, Z. (2007). LEARNING STYLES AND LEXICAL PRESENTATION MODES. *Estudios de Lingüística Inglesa Aplicada (Elia), 7, 53-78., 6,* 53–78. Retrieved from http://www.ncbi.nlm.nih.gov/entrez/query.fcgi?cmd=Retrieve&db=PubMed&dopt=Cita tion&list_uids=19893492
- Katz, J., & Rice, R. (2002). Project syntopia: social consequences of internet use. IT & Society, 1(1), 166–179. Retrieved from http://scholar.google.com/scholar?hl=en&btnG=Search&q=intitle:project+syntopia+Soc ial+consequences+of+internet+use#0
- Kim, Y. H., Kim, D. J., & Wachter, K. (2013). A study of mobile user engagement (MoEN): Engagement motivations, perceived value, satisfaction, and continued engagement intention. *Decision Support Systems*, 56(1), 361–370. http://doi.org/10.1016/j.dss.2013.07.002
- Kim, Y., & Wei, Q., (2011). The impact of learner attributes and learner choice in an agentbased environment. *Computers & Education*, 56(2), 505-514.
- Kim, Y., & Baylor, A. L. (2007). Pedagogical agents as social models to influence learner attitudes. *Educational Technology*, 47(1), 23-28.
- Koohang, A. A. (1989). A study of attitudes towards computers: Anxiety, confidence, liking, and perception of usefulness. *Journal of Research on Computing in Education*, 22(2), 137–150.
- Korning, P., & Hebo, T. (2014). Establishing Internationally Competitive Wind Turbine Component Production in South Africa: A Feasibility Study.
- Kukulska-hulme, A., Norris, L., & Donohue, J. (2015). *Mobile pedagogy for English language teaching : a guide for teachers. ELT Research Papers* (Vol. 14.07). Teaching English.
- Kukulska-Hulme, A., Sharples, M., Marcelo, M., Arnedillo-S'anchez, Inmaculada; Giasemi, Milrad, Marcelo; Arnedillo-S'anchez, I. and V., & Giasemi, V. (2009). Innovation in mobile learning: A European perspective. *International Journal of ...*. Retrieved from http://www.igi-global.com/article/innovation-mobile-learning/2755
- Lai, C.-L., Gwo-Jen Hwang, B., Jyh-Chong Liang, B., Tsai, C.-C., Hwang, G.-J., & Liang, J.-C. (2016). Differences between mobile learning environmental preferences of high school teachers and students in Taiwan: a structural equation model analysis. *Education Tech Research Development*, 64(3), 533–554. http://doi.org/10.1007/s11423-016-9432-y
- Laureano-cruces, A. L., Velasco-santos, P., Mora-torres, M., & Ramírez-rodríguez, J. (2016). Design of Pedagogical agents : the learning styles , the emotions and the color, 421–431.
- Lee, M.-C. (2009). Understanding the behavioural intention to play online games: An extension of the theory of planned behaviour. *Online Information Review*, *33*(5), 849–872. http://doi.org/10.1108/14684520911001873
- Lee, M. K. O., Cheung, C. M. K., & Chen, Z. (2005). Acceptance of Internet-based learning medium: The role of extrinsic and intrinsic motivation. *Information and Management*, 42(8), 1095–1104. http://doi.org/10.1016/j.im.2003.10.007
- Lee, W. J., Huang, C. W., Wu, C. J., Huang, S. T., & Chen, G. D. (2012). The effects of using embodied interactions to improve learning performance. *Proceedings of the 12th IEEE International Conference on Advanced Learning Technologies, ICALT 2012, 12th*, 557–559. http://doi.org/10.1109/ICALT.2012.104

- Leong, L. Y., Ooi, K. B., Chong, A. Y. L., & Lin, B. (2013). Modeling the stimulators of the behavioral intention to use mobile entertainment: Does gender really matter? *Computers in Human Behavior*, 29(5), 2109–2121. http://doi.org/10.1016/j.chb.2013.04.004
- Liaw, S. S. (2008). Investigating students" perceived satisfaction, behavioral intention, and effectiveness of e-learning: A case study of the Blackboard system. *Computers and Education*, *51*(2), 864–873. http://doi.org/10.1016/j.compedu.2007.09.005
- Liu, K., Shuang, Z., Qin, Y., Zhang, J., & Yan, Y. (2007). Mandarin accent analysis based on formant frequencies. *ICASSP, IEEE International Conference on Acoustics, Speech and Signal Processing - Proceedings*, 4, 637–640. http://doi.org/10.1109/ICASSP.2007.366993
- Liu, Y. (2008). An adoption model for mobile learning. *Proceedings of the IADIS International Conference E-Commerce.*, 235–240. Retrieved from http://www.iadis.net/dl/final_uploads/200814C031.pdf
- Logan, S., Medford, E., & Hughes, N. (2011). The importance of intrinsic motivation for high and low ability readers' reading comprehension performance. *Learning and Individual Differences*, 21(1), 124–128. http://doi.org/10.1016/j.lindif.2010.09.011
- Lombard, M., & Ditton, T. (1997). At the Heart of It All: The Concept of Presence -Lombard - 2006 - Journal of Computer-Mediated Communication - Wiley Online Library. *Journal of Computer-Mediated Communication, Temple University, Philadelphia, Pennsylvania, 3*(2), 0. http://doi.org/10.1111/j.1083-6101.1997.tb00072.xView/save
- Lorraine, W. (1999). The effect of individual goal setting and teacher feedback on students' self-regulation and success in a distance learning environment: A case study .Unpublished doctoral dissertation, Florida University.
- Lowenthal, J. N. (2010). Using Mobile Learning: Determinates Impacting Behavioral Intention. American Journal of Distance Education, 24(4), 195–206. http://doi.org/10.1080/08923647.2010.519947
- Lu, L. (2013). 3D Virtual Worlds as Art Media and Exhibition Arenas: Students" Responses and Challenges in Contemporary Art Education. *Studies in Art Education*, 1–3. Retrieved from http://www.questia.com/library/journal/1P3-2979352221/3d-virtualworlds-as-art-media-and-exhibition-arenas
- Luo, Y., & Shenkar, O. (2017). The multinational corporation as a multilingual community: Language and organization in a global context. In: Brannen M.Y., Mughan T. (Eds) Language in International Business. JIBS Special Collections. Palgrave Macmillan, Cham, 59–92. http://doi.org/10.1057/palgrave.jibs.8400197
- Machado, M., & Tao, E. (2007). Blackboard vs. moodle: Comparing user experience of learning management systems. 37th ASEE/IEEE Frontiers in Education Conference S4J-7, (October10-13), 7–12. Retrieved from http://ieeexplore.ieee.org/xpls/abs_all.jsp?arnumber=4417910

MacIntyre, P. D., MacMaster, K., & Baker, S. C. (2001). The Convergence of Multiple Models of Motivation for second language Learning: Cardner,Pintrich, KUHL, and Mccrosky. Retrieved from

https://books.google.co.uk/books?id=7MELVJorM6AC&pg=PA461&dq=The+converge nce+of+multiple+models+of+motivation+for+second+language+learning:+Gardner,+Pi ntrich,+Kuhl,+and+McCroskey.+Motivation+and+second+language+acquisition.&hl=e n&sa=X&ved=0ahUKEwjzxefGnuX

Maenpaa, M. (2012). Co-creation and sharing experiences in mobile communities.

Maldonado, U. P. T., Khan, G. F., Moon, J., & Rho, J. J. (2010). E-learning motivation and educational portal acceptance in developing countries. *Online Information Review*, 35(1), 66–85. http://doi.org/10.1108/14684521111113597

- Manwaring, K. C., Larsen, R., Graham, C. R., Henrie, C. R., & Halverson, L. R. (2017). Investigating student engagement in blended learning settings using experience sampling and structural equation modeling. *Internet and Higher Education*. http://doi.org/10.1016/j.iheduc.2017.06.002
- Massey, A. P., Ramesh, V., & Khatri, V. (2006). Design, development, and assessment of mobile applications: The case for problem-based learning. *Ieee Transactions on Education*, 49(2), 183–192. http://doi.org/10.1109/te.2006.875700
- Mayhew, D. J. (1999). The usability engineering lifecycle. CHI 98 Conference Summary on Human Factors in Computing Systems - CHI '98, 127–128. http://doi.org/10.1145/286498.286575
- Mazlan, M. (2012). Students" Perception of Motivation to Learn: Does an Avatar Motivate? Doctoral Thesis, Durham University. Available at Durham E-Theses Online: http://etheses.dur.ac.uk/3367/. Retrieved from http://etheses.dur.ac.uk/3367/
- Mazlan, M. N. A., & Burd, L. (2011). Does an avatar motivate? *Proceedings Frontiers in Education Conference, FIE*, 1–6. http://doi.org/10.1109/FIE.2011.6142700
- Meena, S. (2014). Forrester Research World Mobile And Smartphone Adoption Forecast, 2014 To 2019 (Global). Retrieved December 15, 2016, from https://www.forrester.com/report/Forrester+Research+World+Mobile+And+Smartphone +Adoption+Forecast+2014+To+2019+Global/-/E-RES118252
- Mertens, D. M. (2014). Research and evaluation in education and psychology : Integrating Diversity with Quantitative, Qualitative, and Mixed Methods 2009. Retrieved from http://0-eds.a.ebscohost.com.library.vu.edu.au/eds/detail?vid=5&sid=2fcbba0a-7dd6-4b6f-8c50-

09aa9f5f9954@sessionmgr4001&hid=4211&bdata=JnNpdGU9ZWRzLWxpdmU=#db= cat02404a&AN=vic.b1948726

Mertens, D. M. (2015). Research and Evaluation in Education and Psychology. Retrieved from

https://books.google.co.uk/books?hl=en&lr=&id=VEkXBAAAQBAJ&oi=fnd&pg=PP1 &dq=Research+Methods+in+Education,+Sixth+Edition&ots=4ZjAOb69sl&sig=PfD3K nTmGRWO0tIenotWotCKH4s#v=onepage&q&f=true

- Michael, R. J. (2016). Introducing Control in an Open Online Course. In 2016 UKACC 11th International Conference on Control. Belfast. http://doi.org/10.1111/bjet.12484
- Mitchell, P., Parsons, S., & Leonard, A. (2007). Using virtual environments for teaching social understanding to 6 adolescents with autistic spectrum disorders. *Journal of Autism and Developmental Disorders*, 37(3), 589–600. http://doi.org/10.1007/s10803-006-0189-8

Ministry of Finance (2015). Online at:

- https://www.mof.gov.sa/english/downloadscenter/pages/budget.aspx [accessed on 28/07/2017].
- Ministry of Education, MOE (2015). Online at:
- http://www.moe.gov.sa/ar/Ministry/Deputy-Ministry-for-Planning-and-Informationaffairs/HESC/Ehsaat/Pages/default.aspx.
- Monahan, T., Mcardle, G., & Bertolotto, M. (2005). Using 3D Graphics for Learning and Collaborating Online. *Proceeding of Eurographics Association 2005*, 33–40.
- Montgomery, D. (2017). Design and Analysis of Experiments Author: Douglas C. Montgomery. Retrieved from https://www.amazon.de/Design-Analysis-Experiments-Author-

Montgomery/dp/B00XV4CQTU/ref=sr_1_11?s=books&ie=UTF8&qid=1479835758&s r=1-11&keywords=design+and+analysis+of+experiments
- Moore, M. G., & Kearsley, G. (2005). Distance education: A systems view, 2nd ed. *Belmont, CA: Wadsworth*.
- Mupfiga, M., Mupfiga, P., & Zhou, T. G. (2017). Enhancing Teaching and Learning through the Use of Mobile Technologies in Zimbabwean Universities. *Journal of Systems Integration*, 8.2, 43. http://doi.org/10.20470/jsi.v8i2.298
- Naismith, L., Lonsdale, P., Vavoula, G., & Sharples, M. (2004). Literature Review in Mobile Technologies and Learning. *Futurelab, Bristol, U.K., Tech., 11*(Book, Whole), 1–25.
- Nassuora, A. (2013). Students Acceptance of Mobile Learning for Higher Education in Saudi Arabia. *International Journal of Learning Management Systems*, 1(1), 1–9. http://doi.org/10.12785/ijlms/010101
- Mayer, R.E. (2005) The Cambridge Handbook of Multimedia Learning. *New York: Cambridge University Press.*
- Naveed, Q. N., Muhammed, A., Sanober, S., Qureshi, M. R. N., & Shah, A. (2017). Barriers Effecting Successful Implementation of E-Learning in Saudi Arabian Universities. *Barriers Effecting Successful Implementation of E-Learning in Sasudi Arabia* University, 12(6), 94–107. Retrieved from http://online-journals.org/index.php/ijet/article/view/7003/4476
- Ng, R., & Lindgren, R. (2013). Examining the effects of avatar customization and narrative on engagement and learning in video games. *Proceedings of CGAMES'2013 USA*, 87–90. http://doi.org/10.1109/CGames.2013.6632611
- Nguyen, Q., Rienties, B., Toetenel, L., Ferguson, R., & Whitelock, D. (2017). Examining the designs of computer-based assessment and its impact on student engagement, satisfaction, and pass rates. *Computers in Human Behavior*, *76*, 703–714. http://doi.org/10.1016/j.chb.2017.03.028
- Nielsen, J. (1995). *Ten Usability Heuristics for user interfaces design. useit.com*. Retrieved from http://www.useit.com/papers/heuristic/heuristic_list.html
- Noble, I., & Bestley, R. (2016). Visual Research: An Introduction To Research Methods In Graphic Design.
- Norris, J. M. (2016). Language Program Evaluation. *The Modern Language Journal*, *100*(S1), 169–189. http://doi.org/10.1111/modl.12307
- NorthernIrelandSocialCareCouncil. (2014). Beginners Guide Icon for Tips and FAQs -NISCC. Retrieved from

https://niscc.info/storage/resources/2014_11_14_complete_how_to_and_faqs_new.

- Omale, N., Hung, W.-C., Luetkehans, L., & Cooke-Plagwitz, J. (2009). Learning in 3-D multiuser virtual environments: Exploring the use of unique 3-D attributes for online problem-based learning. *British Journal of Educational Technology*, 40(3), 480–495. http://doi.org/10.1111/j.1467-8535.2009.00941.x
- Ong, C. S., & Lai, J. Y. (2006). Gender differences in perceptions and relationships among dominants of e-learning acceptance. *Computers in Human Behavior*, 22(5), 816–829.
- Osakwe, J., Dlodlo, N., & Jere, N. (2017). Where learners" and teachers" perceptions on mobile learning meet: A case of Namibian secondary schools in the Khomas region. *Technology in Society*, *49*, 16–30. http://doi.org/10.1016/j.techsoc.2016.12.004
- Osunkoya, T., & Chern, J. (2013). Gesture-Based Human-Computer-Interaction Using Kinect for Windows Mouse Control and PowerPoint Presentation. *Proceedings of 46th. Midwest Instruction and Computing Symposium*, 92–106.
- Park, Y. (2011). A pedagogical framework for mobile learning: Categorizing educational applications of mobile technologies into four types. *International Review of Research in Open and Distance Learning (IRRODL)*, 12.2, 78–102. http://doi.org/10.3394/0380-1330(2006)32
- Parsons, D., Hokyoung, R., & Cranshaw, M. (2006). A Study of Design Requirements for

Mobile Learning Environments. *Advanced Learning Technologies*, 2006. *Sixth International Conference on*, 96–100. http://doi.org/10.1109/ICALT.2006.1652376

- Perrin, D., Downes, S., Muirhead, B., & Perrin, E. (2006). M-Learning A New Paradigm in Education. *International Journal of Instructional Technology and Distance Learning*, *3*(2), 27–34.
- Peterson, M. (2006). Learner interaction management in an avatar and chat-based virtual world. *Computer Assisted Language Learning*, *19*(1), 79–103. http://doi.org/10.1080/09588220600804087
- Piller, F., Vossen, A., & Ihl, C. (2012). From social media to social product development: The impact of social media on co-creation of innovation. *Die Unternehmung*, 1–22. Retrieved from http://papers.ssrn.com/sol3/Papers.cfm?abstract_id=1975523
- Pourhosein Gilakjani, A. (2011). Visual, Auditory, Kinaesthetic Learning Styles and Their Impacts on English Language Teaching. *Journal of Studies in Education*, 2(1), 104–113. http://doi.org/10.5296/jse.v2i1.1007
- Prajapati, B., Dunne, M., Bartlett, H., & Cubbidge, R. (2011). The influence of learning styles, enrolment status and gender on academic performance of optometry undergraduates. *Ophthalmic and Physiological Optics*, 31(1), 69–78. http://doi.org/10.1111/j.1475-1313.2010.00798.x
- Quinn, C. N. (2011). Designing mLearning TAPPING INTO THE MOBILE REVOLUTION FOR ORGANIZATIONAL PERFORMANCE.
- Rebolledo-Mendez, G., Burden, D., & De Freitas, S. (2008). A model of motivation for virtual-worlds avatars. In *Springer-Verlag Berlin Heidelberg* (Vol. 5208 LNAI, pp. 535– 536). http://doi.org/10.1007/978-3-540-85483-8_76
- Reeves, T., & Hedberg, J. (2003). Interactive Learning Systems Evaluation.
- Robert, D. (2012). An animated pedagogical agent for assisting novice programmers within a desktop computer environment. *A Thesis in Coputer Scince, Staffordshire University*, (September). Retrieved from http://ethos.bl.uk/OrderDetails.do?uin=uk.bl.ethos.570540
- Romrell, D., Kidder, L. C., & Wood, E. (2014). The SAMR model as a framework for evaluating mLearning. *Journal of Asynchronous Learning Network*, *18*(2), 1–15.
- Rouse, M. (2015). What is Framework? Definition from WhatIs.com. Retrieved from http://whatis.techtarget.com/definition/framework
- Rubin, H., & Rubin, I. (2012). RESEARCH PHILOSOPHY AND QUALITATIVE INTERVIEWS. In *Qualitative Interviewing: The art of hearing data* (p. 13). Los Angeles: SAGE Publications.
- Saadé, R. G. and Galloway, I. 2008, 'Understanding intention to use multimedia information systems for learning', *Issues in Informing Science & Information Technology*, Vol. 2, pp. 287-96.
- Saunders, B. M., & Tosey, P. (2012). The Layers of Research Design. *Rapport: The Magazine for NLP Professionals*, 58–59.
- Sedgwick, R. (2001). *Education in saudi arabia*. *World Education News and Reviews*. Retrieved from http://www.alimtiaz3.zainph.com/pdf-files/Education in Saudi Arabia.pdf
- SEG Research. (2008). Understanding Multimedia Learning: Integrating multimedia in the K-12 classroom, (September), 17. http://doi.org/10.1007/978-0-387-73337-1
- Seifert, T. (2014). Pedagogical applications of smartphone integration in teaching -Lecturers", students" & pupils" perspectives. In *Proceedings of the 10th International Conference on Mobile Learning 2014.* (pp. 117–124). http://doi.org/10.4018/ijmbl.2015040101
- Seliaman, M., & Al-Turki, M. (2012). Mobile learning adoption in Saudi Arabia. *World Academy of Science, Engineering and Technology*, 6(9), 346–348. Retrieved from

http://www.waset.org/publications/11339

- Sharples, M. (2002). Disruptive devices: mobile technology for conversational learning. International Journal of Continuing Engineering Education and Lifelong Learning, 12(5/6), 504. http://doi.org/10.1504/IJCEELL.2002.002148
- Shashaani, L., & Khalili, A. (2001). Gender and computers: Similarities and differences in Iranian college students" attitudes toward computers. *Computers and Education*, 37(3), 363–375.
- Shneiderman, B., & Plaisant, C. (2004). Designing the User Interface: Strategies for Effective Human-Computer Interaction.
- Singh, H., & Reed, C. (2001). A white paper: Achieving success with blended learning. *American Society for Training & Development, Centra Software*, (March), 1–11. Retrieved from http://www.leerbeleving.nl/wbts/wbt2014/blend-ce.pdf
- Sohn, J.-H., Park, Y.-H., Yoon, C., Woo, R., Park, S.-J., & Yoo, H.-J. (2005). Low-Power 3D Graphics Processors for Mobile terminals. *IEEE Communications Magazine*, 90–99.
- Sonak, B. C., Suen, H. K., Zappe, S. M., & Hunter, M. W. (2002). The Efforts of a Web-Based Academic Record and Feedback System on Student Achievement at the Junior High School Level. Paper Presented at the Annual Meeting of the American Educational Research Association.New Orleans.
- Stoddart, A., Chan, J. Y.-Y., & Liu, G.-Z. (2016). Enhancing successful outcomes of wikibased collaborative writing: a state-of-the-art review of facilitation frameworks. *Interactive Learning Environments*, 24(1), 142–157. http://doi.org/10.1080/10494820.2013.825810
- Stoicescu, M., & Stanescu, M. (2015). Using mobile phone technology to enhance physical education efficiency. *The International Scientific Conference eLearning and Software* for Education, 3, 199–205. http://doi.org/10.12753/2066-026X-13-131
- Tabak, F., & Nguyen, N. T. (2013). Technology Acceptance and Performance in Online Learning Environments: Impact of Self-Regulation. *Journal of Online Learning and Teaching*, 9(1), 116. Retrieved from

http://search.proquest.com.library.capella.edu/docview/1500386147?accountid=27965% 5Cnhttp://wv9lq5ld3p.search.serialssolutions.com.library.capella.edu/?ctx_ver=Z39.88-2004&ctx_enc=info:ofi/enc:UTF-

8&rfr id=info:sid/ProQ%3Aeducation&rft val fmt=info:ofi/fmt:

- Taleb, Z., & Sohrabi, A. (2012). Learning on the Move: The use of Mobile Technology to Support Learning for University Students. In *Social and Behavioral Sciences (ICEEPSY 2012)* (Vol. 69, pp. 1102–1109). Elsevier B.V. http://doi.org/10.1016/j.sbspro.2012.12.038
- Tang, M., & Byrne, R. (2007). Regular versus online versus blended: A qualitative description of the advantages of the electronic modes and a quantitative evaluation. *International Journal on E-Learning*, 6(2), 257–266. Retrieved from http://www.editlib.org/p/21801
- Thomas, L., Ratcliffe, M., Woodbury, J., & Jarman, E. (2002). Learning styles and perfomance in the introductory programming sequence. *Proceedings of the 33rd SIGCSE Technical Symposium on Com- Puter Science Education SIGCSE*, (2), 33–36. Retrieved from http://0-dl.acm.org.wam.seals.ac.za/citation.cfm?id=563352

Thomson, I. (2002). Thomson job impact study: The next generation of corporate learning. *Retrieved July*, (February). Retrieved from http://scholar.google.com/scholar?hl=en&btnG=Search&q=intitle:Thomson+job+Impac

t+study+the+next+generation+of+corporate+learning#0

- Turan, B. (2014). The Opinions of Teachers on the Use of Cartoon Character in the Mathematics Lesson. *Procedia - Social and Behavioral Sciences*, 141, 1386–1391. http://doi.org/10.1016/j.sbspro.2014.05.239
- Unrau, N., & Schlackman, J. (2006). Motivation and its relationship with reading achievement in an urban middle school. The Journal of Educational Research, 100(2), 81e101. http://dx.doi.org/10.3200/JOER.100.2.81-101.
- Väänänen-Vainio-Mattila, K. (2008). Now let's do it in practice: user experience evaluation methods in product development. *CHI 2008 Proceedings Workshops*, 3961–3964. Retrieved from http://dl.acm.org/citation.cfm?id=1358967
- Vasalou, A., Khaled, R., Holmes, W., & Gooch, D. (2017). Digital games-based learning for children with dyslexia : A social constructivist perspective on engagement and learning during group game-play. *Computers & Education*, 114, 175–192. http://doi.org/10.1016/j.compedu.2017.06.009
- Vaughan, N., Cleveland-Innes, M., & Garrison, D. (2014). Teaching in Blended Learning Environments. Retrieved from http://www.colfinder.org/handle/123456789/4850
- Vázquez-Casielles, R., Iglesias, V., & Varela-Neira, C. (2017). Co-creation and service recovery process communication: effects on satisfaction, repurchase intentions, and word of mouth. *Service Business*, *11*(2), 321–343. http://doi.org/10.1007/s11628-016-0311-8
- Venkatesh, V., Morris, G. M., & Ackerman, P. L. (2000). A longitudinal field investigation of gender differences in individual technology adoption decision making processes. *Organizational Behaviour and Human Decision Processes*, 83(1),33-60.
- Venkatesh, V., Morris, M. G., Davis, G. B., & Davis, F. D. (2003). USER ACCEPTANCE OF INFORMATION TECHNOLOGY: TOWARD A UNIFIED VIEW. *MIS Quarterly*, 27(3), 425–478.
- Vermeeren, A., Law, E., & Marianna, V. (n.d.). User experience evaluation methods: current state and development needs. In *Proceedings: NordiCHI* (pp. 521–530). Retrieved from http://dl.acm.org/citation.cfm?id=1868973
- Wang, C.-C., & Yeh, W.-J. (2013). Avatars with Sex Appeal as Pedagogical Agents: Attractiveness, Trustworthiness, Expertise, and Gender Differences. *Journal of Educational Computing Research*, 48(4), 403–429. http://doi.org/10.2190/EC.48.4.a
- Wang, H., Chignell, M., & Ishizuka, M. (2005). Improving the Usability and Effectiveness of Online Learning: How can Avatars help. *Human Factors and*, 769–773. http://doi.org/10.1177/154193120504900706
- Wang, Y. S., Wu, M. C., & Wang, H. Y. (2009). Investigating the determinants and age and gender differences in the acceptance of mobile learning. *British Journal of Educational Technology*, 40(1), 92–118. http://doi.org/10.1111/j.1467-8535.2007.00809.x
- Wiecha, J., Heyden, R., Sternthal, E., & Merialdi, M. (2010). Learning in a virtual world: Experience with using Second Life for medical education. *Journal of Medical Internet Research*, 12(1). http://doi.org/10.2196/jmir.1337
- Wisneski, J. E., Ozogul, G., & Bichelmeyer, B. A. (2017). Investigating the impact of learning environments on undergraduate students" academic performance in a prerequisite and post-requisite course sequence. *The Internet and Higher Education*, 32, 1–10. http://doi.org/10.1016/j.iheduc.2016.08.003
- Wu, J., & Liu, D. (2007). The effects of trust and enjoyment on intention to play online games. *Journal of Electronic Commerce Research*, 8(2), 128–140.
- Yee, N., Bailenson, J. N., Urbanek, M., Chang, F., & Merget, D. (2007). The Unbearable Likeness of Being Digital : Online Virtual Environments. *Cyberpsychology & Behavior*, 10(1), 115–121. http://doi.org/10.1089/cpb.2006.9984

- Yoon, G., & Vargas, P. T. (2014). Know thy avatar: the unintended effect of virtual-self representation on behavior. *Psychological Science*, 25(4), 1043–5. http://doi.org/10.1177/0956797613519271
- Zhang, D., Zhou, L., Briggs, R. O., & Nunamaker, J. F. (2006a). Instructional video in elearning: Assessing the impact of interactive video on learning effectiveness. *Information and Management*, 43(1), 15–27. http://doi.org/10.1016/j.im.2005.01.004
- Zhang, P., & Aikman, S. (2007). Attitudes in ICT Acceptance and Use. In J. A. Jacko (Ed.), *Human-Computer Interaction. Interaction Design and Usability* (pp. 1021–1030). Berlin, Heidelberg: Springer Berlin Heidelberg. http://doi.org/10.1007/978-3-540-73105-4_112

12 Appendices

Appendix A: Research Ethics approval.

- Appendix B: Supervisor's letter to the Saudi Arabian Cultural Bureau in London for the data collection.
- Appendix C: The Saudi Arabian Cultural Bureau in London approval letter.
- Appendix D: The Al-Baha University approval letter to conduct the experiment.
- Appendix E: The pre-post questionnaire of the Experiment 1.
- Appendix F: The pre-post questionnaire of the Experiment 2.
- Appendix G: The copy of the exam/test.
- Appendix H: A release letter from the main supervisor at Al-Baha University for the completion



The Research Ethics approval

Research Ethics Committee



Project Submission Form

NoteAll sections of this form should be completed.Please continue on separate sheets if necessary.

Principal Investigator: Prof. Rachel McCrindle

School: School of system Engineering

Email: r.j.mccrindle@reading.ac.uk

Title of Project: Application of M-learning Using Avatars to Assist the Teaching of English as Part of Higher Education Courses in Saudi Arabia

Proposed starting date: 24/08/2015

Brief description of Project:

This study is conducted to investigate the potential effectiveness of using mlearning to assist the teaching process and the benefits that it can bring to the Higher Education process in Saudi Arabia.

The study will be conducted amongst the students of Saudi Arabian university in order to identify student preferences for, and engagement with, different avatar types as a way of receiving and interacting with course content and the effects this has on their pedagogic performance.

The proposed methodology for this research is a mixed methodology of both qualitative and quantitative methods based on the "Research Onion" model for collecting data and stepping through the different stages required when formulating an effective research approach. A Grounded theory approach is used whereby the hypothesis is built up as the different stages of the research are undertaken. According to Ross and Morrison (2007), the method of experimental design is used to "test hypotheses regarding causation, for example, that a particular instructional strategy leads to better student performance".

This study to advancement of the field of mobile learning delivery by investigating the effectiveness of learning through the use of mobile avatar interfaces in Saudi Arabian Higher Education courses. Further, it will contribute to the field of mlearning interface by the co-creation and customisation of Avatars interface by the students themselves. I confirm that to the best of my knowledge I have made known all information relevant to the Research Ethics Committee and I undertake to inform the Committee of any such information which subsequently becomes available whether before or after the research has begun.

I confirm that I have given due consideration to equality and diversity in the management, design and conduct of the research project.

I confirm that if this project is an interventional study, a list of names and contact details of the subjects in this project will be compiled and that this, together with a copy of the Consent Form, will be retained within the School for a minimum of five years after the date that the project is completed.

Signad	N	2		
			Date:	11-6-15
(Investigator)				
	•		Date:	11-6-15
(Head of School	or auth	orised He	ad of Department)	
			Date:	11-6-15
(Student -where	applica	ble)		

Checklist

1.	This form is signed by my Head of School (or authorised Head of
	Department)

- 2. The Consent form includes a statement to the effect that the project has been reviewed by the University Research Ethics Committee and has been given a favourable ethical opinion for conduct
- 3. I have made, and explained within this application; arrangements for any confidential material generated by the research to be stored securely within the University and, where appropriate, subsequently disposed of securely.
- 4. I have made arrangements for expenses to be paid to participants in the research, if any, OR, if not, I have explained why not.

No expense will be incurred because it's done as a part of my course.

- 5. EITHER
 - (a) The proposed research does not involve the taking of blood samples;

OR

(b) For anyone whose proximity to the blood samples brings a risk of Hepatitis B, documentary evidence of immunity prior to the risk of exposure will be retained by the Head of School or authorized Head of Department.

Signed:

• • • • • •

Date 11/06/15

(Head of School or

authorised Head of Department)

6. EITHER

(a) The proposed research does not involve the storage of human tissue, as defined by the Human Tissue Act 2004;

 \checkmark

V

OR

(b) I have explained within the application how the requirements of the Human Tissue Act 2004 will be met.

- 7. EITHER
 - (a) The proposed research will not generate any information about the health of participants;

OR

(a) If the research could reveal adverse information regarding the health of participants, their consent to pass information on to their GP will be included in the consent form and in this circumstance I will inform the participant and their GP, providing a copy of the relevant details to each and identifying by date of birth

 $\sqrt{}$

 $\sqrt{}$

OR

(c) I have explained within the application why (b) above is not appropriate.

8. EITHER

(a) the proposed research does not involve children under the age of 5;

OR

(b) My Head of School (or authorised Head of Department) has [given details of the proposed research to the University's insurance officer, and the research will not proceed until I have confirmation that insurance cover is in place.

Signed:

Date

(Head of School or authorised Head of Department)

This form and further relevant information (see Sections 5 (b)-(e) of the Notes for Guidance) should be returned, both electronically and in hard copy, to:

Dr Mike Proven Coordinator for Quality Assurance in Research Whiteknights House Email: m.j.proven@reading.ac.uk

You will be notified of the Committee's decision as quickly as possible, and you should not proceed with the project until a favourable ethical opinion has been passed.



School of

[Avoid using personal contact details. If it is necessary to use a mobile, ensure that it is a project specific mobile] Contact address

phone fax email

Appendix A: INFORMATION SHEET

What is the purpose of the study?

The aim of this study is to investigate the benefits that mobile devices can bring to students' learning. The objective of this study is to develop and build an online mobile web application which has been designed with four types of avatar to interact with learnings while delivering their course content. The main purpose of the study is to measure how engaged and motivated students are to use this approach and how effective the approach is with regards to students retaining information that have been taught. A framework will ultimately be produced that provides a checklist and criteria to ensure successful implementation of the technology and to evaluate the engagement and experience of the technology with the avatar, the acceptance levels and measure the performance of the technology.

How are the participants selected?

We are recruiting from students at Al-Baha University, Saudi Arabia who are studying English as part of their degree courses in the faculties of Computer Science, Science and Art faculty.

Participants will need to be able to understand verbal explanations and writing information in English.

What will I be asked to do?

You will be required to fill in a questionnaire before and after receiving a small amount of course content on your mobile phones designed to help you learn elements of English Language. You will also take part in a short quiz about the material presented. Learning material will presented in English but the questionnaires will be supplied in both English and Arabic. The actual experiment will last for 1-2 hours. Some students will be invited to participate further in the co-creating of user interfaces for content delivery.

Researcher (principal): Email: Phone: Researcher (role):

Email:

What data will be collected, and how will it be used?

Data from the questionnaires and quizzes will be analysed using SPSS and will lead to identification of issues that will be explored by the participants in the co-creating workshops.

Where will the studies take place?

The procedure will take place in the classrooms of Al-Baha University, Saudi Arabia where you are studying.

What if I do not wish to complete the study?

Participants is entirely voluntary, and you can withdraw at any time without giving a reason.

Will my data be kept anonymous?

You will be asked to provide your name and contact details, and to sign a consent form so that the University can keep a record of your participation in the study. However, data from the study will be stored, processed, and reported using an anonymous user ID.

Can I learn the results of the study?

If you would like to learn the results at the end of the study, please contact the researcher.

Who are the researcher responsible for this study?

Experiments and study will be conducted by Ali Alowayr (PhD Computer Science) and will be supervised by Prof Rachel McCrindle.

Contact details:

Ali Alowayr

Email address: <u>a.s.a.alowayr@pgr.reading.ac.uk</u>

Mobile: 0507779632

Please feel free to contact us directly if you have questions about this study:

This project has been subject to ethical review, according to the procedures specified by the University Research Ethics Committee, and has been given a favourable ethical opinion for conduct.





Consent Form

1. I have read and had explained to me by Ali Alowayr

The accompanying Information Sheet relating to the project on:

Application of M-learning Using Avatars to Assist the Teaching of English as Part of Higher Education Courses in Saudi Arabia

Contact details: Ali Alowayr Email address: <u>a.s.a.alowayr@pgr.reading.ac.uk</u> Mobile: 0507779632

- 2. I have had explained to me the purposes of the project and what will be required of me, and any questions I have had have been answered to my satisfaction. I agree to the arrangements described in the Information Sheet in so far as they relate to my participation.
- 3. I understand that participation is entirely voluntary and that I have the right to withdraw from the project any time, and that this will be without detriment.
- 4. I agree to the interview/session being video/audio taped and understand that unless further agreement is obtained I will not be personally identified through these activates.
- 5. I understand that this project has been subject to ethical review, according to the procedures specified by the University Research Ethics Committee, and has been given a favourable ethical opinion for conduct.
- 6. I have received a copy of this Consent Form and of the accompanying Information Sheet.

Name:
Date of birth
Signed:
Date:



The Supervisor's letter to the Saudi Arabian Cultural in London for the data collection



Professor Rachel McCrindle School of Systems Engineering University of Reading Whiteknights Reading, RG6 6AY

1st October 2015

To Whom it May Concern,

Mr Ali Alowayr

PhD Title: Application of m-learning with avatars to deliver English courses for Higher Education in Saudi Arabia

This letter is to confirm that I support the visit of my PhD student, Ali Alowayr, to Saudi Arabia for the period of three months, for the purposes of data collection in Albaha University. Ali needs to collect this data as it is an integral part of his PhD.

Yours faithfully

Professor Rachel McCrindle Professor of Computer and Human Interaction National Teaching Fellow Director of Enterprise, School Systems Engineering University of Reading, RG6 6AY r.j.mccrindle@reading.ac.uk



The Saudi Arabian Cultural Bureau in London approval letter





ROYAL EMBASSY OF SAUDI ARABIA cultural bureau london

التاريخ 1436/08/02هـ

إفادة

رقم الملف BHU100

تفيد الملحقية الثقافية بسفارة المملكة العربية السعودية في لندن بأن المبتعث/ علي سعيد عبدالله آل عوير سجل مدني (1044673091) مبتعث من قبل جامعة الباحة لدراسة الدكتوراة بجامعة University of Reading إعتبارا من 1434-11-04 هـ حتى 1437-12-80 هـ.

وقد أعطيت له هذه الإفادة بناءً على طلبه لتقديمها إلى جامعة الباحة.

وتقبلوا فائق التحيات،،،

الملحق الثقافي
بسفارة المه

فيصل بن محمد المهُنا أبا الخيل

35

الرقم : الموافق : التاريخ : التاريخ :

630 Chiswick High Road, London W4 5RY Tel: +44 (0) 20 3249 7000 Fax: +44 (0) 20 3249 7001 E-mail: sacbuk@uksacb.org www.uksacb.org

Appendix D

The Al-Baha University approval letter to conduct the experiment



KINGDOM OF SAUDI ARABIA MINISTRY OF HIGHER EDUCATION **AL-BAHA UNIVERSITY** (042)

Deanship of Scientific Research

الملكة العربية السعودية وزارة التعليم العالي

جامعة الباحة (٠٤٢)

عمادة البحث العلمي

سلمه الله

سعادة عميد كلية العلوم والآداب بالمخواه

السلام عليكم ورحمة الله وبركاته وبعد،،،

يقوم المحاضر/ علي بن سعيد آل عوير المبتعث في بريطانيا بإجراء دراسة بحثية عن طلاب وطالبات السنة التحضيرية بكلية العلوم والآداب بالمخواه ويرغب في تطبيق الدراسة طلاب وطالبات الكلية و جمع بيانات تتعلق بالبحث الحاقي بدرسة لمرحت الدكتوراه.

وعليه آمل تسهيل مهمة الباحث.

وتقبلوا خالص تحياتي وتقديري سس

Per // 15/ 127

الرقم : التاريخ :

المشفوعات :

تَلْيَضُونَ : ٧٢٧٤١١١ - ٧ - ٢٠٩٦٦ فاكس : ٧٢٤٧٢٧٢ - ٧ - ٢٠٩٦٦ الباحة : ص.بَ (١٩٨٨)

Appendix E

The pre-post questionnaire of the Experiment - 1

Part 1: Questionnaire to investigate the effectiveness of delivering course content through

mobile devices using avatar interfaces

Ali Alowayr

Participant No. الرقم الجامعي

I am a PhD Student at the University of Reading, United Kingdom and as a part of my research I am investigating whether mobile devices can be used effectively for learning. This may be in relation to making learning more <u>effective</u>, <u>engaging</u> or <u>convenient</u>. Your answers to this questionnaire will be helpful to my study and ultimately in enhancing the use of mobile devices for learning. I would appreciate you taking the time to complete the following questions. They should take no more than 10 - 15 minutes of your time. Your participation is voluntary and your responses will be confidential and not attributed to any individual.

بحث قهدم منطالب للقتور الفي جامعه وين في يدول من طلي اول في ه هذا است ط علل أي كجزء من ي في حص لي في عيش التخليم علي جلي ا من خلال است خدام جهاز الجوال كأدا لمتوصري ل الدروس ل في يومش لكل مهتم وماهنم من خلال ملئ كل هذه الامتبال ه، سو فتس اعدالي احث على يت عزي زيون مي ذالتخليم ما الالت روزي است خدام اجمز قال جوال لذلكي ه. أو د ان شكر لك وتتك الذي سو فتست غرق ه والهت يقع ازي أخذ 10 - 15 في هذه الوديل لا غك ان المجلسك على هذا الامت طلاعت كون في الي من على من على المي الذي اسم كوليا.

الجزء الأول: أول متدور حولك وعن ج مازجوالك Section 1: About you and your mobile device

What is your gender?

حد چست.	
☐ Male	🗖 Female
ذك	لاثى

- 3. What is your faculty?
- ما موتخرصك مرتق، لا؟

4.

Computer Science 🔲 علوم حلنب	Science 🗖 علوم	Art المحالي	Other please specify: یغر ذلك، حدد	
Which year of study an	e you in?			
في ايس قندر س؟	. —			

First year	Second year	Third year	Fourth year
السرن الأولى	السرن الخرابي ه	السرين الثيالية ه	السرين الرباعه

5. What mobile devices do you have now or have had in the past? Tick all that apply. مانوع ج مازجو ال اللى حال يأو ال في سياق وأن استخدت مان علامة ($\sqrt{)}$

Type of Mobile	Now	Have had
نوع جوالك	لاحالي	سي ان ان ان ان ان ا
Basic-phone		
ل جو ال ل عادي		
IOS smart phone (e.g. iPhone)		
الخيبون		
Android smart phone (e.g. Samsung Galaxy)		
سلمس جگسي		
Windows smart phone (e.g. Nokia Lumia, LG)		
نواڼي او اِل جي (ا ما ا ما ا		
Tablet (e.g. iPad) الإياد		
Tablet (e.g. Galaxy)		
ا مانادر (e.g. Galaxy) ل بالا سينوت		
Other please specify		

6. If you have a mobile device (smart phone or tablet), how long have you had it?

ا حددكفيترة لمتهلاك لُه؟	لي تُ شال ليلاباد و غيره؛	وُ الْذَكْتِي او الْنِبَال	ن كفتتم في ج ماز جو ال، إماال ج	في حال را
Less than 1 year	□ 1-2 years □	2-3 years	☐ More than 3 years	Do not have one
فللمن سرن، واحده	ىيېن 1-2ىرىنە	يين 2-3سرن ه	لمحلی من جسن ه	لاىلىمك جوال

- 7. On average how many hours a day do you use your mobile device? كم يتوسط العربي الي يوجل الي وجل الي وال؟
 - لا المت خدم مبتعلتا ألفتر من 3ساعات بين 2-3ساعه بين 1-2ساعه في ل في ساعه في ل في ساعه المتحدم مبتعلتا المت
- How easy do you find your mobile device(s) to use?
 في في منه د س ها و است خدام ج ماز جوال ك

ہم ج قرر جوں۔		يو معني		
□Very easy	Easy	Fairly easy	Quit complicated	☐ It is too complicated
سەل جدا	سەل	سەل نو عا ما	مڨدنوعا ما	عقد جدا

9. What features of mobile device(s) do you use? Tick all that apply. ما دي مړيزىك ال جوال الۍ تي سي خدم دا؟ التخبر مايي اسب م عالى بو ض ع علامة ($\sqrt{}$).

Usages of mobile devices	Often	Regularly	Some	Seldom	No at all
استخدامات ج هل ال جو ال	البيالخ	بالثنظام	بعض	نادرا ما	ىلىدا
Making phone calls					
إجراء لمحال ماتت طيء					
Text messaging					
ر س ط لن صري ه					
Using Twitter for social purposes					
استخما تتقير للاغراض الاخماعي،					
Using Facebook for fun					
است خدالها غيبي بلغى ل لتغوي ه					
Using WhatsApp for social purposes					
استخدام الهيئس اببللأغراض الابتماعي					
Calendar					
مل <i> في الثق</i> ر م					
Reading articles, books, online content for fun					
قراءة المقالات وللتخسومحت ويات الثقترق للقغوي ه					
Watching videos (e.g. YouTube) for fun					
مشا هدةمقاطان فيهي والوليتيوي وبداليتماعة والتائيف					
Playing games					
الالعب المتنوعيي ه					
Map facility					
خدمت الخرع،ط					
Taking, sending or viewing photos					
المحاث ه والفقش ه ومشاهدة الصرور					
Taking, uploading videos					
اخذمقاالطافيهفيو وات وفعاها					
Browsing the web for fun					
تصفح الاتتارق للتغوي واللتماعه					
Browsing the web to find educational content (e.g.					
reading articles and attending virtual classes or					
tutorials.					
مصفح و مشاهدة الدروسال غيمي، على الانتهزن تلفقراءه المقالات					
وضور الدروس الصفوف اقتتاراضي ه					
Other please list					
غي ذلك حدد					

الجزء الثلاي النظيم عن طويق لاجوال Section 2: Mobile Learning

10. Do you understand what m-learning involves? لى ل ييانف م عن ماذقيُّص بـالتخليم الثلات ربني عن طوق الجوال؟

□ Yes	🗖 No	Partially	□ Not sure
نعم	لا	جفي	غير بةأكد

11. Given the definition of m-learning as "learning across various contexts, through social and content interactions, using personal electronic mobile devices, thereby enabling students to access learning materials", have you had experience of using m-learning?

بالنظر ال يت عريف النظل عن طريق ال جوال نجد المقولي النظليم أنك ال النتر و عسواءمقروء او مسموع الغذي وي من خلل المشاعل ع المتخوى العلمي باست خدام اجمزة لاجو اللذائي والشخصرية، مطيساعد المتخلف ي سمل مالدخل للماده للطي ه والامف اده في ها. الى لهيك اي بخره او تجبه عن هذا المتخليم؟

🗖 Yes, a lot	🗌 Yes, a little	🗌 No	Not sure
نعم، لخفير ا	نعمقلي	لا	غير بتماكد

12. How important do you think the convenience of m-learning is? ما مدى الا هوه ال ذيرينية ويفرى دارة وملاء والتكل ول وجودل عن طوق ال جوال؟

	وال.	ہوں جن کے کوبی ان	ي راح و هارع مواصع الم	ا مدی اد انوب الدیکی دی	~
□ Very important	Important	🗖 Neutral	□ Not important	□ Not sure	
م مم جدا	مقم	مطيد	فجير مەم	غير بتأكد	

13. In your previous and/or current education what kinds of learning have you experienced? Check all that apply.

في در الله الماضي والحالي، ما من و التعليم الذي جربت؟ التحر ملية اسب معكمو ضع علامة igvee v.

Male and female respondents 👘 🛉			للذكور وا ناث			
Mechanism of learning delivery	Regularly	Some	Seldom	Not at all		
طرقة وصريل التعليم	ببالتهظام او نطءما	بعض	نادرا ما	لابابدا		
Face-to-face lectures (traditional course) in class						
ل ج اض رات تلاق اي بي ه وج ه ال وج ه داخ ل اصف						
Face-to-face screen cast with male instructor in class						
صورة المجاضر من خلاللشاش داخلالقاع الدراسي ه						
Online learning (e-learning) via PCs or laptops						
التعليم الالت رزي عن طريق الكمجيتور أو البلاتوب						
Mobile learning (m-learning) via phones or tablets						
التعليم الالتريني عن طريق ج دازال جوال						
Female respondents only			لم	ن الإث فق		
Face-to-face screen casts with female instructor in class صورة المحاضره من خلال شاشه داخلل قاع الدراسي ه						
Other please list غير ن ك حدد						

14. The following are examples of learning related activities that can be done on a mobile device. Tick all those that you have done or think you would do if they were available.

في وإلى ينتُم وتيدكون سبق لك وأن الت خلق ما عن طريق الجوال الخذل كسعض في ملتها في استخدام ما عن طريق ج ماز الجوال حدد أي في مليتهاسب معظموضع علامة (٧).

Activity examples	Have	Would	Might	Would
يلهل معل الى فعالي ات	done	do	do	not do
	سيق و ان		قد	لاتربغ في
	استخدمت	نستخدمه	نتست خدمه	استخدامه
Register for courses				
يتس جي ل مو الى مرل الدراسي				
Check course timetable				
المتشروبي المجلى الجدلو الدر اسي				
Check course syllabus and references information				
التشريبيكعل عصردات ل في وج و مراجع الحر أ الدر اس ي ل غيمي ه				
Listen to lectures as screen casts				
الاستماع الى المحاضر ات				
View educational videos				
مشا مدمق اطلى فيجيبو التتخليبي ه				
Submit course work				
تسليم الاعطل والبواجيات للنغز لييه				
Access library account for reserve or renew books				
الدخول على الحساب للخاص الملقته وت جيد لخات المست عاره				
Access library database to search for books or information				
الدخول على ق اعدة عيانات الملقت الوبينع في المصادر لل في مي ه				
Use social media for education (e.g Facebook)				
استخدام ادوالتليات ولصل الاجتاميجي ف يالتكلي مثمال فعيس بوك				
Pay fees				
فدع الرسوم				
Take tests				
إجراء الخبارات				

section 3: Avatars مثمال أو وانحيال

- 15. Have you heard of the term avatar?
 - لى سي ق وان سم عتب مرطل حال ممم ل أو التعش بدور شخص ما، مم ال المؤم او ال طبيب اللختار وفي الا عنه سي ق وان سم عتب مرطل حال ممم ل أو التعش موي الا عنه منه عنه المعني المعني المنافقة المعني المعني المعني الم غير مم اك لا ي المعني مراجع المحمد المحمد المحمد المعني المعني المحمد المحمد المحمد المحمد المحمد المحمد المحمد
- 16. Do you know what the term avatar means? مانت عرف مانفي عي مصطلح ال مثمال او ليتهشيه؟ Ves INo INot sure Yes في ريتماك د لا يزم
- 17. Given the definition of an avatar as a digital representation of a person that can be static or animated and which might take the form of audio, talking heads or full-figure models, have you had any experience of using avatars?

had any experience of using avatars ? بالنظر إلى يحيف لتجشب منجد أن منطي رقمي أي الحتر وإعصفات الشخص) لحلم (جي يثي كون إمل عن شكل من و مثبته أوت حرك فكم لومكن أذي الذي عن الدي عنها أو يظمر على عنها و رأس المت حد شأو يظمر كالع شائل و اطرف معلى عشراش الكون وحداملك لحم المحمل و شهل للجوال، أو على أجهزة الاعاب لغرض إرش ادوتو عجب المست خدم. لى سبق وأن است خدمت أو رؤيت مثلي ذلك؟

🗆 Yes, a lot	🗖 Yes, a little	🗆 No	□ Not sure
نعم، أفيير ا	نعمق أعيال	لا	غير بتماكد



18. Each of the above examples can be classified as a type of avatars, have you encountered any of these types?

كلا من الأملى، أعلامي لمحن تصريفٍ، ولني وع من ألو اعل مثم ل أو المتثل، ٥٠ الى سيق لك وإن واج مت أيرن وعيش، ط هذه الأروَّاع؟



Section 4: Mobile learning and Avatars' preferences and effective performance

الجزء 4 النجليم عنَّ طوق الجوال و مدىف تركية وتتلي رال مثم ل على أداء الطلاب مقر ونيا ال مثم ا الأنشر رغبة

19. If given a choice would you prefer your course to be delivered through:

الغه الانجليزي، من خلال:	هلهترغب أزينطىم	ل الاختيار،	لو اعطيت

Traditional face to face	☐ Mobile device	Combined approach	□ Not sure
للتابيم للتظبيدي وج الوج ه	جماز الجوال	الاروين معا	في ر متماكد

20. If you were learning via a mobile device how important do you think it is to have an avatar on the m-learning screen rather than just textual delivery to engage you in learning? لو كل التجاييم واسط لاجوال، ما مدى الاهم الذي تتقويب أنه الجد من التواء شال التجايم معلى مما لل علم المعلم الذي تتقويب أنه الجد من التواء شال قل ول التجايم معلى مما لل علم الله على الله على معلم على على معلم مع عدم ظمور صور و لتعلم مع الن على الله على الله على الله على معلم الله على معلم الله على معلم الله على معلم الله على معلم الله على معلم الله على معلم ال الله يجذب الله التحليم في الن على الله على الله على الله على الله على التحليم على معلم الله على معلم الله على م الله يجذب الله على الله على التحليم معلم الله التحليم الله على الله على على معلم عدم ظمور صور و التحليم على ال

غىر ىتجەك م مم جدا م جيد فېر مەم م ەم

21. Which avatar type do you think would be most engaging and enjoyable to learn through on mobile phone? Please rank 1 as the most preferred and carry on to 5 as the least preferred. أي من أواع ل مُمَّالتَستُعقد أو سوف **يبهجك و يجَذَبنك**انتكلم من خلل شاشُ ة للجوال؟ فنت لكق مبتادة من الانكار ريخُ ه برق متاسل سال بالادقام بني 5 للأول ريغه.

ن Male and female respondents 👚 👘	للذكور وال
Type of m-learning delivery on mobile device which would <u>engage</u> me	Rank
ريس من خلال الجوال ال <i>تاي ماتيق</i> د أن ها ستزيد من حماسيل يالتكلم	قيِّم طرق الـــت
Instructor voice only with text (but no image)	
چل اش رح للدرس مع ظور الن صف قط وب دون أي ص و ره	صوت ال
Static photo avatar including text + voice	
صورة العلى الشات مع ظهور الن صال مشر والح قراءه	صوت ود
Talking head avatar including text + video	
صورة ليجل، الهت حرك، مع ظهوررالص ل لقراءه)ملى المقطاع فيدي وي (صوت ود
Static or animated cartoon avatar including (text + speech)	
صورة ل عليهاى شرك شخصي لهارتون مع ظمور النَّص المشرو لحلق راءه	صوت ود
مېلللىغامى للقراءفق ط No avatar text only	لايوجد ،
Female respondents only	ن الإف ف
In general do you have a preference for male or female avatar Male Female Female في مو الإثناني الذكر أو الإثناني؟	No preference] الحالاهما سو ا وتلا ضری لوین

22. Which avatar type do you think would be most effective in helping you to learn the most information and improve your performance of the course through your mobile phone? Please rank 1 as the best avatar for retaining information and 5 as the least effective avatar. أي من أبواع ل مُمَّاليَّنتيَّق أبه سوف يؤثر عاساً عدك على عليمة ذك الم علومات كما أن مسوف يحسن أدانك من خلال المتابع معن طري قش ال ذالجوال؟ مرفع ضراب كقم بتارتيم الالثار يستُي للبرق ما إس لس ل الارق ام يتحى رقم كل الحالي

لذكور وابخات Male and female respondents 👚 👘	J
Type of m-learning delivery on mobile device which would <u>affected</u> me to learn	Rank
طرق المهدريس من خلل الجو ال\لنتي لتتقد وللست شراي جلماعل عنه الميممي	ۊٛۑۜۜؗؗۜؗؗ
Instructor voice only with text (but no image)	
صوت ل ج ل م الشرح للدرس مع ظورن الص ف قط وبدون أي ص ور ه	
Static photo avatar including text + voice	
صوت وصورة العلم الشات مع ظهور الن صرال مشر والح قرراءه	
Talking head avatar including text + video	
صوت وصورة العلم، الهت حرك، مع ظموررالص ل القراءه)مل المقطاع في دي وي (
Static or animated cartoon avatar including (text + speech)	
صوت وصورة لعلع لعلى شك شخصي الحارت ون مع ظهور الن ص المشرو لجلق راءه	
No avatar text only لاي وجد مثلي لل عن من ل لقراء فق ط	
ن المجت فق ط Female respondents only	
In general do you have a preference for male or female avatar Male Female No p تلاضي لوين م الأثنى للذكر أو الأن ي:	oreference الحلامما سو ا

We are particularly interesting in how m-learning could be used to help students learn English language as a core course within their programme of study.

يكمن القمام الملحث ت حييط عيطفية أستنخدام الجوالف يالت تجي المكتي ويساعد للطلاب في عيتهم الدغة الال لجي في ه والتكيت قد قرر ا لل الريف ي خطك الدر الري ه

23. How engaging and enjoyable do you think m-learning using avatars will be compared to traditional face-to-face learning of English would be?

ي طيمى مُمْمَّل قيارن ه	، ه من خلال لاجو ال لاذ <i>ي ي</i> خو	يحقدف ييتاييم للدغ ه الار لجي زي	م ا هو ر ي ان <i>ف</i> ي الجاذبيه والابتهاج الذيين
		رىل <i>ىي؟</i>	ببال في التولي دي وج الوج في الصف ال
□ Very engaging	Quite engaging	Not engaging	□ Not sure
محمس وجذاب جدا	محمس وجذاب	غير محمس	غير نةأكد

24. How effective do you think m-learning with an avatar will be in helping you to learn information and improve your performance of English language compared to traditional face-toface learning?

ما دوي لك في فاعليه التكليم من خلل الجوال بوجود مُمَّل في الشش و مدى <mark>تأثير و الايجابي ع</mark>ليك كتطى م لتحسي ن □ Very effective فعال جدا

Please select by ($\sqrt{}$) an answer that best represent how much you agree or disagree with the following statements ($\sqrt{}$).

	1. Strongly Disagree 2. Disagree 3. Neutral محييد لا افلىق	4. Agree اولىق		ongly Ag اولىقەقور	ree	
No				Represen	t	
	الاعل • Questions	Strongly Agree اولىقىقەرە 5	Agree اولىق 4	Neutral ميچيه ع	Disagree لا اولىق 2	Strongly disagree لا اوْلىقىقىرە 1
25	I think that m-learning could be more <u>convenient</u> to me than learning face-to-face in class أنا اعتقد أنالتهم من خلال لجوال لل غه الانلي زي هي لكن از يك ونف ي ومناسب لي لتشر لم ين علم للحض ورل م حاض التصفيه			3		
26	I think the portability of mobile device plays a strong factor in enabling me to learn anywhere and anytime مان خصري الدرال جو التال عب لحجام مان المحرمون والتي قال جايون ف عيت الجناعين مان جليون ف عي أي المحان وزمان					
27	I think the repeatable and pause features of lessons on the mobile device would be more beneficial to me than a traditional class. اعتى د ل خاصري الإق اف المقت و إعادمتك راتش في ال ق طيك علم عن طولة إل تغيدن ي حيث لا أجد هذه الخوص ف لهيتاً علي إلى طي					
28	I think that m-learning could <u>engage</u> me in learning English lesson more than learning in face-to-face أنا اعقد لايت علم من خلال الجوال <u>سيجذيني في وم</u> ت عل لم لغه الاجليزي. بشكل لكبر لمين علم عن طري قل حضورل صفي ي					
29	I think that m-learning could be a more <u>effective</u> way to learn English language than learning in face-to-face class أن كات ل النت على من خلال الجوالي لخن أزيكوزوسول، جيده فعاله وناجعان ترالم حاض التصافي،					
30	I think having access to the English materials on my mobile device would enhance my motivation to learn English language أنا اعتقد أنالدخول على عرجتوى مقرر الانديزي عن طويق ج ماز جولاي سي عزوزي حضزن مت عليم لل غة الاجليزي					
31	I think m-learning could strengthen my participation when learning English language أنا اعتقد للايتاعلم من خلال الجوالي لمجن أن يقي وي مشارلقي فيتي علم للغة الإجلي زي، من خلال مذاالي، حث					
32	I think having access to the English materials on my mobile device could enhance my learning- content interaction أن م الله الم الله الم الله الم الله الم الله الم الله الله					
33	I think what the avatar interface looks like is important for m-learning ان اعتقدلئینیفییتشک و مظمرال واج الذم حتینی محلی مشال الم علم، دی شی طول علم من خلال الجوال					
34	I think having an image of the instructor on mobile screen is important for helping me to concentrate ان لي ان ال ال ال ال ال ال ال ال ال ال ال ال ال					

No	Questions	Strongly	Agree	Neutral	Disagree	Strongly
	الاعظره	Agree اولىقىقىرە	اولىق	م اي يد	لا اولىق	disagree لا اولىقىقىرە
	I deinte internations with the Dustich materials on	5	4	3	2	1
	I think interacting with the English materials on my device via an avatar would improve my					
35	understanding of the lesson					
55	أنا التقد لأينتاف عل مع حتوى مقرر الاتجليوزي عن طريق ج فإ جوال ي					
	بولاطة المم السوف في من زم ف مركيال وس ال علي في من الم علي عبر المع المركز الم					
	I think having the mobile interface with an avatar					
	would assist me to learn English language					
36	أنا اعتقد اختياء شاشة جوال لامت خدم في مُمْه المعطموف يساعدني طيق ما يتربيا المشالان المن م					
	على قررال غة الانطيري . I think having the mobile interface with an avatar					
	will be fun to learn from					
37	أنا العققد التجواءش شرة جوال للميتخدم فيمي مُتم لو يعجبر شريت فوي هي					
	ي چا چا جا جا جا جا جا جا جا جا جا جا جا جا جا					
	I think having an avatar will <u>improve</u> my					
38	retention of information that I have learnt					
	أن1علتقد أن وجودال مُمْاللمعلم سوفيحسن عملي صلحوي للهعل ومات الميت علمتاي جليما					
	I think female students should be able to choose					
	and learn from whatever screen gender they prefer					
39	to learn from on a mobile device					
39	اعتى انالطلباطيدي المهقدره والجيره في انتخيار بياكان بن سالهعلم					
	الذي فض لولين علم من على لمراش اذاك الن عليهم من خلال					
	لاجوال					
Fem	ale respondents only				فقط	نلإث
No	Questions	Strongly	Agree	Neutral	Disagree	Strongly
	الاعتل،	Agree اولىقىقو ە	اولىق	م اي يد	لا اولىق	disagree لا اولىقىقىرە
		5	4	3	2	1
	I think learning from a female instructor in a face-					
40	to-face class would <u>engage</u> me to learn more than learning via a screen cast of a male instructor					
10	العقاد الذيتاع لم وجوا ل وجوال عل مه داخل المصف و ف يحمسني ان الج ام					
	انشر من النت علم من الهرعل مالير جل من خلال ال جو ال					
	I think learning from a female instructor via a					
	mobile screen cast would be more <u>effective</u> than					
41	learning from a male instructor via a mobile					
41						
	اعتقد لانتتاعلم منالج لمه من خلال شاش الجو ال المتطبي في مال ه <mark>تأثير فعال</mark> التشر لمينت علم من الرجل الفرعل معل تشانشة ال جو ال					
	I find a face-to-face female instructor more					
	engaging to learn from than screen based a male					
42	instructor					
	ان أجبش كما م لاينياع لم منال مع لم ه لايش وج دالوج د داخل لص ف مسيد مسينا البط الثني بالما ما المسيد الما المسيد علمه الثراث					
	ي حمسَقٍ وي جنبَقي لل علَّم المثارَ من ألفاعل مالر جل آلموجودً على ألق الله ه					

43. What do you think would be two potential benefits to you of m-learning?

عددىلىرى أو ألشرىتىتى د ن، م مي المانى من خلال ل جوال 1-

2-

44. What do you think would be two potential negatives to you of m-learning? عدد المضين أو ألشريت يحدد إن المن عي بي المتخلم من خلال لاجو ال 1- 2-

45. Are there any other comments you would like to make about how traditional face to face teaching of English language as a part of your course could be improved or enhanced through the use of m-learning and avatars?

التنسب الي لي من تعليقات أو ملاحظات أخر ديت غب أن خيف واعلى للتعليم التولي دي)وج وال وج (الذيبيت عه الآن والتي وجكن ان طور وا وتجسن وا من خلال المتعلم عن طوق الجول اله حت و يعلى مثل ؟

Thank you for your participating. No comments made will be attributed to any identified individual. We would like to develop more specific avatars for mobile learning. If you would be willing to participate further in this research, please fill in your contact details below

اشكر لك مشاركتك في تعبينه هذه الاستبانه و أود أن اشعرك بأنه لن يفشى أي معلومات خاصة عنك ولن يظهر اسمك الشخصي بملاحظات فرديه. كل ما يريده الباحث هو عمليه تطوير التعليم من خلال الجوال بإضافة ممثل للمعلم على شاشه المستخدم لكي يتفاعل معه. اذا كنت مهتما وترغب في المشاركه في هذا البحث، فمن فضلك املاء بياناتك في الحقول التاليه

Please write your name, email address below and/or your mobile phone number.

فېن لكالځتب لمسمك وعنوان بيميمك و (ق) ق مجوالك التي اري (

Full name:
Email address:
Mobile number:

Part 2: Post-questionnaire to investigate the effectiveness of delivering course content through mobile device using an avatars interfaces. The purpose of this questionnaire is to help the researcher as a part of his PhD therefore, participants are going to do short tests which would not be formal nor connect to their modulus

in mounus.	l
Participant No.	
الرقم آلجامعى	

Now that you have experienced mobile learning in your university study, Please answer the following questions الان اصبح ليميك ال نجر الويت جربه الفتكالم من خلال جاز الجوال. فينض لك أجب عن الاعل الخيالي، Please tick all that apply:

	11 5	C			
No	Questions	A lot	Some	Not at all	Not sure
	الانطئ ه	لتغيرا	بعض	لابابدا	فېر ښكد
	Did you enjoy having content delivered to				
46	you by a mobile device				
	هل استمتعت فع عشم لييت المرحتو يال في مي من خلال ج فل جو الكُ				
	Did the content delivered to you by a mobile				
47	device <u>engage</u> you				
	ملال منظوى الذي أعطى لك من خلال ج هاز جوالك جذبك				
	وتحمست له				

48. If you had more content delivered on your mobile device would you be more motivated to learn English than via traditional teaching (face-to-face)?

، ىت حمس لەا ألىڭىر من	ازجوالك، ەلىتۈكۈن	لك عن طريق ج ه	من المحقوبات ال في من على	ف ي حال لو كان ڧاك مړيدا
				لل الي القلي دي وج وال وج ه ؟
☐ Yes, more motiv	vated $\Box A$	s motivated	□ No, less motivate	d 🔲 Not sure
نعم مقاحمس جداً	 	ىقەحمس	لا غير نټحمس	غير ىتماكد
49. How important do y				
	ہول؟	خلالالجو الاالمح	ىي ملاءمە و راھةالىت خلە من	ما مدى الا مي، الذي تنتق وف
□ Very important	Important	Neutral	□ Not important	□ Not sure
م مم جدا	م ەم	مطيد	غير مەم	غير ىتماكد

50. Do you think accessing course material on your mobile device with an avatar helped you to learn the material better than the same content presented in traditional formats (face-to-face)? التستخدين الدخول على للمُتحوى للحي ميلك قرر للدر لمب عن طريق للجوال للمزود الممُهّل ساعدك على على مالدرس مقادن مبلُ طرو أعطي للطبال طريق ه لتوالي في وج هال وجه؟

			ه وج ۵ لوج ۵ :	لريق ه ليوري ي	لهبال
🗆 Yes	🔲 May be	🗌 No	\Box Not sure		
نعم	التخامال	У	غير ىتمأكد		

51. Which type of teaching do you think enabled you to learn the most information and made you feel you have learnt something? Please rank 1 as the most effective way of learning information and 6 or 7 as the least effective way.

أي من طرق التدريس الى الي منتققد أن استعقد على الاحتفاظ بالمعلومات التقل في توج لتا يقطى مشي ما؟ فينص ل كتيم م الطرق الثالي موضَّع رقم 1 أمام الطريق الالكرف علي موتعلي راي جدي علي علي علي من المراس لا بالار قام الأولف علي متحي رقم 6 لذكور و أللانيات

Male and female respondents 🕇 🕈	ر والإذاث	للذكور
Type of teaching mechanism that I really find	d it <u>affected my performance</u> of learning	Rank
حتفاظ بالمعلومه ورفعت من آدائك في الاختبار	نوع او طريق المتدريس الترييت تقد رادا ساعت كعلى الا	ۊٛڲؘۜؗؗؗؗؗؗ
Traditional class teaching face-to-face in class	لتدري سنالق لي ميه وج وآل وج ه داخل الصف	
No avatar, textual only through mobile device	تاعلم من خلالشااشهالجوال لکے فصقط	
ر الليخلم Text and audio avatar through mobile	الصوت والن)صالمقروء من خلال ج هازال جو البدون في صره	
Static male avatar with text + voice through mo	صوت وصور ةال مظیما الثلبانه من خلال ج هاز bile device	
	ال جو ال معالين صل لق راءه	
Talking head male avatar with text + video	صوت وصور ذالهعلمالة حرك، من خلال ج دازال جو ال مع	
through the mobile device	الن)صالمقروء ايالمقطعالفيديوي	
Animated cartoon avatar with text + speech	صوت وصورةالوعلم على شكىل شخصري الخوات وزمت حركه	
through the mobile device	من خلال ج دازال جو ال معالین صل لقر اءه	
Female respondents only 希	بث فقط	ど
Static female avatar with text + video صليلقراءه	صوت وصورةال مثى مالثلبته من خلال ج مازال جوال معالين	

52. Which type of teaching did you find most engaging and enjoyable to learn from? Please rank as 1 as the most preferred and 6 or 7 as the least preferred way.

أي من طرق التحديس الثالي، وجدتفيها اللاشر جلني، وتمَّعَ، وأَلْكَحْقَقَ، رَغْبَتَ الْيَعْلَمُ مَنْ خَالَ، أَفْنَضُ لَكَتْقِيم الطرق الثالي، ويضع رقم 1 امام المعالمان طريق، تعريس الجالارق الم الأول فنضري لا ورغ تقتص رقم كال الخور و 7 للإزائث

Type of teaching mechanism that I really find	most engaging and enjoyable to learn	Rank
خلالها وزادت من حماسك للتعلم	نوع او طيق التحدي سالت يتنتق د لك رغبت التلعم من .	ۊٚڲؚۜۜۜۜۜؗ
Traditional class teaching face-to-face	طريقةالتدي ستلق لي بي ه وج ه ألوجه داخل الصف	
No avatar, textual only through mobile device	بدون مُمْثال»تاعلم من خلال نشله الجوالانصريفِماقط	
یورہ للکام Text and audio avatar through mobile	الصوت وللنصالمقروء من خلال ج هازالجوالبدون لمي ص	
للقراءه Static male avatar with text + voice	صوت وصور ةالهع لمالثقبته من خلال جوز الجو ال معالين م	
Talking head male avatar with text + video	صوت وصور ةالهعلالم،تحرك، من خلال ج ،ازالجوال مع	
through the mobile device	النيصالمقروء ايال قيط عاليديوي	
Animated cartoon avatar including text + speech	صوت وصور ةالوعل معلى شرك لش خصري الخوات ون	
through the mobile device	ىقەحرىكە من خلال جەازال جوال معالىن صرلىلقىر اءە	
Female respondents only 希	ث فقط	نلا
Static female photo avatar with text + voice through	gh the mobile device	
صل القار العلاي المطلقة فق ط	صوت وصور ةاله علم مالثلبت، من خلال ج مازال جو ال مع الن	

Having tried m-learning please select by $(\sqrt{})$ an answer that best represent how much you agree or disagree with the following statements. (٧) من مدي لفي قامت عنه منه منه منه العرب المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني معني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني معني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني ال

1. Strongly Agree	2. Agree	3. Neutral	4. Disagree	5. Strongly Disagree
افلقىقوه	اولق	محليد	لا اولىق	لا اولىقىقوە

No				Represen	t	
	Questions الایجال ه	Strongly Agree اۇلىقىقورە 5	Agree اولىق 4	Neutral م لي د 3	Disagree لا اولىق 2	Strongly disagree لا اولىقىقىرە 1
53	I think the m-learning would be more <u>convenient</u> to me than learning face-to-face in class أن التقود ان لتعقيد من خلال الحق الأن لمج الانطي ذي هي لكن ان ولكون مفيد ومناسب لي لشار لم ين علم المحض ورل م حاض التصفي ه					
54	I think the repeatable and pause features of lessons on the mobile device would be more <u>convenient</u> to me than a traditional class. اعتقاد ان خاصري، الايقاف المرق ت و إعاد متك رات شريغ ال ق طين علم عن طري ق الجو ال كان ت الخشر فانده و ملاءما ي حيث لا اجد هذه ال خواص ف اعينا على احف ي					
55	I think the portability of mobile device plays a strong factors in enabling me to learn anywhere and anytime مازال جوالت لعب الحاري مال محمول تظان قرللج مازال جوالت لعب الحاري من المان عب تعلي مان و زمان					
56	I think that the m-learning would be a more <u>effective</u> way to learn English language than learning face-to- face in classes أنا اعتقاد لايتاعلم من خلال الجوالكان وسول، جيده فعاله وناجلا علم اللغة الاجليزية المشر في حضور المحض التصافي،					
57	I think having access to the English materials on my mobile devices would enhance my motivation to learn English language أنا التقد انالدخول على حتوى مقرر الاجليزي عن طريق ج از جوالي عززوقوى الخرالي غة الاجليزي					

No	الایکل، Questions	Strongly Agree اۇلىقىرە	Agree اولىق	Neutral مبچید	Disagree لا اولىق	Strongly disagree لا اولقيقوه
58	I think m-learning would strengthen my participation when learning English language أن اعتقد لينتاعلم من خلال الجو السوف عقووي من مش الرايخي ف ي الت التي م إذا لك المرحتوى هومق رال لي غة الاندلجي زي		4	3	2	
59	I think m-learning will increase my understanding of the learning material than face-face in class ن أيتخ لايتاعلم من خلال ال جو السفيو ي رفيون ي من من علمي يل غه الاجلي زي مبشلك الجبر المنتاعلم عن طورة الحض الرس في					
60	I think having access to materials on mobile devices would enhance content-learner interaction ناچلتىقد أنالدخول على عرصتوى حقار الانلچيزي عن طريق ج «از روالى عززن متضلحالي ماللى مرتسو على يحلي ي لمقر رال لغة الانلچيني ة	Î				
61	I think what the avatar interface looks like is important for m-learning ن م مثالياه على من خلال الجو اله سيب جلنييتموشده ليتباله مك مي على من خلال الجو الم سيب جلنييتموشره ليتباله مك عل مين	1				
62	I think having an instructor avatar would cause me ان التقاد ان وجودال تیش میساله عل معلی to lose concentration شاش تال جو ال المنتخلی بی مس و فی سی بدل فی قدان الت دلی بی فی ال در س					
63	I think having an m-learning interface with an avatar would positively assist me to learn English أن التقويد التي واعش ش ة جوال ل ميت خدم في من مثم ال الم عل ميس وف بس محدن ي بي جلي اعلى عيد لم ي ل مقتوى در سال ل غه الانكي ي	(4				
64	I think having an interface with an avatar will be fun to learn from أن التقد التحواء شرش ة جوال للمتخدم في مُمْ ليع بير شريت فوي هي نوعلم في ه					
65	I would prefer to learn in the normal traditional face-to-face class لناس وفضاض للمتاعلم من خلال حض ورال محاض رالتص في وج ما					
66	I would prefer having an m-learning interface أَنْطَاضَ لِلْسَاعِلَم مِن خلال (e.g text only) شَاضَعَالَم مِن خلال شاشقالجو ال النصري مفقط من دون صوت أو صور لها معلم					
67	I would prefer to learn through mobile devices including text + voice via mobile device (but no image) شرر حالم علم دلك ن بلاحت و يع لى عن من خلال الدي حت و يع لي عن من وصوت					
68	I would prefer to learn with a static male avatar via mobile device with text + voice انظاض لا المتحلم من خلال ج هازال جو ال لاذي حتوي على صور هاله علم المثلبت، مع ظمور الصوت والص المشروح					
69	I would prefer to learn with a talking head male في المعنون العام من خلالال جو الالذي ي محتوي على صور اله علمالهت حرك خدورال صوت والن صالمشرو الحال في دي و مع	ف				
70	I would prefer to learn with a cartoon photo avatar via mobile devices with text + speech نظاطرالما علم من خلالال جو ال الذي حتوي على صور اله علم على شرك لش خصري الخوات وزيمت حرك مع ظ مورال صوت والن صرال مشرر و ح	١				
71	I plan to continue using my mobile device for receiving educational content (if available). ان أعز هو أن ويعلى كمال ملت خدام للجو الكتي في ما ذاك ان تتباحا					
72	I believe extending the concept of the avatar to other courses would be useful. ن اؤمن ان وجود للمثلف ي لل قرر رات الاخرى سينيويد جدا					

Agree التخطن التعنية <	ıgly
73 I think having an avatar approach would increase my understanding of information that I have learnt not provide the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure of the structure	
73 my understanding of information that I have learnt ان تلكت قد أن طوق، فل ممثل للتي بلتي خدمت سال قال على مرك مثل المعلى فعت وزادت أي ف هي لله عل ومات للتي على مثل فعت وزادت أي ف هي لله عل ومات للتي على على مثل المعلم المعلم المعلم المعلم المعلم المعلم المعلم المعلم المعلم المعلم فض العقد أن طوق، فل ممثل للتي يلت خدمت سابق النتاجيم ولك ممثل المع على فعت وزادت المحت المعلم المعلم المعلم المعلم المعلم المعلم المعلم المعلم المعلم معت على مثل التي يلت خدمت سابق النتاجيم ولك ممثل المعلم على فض تل العقد أن طوق، فل ممثل التي يلت خدمت سابق النتاجيم ولك ممثل المعلم المعلم المعلم التي على مات المعلم المعلم المعلم المعلم المعلم المعلم المعلم المعلم التي على مات المعلم المعلم المعلم التي المعلم المعلم المعلم المعلم المعلم المعلم المعلم br>المعلم المعلم المعلم المعلم المعلم المعلم المعلم المعلم المعلم المعلم المعلم المعلم المعلم المعلم المعلم المعلم المعلم المعلم المعلم المعلم المعلم المعلم المعلم br>المعلم المعلم br>المعلم المعلم المعلم المعلم المعلم المعلم المعلم المعلم المعلم الم	
73 أن توت قون المعتمان التي الميت خدمت سياف تك علي موك ممثاليا له على ما تشخص علي تستخدمت سياف تك علي موك ممثاليا له على وما تشخص علي تستخدمت سياف تك علي معنا علي تستخدمت علي تستخدمت علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المتالي المتالي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد علي المعتاد	
74 آن محت وزادت نؤ ف هي لله عل ومات لؤيت علمت وعلم وقام بالسام على م 74 I think having an avatar approach would increase my retention of information that I have learnt i degs of back by the system of information that I have learnt i degs of back by the system of information that I have learnt i degs of back by the system of information that I have learnt i degs of back by the system of information that I have learnt i degs of back by the system of information that I have learnt i degs of back by the system of information that I have learnt i degs of back by the system of information information that I have learnt i degs of back by the system of information of information that I have learnt i degs of back by the system of information of information that I have learnt i degs of back by the system of the system of information of information that I have learnt i degs of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the sys	
I think having an avatar approach would increase my retention of information that I have learnt أن ا اعتقاد أن طوق ال مثل للذي بلت خدمت سياق النتائيم وك مثل للهاعلم فضعت وزادت أم تت فاطع بلل على و مات لليت علمت I think female students should be able to choose and learn from whatever screen gender they prefer to learn from on the mobile device و الت أوي و ال عيد ف و العيد و ف مثل اله على النائين على مات اعتقاد ان الطلب ات ك از الي على مات الي تعالي من اله على مال المتحال اله عن دما الخالية تعاليم من خلال ال جو ال ي فض لولايت على من در مالكارين عليم من خلال ال جو ال Female respondents only	
74 my retention of information that I have learnt مالله علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية علي المعالية المعالية علي المعالية علي المعالية علي المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعالية المعاليقالية المعالية المعالية المعالية المعالية المعالية ا	
 أن اعتقاد أن طوق ه الممثل لتني بليت خدمت رياق ألت في م وك مثل للم علم أن اعتقاد م وف مثل للم علم أن اعتقاد من وف مثل للم علم أن اعتقاد من على مت وزادت لم ختف اطيب الع لو مات ليت علمت I think female students should be able to choose and learn from whatever screen gender they prefer to learn from on the mobile device اعتقاد المالي المالي الي على وال عربة ف عي اعتيار عن سالم علم الذي العقاد من على من المالي على من المالي على من المالي على من المالي على من المالي على من المالي على من المالي على من المالي على من المالي على من المالي على من المالي على من المالي على من على من المالي على من على من المالي على من المالي على من المالي على من المالي على من المالي على من على من المالي على من على من على من على من المالي على من على من على من المالي على من على من على من على من على من المالي على من على من على من على من على من على من المالي على مالي على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من على من	
ن المحقود ان طريق دان طريق من طريق من طريق من طريق من طريق من طريق من طريق من طريق الفاطي مرتبع من المحتوي المعالي من المحتوي المعالي من المحتوي المعالي من المحتوي المعالي من المحتوي المعالي من المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتو المحتو المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي المحتوي ال المحتوي المحتوي ا	
I think female students should be able to choose and learn from whatever screen gender they prefer to learn from on the mobile device القرد ان الطلب انتكان لدي والم تعدد و لل جود ف عد الجي الرخي على المرالذي المحال الحوال المحال الحوال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال المحال	
and learn from whatever screen gender they prefer to learn from on the mobile device انتخاب النال الطلب التكان الذي والم حين و التجريد من المرحل مال ذي يفض لولينتا علم من وعلى الل الش معن دم التالين تاعليم من خلال ال جو ال Female respondents only	
75 to learn from on the mobile device التقويد انالطليات كان لذي علم القريرة و لل جوره فسي انتخبار جن سالوع علم الذي الفض ليل يشاعلم من وعلى اللي الله معن دم الكارينا عليم من خلال الجو ال Female respondents only	
اعتقى دانال طلب اتكانان له يوالم قدر و لل جريد ف مي انتخيار جن سالم علم الذي يفضل بولينه علم من معلى اللي الله معن دما كالينه اعليم من خلال ل جو ال Female respondents only	
يفض لولاين علم من معلى اللي الله من عندم الخلاية عليم من خلال الجو ال Female respondents only	
Female respondents only	
I would like to learn with a static female avatar	
via a mobile device with text + voice	
ان الفاض القابيم من خلال الدي حتو رجل مور مال جلامه الثلبة،	
مع ظهور الناص في اص وت	
I think learning from a female instructor in a face-	
to-face class would <u>engage</u> me to learn more than	
77 learning via a screen cast of a male instructor	
انتخاب المعني و المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني المعني ال	
الثار من التعام من الهاعل مالير جل من خلال الجوال	
I think learning from a female instructor via a	
mobile screen cast would be more <u>effective</u> than	
learning from a male instructor via a mobile	
78 screen cast	
انتقاد النتاع لم منالج لم من خلال ش الأس الت في عي هاده ت ناثير فعال	
التشر لمن علم من الرجال في علم على عن الرجو ال	
I find a female face to face instructor more	
engaging to me than learning via screen cast of 79 male instructor at the class	
انا أجبشكل عام لاينتاعلم منال علم لايشى وجەل وجە <mark>يحمسنى</mark> ويجذبنى ل تېل منالوعلمال جل عبرالاي شامال م _{لاك} ە	

80. How well do you think you have done in the test using your favourite avatar?

المتنتعقد أن أحد مثل ي لعلم **ساعدك في رفع أدائك** وتُحصِّي لك العلمي في الاحتبار . التحتوة يورا لمُستولك الذيَّيتنعق د لك قتقت ؟ Average Poor Good Do not know Great

لا اعلم ضريج توسط جيد متهاز Having tried m-learning, would you like to experience it further, please tick all that apply: بعد أن قمت من التابعيم من خلال التابعيم عن طريق الجوال، هايتار غدف حيتك رار هذه الطرق من التابعيم مرتق لا، ضرع علامه (٧). أمام مليتياسب معك

					امام مال الس
No	Questions	Yes	May be	No	Not sure
	الاعظل ه	نعم	الخث مال	لا	غېر ښكد
	Would you like to undertake future courses with				
81	integrated mobile learning for English language				
	ملت رغيفي انتدرس ميتق لال قرر رات الأن لجي ذي عن طريق للجوال				
	Would you like to undertake future courses with				
82	integrated mobile learning for other courses				
	ملت رغيفي انتدرس ميت ق لالي قررات الآخرى عن طريق ل جوال				
	Would you recommend m-learning with an avatar				
83	interface to other students				
	مل سين صب هذه ال طريق إول طلاب الأخرون ال علم من خلال الجو ال ال مح مول مع				
	وجود مم المع علم لغرض لاشرح والمتهو تجي موشد الانتباه				

85. What did you dislike most about learning with an avatars? ما مو ألش شي لي عجك ولمت ريخ، من خلال عليه مك مع وجود ال مثمل؟

86. What do you feel was engaging about m-learning using avatars?

صف مدىان جنبكواستنتهك الذي شعر يتفيه عيدماً أخذت مذه الدروس العلميه مع وجود مثمال العظم؟

87. How could the avatars you used be improved?

ما مويريك فيتطوير مثمال لكم لعملي استخدمت؟

88. Are there any other comments you wish to make about your experience of receiving course content with an avatar?

هل فاك أي ملاحظات تتنى أن خرف امن أجل تحسي فستوى ممثل لع لع لك يوسا التقال لع ل وات ؟

Thank you for your participation. No comments made will be attributed to any identified individual. We would like to develop specific avatars that are engaging to you for mobile learning. In order to do this the researcher will run a workshop where learners will be given the opportunity to express their views and ideas about how the avatars should be developed.

اشكرك على مشاركتك في هذه الدراسه البحثيه واحيطك علماً بانه لن يفشى أي معلومات خاصة عنك ولن يظهر اسمك الشخصي بملاحظات فرديه. كل ما يريده الباحث هو عمليه تطوير التعليم من خلال الجوال بإضافة ممثل للمعلم على شاشه المستخدم. من اجل ان نقوم بذلك فان الباحث يتطلع الى انشاء ورشه عمل والتي من خلالها يتاح للمتعلمين الفرصه لإعطاء وجهات نظرهم والمشاركه بأفكارهم والتي ستوخذ بعين الاعتبار والجديه في تطوير واعاده تصميم هذه الواجهات لعرض الدروس العلميه. إذا كنت ترغب في المشاركه في هذه الورشه، فاكتب بياناتك في التاليه.

If you are willing to take part in the workshop, please write your name, email address below and/or your mobile phone number. هنخ القواص معك معن وازي يله الغيارف ي التعليه وقرح وال لك مع يعن وازي يله المع

Full name:

Appendix F

The pre-post questionnaire of the Experiment - 2

Thank you for taking part in this experiment. This second set of experiments builds upon an initial experiment to investigate whether mobile devices can make English language learning more effective and engaging. I would appreciate you taking the time to complete the following questions. They should not take more than 10 - 15 minutes of your time. Your participation is voluntary and your responses will be confidential and not attributed to any individual.

Section 1: About you and your mobile device الرجزء ا . ول أنوئ مع على ومثلك

- 1. What is your student number? كستب اس في الثلاثيب الث
- 2. What is your gender? حد ښك؟ □ Male □ Female

<u>څی</u> دکر

3. What is your faculty?



In your previous and/or current education, have you experienced m-learning?
 Yes
 No
 Maybe

Section 2: Your views about how effective and engaging you find different styles of mlearning delivery. Please tick the box that most closely represents your views.

		Represent					
No	الاعل، Questions	Strongly Agree اولىقىقىرە	Agree اولاق	Neutral م ا ييد	Disagree لا اولىق	Strongly disagree لا اولىقىقرە	
		5	4	3	2	1	
	I liked having an m-learning interface with: انا أفضل التعلم عن طريق شاشة الجوال اذا كانت تحتوي على:						
5	Voice (audio) media instruction صوت لچلم مع ظمیر النص ال شروح						
6	Video media instruction صور ةوصوت العلى شرك لفيجيي						
7	Cartoon media instruction مېلى للىچرىچارى شا <u>ن</u> ترلوخصىيە كىتىنىيە مع ظمور لماسوت والىن						
	With regards to self-assessment and feedback int التقييم الذاتي والتغديه الراجعه	teraction					
8	Having immediate feedback improved my interaction with the learning process نا أش عرب أن لك غيف الراجع التقطيع ساعين على المقاعل مع الم حتوى						
9	Having commands to continue with or review the lesson helped me to interact with the learning process أزر الابت الخم ليتون ال ال لي شريح ملك الي الاجبة الاجبة صرح و ال عودل لدس السريد فف ي حال ك ل تا الاجبة خاطئه جلتني ق ادرعلى المشاعل ع الم يتوى التخليم						
10	Having m-learning with a voice, video or cartoon interface assisted me in the learning process. أن المتحدد أن المتحلم عن طوق الجوال الذي حتوي على أحد الوسطط المثلث ه)الصري للى في وأال شخصي الحر توي هو أن ه ساع في علي المتحلم						

	I believe having access to the English materials	on my mo	bile devi	ces with t	he <u>audio</u>	
	instructions interface: أنا اعتقد ان الدخول على درس اللغه الإنجليزيه من خلال جهاز جوالي عندما تكون الشاشه <u>صوتيه</u> مع النص انه:					
11	Enhanced my motivation to learn English language					
11	عز زن يو حدرن ي لطل م ال غ م الانجليزي م					
12	Engaged me in the English language lesson زاد من ج <i>لسي ل</i> ظلم ا ل غ، الانجليزي،					
13	Enabled me to interact with the learning process جلن يق ادرع لى القداع ل ج المضي النظاي مي					
14	Assisted my learning performance سالح في ف عس تواي العلمي في الاستبار					
15	Positively helped me to learn English language لمطره الإيجابيعل يفي ينتخل مال غه الانجابي ه					
	I believe having access to the English materials	on my mo	bile devi	es with t	he <u>video</u>	1
	instructions interface: هاز جوالي عندما تكون الشاشه <i>فيدويه</i> كصوره وصوت المعلم مع	، من خلال حا	فه الانحليز با	لم در س اللغ	ان الدخول ع	أنا اعتقد
	النص الله: <u>يوني</u> الله: النص الله: النص الله: النص الله: النص الله: النص الله: النص الله: النص الله: النص الله: ال		-		-	
		Strongly Agree اولىقىقوە	Agree اولىق	Neutral م ا ييد	Disagree لا اولىق	Strongly disagree لا اولىقىقىرە
	Enhanced my motivation to learn English	5	4	3	2	1
16	language عززن يو حفرن ي لظلى الغ، الانجليزي،					
17	Engaged me in the English language lesson زاد من ج <i>لسي ل</i> ظلم ا ل غ، الانجليزي،					
18	Enabled me to interact with the learning process جان،يق ادريل، القداعل ج المضيء المتخليمي					
19	Assisted my learning performance سالحي في في صحي تواي العلميفي الاستبار					
20	Positively helped me to learn English language ىتْرىماريار ماليغام النغ الانجليني ه					
	I believe having access to the English materials	on my mo	bile devi	ces with t	he <u>cartoo</u>	<u>n</u>
	instructions interface: أنا اعتقد أن الدخول على درس اللغه الانجليزيه من خلال جهاز جوالي عندما تكون الشاشه على شكل <i>شخصيه كرتوينه</i> إنه:					أنا اعتقد
21	Enhanced my motivation to learn English language عززن يو حفرزي لظل م ال غ م الانجليزي ه					
22	Engaged me in the English language lesson زاد من جسي لطان مالغ الانجليزي ه					
23	Enabled me to interact with the learning process جان يق ادرعلى المشاعل ع المضي المتظلي مي					
24	Assisted my learning performance سائحي في في محسنولي العلميفي الامتبار					
25	Positively helped me to learn English language ىتُوعِل <i>ي إي ح</i> ليا لظي م ال غ ه الانجلي في ه					
	Based on your experience of m-learning from th	is experin	nent:			
26	I would recommend to others the use of m- learning interfaces to deliver m-learning					

27	I would use m-learning for English language frequently if it was available			
28	I would use m-learning in my studies for other courses if it was available			

Which interface style of interaction did you prefer?

29. In which order did you find the lessons most engaging:

- O Audio then Video then Cartoon
- O Audio then Cartoon then Video
- O Video then Cartoon then Audio
- O Video then Audio then Cartoon
- O Cartoon then Audio then Video
- O Cartoon then Video then Audio
- 30. Which order do you think has helped you learnt most effectively:
 - O Audio then Video then Cartoon
 - O Audio then Cartoon then Video
 - O Video then Cartoon then Audio
 - O Video then Audio then Cartoon
 - O Cartoon then Audio then Video
 - O Cartoon then Video then Audio

Section 3. Further views of finding from the first experiment about m-learning

معلومات إضافيه ووجهات نظر من التجربه السابقه حول التعليم من خلال الجوال توصلنا إلى الأتي:

In the earlier experiments we found that:

31. More male students than female students preferred m-learning to traditional face-to-face lecturing. Why do you think this might be the case?

النثير منال طلاحيفض لون التعلم عن طريق لاجول التعليم عمق ان مبالطال استال في ف ص ل التلجيم التطي دي وج وال وجه. ما ووالسبب الذيت تقدلهذلك؟

32. Both male and female students particularly liked learning with a video interface. Why do you think this might be the case?

الجيمع الطلاب والطالب ات رينجوا المتعلم عن طريق واج هات ل جو ال فيدي هي ه ول ها هي الاض ل مقارز بمال واج هات الاخرى س والن صري ه أ الص وي ه، م ل ب ال ذي تقيمه ل ذلك؟

33. In the first experiment, audio was the least preferred but was found to enable the students to perform best. Why do you think this might be the case?

ف لليت جي لماسياق متحص لن اعلى أن الدرس للصري يكان هو الأولى فضيلا ورغ مليكن لاغ يب أن مج علي تعليم الطلاب والطالب انف ي الاختبار هي المجلى ما السيب الذي تقتي ولذلك؟

34. Female students preferred female rather than male audio. Why do you think this might be the case? Do you think it would be the same for a video presentation (for female students only).

يوصرن إلى أن الطال ليف ضرارون التعلم عن طرق الشراش والصروي الل من مالمر أو ألفش نم الشرش الصروي لطع لم الرجل، ما السبب الذي يستخيب لذلك؟ و مايتوق عين أينغس الوض عم الشراش الوف عن وي.

35. Science students liked using m-learning more than Arts students, however, Arts students intended to continue using m-learning after the experiment more than Science students. Why do you think this might be the case?

طلاب وطالب،ات المتخصص العلمي فض في است خدام ل جول المتعلي مي ألتشر من طلاب وطالب اللىت خصص الألبي، ليكن الأب ي عز مورج لى است خدام ال لجواف المتحل مالتشر من طلاب وطالب اللى تض ص العلمي ما السبب الذي يتقيعه لذلك؟

36. Male students rated the convenience of m-learning higher than female students. Why do you think this might be the case?

الطلابةي مواملي ما التبخدام لاجو الحياي المتكلم أكثار في الطالبات؟ مل سبب الذي يقتع الذك؟

37. Can you suggest any improvements to the way in which learning can be delivered via an audio, video or cartoon interface?

ملتست طيع أفتقترح أييت طور اليمى طرق شرحال دروس التظليمي مسواء الصوي لمى فيدي وه أو اليرت ويكلى شاش ه الجوال؟

38. Is there anything else would like to add about m-learning interfaces or m-learning in comparison to face-to-face teaching in general?

omparison to face-to-tace teacning in general . مل فناك لي شري آخرت يد أن يضيف ول شاش ةال جوال التظليمي، أو حول التظليم عن طوق ال جو المقارن بعالتظلم التي يدي ال وج مل وجه؟

39. If given a choice would you prefer your course to be delivered through:

لو أعطيت الاختيار، ولفتر عب انتظام الع، الأنجايزي من خلال:							
Traditional face to face	☐ Mobile device	Combined approach	□ Not sure				
للت لي التظيدي وج ه ألوجه	جاز الجوال	الثنوين معاً	فجير بتمأكد				

40. Maybe you have chosen the traditional face-t-face learning ,however, if you have given the choice would you prefer your course to be delivered through:

E-learning via PCs or laptops M-learning via phones or tablets No preference



The copy of the exam/test

Find verbs in the two brackets and choose the right answer which linked to each tenses' grammar

Present simple

- 1. You English language every class. (Speak). Negative
 - \circ are not speak
 - $\circ \quad \text{does not speak} \\$
 - \circ do not speak
 - $\circ \quad \text{do not speaks} \quad$

2. She Tennis every week. (Play). Positive

- o is play
- o plays
- $\circ \ \ does \ play$
- o play
- 3. Theythe school at 3.00pm every day. (Leave). Positive
 - $\circ \ \ \text{do leave}$
 - o leaves
 - \circ are leave
 - \circ leave
- 4. It at 8 o'clock every morning. (Start). Negative
 - \circ does not start
 - \circ is not start
 - $\circ \ \ \text{does not starts}$
 - o not starts

5. I to eat Chicken Burger every night. (Like). <u>Negative</u>

- \circ am not like
- $\circ~$ do not likes
- \circ not like
- \circ do not like

Past simple

- 1. You English language yesterday. (Speak). Negative
 - o did not speaked
 - \circ did not speak
 - \circ did not speaks
 - \circ are not speak
- 2. SheTennis last month. (Play). Positive
 - \circ did paly
 - o play
 - \circ is played
 - \circ played
- 3. They the school at 2.00pm yesterday. (Leave). Positive
 - o leave
 - \circ did leave
 - \circ did leaved
 - \circ leaved
- 4. It at 8.00 o'clock in the morning. (Start). Negative
 - did not start
 - \circ is not started
 - \circ did not started
 - \circ does not start
- 5. I the chicken burger last night. (Like). Negative
 - \circ did not like
 - \circ not liked
 - \circ am not like
 - \circ did not liked

Present continuous

- 1. You English language right now.(Speak).Negative
 - \circ do not speaking
 - \circ are not speaking
 - \circ not speaking
 - \circ are speaking not
- 2. Shetennis at the moment. (Play). Positive
 - \circ is playing
 - o playing
 - \circ plays
 - o does playing
- 3. They for their exams just now. (Leave). Positive
 - studying
 - $\circ \quad \text{do studying} \quad$
 - $\circ \quad \text{ are study} \quad$
 - \circ are studying
- 4. Itto move right now. (Start). Negative
 - o is not start
 - o does not starting
 - is not starts
 - \circ is not starting
- 5. I to eat chicken burger at the moment. (Like). Negative
 - \circ am do not liking
 - do not liking
 - \circ am not liking
 - \circ not liking



A release letter from the main supervisor at Al-Baha University for the completion of the data collection



الملكة العربية السعودية

وزارة التعليم العالي

جامعة الباحية (٠٤٢)

الرقم :

KINGDOM OF SAUDI ARABIA MINISTRY OF HIGHER EDUCATION AL-BAHA UNIVERSITY (042)

> إلى من يهمه الأمر اسم المبتعث: علي سعيد عبدالله آل عوير. الكلية : كلية علوم الحاسبات و تقنية المعلومات. الجامعة : جامعة الباحة . افيد سعادتكم أن المبتعث المدون اسمه أعلاه قام بدارسة ميدانية وجمع البيانات والمعلومات في مقر الجامعة في مدينة الباحة، و التي تتعلق بمجال بحثه في موضوع :

(Application of M-learning Using Avatars to Assist the Teaching of English as Part of Higher Education Courses in Saudi Arabia)

لمرحلة الدكتوراه في جامعة ريدنج و قد بدأت الدراسة في ٢٠١٥/١٠/١٦ و انتهت في لمرحلة الدكتوراه في جامعة ريدنج و قد بدأت الدراسة في ٢٠١٥/١٠/١٦ م. هذا و قد تم إصدار هذا الخطاب بناء على طلب منه ، و ذلك لتقديمه إلى (الملحقية الثقافية السعودية بلندن). التاريخ : ١٦ / ١٠ / ٢٠١٦م

المشرف على البحث د. عبدالله محمد الخطاب قسم علمه الماسي – كلية العلوم والاداب بالمخواة باحة

المشفوعات :



المتاريخ :

تليفون : ٧٢٧٤١١١ - ٧ - ٢٢٩٦٠ فاكس : ٧٢٤٧٢٧٢ - ٢٠٩٦٦ الباحة : ص.ب (١٩٨٨) Tel.: 00966 7 7274111 fax: 0966 7 7247272 Al-Baha P.O.Box (1988)